## SIGNAL TRANSDUCTION MECHANISMS FOR STEM CELL DIFFERENTIATION INTO CARDIOMYOCYTES

## By

## PETER SAAH HUMPHREY

A thesis submitted in partial fulfilment of the requirement of the University of Hertfordshire for the degree of Doctor of Philosophy. This programme of research was carried out in the Life Sciences Department within the Health and Human Sciences Research Institute of the University of Hertfordshire

September 2008

#### ABSTRACT

Cardiovascular diseases are among the leading causes of death worldwide and particularly in the developed World. The search for new therapeutic approaches for improving the functions of the damaged heart is therefore a critical endeavour. Myocardial infarction, which can lead to heart failure, is associated with irreversible loss of functional cardiomyocytes. The loss of cardiomyocytes poses a major difficulty for treating the damaged heart since terminally differentiated cardiomyocytes have very limited regeneration potential. Currently, the only effective treatment for severe heart failure is heart transplantation but this option is limited by the acute shortage of donor hearts. The high incidence of heart diseases and the scarcity donor hearts underline the urgent need to find alternative therapeutic approaches for treating cardiovascular diseases.

Pluripotent embryonic stem (ES) cells can differentiate into functional cardiomyocytes. Therefore the engraftment of ES cell-derived functional cardiomyocytes or cardiac progenitor cells into the damaged heart to regenerate healthy myocardial tissues may be used to treat damaged hearts. Stem cell-based therapy therefore holds a great potential as a very attractive alternative to heart transplant for treating heart failure and other cardiovascular diseases. A major obstacle to the realisation of stem cell-based therapy is the lack of donor cells and this in turn is due to the fact that, currently, the molecular mechanisms or the regulatory signal transduction mechanisms that are responsible for mediating ES cell differentiation into cardiomyocytes are not well

i

understood. Overcoming this huge scientific challenge is absolutely necessary before the use of stem cell-derived cardiomyocytes to treat the damaged heart can become a reality.

Therefore the aim of this thesis was to investigate the signal transduction pathways that are involved in the differentiation of stem cells into cardiomyocytes. The first objective was the establishment and use of cardiomyocyte differentiation models using H9c2 cells and P19 stem cells to accomplish the specific objectives of the thesis. The specific objectives of the thesis were, the investigation of the roles of (i) nitric oxide (ii) protein kinase C (PKC), (iii) p38 mitogen-activated protein kinase (p38 MAPK) (vi) phosphoinositide 3-kinase (PI3K) and (vi) nuclear factor-kappa B (NF-kB) signalling pathways in the differentiation of stem cells to cardiomyocytes and, more importantly, to identify where possible any points of convergence and potential cross-talk between pathways that may be critical for differentiation to occur.

P19 cells were routinely cultured in alpha minimal essential medium (α-MEM) supplemented with 100 units/ml penicillin /100 µg/ml streptomycin and 10% foetal bovine serum (FBS). P19 cell differentiation was initiated by culturing the cells in microbiological plates in medium containing 0.8 % DMSO to form embryoid bodies (EB). This was followed by transfer of EBs to cell culture grade dishes after four days. H9c2 cells were cultured in Dulbecco's Modified Eagle's medium (DMEM) supplemented with 10% FBS. Differentiation was initiated by incubating the cells in

medium containing 1% FBS.

ii

In both models, when drugs were employed, they were added to cells for one hour prior to initiating differentiation. Cell monolayers were monitored daily over a period of 12 or 14 days. H9c2 cells were monitored for morphological changes and P19 cells were monitored for beating cardiomyocytes. Lysates were generated in parallel for western blot analysis of changes in cardiac myosin heavy chain (MHC), ventricular myosin chain light chain 1(MLC-1v) or troponin I (cTnI) using specific monoclonal antibodies.

H9c2 cells cultured in 1% serum underwent differentiation as shown by the timedependent formation of myotubes, accompanied by a parallel increase in expression of both MHC and MLC-1v. These changes were however not apparent until 4 to 6 days after growth arrest and increased with time, reaching a peak at day 12 to 14. P19 stem cells cultured in DMSO containing medium differentiated as shown by the timedependent appearance of beating cardiomyocytes and this was accompanied by the expression of cTnl.

The differentiation of both P19 stem cells and H9c2 into cardiomyocytes was blocked by the PI3K inhibitor LY294002, PKC inhibitor BIM-I and the p38 MAPK inhibitor SB2035800. However when LY294002, BIM-I or SB2035800 were added after the initiation of DMSO-induced P19 stem cell differentiation, each inhibitor failed to block the cell differentiation into beating cardiomyocytes. The NF-kB activation inhibitor, CAPE, blocked H9c2 cell differentiation into cardiomyocytes. Fast nitric oxide releasing donors (SIN-1 and NOC-5) markedly delayed the onset of differentiation of H9c2 cells into cardiomyocytes while slow nitric oxide releasing donors (SNAP and NOC-18) were less effective in delaying the onset of differentiation or long term differentiation of H9c2

iii

cells into cardiomyocytes. Akt (protein kinase B) is the key downstream target of PI3K. Our cross-talk data also showed that PKC inhibition and p38 MAPK inhibition respectively enhanced and reduced the activation of Akt, as determined by the phosphorylation of Akt at serine residue 473.

In conclusion, PKC, PI3K, p38 MAPK and NF-kB are relevant for the differentiation of stem cells into cardiomyocytes. Our data also show that the PKC, PI3K and p38 MAPK signalling pathways are activated as very early events during the differentiation of stem cells into cardiomyocytes. Our data also suggest that PKC may negatively regulate Akt activation while p38 MAPK inhibition inhibits Akt activation. Our fast NO releasing donor data suggest that nitric oxide may negatively regulate H9c2 cell differentiation.

#### ACKNOWLEDGMENT

My thanks and appreciation go to my principal supervisor Professor Anwar A Baydoun (University of Hertfordshire, UK) for giving me this unique opportunity to work in his laboratory and for the constant attention, guidance and help that I received from him in this project. My thanks also go to my second supervisor Professor John M Walker (University of Hertfordshire, UK) for all the help that he gave me.

I also extend my thanks to Professor Jeremy D Pearson (King's College London, UK) for his valuable suggestions and advice during the entire period of the project.

Thank you to Dr Oscar J Abilez (Stanford University, USA) for the valuable tips on P19 stem cell culture media formulation and differentiation.

Thanks you to my colleagues (Dr Shorifun N Thakur, Dr Arturo Sanz Sanz, Mrs Maria Alexandra Nunes Costa, Mr Edmund D Garr and Mrs Marzieh Zamani) for their every day help in the lab and also to all the Technicians in the Life Science Department.

My special thanks go to my family for their strong support and understanding.

I am also thankful to the University of Hertfordshire, Hatfield, Hertfordshire, UK for providing the studentship which completely covered the cost of the PhD project and all other expenses associated with the project.

## TABLE OF CONTENTS

ABSTRACT	i
ACKNOWLEDGMENT	v
CONTENTS	vi
LIST OF FIGURES	xvi
ABBREVIATIONS	xxiv

### **CHAPTER 1.0: INTRODUCTION**

1	INTRODUCTION	2
1.1	Definition of stem cells	2
1.2	Potential benefits of stem cell research	3
<b>1.3</b>	Classification of stem cells	<b>3</b>
1.3.1	Embryonic stem (ES) cells	4
1.3.2	Adult Stem Cells	8
1.4	Stem cell niche	8
1.5	Bone marrow stem cells	10
<b>1.6</b> 1.6.1 1.6.2	Potential advantages and disadvantages of ES and adult stem cells Embryonic stem cells Adult stem cells	<b>12</b> 12 14
1.7	Embryonic stem cell culture models	15
<b>1.8</b>	<b>Differences between human and mouse embryonic stem cells</b>	<b>16</b>
1.8.1	Differences in leukaemia inhibitory factor requirement	16

1.8.2	Differences in cell morphologies and growth kinetics	17
1.8.3	Differences in stem cell specific markers	17
1.9	Stem cell self-renewal and pluripotent state	19
1.9.1	Leukaemia inhibitory factor signalling and stem cell self-renewal	19
1.9.2	Leukaemia inhibitory factor signalling: STAT cascade	21
1.9.3	Leukaemia inhibitory factor signalling: ERK/MAPK cascade	22
1.9.3.1	Negative regulation of LIF signalling by SHP-2 and SOCS	25
1.10	Regulation of stem cell self-renewal and pluripotency	28
1.10.1	LIF signalling and pluripotent transcription factors	28
1.10.2	Octamer-binding protein -3/4	29
1.10.3	Nanog	30
1.10.4	Sox2	31
1.10.5	Cooperative Interaction of Oct-4, Sox2 and Nanog	31
1.11	Embryonic stem cell differentiation protocols	33
1.11.1	Embryoid body formation	33
1.11.2	Formation of EB by the Hanging Drops Method	34
1.11.3	Embryonic stem cell differentiation into cardiomyocytes	34

## CARDIAC TRANSCRIPTION FACTORS AND STEM CELL DIFFERENTIATION INTO CARDIOMYOCYTES

1.12	Cardiac transcription factors and stem cell differentiation into cardiomyocytes	37
1.13	GATA TRANSCRIPTION FACTORS	38
1.13.1	Roles of GATA transcription factors in cardiogenesis	41
1.13.2	Role of GATA-4 in cardiogenesis	41
1.13.3	Role of GATA-5 and GATA-6 in cardiogenesis	44
1.13.4	Cooperative Interaction between GATA-4 and GATA-6	45
1.13.5	Partners or Targets for GATA-4	46
1.13.6	GATA-4 and Upstream Protein Kinases	47
1.13.7	Summary for GATA transcription factors	48
1.14	CARDIAC HOMEOBOX TRANSCRIPTION FACTOR	49
1.14.1	Role of the transcription factor Nkx2.5 in cardiogenesis	50
1.14.2	Nkx2.5 role in stem cell differentiation into cardiomyocytes	53
1.14.3	Summary for the Roles of the Homeobox Nkx2.5	53

1.15	T-BOX TRANSCRIPTION FACTORS	54
1.15.1	Role of T-box transcription factors in cardiogenesis	54
1.15.2	Mechanisms of action of T-Box transcription factors	55
1.15.3	Roles of Tbx2 and Tbx5 in heart chamber development	55

#### EXTRACELLULAR SIGNALLING PROTEINS AND STEM DIFFERENTIATION INTO CARDIOMYOCYTES

1.16	ROLES OF EXTRACELLULAR SIGNALLING PROTEINS IN STEM DIFFERENTIATION INTO CARDIOMYOCYTES	58
1.17	BONE MORPHOGENETIC PROTEINS	59
1.17.1	BMP receptor Intracellular signalling and Smad proteins	60
1.17.2	Role of BMPs in stem cell differentiation into cardiomyocytes	63
1.17.3	Role of BMP and MAPK signalling in cardiomyocyte differentiation	64
1.17.4	Role of BMP and Smad Signalling in Cardiomyocyte	
	Differentiation	64
1.17.5	Crosstalk between Smad and MAPK signalling	65
1.17.6	Summary of bone morphogenetic proteins	65
1.18	Wnt Proteins	67
1.18.1	Wnt Signalling pathways	67
1.18.1.1	Wnt canonical (Wnt/β-catenin) pathway	68
1.18.1.2	Non-canonical Wnt/Ca2+ signalling pathway	71
1.18.1.3	Non-canonical Wnt/PCP signalling pathway	73
1.18.2	Classes of Wnt proteins	75
1.18.3	Wnt inhibitors	75
1.18.3.1	Endogenous Wnt inhibitors: secreted frizzled-related proteins	75
1.18.3.2	Endogenous Wnt inhibitors: Dickkopf (Dkk) proteins	76
1.18.4	Frizzled receptor family in the mesoderm	76
1.18.5	Roles of Wnt roles in cardiogenesis	77
1.18.6	Roles of the canonical Wnt/ $\beta$ -catenin signalling in	
	ES cell differentiation into cardiomyocyte	77
1.18.7	Summary of Wht Proteins	81

## 1.19Overview of extracellular signalling molecules and transcription factors in<br/>cardiogenesis and stem cell differentiation into cardiomyocytes82

#### MECHANISMS OF HEART DEVELOPMENT

1.20	Mechanisms of heart development	86
1.20.1	Origin of cardiac progenitor cells	87
1.20.1.1	First heart field (FHF)	87
1.20.1.2	Second heart field (SHF)	88
1.20.2	Parallel between heart formation and ES cell differentiation	
	into cardiomyocytes	89
1.20.3	Summary of mechanisms of heart development	92

#### STEM CELL RESEARCH AND CARDIOVASCULAR DISEASES

1.21	Stem cell research and cardiovascular diseases	94
1.21.1	Stem cell therapy for infarcted hearts	97
1.21.2	Challenges facing stem cell-based therapy	102
1.21.2.1	Challenge of cardiac-specific differentiation of stem cells	102
1.21.2.2	Challenge of post-transplantation tumour formation	103
1.21.2.3	Summary of Stem cell research and cardiovascular diseases	106
1.22	Aim of the PhD project	107

### **CHAPTER 2.0: MATERIALS AND METHODS**

2	Materials and Methods	111
2.1	Resuscitation and culture of H9c2 cells from frozen stock	111
2.2	Resuscitation and culture of P19 stem cell from frozen stock	112
2.3	Routine cell culture	113
2.3.1	Routine cell culture of H9c2 cells	113
2.3.2	Routine cell culture of P19 stem cells	113
2.4	Subculture of cells	114
2.5	Cryopreservation of cells	114
2.6	Quantification of cells	115
2.7	Cell viability (MTT) assay	118
2.8	Differentiation of H9c2 cells into cardiomyocytes	121
2.8.1	Initiation of H9c2 Cell differentiation into cardiomyocytes	121
2.8.2	Effects of drugs (signalling pathway inhibitors) on	
	the differentiation of H9c2 cells into cardiomyocytes	122

<b>2.9</b> 2.9.1 2.9.2	<b>Differentiation of P19 Stem cells into beating cardiomyocytes</b> Initiation of P19 cell differentiation into cardiomyocytes Effects of drugs (signalling pathway inhibitors) on P19 stem	<b>123</b> 123
C 2.9.3	cell differentiation into cardiomyocytes Effects of delayed inhibition of signalling pathways on P19 stem cell	124
	differentiation into beating cardiomyocytes	125
2.10	Cross-talks between PI3K signalling pathway and	100
2.10.1	Time course of Akt phosphorylation at serine 473 residue	128
2 10 2	in H9c2 cells differentiating into cardiomyocytes	128
2.10.2	phosphorylation at serine 473 residue in H9c2 cells	
	differentiating into cardiomyocytes	129
2.11	Total protein extraction for western blotting	131
<b>2.12</b> 2.12.1	Total protein determination in cell lysates using the BCA assay Preparation of bovine serum albumin (BSA) standards	132
<b>2.12</b> 2.12.1	Total protein determination in cell lysates using the BCA assay Preparation of bovine serum albumin (BSA) standards for the BCA assay	<b>132</b> 133
<ul><li><b>2.12</b></li><li>2.12.1</li><li>2.12.2</li></ul>	Total protein determination in cell lysates using the BCA assay Preparation of bovine serum albumin (BSA) standards for the BCA assay Procedure of the BCA assay	1 <b>32</b> 133 133
<ul><li>2.12</li><li>2.12.1</li><li>2.12.2</li><li>2.13</li></ul>	Total protein determination in cell lysates using the BCA assay Preparation of bovine serum albumin (BSA) standards for the BCA assay Procedure of the BCA assay Western blot analysis	132 133 133 133
<ul> <li>2.12</li> <li>2.12.1</li> <li>2.12.2</li> <li>2.13.1</li> <li>2.13.1</li> <li>2.13.2</li> </ul>	Total protein determination in cell lysates using the BCA assay Preparation of bovine serum albumin (BSA) standards for the BCA assay Procedure of the BCA assay Western blot analysis Sample preparation for western blot analysis Sodium dodecyl sulphate polyacrylamide	<b>132</b> 133 133 <b>137</b> 137
<ul> <li>2.12</li> <li>2.12.1</li> <li>2.12.2</li> <li>2.13.2</li> <li>2.13.1</li> <li>2.13.2</li> </ul>	Total protein determination in cell lysates using the BCA assay Preparation of bovine serum albumin (BSA) standards for the BCA assay Procedure of the BCA assay Western blot analysis Sample preparation for western blot analysis Sodium dodecyl sulphate polyacrylamide gel electrophoresis (SDS PAGE)	<b>132</b> 133 133 <b>137</b> 137 137
<ul> <li>2.12</li> <li>2.12.1</li> <li>2.12.2</li> <li>2.13</li> <li>2.13.1</li> <li>2.13.2</li> <li>2.13.3</li> </ul>	Total protein determination in cell lysates using the BCA assay Preparation of bovine serum albumin (BSA) standards for the BCA assay Procedure of the BCA assay Western blot analysis Sample preparation for western blot analysis Sodium dodecyl sulphate polyacrylamide gel electrophoresis (SDS PAGE) Transfer of proteins from gel to PVDF Membrane	<b>132</b> 133 133 <b>137</b> 137 137 139
<ul> <li>2.12</li> <li>2.12.1</li> <li>2.12.2</li> <li>2.13.1</li> <li>2.13.1</li> <li>2.13.2</li> <li>2.13.3</li> <li>2.13.4</li> </ul>	Total protein determination in cell lysates using the BCA assay Preparation of bovine serum albumin (BSA) standards for the BCA assay Procedure of the BCA assay Western blot analysis Sample preparation for western blot analysis Sodium dodecyl sulphate polyacrylamide gel electrophoresis (SDS PAGE) Transfer of proteins from gel to PVDF Membrane Blocking of the membrane	<b>132</b> 133 133 <b>137</b> 137 137 139 141
<ul> <li>2.12</li> <li>2.12.1</li> <li>2.12.2</li> <li>2.13</li> <li>2.13.1</li> <li>2.13.2</li> <li>2.13.3</li> <li>2.13.4</li> <li>2.13.5</li> </ul>	Total protein determination in cell lysates using the BCA assay Preparation of bovine serum albumin (BSA) standards for the BCA assay Procedure of the BCA assay Western blot analysis Sample preparation for western blot analysis Sodium dodecyl sulphate polyacrylamide gel electrophoresis (SDS PAGE) Transfer of proteins from gel to PVDF Membrane Blocking of the membrane Western blot antibody staining	<b>132</b> 133 133 <b>137</b> 137 137 139 141 141
<ul> <li>2.12</li> <li>2.12.1</li> <li>2.12.2</li> <li>2.13</li> <li>2.13.1</li> <li>2.13.2</li> <li>2.13.3</li> <li>2.13.4</li> <li>2.13.5</li> <li>2.13.6</li> </ul>	Total protein determination in cell lysates using the BCA assay Preparation of bovine serum albumin (BSA) standards for the BCA assay Procedure of the BCA assay Western blot analysis Sample preparation for western blot analysis Sodium dodecyl sulphate polyacrylamide gel electrophoresis (SDS PAGE) Transfer of proteins from gel to PVDF Membrane Blocking of the membrane Western blot antibody staining Detection of Protein Bands Using ECL Reagents	<b>132</b> 133 133 <b>137</b> 137 137 139 141 141 142
<ul> <li>2.12</li> <li>2.12.1</li> <li>2.12.2</li> <li>2.13.1</li> <li>2.13.2</li> <li>2.13.3</li> <li>2.13.4</li> <li>2.13.5</li> <li>2.13.6</li> <li>2.13.7</li> </ul>	Total protein determination in cell lysates using the BCA assay Preparation of bovine serum albumin (BSA) standards for the BCA assay Procedure of the BCA assay Western blot analysis Sample preparation for western blot analysis Sodium dodecyl sulphate polyacrylamide gel electrophoresis (SDS PAGE) Transfer of proteins from gel to PVDF Membrane Blocking of the membrane Western blot antibody staining Detection of Protein Bands Using ECL Reagents Quantification of protein bands	<b>132</b> 133 133 <b>137</b> 137 137 139 141 141 142 143
<ul> <li>2.12</li> <li>2.12.1</li> <li>2.12.2</li> <li>2.13.1</li> <li>2.13.2</li> <li>2.13.3</li> <li>2.13.4</li> <li>2.13.5</li> <li>2.13.6</li> <li>2.13.7</li> <li>2.14</li> </ul>	Total protein determination in cell lysates using the BCA assay Preparation of bovine serum albumin (BSA) standards for the BCA assay Procedure of the BCA assay Western blot analysis Sample preparation for western blot analysis Sodium dodecyl sulphate polyacrylamide gel electrophoresis (SDS PAGE) Transfer of proteins from gel to PVDF Membrane Blocking of the membrane Western blot antibody staining Detection of Protein Bands Using ECL Reagents Quantification of protein bands Data Analysis	13 13 13 13 13 13 13 14 14 14 14 14 14

## CHAPTER 3.0: ESTABLISHMENT OF CARDIAC DIFFERENTIATION MODELS

3.1	INTRODUCTION	145
3.1.1	H9c2 cell line model of cardiomyocyte differentiation	145
3.1.2	P19 cell line model of cardiomyocyte differentiation	146
3.2	MATERIALS AND METHODS	148
3.2.1	Culture and differentiation of H9c2 Cells	148
3.2.2	Culture and differentiation of P19 stem cells	148
3.2.3	Western blot analysis	149

3.4 3.5	DISCUSSION Summary	163 172
3.3.4	Differentiation of P19 stem cells into beating cardiomyocytes	161
3.3.3	Establishment and characterization of P19 stem cell differentiation model	158
3.3.2	Differentiation of H9c2 cells into cardiomyocytes	154
3.3.1	Establishment and characterization of H9c2 Cell differentiation model	150
3.3	RESULTS	150
3.2.4	Data analysis	149

## CHAPTER 4.0: ROLE OF NITRIC OXIDE IN CARDIOMYOCYTE DIFFERENTIATION

4.1	INTRODUCTION	175
4.1.1	Physiological functions of nitric oxide	175
4.1.2	Biosynthesis of nitric oxide	175
4.1.3	Molecular or physiological targets of nitric oxide	176
4.1.4	Nitric oxide and stem cell differentiation	177
4.2	MATERIALS AND METHODS	181
4.2.1	Culture and differentiation of H9c2 Cells	181
4.2.2	Cell viability assay	182
4.2.3	Western blot analysis	182
4.2.4	Data analysis	182

4.3	RESULTS	183
4.3.1	Effects of the NO donor NOC-5 on H9c2 cell differentiation into cardiomyocytes	183
4.3.2	Effects of the NO donor SIN-1 on H9c2 cell differentiation into cardiomyocytes	186
4.3.3	Effects of the NO donor NOC-18 on H9c2 cell differentiation into cardiomyocytes	189
4.3.4	Effects of the NO donor SNAP on H9c2 cell differentiation into cardiomyocytes	192
4.3.5	Effects of 8-Bromo-cGMP on H9c2 cell differentiation into cardiomyocytes	195
4.3.6	Viability of H9c2 cells in various concentration of NOC-5	198
4.3.7	H9c2 Cell Viability in Various concentration of SIN-1	199
4.3.8	H9c2 Cell Viability in Various concentration of NOC-18	200
4.3.9	H9c2 Cell Viability in Various concentration of SNAP	201

#### 4.4 DISCUSSION

4.5 Summary

## CHAPTER 5.0: ROLE OF PROTEIN KINASE C IN CARDIOMYOCYTE DIFFERENTIATION

5.1	INTRODUCTION	212
5.1.1	Structural features of protein kinase C isoforms	212
5.1.2	Generation of PKC second messengers	214
5.1.3	Differential requirement of second messengers by PKC isoforms	214
5.1.3.1	Conventional PKC (cPKC) isoforms	215
5.1.3.2	Novel PKC (nPKC) isoforms	215
5.1.3.3	Atypical PKC (aPKC) isoforms	215
5.1.4	PKC activation by phosphorylation,	216
5.1.5	PKC activation by second messengers	217
5.1.6	PKC isoforms and cell differentiation	218
5.1.7	PKC isoforms and stem cell differentiation	219
5.2	MATERIALS AND METHODS	220
5.2.1	Culture and differentiation of H9c2 Cells	220
5.2.2	Culture and differentiation of P19 stem cells	221
5.2.3	Cell viability assay	221
5.2.4	Western blot analysis	221
5.2.5	Data analysis	222
5.3	RESULTS	223
5.3.1	Effects of protein kinase C inhibition on H9c2 cell differentiation into	
	cardiomyocytes	223
5.3.2	Viability of H9c2 cells in $10\mu$ M of the protein kinase C	
	inhibitor, BIM-I	226
5.3.3	Effects of the protein kinase C inhibition on P19 stem	
	cell differentiation into beating cardiomyocytes	228
5.3.4	Effects of delayed inhibition of protein kinase C on P19 stem	000
	cell differentiation into beating cardiomyocytes	232
5.4	DISCUSSION	235
5.5	Summary	239

### CHAPTER 6.0: ROLE OF p38 MAP KINASE IN H9C2 AND P19 STEM CELL DIFFERENTIATION INTO CARDIOMYOCYTES

6.1	INTRODUCTION	242
6.1.1	Mitogen-activated protein kinase	242
6.1.2	Mitogen-activated protein kinase family	245
6.1.2.1	P38 MAPK Pathway	245
6.1.2.2	Extracellular signal-regulated kinase (ERK)	246
6.1.2.3	C-Jun N-terminal kinase/stress-activated protein kinase	247
6.1.2.4	MAPK and stem cell differentiation	248
6.2	MATERIALS AND METHODS	250
6.2.1	Culture and differentiation of H9c2 Cells	250
6.2.2	Culture and differentiation of P19 stem cells	250
6.2.3	Cell viability assay	251
6.2.4	Western blot analysis	251
6.2.5	Data analysis	252
6.3	RESULTS	253
6.3.1	Effects of the p38 MAPK inhibition on H9c2 cell differentiation into	
	cardiomyocytes	253
6.3.2	Viability of H9c2 cells in 10µM of the p38 MAPK inhibitor	
	SB203580	256
6.3.3	Effects of p38 MAPK inhibition on P19 stem cell differentiation	250
	Into cardiomyocytes	200
6.4	DISCUSSION	261
6.5	SUMMARY	265

### CHAPTER 7.0:ROLE OF PHOSPHOINOSITIDE 3-KINASE IN H9c2 AND P19 STEM CELL DIFFERENTIATION INTO CARDIOMYOCYTES

<b>7.1</b>	INTRODUCTION	<b>268</b>
7.1.1	Phosphoinositide 3-kinases	268
7.1.2	PI3K and stem cell differentiation into cardiomyocytes	271
<b>7.2</b>	MATERIALS AND METHODS	<b>272</b>
7.2.1	Culture and differentiation of H9c2 Cells	272
7.2.2	Cross-talk studies	273

7.2.3	Culture and differentiation of P19 stem cells	273
1.2.4	Cell viability assay	274
7.2.5	Western blot analysis	274
7.2.6	Data analysis	274
7.3	RESULTS	275
7.3.1	Effects of the PI3K Inhibition on H9c2 Cell Differentiation	
	into Cardiomyocytes	275
7.3.2	Effects of LY294002 (20 $\mu$ M) on the viability of H9c2 cells	278
7.3.3	Effects of PI3K inhibition on P19 stem cell differentiation	
	into beating cardiomyocytes	280
7.3.4	Effects of delayed inhibition of PI3K on P19 stem cell	
	differentiation into beating Cardiomyocytes	284
7.4	CROSS-TALK STUDIES	287
7.4.1	Time course of Akt phosphorylation following the	
	initiation of H9c2 cell differentiation into cardiomyocytes	
	by 1% serum	287
7.4.2	Effects of PI3K inhibition on Akt phosphorylation at	
	serine 473 in H9c2 cells differentiating into cardiomyocytes	289
7.4.3	Effects of protein kinase C inhibition on Akt phosphorylation	
	at serine 473 in H9c2 cell differentiating into cardiomvocytes	291
7.4.4	Effects of p38 MAPK inhibition on Akt phosphorylation	
	at serine 473 in H9c2 cell differentiating into cardiomyocytes	293
7.5	DISCUSSION	295
		200
7.6	SUMMARY	310

## CHAPTER 8.0: ROLE OF NF-kB IN H9C2 CELL DIFFERENTIATION INTO CARDIOMYOCYTES

<b>8.1</b>	<b>INTRODUCTION</b>	<b>314</b>
8.1.1	NF-κB signalling and stem cell differentiation	316
<b>8.2</b>	MATERIALS AND METHODS	<b>318</b>
8.2.1	Culture and differentiation of H9c2 Cells	318
8.2.2	Cell viability assay	319
8.2.3	Western blot analysis	319
8.2.4	Data analysis	319

8.3	RESULTS	320
8.3.1	differentiation into cardiomyocytes	320
8.3.2	Viability of H9c2 cells in 30µM of NF-kB activation inhibitor CAPE	323
8.4	DISCUSSION	325
8.5	SUMMARY	329

#### **CHAPTER 9.0: GENERAL DISCUSSION**

9.1	Establishment of cardiac differentiation models	331
9.2	Effects of nitric oxide donors on H9c2 differentiation into cardiomyocytes	336
9.3	Effects of PKC inhibition on H9c2 and P19 stem cell differentiation into cardiomyocytes	339
9.4	Effects of p38 MAPK inhibition on H9c2 and P19 stem cell differentiation into cardiomyocytes	341
9.5	Effects of the PI3K Inhibition on P19 stem and H9c2 Cell differentiation into cardiomyocytes	344
9.6	Effects of the inhibition of the activation of NF-kB on H9c2 cell differentiation into cardiomyocytes	351
9.7	CONCLUSION	353
FURT	HER WORK	355
REFERENCES		359

## LIST OF FIGURES

## **CHAPTER 1.0: INTRODUCTION**

Figure 1.10	Pluripotent embryonic cell lines from mammals	7
Figure 1.11	LIF signalling and stem cell self-renewal.	24
Figure 1.12	Negative regulation of LIF signalling by socs proteins.	27
Figure 1.13	Structural similarity among the GATA transcription factors.	40
Figure 1.14	BMP signalling pathway.	62
Figure 1.15	Activation of Wnt canonical pathway.	70
Figure 1.16	Activation of Wnt/ca2+ non-canonical pathway.	72
Figure 1.17	Activation of Wnt/PCP non-canonical pathway	74
Figure 1.18	Extracellular signalling molecules and transcription factor	84
	position in cardiogenesis.	

### **CHAPTER 2.0: MATERIALS AND METHODS**

Figure 2.10	A schematic diagram (cartoon) of a haemocytometer	117
Figure 2.11	A Representative BCA Standard Curve	136
Figure 2.12	Semi-dry transfer of protein from gel to PVDF	141

# CHAPTER 3.0: ESTABLISHMENT OF CARDIAC DIFFERENTIATION MODELS

Figure 3.10	Morphology of H9c2 cell (undifferentiated) in routine culture	152
Figure 3.11	Morphology of undifferentiated H9c2 cell in culture	153
Figure 3.12	Morphology of differentiated H9c2 cell in culture	154
Figure 3.13	Expression of myosin heavy chain in undifferentiated H9c2 cells	156
Figure 3.14	Changes in the expression of myosin heavy chain in	157
	differentiated H9c2 cells	
Figure 3.15	Changes in the expression of ventricular myosin chain light chain In differentiated H9c2 cells	158
Figure 3.16	Normal growth pattern of P19 embryonal carcinoma stem	160
	Cells in culture	
Figure 3.17	Formation of embryoid bodies from P19 embryonal carcinoma	161
	Stem cells in culture	
Figure 3.18	DMSO-induced differentiation of P19 stem cells into beating cardiomyocytes.	163

## CHAPTER 4.0: ROLE OF NITRIC OXIDE IN CARDIOMYOCYTE DIFFERENTIATION

Figure 4.10	Effects of the NO donor NOC-5 on H9c2 cell differentiation into cardiomyocytes: morphological changes. control H9c2 cells were cultured	185
Figure 4.11	Effects of the NO donor NOC-5 on H9c2 cell differentiation into cardiomyocytes: cardiac ventricular myosin chain light 1 expression	186
Figure 4.12	Effects of the NO donor SIN-1 on H9c2 cell differentiation into cardiomyocytes: morphological changes.	188
Figure 4.13	Effects of NO donor SIN-1 on H9c2 cell differentiation into cardiomyocytes: cardiac ventricular myosin chain light 1 expression	189
Figure 4.14	Effects of the NO donor NOC-18 on H9c2 cell differentiation into cardiomyocytes: morphological changes	191
Figure 4.15	Effects of the NO donor NOC-18 on H9c2 cell differentiation into cardiomyocytes: cardiac ventricular myosin chain light 1 expression	192
Figure 4.16	Effects of the NO donor SNAP on H9c2 cell differentiation into cardiomyocytes: morphological changes	194
Figure 4.17	Effects of the NO donor SNAP on H9c2 cell differentiation into cardiomyocytes: cardiac ventricular myosin chain light 1 expression	195
Figure 4.18	Effects of 8-bromo-cGMP on H9c2 cell differentiation into cardiomyocytes: morphological changes	197
Figure 4.19	Effects of 8-bromo-GMP on H9c2 cell differentiation into cardiomyocytes: ventricular myosin chain light 1 expression	198
Figure 4.20	Viability of H9c2 cells in the presence of various concentration of NOC-5	199
Figure 4.21	Viability of H9c2 cells in the presence of various concentrations of SIN-1	200

Figure 4.22	Viability of H9c2 cells in the presence of various	201
	concentration of NOC-18	
Figure 4.23	Viability of H9c2 cells in the presence of various	202
	concentration of SNAP	

#### CHAPTER 5.0: ROLE OF PROTEIN KINASE C IN CARDIOMYOCYTE DIFFERENTIATION

- **Figure 5.10** A schematic diagram showing the structure of conventional, 214 novel and atypical PKCs.
- Figure 5.11Effects of protein kinase C inhibition on H9c2 cell225differentiation into cardiomyocytes: morphological changes
- Figure 5.12 Effects of protein kinase C inhibition on H9c2 cell 226 differentiation into cardiomyocytes: cardiac ventricular myosin chain light 1 expression
- **Figure 5.13** Viability of H9c2 cells in the presence of 10µM of BIM-I 228
- **Figure 5.14** A schematic diagram of the effects of protein kinase C 230 Inhibition on DMSO-induced differentiation of P19 stem cells into cardiomyocytes.
- **Figure 5.15** Effects of protein kinase C inhibition on DMSO-induced 231 differentiation of P19 stem cells into cardiomyocytes
- Figure 5.16 Effects of protein kinase C inhibition on P19 stem cell 232 differentiation into cardiomyocytes: cardiac troponin I expression
- **Figure 5.17** A schematic diagram of the effects of delayed inhibition of 234 protein kinase C on P19 stem cell differentiation into beating cardiomyocytes.

Figure 5.18 Effects of delayed addition of the protein kinase C inhibition 235

on P19 stem cell differentiation into beating cardiomyocytes

## CHAPTER 6: ROLE OF p38 MAP KINASE IN H9C2 AND P19 STEM CELL DIFFERENTIATION INTO CARDIOMYOCYTES

Figure 6.10	MAPK signalling pathway architecture.	245
Figure 6.11	Effects of p38 MAPK inhibition on H9c2 cell differentiation into Cardiomyocytes: Morphological changes.	255
Figure 6.12	Effects of p38 MAPK inhibition on H9c2 cell differentiation into Cardiomyocytes: cardiac ventricular myosin light 1 expression.	256
Figure 6.13	Viability of H9c2 Cells in the presence of the p38 MAPK inhibitor, SB203580.	258
Figure 6.14	A schematic diagram of the effects of p38 MAPK inhibition on P19 stem cell differentiation into cardiomyocytes	260
Figure 6.15	A schematic diagram of the effects of delayed inhibition of p38 MAPK on P19 stem cell differentiation into cardiomyocytes.	260
Figure 6.16	Effects of p38 MAPK inhibition on P19 stem cell differentiation into cardiomyocytes: troponin I expression.	261

### CHAPTER 7.0: ROLE OF PHOSPHOINOSITIDE 3-KINASE IN H9C2 AND P19 STEM CELL DIFFERENTIATION INTO CARDIOMYOCYTES

Figure 7.10	Effects of the PI3K inhibition on H9c2 cell differentiation into cardiomyocytes: morphological changes	277
Figure 7.11	Effects of the PI3K inhibition on H9c2 cell differentiation into cardiomyocytes: cardiac ventricular myosin chain light 1 expression	278
Figure 7.12	The viability of H9c2 cells in the presence of the PI3K inhibitor, LY294002	280
Figure 7.13	A schematic disgram of the effects of the PI3K inhibition on DMSO-induced differentiation of P19 stem cells into cardiomyocytes.(schematic )	282
Figure 7.14	Effects of PI3K inhibition on DMSO-induced differentiation of P19 stem cells into beating cardiomyocytes	283
Figure 7.15	Effects of PI3K inhibition on P19 stem cell differentiation into cardiomyocytes: cardiac troponin I expression	284
Figure 7.16	A schematic diagram of the effects of delayed inhibition of PI3K on P19 stem cell differentiation into beating Cardiomyocytes.	286
Figure 7.17	Effects of delayed inhibition of PI3K on P19 stem cell differentiation into beating cardiomyocytes	287
Figure 7.18	Time course of Akt phosphorylation following the initiation of H9c2 cell differentiation into cardiomyocytes by 1% serum	289
Figure 7.19	Effects of the PI3K inhibitor LY29402 on Akt-ser473 phosphorylation in H9c2 cells differentiating into cardiomyocytes	291
Figure 7.20	Effects of protein kinase C inhibition on Akt phosphorylation at serine 473 residue in H9c2 cells differentiating into cardiomyocytes	293
Figure 7.21	Effects of protein p38 MAPK inhibition on Akt phosphorylation at serine 473 residue in H9c2 cells differentiating into cardiomyocytes	295
Figure 7.22	PI3K and PKC regulate stem cell differentiation into cardiomyocytes through possible Akt-independent	308

mechanisms

*Figure 7.23* PI3K and p38 MAPK regulate stem cell differentiation into 310 cardiomyocytes through possible Akt-independent mechanisms

### CHAPTER 8.0: ROLE OF NF-kB IN H9C2 CELL DIFFERENTIATION INTO CARDIOMYOCYTES

Figure 8.10	Effects of the inhibition of NF-kB activation on cardiomyocyte differentiation: morphological changes	322
Figure 8.11	Effects of the inhibition of NF-kB activation on cardiomyocyte differentiation: ventricular myosin chain light 1expression	323
Figure 8.12	The viability of H9c2 cells in the presence of cape.	325

#### **CHAPTER 9.0: GENERAL DISCUSSION**

**Figure 9.10** PI3K, PKC and p38 MAPK regulate stem cell differentiation 351 into cardiomyocytes through possible Akt-independent mechanisms.

APPENDIX	430
APPENDIX I: CELL LINES	430
Appendix I (a):H9c2 (2-1) Cell Line	430
Appendix I (b):P19 Embryonal Carcinoma Stem Cell Line	430
APPENDIX II: CELL CULTURE REAGENTS	431
APPENDIX III: NITRIC OXIDE DONORS AND 8-BROMO-CGMP	432
Appendix III (a): NOC-5	432
Appendix III (b): NOC-18	433
Appendix III (c): SNAP	433
Appendix III (d) : SIN-1, Hydrochloride	434
Appendix III (e): Guanosine 3',5'-cyclic Monophosphate	434
APPENDIX VI: INHIBITORS	435
Appendix VI (a): BisindolyImaleimide I	435
Appendix VI (b): LY294002	435
Appendix VI (c): SB203580	436
Appendix VI (d): CAPE	436
APPENDIX VII: ANTIBODIES	437
Appendix VII (a): Monoclonal Anti-Cardiac Myosin Light Chain I Antibody	
(MLC-1v) Antibody	437
Appendix VII (b): Monoclonal Anti-Cardiac Myosin Heavy Chain	
(α & β isoforms) Antibody	438
Appendix VII (c): Monoclonal Anti-cardiac Troponin I Antibody	439
Appendix VII (d): Monoclonal Anti- phospho-Akt (Ser473) Antibody Appendix VII (e): Monoclonal Anti-β-Actin antibody conjugated to	440
horseradish peroxidise (HRP)	441
APPENDIX VIII: PREPARATION OF STOCK SOLUTIONS	442
Calculation of mole of a compound	442
Calculation of required volume for dissolving a compound	442
An Example Calculation	443

## ABBREVIATIONS

ΑΚΤ	Protein kinase B
ASK	apoptosis stimulating kinase
APS	Ammonium persulfate
ATF2	Activating Transcription factor 2
ATRA	All- <i>trans</i> retinoic acid
BCA	Bicinchoninic acid
ВМР	Bone Morphogenetic Protein
BMPR	Bone Morphogenetic Protein Receptor
BSA	Bovine serum albumin
CO <sub>2</sub>	Carbon dioxide
CRD	Cysteine-rich domain
DAG	1, 2-diacyglycerol
DEA/NO	2-(N, N-diethylamino)-diazenolate-2-oxide
DLK	Dual leucine zipper bearing kinase.
DMEM	Dulbecco's Modified Eagle's Medium
DMSO	Dimethylsulfoxide
EB	Embryoid body
ECL	Enhanced chemiluminescence
ELK-1	E-26-like protein 1
ERK	Extracellular signal-regulated kinase

ES cell	Embryonic Stem Cell
FBS	Foetal Bovine Serum
Grb2	Growth factor receptor-bound protein 2
Gab1	Grb2-associated binding protein 1
FOG	Friend Of GATA
ICM	Inner cell mass
IP3	Inositol (1,4,5)tris-phosphate
JAK	Janus Kinase
JNK	c-Jun N-terminal kinase
LIF	Leukaemia Inhibitory Factor
LIFRβ	Leukaemia Inhibitory Factor Receptor
MADS	An acronym for MCM1, Agamous, Deficiens SRF
МАРК	Mitogen activated protein kinase
МАРКК	MAP kinase kinase
МАРККК	MAP kinase kinase
МАРКККК	MAP kinase kinase kinase
МЕК	Mitogen-activated Protein/Extracellular Signal-regulated Kinase Kinase
MCM1	Minichromosomal Maintenance protein 1
MSC	Mesenchymal stem cells
MEF	Mouse Embryonic Fibroblast
MEF2	Myocyte Enhancer Factor 2
MEF2A	Myocyte Enhancer Factor 2 A

MEF2C	Myocyte Enhancer Factor 2 C
MEF2D	Myocyte Enhancer Factor 2 D
MEKK 1/4	Mitogen ERK kinase kinase, 1/4
MLK3	Mixed lineage kinase 3
MNK1	MAPK signal-activating kinase 1
MSK	MAPK/SAPK-activated kinase
МТТ	(3-[4, 5-dimethylthiazol-2-yl] 2, 5-diphenyl-tetrazolium bromide
Nanog	Tir <i>nan Og</i> , a mythical celtic land of the ever young
NF-AT	Nuclear Factor of Activated T cells
NF-kB	Nuclear Factor -kappa B
NO	Nitric oxide
NOC-18	2, 2'-(Hydroxynitrosohydrazino)bis-ethanamine
NOC-5	3-[2-Hydroxy-1-(1-methylethyl)-2-nitrosohydrazino]-1-propanamine
р38 МАРК	p38 Mitogen activated protein kinase
РМА	Phorbol 12-myristate-13-acetate
PAPA/NO	(Z)-1-[N-(3-ammoniopropyl)-N-(n-propyl)amino]diazen-1-ium-1,2-diolate
PBS	Phosphate buffered saline
PDK1	3-Phosphoinositide-Dependent Protein Kinase 1
PDK2	3-Phosphoinositide-Dependent Protein Kinase 2
РІЗК	Phosphatidylinositol-3 Phosphate Kinase
PIP2	Phosphatidylinositol (1, 4) bis-phosphate
РКВ	Protein Kinase B

РКС	Protein Kinase C
PVDF	Polyvinylidene fluoride
RPM	Revolutions per minutes
SAPK	Stress activated protein kinase
SDS	Sodium dodecyl sulphate
SDS PAGE	Sodium dodecyl sulphate polyacrylamide gel electrophoresis
SH-2	Src Homology Domain 2
SHP2	Sh-2-Containing Tyrosine Phosphatase
SIN-1	3-Morpholinosydnonimine
SMAD	Similar To Mothers Against Decapentaplegic
SNAP	(±)-S-Nitroso-N-acetylpenicillamine
SOCS	Suppressors Of Cytokine Signalling
SOS	Son Of Sevenless
SSEA	Stage specific embryonic antigen
STAT	Signal Transducers And Activators Of Transcription
ТАК	transforming growth factor- $\beta$ -activated kinase
TRA-1-60	
TRA-1-81	Battle of <u>Tra</u> falgar (i.e. named after the Battle of <u>Tra</u> falgar)
TBS	Tris buffered saline
TEMED	N,N,N',N'-tetramethylethylenediamine

## CHAPTER 1.0 INTRODUCTION

#### **1** INTRODUCTION

#### 1.1 Definition of stem cells

A stem cell is an unspecialised or undifferentiated cell from the embryo (i.e. embryonic stem cell) or from adult or foetus tissue (i.e. adult stem cell). An embryonic stem cell has two important fundamental properties: unlimited ability to divide and generate new stem cells (i.e. self-renewal) without going into senescence and under appropriate physiological or experimental conditions, can specialise or differentiate into other cell types (Evans and Kaufman, 1981; Martin, 1981; Thomson et al., 1998). Adult stem cells, like ES cells, are capable of self-renewal and can differentiate into various cell types. However, unlike ES cells, adult stem cells (at most) are only multipotent (i.e. are not pluripotent) and also have limited self-renewal ability. Self-renewal and pluripotency (multipotency for adult stem cells) allow stem cells to serve as building blocks in developmental processes in human and animals. In the adult human or animal, stem cells also maintain tissue homeostasis by providing a continuous supply of new cell types to replace highly differentiated cells that have become damaged or simply turn over as part of the normal physiology of organs or tissues regeneration. Embryonic stem cells are the only non-transformed (i.e. stable diploid karyotype) mammalian cells that can be continuously propagated in vitro.

#### 1.2 Potential benefits of stem cell research

In the future, as our understanding of the essential properties and behaviour of stem cells increase, it may be possible to use them for cell-based therapies to treat human diseases (e.g. diabetes and heart diseases), to have more understanding of human development and developmental-related diseases (such as birth defects), the aging process and aging-related diseases (e.g. Parkinson's disease). Stem cell research will also allow for the use of stem cell-derived specialised cell types for the screening of new drugs and other chemical compounds. Stem cell research therefore has huge scientific and medical importance.

#### 1.3 Classification of stem cells

On the basis of their origin, human and animal stem cells can be classified into two categories: somatic or adult stem cells and embryonic stem cells. Three types of mammalian embryonic stem cells have been derived: embryonic stem (ES) cells (Evans and Kaufman, 1981; Nagy *et al.*, 1990; Brook and Gardner, 1997; Thomson *et al.*, 1998), embryonic carcinoma (EC) Cells (Martin, 1981; McBurney and Rogers, 1982) and embryonic germ (EG) cells (Shamblott *et al.*, 1998; Turnpenny *et al.*, 2003; Liu *et al.*, 2004) (Figure 1.10) On the basis of their developmental or differentiation potential ES cells can be further subdivided into the following types: totipotent, pluripotent, multipotent and unipotent stem cells. Totipotent stem cells (i.e. the zygote) can differentiate into all cell types of body. The three types of embryonic stem

cells (ES, EC and EG) originate from the zygote (Figure 1.10). Pluripotent stem cells (i.e. cells of the inner cell mass at the blastocyst stage) can differentiate into any cell type of the three primary germ layers i.e. endoderm (e.g. pancreatic cells, thyroid cells and lung alveolar cells), mesoderm (e.g. cardiac muscle cells, skeletal muscle cells, kidney tubule cells) and the ectoderm (e.g. epidermal tissues, neuron and pigment cells). Multipotent stem cells (e.g. haematopoietic cell) can differentiate into several but not all cell types. Unipotent stem cells (e.g. spermatogenic stem cells) give rise to only one cell type.

#### 1.3.1 Embryonic stem (ES) cells

Embryonic stem cells are pluripotent cells derived from the inner cell mass of embryos at the blastocyst stage (3 to 5-day old embryo) (Evans and Kaufman, 1981; Martin, 1981; Doetschman *et al.*, 1985; Nagy *et al.*, 1990; Brook and Gardner, 1997; Thomson *et al.*, 1998) (Figure 1.10). When returned to the embryonic environment by transfer into a host blastocyst or aggregation with blastomere-stage embryos, cultivated ES cells reincorporate into normal embryonic development and have the full potential to develop into all lineages of the embryo proper (Bradley *et al.*, 1984; Nagy *et al.*, 1990).

The blastocyst includes three structures: the trophoblast, blastocoel and inner cell mass (ICM). The trophoblast is the layer of cells that surrounds the blastocyst. The blastocoel is the hollow cavity inside the blastocyst. The inner cell mass is a group of cells (approximately 30 cells) at one end of the

blastocoels (Figure 1.10). The ICM can be grown on mitotically arrested (mitomycin C treated or gamma irradiated) adherent mouse embryonic fibroblasts (MEF) or feeder layer as it is referred to. Both mitomycin-C and gamma irradiation induce DNA damage and cause cell cycle. Mitomycin-C binds to DNA and causes DNA cross-linking (Rink et al., 1996; Volpato et al., 2005; Nieto et al., 2007). This interferes with DNA replication, RNA transcription and protein synthesis. Gamma irradiation treatment induces DNA damage irradiation (Bishay et al., 2000; Ibuki et al., 2003). Therefore the treatment of cells with either mitomycin C or gamma arrests the cell growth or proliferation. The feeder layer cells provide a sticky surface for the ES cells and also release nutrients into the culture medium. In developing tissues, a stem cell from the blastocyst is capable of giving rise to the multiple specialised cell types. ES cells have been differentiated into various cell types including cardiomyocytes (Doetschman et al., 1985; Kehat et al., 2001), skeletal and smooth muscle cells (Rohwedel et al., 1994; Drab et al., 1997), neuronal cells (Bain et al., 1995; Bruneau et al., 1999; Reubinoff et al., 2001; Zhang et al., 2001), β islet pancreatic cells (Assady et al., 2001), dipocytes (Dani et al., 1997), endothelial cells (Levenberg et al., 2002) and several other tissue types.

The other types of embryonic stem cells are embryonic (embryonal) carcinoma cells and embryonic germ (EG) cells. Embryonal carcinoma (EC), like ES, are fully pluripotent cells derived from malignant teratocarcinomas rather than, directly from , the ICM (Martin, 1981; McBurney and Rogers, 1982) (Figure 1.10). Embryonic germ (EG) cells are also pluripotent and are

derived from primordial germ cells of foetal gonadal or genital ridges (Shamblott *et al.*, 1998; Turnpenny *et al.*, 2003; Liu *et al.*, 2004) (Figure 1.10).

Mouse EG cells, like the ES cells, are cultured on MEF or in the presence of leukaemia inhibitory factor (LIF). Leukaemia inhibitory factor is a differentiation inhibitor that keeps stem cells in the undifferentiated state. In contrast to both ES and EG cells that require culture on embryonic fibroblast feeder layer cells or in the presence of LIF, EC cells (e.g. P19 cell line) are embryonic fibroblast feeder-cell independent (Bradley *et al.*, 1984).



*Figure 1.10: Pluripotent embryonic cell lines from mammals*: Embryonic carcinoma (EC) cells, embryonic stem (ES) cells and embryonic germ (EG) cells. The ES cells are derived from the ICM and epiblast. EC cells are derived from germ cell tumours or teratocarcinomas. EG cell lines are derived from primordial germ cells of gondal or genital ridge of embryos. Taken from Boheler *et al.*, 2002.

#### 1.3.2 Adult Stem Cells

Adult stem cells are undifferentiated cells found among differentiated cells in tissues. They can be isolated from foetal (after gastrulation) or adult tissues. Amniotic fluid stem cells are adult stem cells from foetal tissues (Perin *et al.*, 2008). Unlike ES cells which are pluripotent, adult stem cells are only multipotent and are generally believed to be programmed to differentiate into specialised cell types of the tissues or organs in which they reside. The primary role of adult stem cells in a living organism is to maintain tissue homeostasis and repair. They maintain homeostasis by generating new cells to replace terminally differentiated cells that are no longer functional. Adult stem cells have been isolated from a wide range of tissues or organs including the heart (Laugwitz *et al.*, 2005; Moretti *et al.*, 2007), bone marrow (Pittenger *et al.*, 1999; Reyes *et al.*, 2001), brain (Clarke *et al.*, 2000), liver (Yang *et al.*, 2002), skin (Toma *et al.*, 2001), adipose (Zuk *et al.*, 2002), skeletal muscle (Jackson *et al.*, 1999) and blood (Zhao *et al.*, 2003).

#### 1.4 Stem cell niche

The decision between stem cell self-renewal and differentiation is thought to be a highly regulated process under the control of molecular and cellular factors. This *in vivo* regulatory microenvironment is known as a stem cell niche. Stem cell number, cell division, self-renewal and differentiation are likely to be controlled by both extrinsic factors and intrinsic factors. Extrinsic factors, including Wnt (Lam *et al.*, 2006; Aicher *et al.*, 2008; Fleming *et al.*, 2008), BMP (Kosinski *et al.*, 2007), Notch (Alexson *et al.*, 2006; Song *et al.*,
2007) and intrinsic factors, including upstream controllers such as the unpaired (upd) gene (Kiger et al., 2001; Tulina and Matunis, 2001; Boyle et al., 2007) and transcription factors, such as Early Growth Response 1(EGR1) (Min et al., 2008), Pharynx and Intestine in Excess (PIE-1) (Mello et al., 1996; Seydoux et al., 1996) play roles in stem cell niche. Stem cell niches have been proposed to exist in several adult tissues, including the haematopoietic system (Calvi et al., 2003; Zhang et al., 2003b; Arai et al., 2004), neural tissues (Shen et al., 2004) and in epithelial tissues including the qut epithelium (He et al., 2004) and the skin (Tumbar et al., 2004). In vivo, the stem cell niche favours the maintenance of the stem cells in the undifferentiated state. The ultimate decision of a stem cell to remain in the undifferentiated state or to differentiate into specific cell types may depend upon the expression of several regulatory molecules, including transcription factors within the niche. The expression of these regulatory molecules may be induced either by intrinsic or extrinsic events or by both. A clear understanding of how the decision between stem cell self-renewal and differentiation is made will make it easier to culture adult stem cells in the differentiated state and also to direct their differentiation into specific cell types using the appropriate conditions. This clear understanding may be part of the first steps in the direction toward the realisation of the huge therapeutic potentials of adult stem cells.

#### 1.5 Bone marrow stem cells

The bone marrow (BM) is a good source of adult stem cells. The bone marrow contains at least two kinds of stem cells: haematopoietic stem cells (HSC) and mesenchymal stem cells (MSC). The HSC give rise to all the blood cells: red blood cells, B lymphocytes, T lymphocytes, natural killer cells, neutrophils, basophils, eosinophils, monocytes, macrophages, and platelets (Spangrude et al., 1988; Okada et al., 1991; Bhatia et al., 1997; Yang et al., 2005). Mesenchymal stem cells have the potential to differentiate into multiple lineages but they are rare (i.e. 1 in 10,000 of nucleated BM cells). Mesenchymal stem cells express several markers but none of the markers expressed are specific to their identification. The identification of MSC is therefore based on a composite marker expression and a lack or absence of haematopoietic stem cell marker expression such as CD45 (Lu et al., 2005; Polisetty et al., 2008), CD34 (Conrad et al., 2002; Polisetty et al., 2008), CD14 (Lu et al., 2005) or CD11 (Polisetty et al., 2008). Mesenchymal stem cells also do not express some co-stimulatory molecules such as CD80 (Sun et al., 2003; Lu et al., 2005), CD86 (Lu et al., 2005; Liu et al., 2006) or CD40 (Liu et al., 2006). Human MSC can however express CD105 (Barry et al., 1999; Conrad et al., 2002; Roura et al., 2006; Odabas et al., 2008), CD73 (Barry et al., 2001; Trivedi and Hematti, 2007; Odabas et al., 2008), CD44 (Oswald et al., 2004; Yanada et al., 2006; Zhu et al., 2006), CD90 (Oswald et al., 2004; Wiesmann et al., 2006) and some adhesion molecules (e.g. CD106 (Polisetty et al., 2008), CD166 and CD29 (Oswald et al., 2004; Polisetty et al., 2008)).

Mesenchymal stem cells have the potential to contribute to the regeneration of various mesenchymal tissues including bone (osteocytes), cartilage cells (chondrocytes) (Worster et al., 2001; Bai et al., 2004), ligament, tendon, fat (adipocytes) (Scavo et al., 2004) and smooth muscles (Jeon et al., 2006; Wang et al., 2006). In addition to their differentiation into mesenchymal tissues, recent research suggests that MSC can also differentiate into cardiomyocytes (Hakuno et al., 2002; Li et al., 2006; Wang et al., 2006; Tomita et al., 2007; Muscari et al., 2008) and this can be achieved 5-azacytidine (Fukuda, 2001; Xu et al., 2004b; experimentally using Antonitsis et al., 2007). 5-azacytidine is a nucleoside analog of cytidine. Carbon 5 on the pyrimidine ring of 5-azacytidine is replaced by a nitrogen atom. Therefore when incorporated into DNA, 5-azacytidine cannot be methylated thus leaving the new synthesized genomic DNA largely in the unmethylated state (Jones and Taylor, 1980; Taylor and Jones, 1982). DNA methylation is one of the key mechanisms of gene regulation. The incorporation of 5-azacytidine into DNA can activate gene expression and also alter the differentiation of some eukaryotic cells (Jones and Taylor, 1980; Jones et al., 1983). The successful cardiac differentiation of the MSC was noted by expression of mRNA for Nkx2.5, GATA-4, TEF-1, and MEF2C (Myocyte Enhancer Factor 2 C) before and further expression of MEF2A and MEF2D after 5-azacytidine treatment. Cardiac proteins including atrial natriuretic peptide (ANP) were also detected (Fukuda, 2001). The analysis of their contractile protein, such as myosin and  $\alpha$ -actin, indicated that their phenotype was similar to foetal ventricular cardiomyocytes (Fukuda, 2001).

The differentiation of MSC into cardiomyocytes as determined by the expression of  $\alpha$ -cardiac actin,  $\beta$ -myosin heavy chain and cardiac troponin-T by human MSC has also been demonstrated (Xu et al., 2004b; Antonitsis et al., 2007). An important observation in these studies is that both 5-azacytidinetreated cells and non-treated cells expressed cardiac specific genes (Fukuda, 2001; Antonitsis et al., 2007). It was further shown that cardiac specific genes were not expressed in undifferentiated MSC that were not being expanded in culture (Antonitsis et al., 2007). This suggests that while MSC can differentiate into cardiomyocytes, they do not do so randomly or spontaneously but only in the presence inducing factors (e.g. 5-azacytidine or conditions such as being expanded in culture). While the physiological stimulus that drives MSC differentiation into cardiomyocytes is not yet known, these results raise a possibility that MSC (or cardiomyocytes derived from bone marrow sources) can potentially be valuable for the treatment of myocardial infarction. A better understanding of the MSC differentiation into cardiomyocytes may allow the use of MSC-derived cardiomyocytes to replace cardiomyocytes in the damaged heart.

# 1.6 Potential advantages and disadvantages of ES and adult stem cells

# 1.6.1 Embryonic stem cells

Embryonic stem cells are obtained from embryos and can be expanded in culture for a very long time. Pluripotent ES cells can undergo flexible differentiation into many different cell types. However, the pluripotent nature of

ES cells means that they are difficult to differentiate into a uniform or homogeneous population of one cell type. Due to their pluripotent capacity, undifferentiated ES cells transplanted into target tissues may form teratomas. Apart from the risks of tumour formation, ES cell-derived target cell types may also cause an immune reaction in the recipient, thus, causing the rejection of transplanted cells. Apart from the above scientific issues with ES cells that have to be resolved, there is also an ethical dimension to human ES cell research and potential future use. Currently available stem cell technologies, for the derivation of pluripotent human ES cells, require the "destruction" of human embryos at the blastocyst stage (Thomson et al., 1998). Therefore for some people, ES cell research and potential use may pose ethical problems. However the creation of induced pluripotent stem cells (iPS) from adult human cells (Takahashi et al., 2007; Yu et al., 2007) may offer a viable alternative to the human embryo as a source for pluripotent stem cells. Octamer-binding protein (Oct-3/4), a transcription and pluripotency marker, is expressed in embryonal carcinoma cells, embryonic stem cells and embryonic germ cells (Thomson et al., 1998; Hansis et al., 2000; Reubinoff et al., 2000). Human amniotic fluid stem cells have also been shown to express Oct-3/4. Human amniotic fluid stem cells can be obtained without the destruction of human embryos (Prusa et al., 2003; De Coppi et al., 2007; Perin et al., 2008; Zheng et al., 2008b). In the future iPS and adult stem cells, like amniotic fluid stem cells with higher pluripotency, may offer viable alternatives to the human embryo as a source for pluripotent stem cells.

#### 1.6.2 Adult stem cells

Adult stem cells are rare in mature tissues and methods for harvesting pure adult stem cells and growing them culture are difficult. Therefore adult stem cells unlike ES are relatively more difficult to obtain in large numbers.

For research purposes, the need for a pure population of adult stem cells is a critical consideration because the reliability of experimental results from adult stem studies may be compromised by the presence of other cell type in the cell culture or stem cell preparation from adult tissues. The presence of non-stem cell populations in isolated adult stem cell population can lead to the wrong interpretation of results, especially if the contaminating cell type has or express the phenotypes of the final or target cell type for the differentiation studies.

Adult stem cells exist *in vivo* and this means that they must be able to selfrenew in order to be able to participate in the maintenance of long term tissue homeostasis in human and animals, but isolated adult stem cells cannot undergo long-term self-renewal *in vitro*. The renewal capacity of adult stem is therefore less than that of ES cells hence adult stem cells may not live as long as ES cells in culture. For future stem cell therapies, the rarity of adult stem cells and their low self-renewal capacity are very important considerations, since a large number of cells may be needed on a routine basis for patient treatment.

Adult stem cells are programmed to differentiate into cell types of the tissues in which they are found. However emerging evidence suggest that under appropriate conditions, animal and human adult stem cells can differentiate into cell types of tissues other than that of their origin. Human MSC have been differentiated into cardiomyocytes (Grauss et al., 2007), chondrocytes (Bai et al., 2004), adipocytes (Scavo et al., 2004) and smooth muscle (Jeon et al., 2006). Mouse MSC have been differentiated into cardiomyocytes (Kudo et al., 2003) and neurons (Tropel et al., 2006). Rat MSC have been differentiated into neurons (Lou, et al. 2003) and smooth muscle (Wang et al., 2006) and horse MSC have been differentiated into chondrocytes (Worster et al., 2001). However, generally, adult stem cells have less differentiation potentials then ES cells and may, therefore, be more difficult to differentiate into cell of other tissue other than the tissue of origin. A key potential advantage of using adult stem cells is that if the patient's own cells are expanded in culture and then re-introduced into the patient, the risk of immune rejection of the transplanted cells may be avoided. In terms of ethical considerations, adult stem cell research and their therapeutic use may be more acceptable than ES cells, since adult stem cells can be isolated without any potential harm to the donors.

# 1.7 Embryonic stem cell culture models

Embryonic stem cells are cultured at low density on a mitotically inactivated embryonic fibroblast feeder layer or in the presence of LIF or both. Under these culture conditions, the stem cells are prevented from undergoing

undirected spontaneous differentiation processes thus maintaining their stem cell pluripotency (Evans and Kaufman, 1981; Martin, 1981; Smith *et al.*, 1988; Williams *et al.*, 1988; Niwa *et al.*, 1998).

A huge amount of what we know about stem cell and stem differentiation comes from studies using mouse ES cell models. Experience with human ES cells is relatively new and limited, but comparison between human and mouse models using the limited data from human have revealed important differences between the two models. These differences include the cell culture requirements to keep the stem cells in the undifferentiated state, morphology, growth kinetics and antigenic phenotypes or stem cell specific molecular markers.

# 1.8 Differences between human and mouse embryonic stem cells

#### 1.8.1 Differences in leukaemia inhibitory factor requirement

The leukaemia inhibitory factor (LIF) (fully discussed below under LIF signalling) alone is sufficient, even in the absence of feeder layer cells, to maintain mouse stem cell cultures in the undifferentiated state. However LIF alone is not sufficient to prevent the differentiation of human (and other primates) ES cells (Schuringa *et al.*, 2002). This means that human ES cells, unlike mouse ES cells, are maintained in an undifferentiated state through a signalling pathway(s) that may be independent of or in addition to LIF signalling.

#### 1.8.2 Differences in cell morphologies and growth kinetics

Mouse ES cells grow in tightly attached rounded masses while human ES cells form relatively flat compact colonies with high nuclear to cytoplasm ratio and prominent nucleoli (Thomson *et al.*, 1998; Reubinoff *et al.*, 2000; Park *et al.*, 2004). Human colonies can be easily dissociated into single cells with conventional mechanical and enzymatic techniques. The population doubling time of human ES cells of approximately 36 hours (Hovatta *et al.*, 2003; Park *et al.*, 2004) is significantly longer than that of mouse ES cells (approximately 10-16 hours) (Viswanathan *et al.*, 2002; Berrill *et al.*, 2004; Fok and Zandstra, 2005; Greenlee *et al.*, 2005).

# 1.8.3 Differences in stem cell specific markers

In addition to morphological differences, human and mouse pluripotent stem cells differ in their expression of a number of antigenic phenotypes or stem cell specific molecular markers. Common stem cell markers that are used in characterisation of ES cells include the glycolipid surface Stage Specific Embryonic Antigens (SSEA-1, SSEA-3, SSEA-4) and the keratan sulfate-related antigens (TRA-1-60 and TRA-1-81) (Solter and Knowles, 1978; Gooi *et al.*, 1981; Ozawa *et al.*, 1985; Krupnick *et al.*, 1994; Draper *et al.*, 2002). The TRA antigens (or antibodies) were named after the Battle of <u>Tra</u>falgar. Undifferentiated primate (human and nonhuman primates) ES cells express the specific cell surface markers: SSEA-3, SSEA-4, TRA-I-60 and TRA-1-81 (Draper *et al.*, 2002; Henderson *et al.*, 2002). They also express alkaline phosphatase. However differentiated primate ES cells strongly expressed

SSEA-I (Krupnick et al., 1994; Thomson et al., 1995). Differences between mouse and human is shown by the fact that mouse inner cell mass cells, undifferentiated ES cells and EC cells express SSEA-1 but do not express SSEA-3, SSEA-4, TRA-1-60, or TRA-1-81, in contrast to undifferentiated primate ES cells. SSEA-1 appears during the late cleavage stages of mouse embryogenesis. SSEA-1 is strongly expressed by undifferentiated mouse ES cells (Solter and Knowles, 1978; Gooi et al., 1981; Ozawa et al., 1985; Krupnick et al., 1994). Differentiated murine ES cells are characterised by the loss of SSEA-1 expression (Cui et al., 2004). The loss of SSEA-1 expression may be accompanied in some instances by the appearance of SSEA-3 and SSEA-4. In contrast to undifferentiated mouse ES cells, undifferentiated human ES cells are characterised by the expression of SSEA3, SSEA4, TRA-1-60, and TRA-1-81, and by the lack of SSEA1 (Draper et al., 2002; Henderson et al., 2002). The differentiation of human ES and EC cells is characterised by down-regulation of SSEA-3 and SSEA-4 (Draper et al., 2002). In contrast to the differential expression of the above stem cell markers, both mouse and human embryonic stem cell express octamerbinding transcription factor-3/4 (Oct-3/4) (Thomson et al., 1998; Hansis et al., 2000; Reubinoff et al., 2000), Nanog (see section 1.10.3) (Chambers et al., 2003; Mitsui et al., 2003; Hart et al., 2004; Sato et al., 2004) and have high telomerase activity (Thomson et al., 1998; Amit et al., 2000; Armstrong et al., 2000; Yang et al., 2008). Stem cell markers are proteins found in stem cells (and rarely in other cells) and as such they have specific physiological functions. For example, Oct-3/4 is a transcription factor that is expressed in totipotent and pluripotent stem cells of the pregastrulation embryo and is

required to sustain stem cell self-renewal and pluripotency (Rosner *et al.*, 1990; Palmieri *et al.*, 1994; Nichols, 1998; Niwa *et al.*, 2000). While the physiological functions of many stem cell markers are not currently known, stem cell markers are important research tools for the identification of undifferentiated stem cells and the characterization of differentiated cell types.

# 1.9 Stem cell self-renewal and pluripotent state

While research effort has been directed toward clearly understanding the mechanisms that are involved in stem cell differentiation into various cell types, it is also equally important to understand the regulatory mechanisms that are responsible for ES cell self-renewal and pluripotency. A clear understanding of the key self-renewal/pluripotency regulatory mechanisms can lead to better differentiation protocols and this may lead to the realisation of ES cells full therapeutic potentials. While intracellular signalling events that are involved in the maintenance of ES cell self-renewal and pluripotency are complex and not yet fully understood, research data point to the involvement of various extrinsic factors (e.g. LIF, BMP, Wnt) and key ES cell specific factors (e.g. Nanog, Oct-4, Sox2).

## 1.9.1 Leukaemia inhibitory factor signalling and stem cell self-renewal

In cell culture medium, pluripotent ES cells undergo spontaneous differentiation to generate the three germ layers: endoderm, mesoderm and ectoderm leading to a mixed population of cell lineages and cell types (Amit

*et al.*, 2000; Schuldiner *et al.*, 2000). The presence of leukaemia inhibitory factor (LIF) or closely related cytokines is required in culture, in order to prevent this spontaneous differentiation (Smith and Hooper, 1987; Smith *et al.*, 1988; Williams *et al.*, 1988; Niwa *et al.*, 1998). LIF, a multifunctional cytokine of the interleukin-6 (IL-6) cytokine family, sustains ES cell self-renewal and pluripotency by signalling mechanisms that are not yet fully understood.

Leukaemia inhibitory factor acts on a cell surface receptor complex that is composed of two chains [(i.e. an 80-kDa LIF-binding  $\alpha$ -chain and a 130-kDa signal transducer  $\beta$ -chain (gp130)]. The gp130 chain is a common receptor subunit that is shared by receptors for other cytokines of the IL-6 family, including IL-6, IL-11, cardiotrophin-1, oncostatin M, and ciliary neurotrophic factor (Hibi *et al.*, 1990; Gearing *et al.*, 1991; Gearing and Bruce, 1992; Ip *et al.*, 1992; Davis *et al.*, 1993; Murakami *et al.*, 1993). Interleukin-6 can substitute for LIF in mouse ES cells. Mouse ES cells can be maintained in the undifferentiated state using a combination of interleukin-6 and soluble interleukin-6 receptor (IL-6/sIL-6R) (Nichols *et al.*, 1994; Yoshida *et al.*, 1994; Yoshida *et al.*, 1996)

The binding of LIF induces the dimerisation of the receptor chains to form a high affinity heterodimer complex (Davis *et al.*, 1993; Murakami *et al.*, 1993). The high affinity receptor complex formation leads to the recruitment and activation of cytosolic tyrosine kinases of the Janus kinase (JAK) family (i.e. JAK-1, JAK-2, JAK-3 and Tyk2). The activated JAK cross phosphorylate themselves (Narazaki *et al.*, 1994; Stahl *et al.*, 1994) and then phosphorylate

the tyrosine residues on the intracellular domain of gp130 (Stahl *et al.*, 1994; Nakamura *et al.*, 1998). This phosphorylation by JAKs creates binding or docking sites (i.e. the phosphorylated tyrosine residues) on the receptor for the binding of proteins containing the Src-homology 2 (SH2) domain. The creation of docking sites on the gp130 receptor, is followed by the activation of two major signalling cascades: JAK-STAT which promotes ES cell selfrenewal (Boeuf *et al.*, 1997; Niwa *et al.*, 1998; Matsuda *et al.*, 1999; Raz *et al.*, 1999) and SHP2-Ras/Raf/ERK/MAPK which promotes ES cell differentiation (Burdon *et al.*, 1999). This means that the balance between the JAK-STAT cascade and the SHP-2-ERK cascade may be crucial in determining whether ES cells differentiate or remain the differentiate state.

# 1.9.2 Leukaemia inhibitory factor signalling: STAT cascade

Docking sites on gp130 are required for LIF/STAT signalling to maintain ES cell self-renewal (Niwa *et al.*, 1998). The STAT (signal transducer and activator of transcription) protein family include STAT1, 2, 3, 4, 5A, 5B and 6. Non-phosphorylated latent STAT proteins are activated to become active transcription factors by phosphorylation. STAT3 is the major or critical mediator of LIF signalling that maintain ES cells in the differentiated state. Over-expression of STAT3F (dominant negative mutant form of STAT3 with tyrosine 705 changed to phenylalanine) in ES cells growing in the presence of LIF can cause loss of pluripotency and enhanced cell differentiation (Boeuf *et al.*, 1997; Niwa *et al.*, 1998). STAT3 activation is required and sufficient to maintain mouse ES cells in the undifferentiated state (Matsuda *et* 

*al.*, 1999; Raz *et al.*, 1999). It has been shown that the inhibition of STAT3 activation blocked ES cell self-renewal and promoted differentiation (Niwa *et al.*, 1998).

When STAT3 docks on the cytosolic domain of the receptor via SH-2phosphotyrosine interactions, it (i.e. STAT3) is phosphorylated by receptorbound JAKs (or other nonreceptor tyrosine kinases). STAT3 phosphorylation at Tyr705 is required for dimerisation and nuclear translocation because overexpression of STAT3F has been shown to block the activation of endogenous STAT3 in myeloid leukemia MI cells (Minami *et al.*, 1996; Nakajima *et al.*, 1996). The activated STAT factors dissociate from the receptor complex, undergo homodimerisation or heterodimerisation and then translocate into the nucleus where they bind to elements in the promoters of STAT responsive or target genes to control their transcription (Figure 1.11)

#### 1.9.3 Leukaemia inhibitory factor signalling: ERK/MAPK cascade

The SHP-2-ERK cascade is activated by docking of growth factor receptorbound protein 2 (Grb2) on phosphorylated SHP2 on the activated gp130 receptor. Grb2 and the son of sevenless (SOS) form a complex. The complex formation leads to the activation of Ras which in turn activates Raf. Many other pathways also use Ras signalling to achieve various outputs or functions. It is not yet fully understood how specificity is generated. Specificity of Ras action may be determined by cell type or the presence (or absence) of

specific signal-activated and tissue-restricted signalling molecules (e.g. transcription factors).

Ras-activated Raf activates mitogen-activated protein kinase kinase (MEK) which in turn finally activates ERK (Figure 1.11). Activated SHP-2-ERK cascade antagonises the STAT3 cascade to promote stem cell differentiation. ERK1 and ERK2 play a role in mediating mitogenic responses of cells to growth factors (Pages et al., 1993; Cowley et al., 1994; Weber et al., 1997). SHP-2-ERK activation is also not unique to controlling the balance between ES cell renewal and differentiation. However, SHP-2-ERK activation has been linked to the regulation of mouse ES cell self-renewal. SHP-2-ERK signalling can repress the expression of Nanog in mouse ES cells (Hamazaki et al., 2006). A dominant-negative mutant of Ras has been found to suppress the expression of GATA-4, GATA-6 and other markers of extraembryonic endoderm differentiation. It has also been reported that a constitutively active Ras promoted ES cell differentiation and this promoted ES cell differentiation was inhibited by the application of a MEK inhibitor (U0126) (Yoshida-Koide et al., 2004). Similarly it has been shown that the inhibition of ERK signalling promotes self-renewal of mouse ES cells (Burdon et al., 1999).



*Figure 1.11: LIF signalling and stem cell self-renewal.* This diagram shows a simplified view of LIF-induced signalling events. LIF binding to the receptor leads to recruitment of JAK and the activation of STAT. STAT dimerise, translocate to the nucleus and switch on the expressions of cytokine (e.g., LIF) target genes. In the case of LIF binding to the receptor, in stem cells, genes that keep the stem cells in the undifferentiated state are activated. Abbreviations: JAK (Janus kinase); MEK (MAPK kinase); P (Phosphate group); SOS (Son of sevenless) and STAT (Signal transducers and activators of transcription); Sh-2-Containing Tyrosine Phosphatase (SHP2),

#### **1.9.3.1** Negative regulation of LIF signalling by SHP-2 and SOCS

The gp130-dependent signalling pathway is regulated by action of Srchomology tyrosine phosphatases 2 (SHP-2) and suppressor of cytokine signalling (SOCS) proteins (Figure 1.12). SHP-2 recruited to the phosphorylated gp130 receptor, in addition to their activation of ERK cascade, negatively regulate gp130-induced also STAT3 activation by dephosphorylation of phosphotyrosines (Anhuf et al., 2000; Ohtani et al., 2000). In SHP2 signal-deficient mice, gp130-induced STAT3 activation showed prolonged gp130-induced STAT3 activation (Ohtani et al., 2000). Suppression of SHP-2 signalling promotes mouse ES cell self-renewal (Burdon et al., 1999).

The eight SOCS proteins (SOCS1-SOSC7 and CIS) are characterised by a conserved C-terminal SOCS box, a central SH2 domain and a variable N-terminal (Hilton *et al.*, 1998). It has been shown that over-expression of SOCS3 can completely suppress LIF-induced signalling events (Yasukawa *et al.*, 2001). Over-expression of SOCS3 in a trophoblast stem cell line (Rcho-1) also suppressed the cell differentiation while SOCS3-deficient Rcho-1 cells were found to rapidly differentiate thus demonstrating that SOCS3 negatively regulates the ES cell differentiation (Takahashi *et al.*, 2003).

The mechanism by which SOCS inhibit gp130 receptor signalling is not yet entirely clear. SOCS proteins can be recruited to specific phosphotyrosine residues in activated gp130 receptor or can bind to receptor bound-JAKs and inhibit LIF signalling. For example it has been shown that SOCS1 can inhibit

the activities of JAK1, JAK2 and JAK3 by direct interaction (Endo *et al.*, 1997; Nicholson *et al.*, 1999). In contrast SOCS3 inhibitory activity may depend on its direct interaction with the phosphorylated gp130 receptor (Nicholson *et al.*, 2000). The gp130 receptor activation is known to induce rapid and transient up-regulation of the expression of SOCS proteins. The SOCS proteins then inhibit the signalling pathways that initially led to their production in a negative feedback loop (Krebs and Hilton, 2001).

The importance of the negative regulatory role of the SOCS proteins may be the reason why LIF can keep mouse ES cells but not human (including other primates) ES cells in the undifferentiated state. It has been shown that human but not mouse ES cells are rich in SOCS proteins (e.g. SOCS-1) (Schuringa *et al.*, 2002; Sato *et al.*, 2003).The negative regulation by SOCS proteins and SHP-2 ensure that LIF/STAT signalling has the appropriate magnitude and the durations.



*Figure 1.12: Negative regulation of LIF signalling by socs proteins.* Cytokine binding to the receptor activates the JAK-STAT pathway, leading to the induction of SOCS protein. The induced SOCS proteins act in a negative feedback loop to inhibit LIF signalling either by interacting with JAK or with the phosphorylated gp130 receptor. Redrawn from (Krebs and Hilton, 2001) with modifications. Abbreviations: JAK (Janus kinase); STAT (Signal transducers and activators of transcription); SOCS (suppressor of cytokine signalling)

# 1.10 Regulation of stem cell self-renewal and pluripotency

## 1.10.1 LIF signalling and pluripotent transcription factors

LIF signalling post STAT3 is poorly understood. Effectors downstream of STAT3 are poorly understood. Whether LIF signalling has any effect on key transcription factors that are responsible for ES cell self-renewal and pluripotency, is not yet clearly established. Evidence of direct connection between LIF/STAT signalling (i.e. post STAT3 activation) and ES cell self-renewal associated transcription factors (such as Nanog, Sox2 and Oct-4) is still missing. Information linking LIF/STAT signalling to these transcription factors can lead to the better understanding of LIF effects on ES cells and may also lead to the development of efficient stem cell culture and differentiation protocols.

LIF/STAT3 signalling has been given a prominent role in the maintenance of ES self-renewal and pluripotency in many studies (Niwa *et al.*, 1998). However LIF signalling pathway is not essential for the maintenance of ICM and human ES cell self-renewal and pluripotency. For example Nanog is capable of maintaining ES cell self-renewal independently of LIF/STAT3 signalling (Chambers *et al.*, 2003; Mitsui *et al.*, 2003). LIF is also not specific to ES cells because it can act on a wide range of cell types. LIF can mediate growth arrest and apoptosis in myeloid leukemia MI cells (Minami *et al.*, 1996). Moreover LIF also does not exert a consistent effect on all cells. LIF can inhibit differentiation (i.e. promotes cell self-renewal) or LIF can induce differentiation (i.e. inhibit cell self-renewal). Both of these two mutually

exclusive processes (differentiation and self-renewal) require the activation of LIF signalling so the presence (or absence) of a set of genes (which may or may not be unique to the target cell type) with which LIF can interact, may ultimately determine whether LIF will promote self-renewal or induce of differentiation. In fact, ES cell pluripotency is thought to be maintained by a few key transcription factors including Oct-4, Nanog, Sox2 and other. These factors bind to regulatory sequences in the promoters of self-renewal and pluripotency specific genes and control their expression, thus, regulating ES cells self-renewal and pluripotency.

# 1.10.2 Octamer-binding protein -3/4

Octamer-binding protein -3/4 (Oct-3/4) belongs to POU (PIT/OCT/UNC) family of transcription factors. It binds to octameric sequence AGTCAAAT in target genes. Oct-4 is expressed in totipotent and pluripotent stem cells of the pregastrulation embryo and the germ cell lineage (Okamoto *et al.*, 1990; Rosner *et al.*, 1990; Yeom *et al.*, 1996; Nichols, 1998). When the embryo differentiates into the endoderm and mesoderm, Oct-4 expression is downregulated (Rosner *et al.*, 1990). Oct-4 is expressed in the ICM but not in the trophectoderm (Palmieri *et al.*, 1994). Oct4-deficient embryos develop normally to the blastocyst stage, but their ICM cells are not pluripotent (Nichols, 1998). The trophectoderm (TE) is the first differentiated cell lineage or tissue of the mammalian blastocyst. The TE surrounds the ICM. Forced repression of Oct-4 has been shown to induce the loss of pluripotency in ES cells causing their differentiation into the TE lineage (Niwa *et al.*, 2000). The

above study results strong evidence in support of the requirement of Oct-4 for ES cell pluripotency.

The precise levels of Oct-4 are critical for proper maintenance of ES selfrenewal and pluripotency. Repression of Oct-4 (~50% of normal level) induces differentiation toward the trophectoderm formation (i.e. differentiation) while over-expression of Oct-4 (~150% of normal level) induces differentiation mainly toward extraembryonic endoderm (Niwa *et al.*, 2000). It is generally accepted that Oct-4 expression is unique to ES cells. However this view is been increasingly challenged by several studies that reported the expression of Oct-4 in the MSC of human (D'Ippolito *et al.*, 2004; Izadpanah *et al.*, 2006), rhesus monkey (Izadpanah *et al.*, 2006) and mouse (Pallante *et al.*, 2007). MSC "pluripotency" was demonstrated by the expression of Oct-4 and also by MSC ability to differentiate into beating cardiomyocytes and also expressed cardiac specific protein cardiac genes including Nkx2.5, cTnT,  $\alpha$ cardiac actin, MLC-2v, Cx43, and Cx40 (Pallante *et al.*, 2007). These results suggest that Oct-4 critical roles in pluripotency may not be unique for ES cells but may also apply to adult stem cells.

## 1.10.3 Nanog

Nanog was named after Tir nan Og, a mythical celtic land of the ever young, by Chamber *et al.*, 2003, in recognition of its ability to maintain ES cell selfrenewal and Pluripotent state. Nanog is both necessary and sufficient for selfrenewal and pluripotency of mouse epiblast and ES cells (Chambers *et al.*,

2003; Mitsui *et al.*, 2003). It has been found that Nanog-deficient ICM can not generate epiblast and also that nanog-deficient ES cells can not maintain pluripotency (Mitsui *et al.*, 2003). Over-expression of Nanog is sufficient to maintain ES cell self-renewal independently of LIF signalling (Chambers *et al.*, 2003; Mitsui *et al.*, 2003). These studies demonstrate that Nanog is required for the maintenance of pluripotency in both ICM and ES cells.

#### 1.10.4 Sox2

Sox2, like Oct4, is expressed in the ICM, epiblast, and germ cells (Avilion *et al.*, 2003). Sox2 and Oct-4 synergistically bind in the enhancers of several genes (Yuan *et al.*, 1995; Ambrosetti *et al.*, 1997; Okuda *et al.*, 1998; Nishimoto *et al.*, 1999; Ambrosetti *et al.*, 2000; Boyer *et al.*, 2005; Tan *et al.*, 2007). Octamer-Sox enhancers can regulate the expression of pluripotent stem cell-specific genes, including Nanog, Oct-4 and Sox2 itself. It has been shown that forced expression of Oct-4 can rescue the pluripotency of Sox2-null ES cells (Masui *et al.*, 2007). These results suggest that Sox2 is dispensable for the activation of Oct–Sox enhancers but is essential for maintaining the correct levels of Oct-4 in order to stabilise ES cells in a proper pluripotent state (Masui *et al.*, 2007)

## 1.10.5 Cooperative Interaction of Oct-4, Sox2 and Nanog

The cooperative interaction of Oct-4, Sox2 and Nanog is required in early development and also for the maintenance of the pluripotent and

undifferentiated states. Cooperation of Oct-4 and Sox2 has been detected in the enhancers of several genes including fibroblast growth factor (fgf)-4 (Yuan *et al.*, 1995; Ambrosetti *et al.*, 1997; Ambrosetti *et al.*, 2000), undifferentiated embryonic cell transcription factor 1 (UTF1)(Nishimoto *et al.*, 1999; Boyer *et al.*, 2005), Fbox15 (Tokuzawa *et al.*, 2003) and Nanog (Kuroda *et al.*, 2005; Rodda *et al.*, 2005).

UTF1 is expressed in human and mouse ES cells and the ICM (Okuda *et al.*, 1998; Boyer *et al.*, 2005; Tan *et al.*, 2007) but its transcription stops at the onset of differentiation (Okuda *et al.*, 1998). Similarly the expressions of Oct-4, Sox-2 and UTF1 are down-regulated following the induction of differentiation in ES cells (Nishimoto *et al.*, 1999; Boyer *et al.*, 2005).. The simultaneous down-regulation of Oct-4, Sox-2 and UTF1 suggest that UTF1 expression may be controlled by synergistic binding of Oct-4 and Sox-2 to an Oct–Sox regulatory sequence in UTF1 (Nishimoto *et al.*, 1999; Boyer *et al.*, 2005). Like UTF1, fgf-4 gene expressions in the ICM and in EC cells requires the cooperative activity of Sox-2 and Oct-4 (Yuan *et al.*, 1995; Ambrosetti *et al.*, 1997; Ambrosetti *et al.*, 2000). The cooperative activity of Sox-2 and Oct-4 may also be required for the transcription of Nanog. It has been shown that specific knockdown of Oct4 and Sox2 mRNA by RNA interference inhibited the expression of Nanog (Rodda *et al.*, 2005).

The above data suggest that the combinatorial regulatory actions of Oct-4 and Sox2 may be a general mechanism by which Sox-2 and Oct-4 can control the expression of several genes, especially those that are involved in the

maintenance of ES cell self-renewal and pluripotent state. Oct4, Sox2, Nanog and possibly other factors may be part of a very complex auto-regulatory network that is required to balance ES cell decisions between self-renewal and differentiation.

# **1.11 Embryonic stem cell differentiation protocols**

# 1.11.1 Embryoid body formation

The most common method used for *in vitro* differentiation is to culture ES cells in non-treated dishes (e.g. Petri dishes) and in absence of a feeder layer or LIF. Standard non-treated culture plates (e.g. microbiological plates) are usually sufficient for this purpose. ES cells that are cultured under these conditions form multicellular aggregates called embryoid bodies (EBs) and spontaneously and irreversibly differentiate into cell types of the three embryonic germ layers (ectoderm, mesoderm and endoderm) (Evans and Kaufman, 1981; Martin, 1981; Doetschman *et al.*, 1985; Smith *et al.*, 1988; Williams *et al.*, 1988; Niwa *et al.*, 1998). The three-dimensional structure of EBs is thought to initiate signalling that induces the differentiation of ES cells into cell types of embryonic germ layers (Keller, 1995; Itskovitz-Eldor *et al.*, 2000). The differentiation of ES cells into specific subsets of cells may be induced by treatment with specific factors (e.g. growth factors).

#### 1.11.2 Formation of EB by the Hanging Drops Method

Different protocols have been used in mouse ES cell culture for initiating the formation of EBs. One of the widely used methods is the "hanging drops" technique. In this technique, a definite number of mouse ES cells (usually 400) are cultivated in hanging drops (droplet of medium on the inside of a culture dish for a short period (e.g. 2 days). The EBs are then collected and further cultured for an additional period (e.g. 4-5 days) in suspension and then plated on gelatin-coated culture dishes for further differentiation. A simpler method than the "hanging drop method" is the liquid suspension culture. In the method, a definite number of stem cells are seeded in bacterial-grade dishes in an appropriate growth medium and left to form EBs. After the EB formation, they are then transferred to cell culture-grade dishes.

## 1.11.3 Embryonic stem cell differentiation into cardiomyocytes

The fundamental question of whether stem cells can actually differentiate into functional cardiomyocytes has been scientifically and convincingly answered. Under the appropriate experimental conditions, mouse and human ES cells can be differentiated into cardiomyocytes (Wobus *et al.*, 1991; Xu *et al.*, 1998; Kehat *et al.*, 2001; He *et al.*, 2003; Mummery *et al.*, 2003). The differentiation of ES cells into cardiomyocytes can be monitored by expression of several cardiac specific structural proteins including tropomyosin (Moore *et al.*, 2004),  $\alpha$ -actinin (Denning *et al.*, 2006), cardiac troponin I (cTnI) (Kehat *et al.*, 2001), cardiac troponin T (cTnT) (Kehat *et al.*, 2001), atrial natriuretic peptide (ANP) (Denning *et al.*, 2006), cardiac myosin

light chain isoforms (MLC-2v and MLC-2a) (Kehat *et al.*, 2001; Moore *et al.*, 2004), MLC-1v & MLC-1a and cardiac myosin heavy chain isoforms ( $\alpha$ -MHC and  $\beta$ -MHC) (Kehat *et al.*, 2001; Denning *et al.*, 2006; Anderson *et al.*, 2007). ES cell differentiation into cardiomyocytes can be also be tracked by the expression of several cardiac specific transcription factors including Nkx2.5, GATA-4 (Kehat *et al.*, 2001; Nakamura *et al.*, 2003; Anderson *et al.*, 2007), and myocyte enhancer factor (MEF2C) (Denning *et al.*, 2006; Anderson *et al.*, 2007).

# CARDIAC TRANSCRIPTION FACTORS AND STEM CELL DIFFERENTIATION INTO CARDIOMYOCYTES

# 1.12 Cardiac transcription factors and stem cell differentiation into cardiomyocytes

Cardiac development requires the interactions of many cardiac transcription factors. Cardiac transcription factors regulate cardiac specification and differentiation of cardiac cell types by controlling the expression of various cardiac genes that code for structural proteins or regulatory proteins. From Drosophila to man, cardiac specification and differentiation is guided by a complex transcriptional regulatory network that include the GATA family of cardiac-specific transcription factors, homeobox transcription factor (Csx/Nkx2.5), MEF2 transcription factors and the T-box transcription factors. The roles of GATA and the Nkx2.5 transcription factors are discussed below as examples of the complex roles of transcription factors in cardiomyocyte differentiation

# **1.13 GATA TRANSCRIPTION FACTORS**

GATA transcription factors belong to a family of proteins that contain either one or two highly conserved zinc finger DNA-binding domains. The family name "GATA" is based on the a recognition of the consensus DNA sequence (T/A)<u>GATA</u>(A/G) (Orkin, 1992; Lowry and Atchley, 2000). Within the promoters and enhancers of target genes, the GATA family members bind this consensus sequence with high efficiency and specificity (Orkin, 1992). The zinc finger motif in the GATA DNA binding domain is in the form Cys-X<sub>2</sub>-Cys-X<sub>17-20</sub>-Cys-X<sub>2</sub>-Cys followed by a basic region (Patient and McGhee, 2002). In animals, where GATA factors have been shown to play critical roles in development, differentiation and control of cell proliferation, the GATA DNA binding domain adopts the form Cys-X<sub>2</sub>-Cys-X<sub>17</sub>-Cys-X<sub>2</sub>-Cys (Patient and McGhee, 2002). The C-terminal zinc finger and the adjacent basic domain are required for DNA consensus sequence recognition and binding to the DNA (Tsai *et al.*, 1989; Lee *et al.*, 1991; Morrisey *et al.*, 1996) (Figure 1.13).

Six members of the GATA family of transcription factors (GATA-1 to -6) have been identified in vertebrates and a wide range of other eukaryotic organisms including plants (Reyes *et al.*, 2004), insects, and echinoderms (Lowry and Atchley, 2000) and fungi (Maxon and Herskowitz, 2001). Based on sequence homology and expression pattern, GATA transcription factors have been divided into two subgroups. The first subgroup of GATA proteins (GATA-1, -2 and -3) plays important roles mainly in haematopoiesis. GATA-2 is indispensable for development of definitive haematopoiesis (Mouthon *et al.*, 1993; Labbaye *et al.*, 1995; Orlic *et al.*, 1995; Tsai and Orkin, 1997;

Shimizu and Yamamoto, 2005). The second subgroup includes GATA-4, -5 and -6. These are found predominantly in the heart, digestive system and the extraembryonic endoderm. GATA-4, -5 and -6 bind to similar consensus sequences in cardiac genes (Charron *et al.*, 1999). There are confirmed phosphorylation sites in GATA -4 ( but not in GATA -5 or -6) (Grepin *et al.*, 1994; Ip *et al.*, 1994; Jiang and Evans, 1996) (Figure 5).



*Figure 1.13: Structural similarity among the GATA transcription factors.* Domains and the putative phosphorylation sites with their respective kinases are shown. The \* denotes a confirmed MAPK phosphorylation site. The percentage shown represents the degree of homology for each protein in comparison to GATA-4. ZF = zinc finger; TAD = transactivation domain; DBD =DNA binding domain; NLS = nuclear localisation signal, FOG = Friend of GATA) and MADS is an acronym that refers to the genes in which the sequence element was first identified: <u>MCM1</u> (minichromosomal maintenance protein 1) in *Saccharomyces* cerevisiae, <u>Agamous</u> from Arabidopsis thaliana, <u>Deficiens</u> from the snapdragon Antirrhinum majus and <u>SRF</u> from human. Taken from Temsah *et al.* (Temsah *et al.*, 2005)

#### 1.13.1 Roles of GATA transcription factors in cardiogenesis

In the developing embryo, the expression of GATA-4, -5 and -6 is detected and persists later in the developing heart. GATA-6 gene expression is restricted to the precardiac mesoderm, the embryonic heart tube and the primitive gut (Morrisey *et al.*, 1996). GATA-4 and GATA-6 transcript localisation overlap throughout the developing cardiogenic region (Jiang and Evans, 1996). GATA-5 is also expressed in cardiac crescent prior to formation of the primordial heart tube. After the tube heart, the expression of GATA-5 is then seen in both endocardium and myocardium (Laverriere *et al.*, 1994). GATA-4 is a potent transcriptional regulator of the promoters of several cardiac specific genes including ANP, BNP (Grepin *et al.*, 1994; Charron *et al.*, 1999; McBride and Nemer, 2001),  $\alpha$ -MHC, cTnl (Molkentin *et al.*, 1997; Murphy *et al.*, 1997; Di Lisi *et al.*, 1998; Charron *et al.*, 1999), and cTnC (Ip *et al.*, 1994)and MLC (McGrew *et al.*, 1996). All the above are strong evidence to suggest that these transcription factors play key roles in cardiogenesis.

# 1.13.2 Role of GATA-4 in cardiogenesis

Homozygous GATA4 null (GATA-4<sup>-/-</sup>) mice arrested and died between 7.0 and 10.5 days post coitum (dpc) because of severe developmental abnormalities including ventral foregut closure and cardiac bifida (Kuo *et al.*, 1997; Molkentin *et al.*, 1997; Xin *et al.*, 2006). The GATA-4<sup>-/-</sup> embryos showed severe defects in the "rostral-to-caudal" and "lateral-to-ventral folding" (Kuo *et al.*, 1997). The embryos of these GATA-4<sup>-/-</sup> mice developed splanchnic mesoderm that differentiated into primitive cardiomyocytes that expressed

contractile proteins (Kuo *et al.*, 1997; Molkentin *et al.*, 1997). However the GATA-4 null embryos lacked primitive heart tubes because the procardiomyocytes failed to migrate to the ventral midline to form a linear heart. The GATA-4<sup>-/-</sup> cardiomyocytes and endocardial cells were still able to populate the hearts of GATA-4<sup>-/-</sup>-C57BL/6 chimeric mice(Kuo *et al.*, 1997). The two bilaterally symmetric promyocardial primordia in the mutants did not migrate ventrally but remained lateral and formed two independent heart tubes containing differentiated cardiomyocytes (Molkentin *et al.*, 1997).These results suggest a role for GATA-4 in cardiac morphogenesis but not in cardiomyocyte lineage specification.

Small interfering RNAs (siRNA) that specifically target GATA-4 in cardiac mesodermal cells suppressed the expression of N-cadherin mRNA without affecting the expression of the other cardiac marker mRNAs however it caused cardia bifida in chick embryos (Zhang *et al.*, 2003a). The selective suppression of N-cadherin, one of the genes essential for the single heart formation, with the lack of effect on cardiac specific markers by GATA-4 siRNA also suggests that GATA-4 may not be essential for cardiomyocyte commitment during cardiogenesis.

Similarly, the inhibition of GATA-4 using GATA-4 antisense transcripts blocked P19 stem cell cardiac differentiation at the premyocardial (cardioblast) stage but brachyury T was also found to be expressed. Brachyury T is a T-box transcription factor (Kispert *et al.*, 1995) and is necessary for the establishment of the mesoderm in vertebrates (Schulte-

Merker *et al.*, 1992). Mesodermal lineage gives rise to cardiomyocytes. The expression of brachyury T suggests that mesodermal lineage commitment occur in the absence of GATA-4 (Grepin *et al.*, 1997) (Grepin *et al.*, 1997). The expression of brachyury T in absence of GATA-4 is further strong supporting evidence that GATA-4 is not essential for the initiation of cardiac lineage commitment.

Zebrafish embryos depleted of GATA-5 and GATA-6 have no heart. In contrast, embryos depleted of GATA-4 in the presence of either functional GATA-5 or GATA-6 develop defective heart tubes (Holtzinger and Evans, 2007) rather than total blockage of cardiomyocyte specification. These results also support an essential role for GATA-4 in cardiac morphogenesis but not in cardiomyocyte lineage specification.

All the above data when considered together, suggest that GATA-4 may be required for the events that control normal cardiac morphogenesis after the early stages of cardiomyocyte commitment but may not be essential for cardiomyocyte commitment during the early stages of cardiogenesis. The lack of normal cardiac morphogenesis in the absence of functional GATA-4 but in the presence of intact GATA-5 and GATA-6 also suggest that there may not be functional redundancy between GATA-4 and GATA-5 or between GATA-4 and GATA-6.

#### 1.13.3 Role of GATA-5 and GATA-6 in cardiogenesis

GATA-6 is expressed at the blastocyst stage in part of the inner cell mass and in the trophectoderm (Koutsourakis *et al.*, 1999). This suggests that GATA-6 may have a role to play in the very early events of heart development. Inactivation of GATA-6 gene in mice is lethal at gastrulation, with differentiation defects in the ectoderm and part of the visceral endoderm (Koutsourakis *et al.*, 1999). GATA-6 is expressed in cardiac mesoderm before gastrulation (Peterkin *et al.*, 2003). In both *Xenopus* and zebrafish embryos, the injection of anti-sense oligonucleotides to specifically block the translation of GATA-6 protein severely reduced the expression of cardiac proteins in embryos (Peterkin *et al.*, 2003).

The fact that zebrafish embryos lacking GATA-5 and GATA-6 did not develop hearts (i.e. lack of cardiomyocyte specification) (Holtzinger and Evans, 2007) but those lacking only GATA-4 in the presence of a functional GATA-5 or GATA-4 can developed defective tubular hearts (i.e. cardiomyocyte specification) suggest that GATA-5 or GATA-6 may be essential and also perform redundant functions during cardiomyocyte specification. Similarly, the ablation of both GATA-5 and GATA-6 resulted in complete lack of the expression of Nkx2.5, vMHC and MLC-2 in zebrafish embryos (Peterkin *et al.*, 2007), thus also suggesting important roles for either GATA-5 or GATA-6 or both. Embryos null for both GATA-4 and GATA-6 (GATA4<sup>-/-</sup>GATA6<sup>-/-</sup>) completely lack hearts (Zhao *et al.*, 2008).These studies point to non-essential roles for GATA-4 but possible essential roles for GATA-5 or GATA-6. It is beginning to emerge that, in contrast to the more prominent roles that
are attributed to GATA-4 in literature, GATA-5 or GATA-6 rather than GATA-4 may have essential roles to play in the induction of cardiomyocyte differentiation.

#### 1.13.4 Cooperative Interaction between GATA-4 and GATA-6

No single GATA transcription factor has been found to be both necessary and sufficient for cardiogenesis or cardiomyocyte specification. This along with their co-expression and localisation, developmental regulation during cardiogenesis and their structural similarities had led to the suggestion of possible functional redundancy among cardiogenic GATA factors. However instead of a possible redundancy of GATA-4 and GATA-6 function, several studies (as discussed above) point to a possible cooperative interaction between GATA-4 and GATA-6. GATA-4 and -6 have similar potency in activating cardiac genes including ANP, BNP (Charron et al., 1999) and troponin C (Morrisey et al., 1996). Furthermore the down-regulation of either GATA-4 or GATA-6 proteins in primary cardiomyocytes resulted in reduced mRNA expression of several cardiac genes including BNP,  $\alpha$ -MHC,  $\beta$ -MHC and cTnI (Charron et al., 1999). Mice that are heterozygous for either a GATA4 or GATA6 null allele are normal but compound heterozygosity of GATA4 and GATA6 is lethal by E13.5 (Xin et al., 2006). These results suggest a possible cooperative interaction between GATA-4 and GATA-6 the regulation of cardiac specific gene expression.

#### 1.13.5 Partners or Targets for GATA-4

GATA-4 interacts with Nkx2.5, a key cardiac transcription factor. ANP gene is a direct downstream target for Nkx2.5 and DNA binding sites for this transcription factor are found in the ANP promoter (Durocher *et al.*, 1996). GATA-4 physically associates with Nkx2.5 to cooperatively activate the transcription of cardiac-restricted genes including ANP (Durocher *et al.*, 1996; Durocher *et al.*, 1997; Durocher and Nemer, 1998; Lee *et al.*, 1998; Sepulveda *et al.*, 1998). BMP activation of cardiac-specific proteins appears to be mediated by the transcription factors GATA-4 and Nkx-2.5. Application of BMP-2 *in vivo* induced ectopic expression of Nkx2.5 and GATA-4, 5 and 6 (Merika and Orkin, 1993; Zhu *et al.*, 1997; Svensson *et al.*, 1999).

More supporting evidence for crosstalk between GATA-4 and Nkx2.5 comes from experimental data which showed that simultaneous over-expression of Nkx2.5 and GATA-4 in P19Cl6[noggin] cells induced cardiac differentiation while differentiation was not observed when either Nkx2.5 or GATA-4 alone was over-expressed in P19Cl6noggin cells (Monzen *et al.*, 1999). P19Cl6noggin over-express the noggin gene. The noggin gene codes for a family of secreted proteins that bind to BMP and inhibit their interaction with their receptor thus blocking their signalling roles (Zimmerman *et al.*, 1996).

Besides Nkx2.5, GATA-4 also associates with other transcription factors including NFAT( Nuclear Factor of Activated T cells) (Suzuki *et al.*, 1999; Diedrichs *et al.*, 2004; Jiang *et al.*, 2007), SRF (Belaguli *et al.*, 2000; Moore *et al.*, 2001), p300 (Slepak *et al.*, 2001; Yanazume *et al.*, 2003), MEF2 (Morin *et al.*, 2001)

*al.*, 2000) and Friend of GATA-2 (FOG-2) (Svensson *et al.*, 1999). GATA-4 association with these other transcription factors that are targets of several signalling pathways underlies the complex mechanisms by which GATA proteins can regulate many important processes during cardiogenesis.

# 1.13.6 GATA-4 and Upstream Protein Kinases

Phosphorylation is a key regulatory mechanism in many biological systems. The location of numerous phosphorylation sites in GATA-4 protein (Figure 1.13) suggests that it is a direct downstream target for several kinases, including the MAP Kinases ERK (Liang et al., 2001). It has been shown that RhoA potentiates the transcriptional activity of GATA-4 via a p38 MAPKdependent pathway that phosphorylates GATA-4 activation domains. It was also shown that GATA binding sites mediate RhoA activation of target cardiac promoters (Charron et al., 2001). Over-expression of TAK1 [a mitogenactivated protein kinase kinase kinase (MAPKKK)] can restore the ability of P19CL6noggin cells to differentiate into cardiomyocytes (Monzen et al., 1999). The over-expression of the dominant negative form of TAK1 in parental P19CL6 cells blocked cardiomyocyte differentiation. While the overexpression of Nkx2.5 or GATA-4 alone can not restore the ability of P19CL6noggin cells to differentiate into cardiomyocytes, the over-expression of both GATA-4 and Nkx2.5 can rescue P19CL6noggin ability to differentiate into cardiomyocytes (Monzen et al., 1999). These results suggest that TAK1, Nkx-2.5 and GATA-4 may cooperatively interact in BMP induced cardiogenic signalling.

#### 1.13.7 Summary for GATA transcription factors

GATA transcription factors are key regulator molecules in animal development. GATA-1, 2 and 3 are relevant in the haematopoietic system while GATA-4, 5 and 6 have roles to play in cardiogenesis. The promoters of many cardiac specific proteins including ANF, BNF, cTnl, α-MHC, MLC contain consensus sequence for GATA-4 in their genes. GATA-4 also interacts with the specific cardiac transaction factor Nkx2.5 and other transcription factors including p300, FOG-2 and NFAT. Great prominence is attributed to the role of GATA-4 in cardiomyocyte commitment but emerging evidence suggests that GATA-4 may not be essential for cardiomyocyte commitment. GATA-4 may play an important role in the events that control normal cardiac morphogenesis after the early stages of cardiomyocyte commitment or cardiogenesis. Increasingly important roles for GATA-5 or GATA-6 or both are also emerging. Finally, all the research data or evidence about the roles of GATA transcription factors in cardiogenesis in vertebrates (e.g. mouse, Xenopus and Zebrafish Chicken) strongly supports the existence of critical roles for GATA-4, 5 and 6 during normal cardiogenesis.

# **1.14 CARDIAC HOMEOBOX TRANSCRIPTION FACTOR**

The discovery of the tinman gene in Drosophila contributed enormously to our understanding of some of the molecular events that are involved in the vertebrate heart development. Tinman expression is essential for the specification of cardiac cells and the formation of Drosophila heart (Kim and Nirenberg, 1989; Azpiazu and Frasch, 1993; Bodmer, 1993; Zaffran and Frasch, 2005).

Csx/Nkx2.5 [i.e. Cardiac-Specific homeobox (Csx) or the fifth vertebrate gene identified in the NK2 homeobox gene family (Nkx2.5)] is a member of the NK homeobox gene family that is conserved in evolution. The gene products act as DNA-binding transcriptional activators (Kim and Nirenberg, 1989). Each NK2 protein consists of a highly conserved structure that is composed of an N-terminal TN domain, the homeodomain and the NK-2specific domain (NK2-SD). Four NK genes (i.e. NK1, NK2, NK3, and NK4) were originally identified in Drosophila. The equivalent of Drosophila tinman gene in mammal is Csx/Nkx2.5 (i.e. Nkx2.5) (Komuro and Izumo, 1993; Lints et al., 1993). Nkx2.5 is a universal marker for early heart development and its homeodomain has a helix-turn-helix motif that binds to DNA sequences. The specific consensus DNA sequence is 5' T(C/T) AAGTG 3'. Nkx2.5 is expressed in a wide range of species including XNkx2.3 and XNkx2.9 in Xenopus (Tonissen et al., 1994; Evans et al., 1995; Cleaver et al., 1996; Newman and Krieg, 1998), cNkx2.3 and cNkx2.8 in chicken (Buchberger et al., 1996; Boettger et al., 1997; Brand et al., 1997; Reecy et al., 1997), nkx2.3 and nkx2.7 in zebrafish (Lee et al., 1996) and Nkx2.6 in mouse (Biben and

Harvey, 1997; Nikolova *et al.*, 1997). Nkx2.5 is expressed in the human heart (Kasahara *et al.*, 2000; Inga *et al.*, 2005; Zheng *et al.*, 2008a). The loss of function of human NKX2.5 (due to mutations) is associated with congenital heart diseases (Kasahara *et al.*, 2000; Inga *et al.*, 2005).

# 1.14.1 Role of the transcription factor Nkx2.5 in cardiogenesis

Nkx2.5 is one of the earliest markers of heart field mesoderm in Drosophila and vertebrates. In the mouse, the expression of Nkx2.5 starts early (about 7.5 dpc) in the precardiac mesoderm. This early expression of Nkx2.5 in the heart progenitor cells in the first heart field (FHF) and the second hearts field (SHF) continues at a high level in the heart through adulthood (Komuro and Izumo, 1993; Lints et al., 1993; Lee et al., 2004). The specification of cardiogenic cells and the development of the Drosophila dorsal vessel (the insect equivalent of the mammalian heart) requires tinman (Kim and Nirenberg, 1989; Azpiazu and Frasch, 1993; Bodmer, 1993). Drosophila with mutated tinman are unable to express myosin and the precardiac markers, zfh-1 and eve in the heart (Azpiazu and Frasch, 1993; Bodmer, 1993). Wildtype tinman expression can partially rescue the mutant phenotype, but widespread expression of tinman does not result in ectopic expression of cardiac markers (Bodmer, 1993). This suggests that tinman alone is not sufficient to initiate the regulatory pathway leading to heart formation. Tinman may require the cooperation of additional regulatory proteins. Indeed as already mentioned above Nkx2.5 needs cooperative interaction with other factors such as GATA-4. Cell culture studies suggest that the Nkx2.5 protein

may act together with other transcription factors to regulate the expression of cardiac genes. For example, the ability of Nkx2.5 to activate transcription from the cardiac  $\alpha$ -actin promoter is strongly increased by the presence of the MADS domain protein, serum response factor (SRF) (Chen and Schwartz, 1996). Even higher levels of transcriptional activation are achieved from this promoter when Nkx2.5, SRF, and the zinc finger protein GATA-4 are all present in the same cell (Sepulveda *et al.*, 1998).

The over-expression of either XNkx2.5 or XNkx2.3 in *Xenopus* (Cleaver *et al.*, 1996) or zebrafish (Chen and Fishman, 1996) embryos caused enlarged hearts which were perfectly normal in all other features. The enlarged heart was due to the thickening of the myocardium caused by an increase in the overall number of myocardial cells (hyperplasia) (Cleaver *et al.*, 1996). On the other hand the expression of inhibitory or dominant negative mutants of either XNkx2.3 or XNkx2.5 in the *Xenopus* embryos resulted in complete lack of cardiac gene expression and complete absence of heart formation (Fu *et al.*, 1998; Grow and Krieg, 1998). This inhibition can be rescued by expression of wild-type Nkx2.5 (i.e. tinman) sequences (Grow and Krieg, 1998) thus underlining the importance of the role of Nkx2.5 in cardiogenesis. The gain-of-function results by Nkx2.5 over-expression and the inhibitory effects of dominant negative Nkx2.5 suggest that Nkx2.5 has an important role in cardiogenesis.

Nkx2.5 knockout mice die *in utero* shortly after looping morphogenesis. Heart tube formation was normal in mutant embryos, but looping morphogenesis, a

critical determinant of heart formation was not initiated at the linear heart tube stage (8.25-8.5 days p.c.) (Lyons et al., 1995). It has also been found that hearts of homozygous null embryos underwent looping but showed arrest of cardiac development and poor development of blood vessels (Tanaka et al., 1999). Commitment to the cardiac muscle lineage was not compromised (Lyons et al., 1995). The ventricular specific marker, MLC-2v was not expressed in mutant hearts or in mutant ES cell-derived cardiomyocytes, but β-MHC, another ventricular specific marker was expressed in the mutant mouse hearts. Cardiac development is multi-step process. The arrest of cardiac development in the Nkx2.5 knockout mice suggests that Nkx2.5 may be required for the expression of several essential factors in the early stages of the developing heart. However the expression of  $\beta$ -MHC in the mutant hearts indicated that not all ventricular-specific genes depend on Nkx2.5 for their expression (Lyons et al., 1995). But the data however demonstrate that Nkx2.5 is required for cardiogenesis. In addition, many studies have shown that the lack of Nkx2.5 compromised or reduces the expression of several cardiac specific genes including the transcription factors eHAND (Biben et al., 1997), Msx2, MEF2C, N-Myc (Tanaka et al., 1999), CARP (Zou et al., 1997) and Irx4 (Bruneau et al., 2000) and many other cardiac genes including ANF, BNP, MLC-2v (Lyons et al., 1995; Tanaka et al., 1999). Cardiac transcription factors are critical players in cardiogenesis and the fact that many may be under the control of Nkx2.5 is an indication of the importance of Nkx2.5 in this process.

#### 1.14.2 Nkx2.5 role in stem cell differentiation into cardiomyocytes

P19 stem cells differentiate into cardiomyocytes in 1% DMSO (McBurney *et al.*, 1982; Habara-Ohkubo, 1996; Angello *et al.*, 2006). The expression of a dominant-negative mutant of Nkx2.5 blocked this differentiation (Jamali *et al.*, 2001). The inability of the cells to form cardiomyocytes was accompanied by the loss of endogenous expression of cardiac genes including GATA-4 and MEF2C. This indicated that the function of Nkx2.5, as well as being critical for cardiogenesis *in vivo*, is also critical for stem cell differentiation into cardiomyocytes.

#### 1.14.3 Summary for the Roles of the Homeobox Nkx2.5

Research data from Drosophila, mouse, *Xenopus* and zebrafish support important regulatory roles for Nkx2.5 in cardiogenesis and the functions of the normal heart. Nkx2.5 interacts with other transcription factors including GATA transcription factors (as discussed earlier) and T-box factors (discussed below) and serum response factor. Nkx2.5 is also involved in the roles of cardiogenic extracellular signalling proteins including BMPs and Wnts (discussed below). Results from both loss and gain of function experiments have established Nkx2.5 as a very important cardiac transcription factor and support a transcriptional regulatory role of Nkx2.5 over a cascade of other cardiac transcription factors and cardiac specific genes.

# **1.15T-BOX TRANSCRIPTION FACTORS**

The T-box proteins are mainly of importance during the specification of the chamber myocardium after the linear tube heart has been formed. Their roles come into play after cardiac lineage commitment. They are therefore discussed very briefly in this introduction.

The T-box proteins belong to a family of transcription factors that are essential for early cardiac lineage determination, chamber specification and valvuloseptal development. This protein family is characterised by the presence of a highly conserved DNA-binding domain (i.e. the T-box). Seventeen T-box proteins been described in vertebrates (Naiche *et al.*, 2005). Tbx1, Tbx2, Tbx3, Tbx5, Tbx18 and Tbx20 are expressed in vertebrate embryonic hearts and play specific roles in the heart development.

#### 1.15.1 Role of T-box transcription factors in cardiogenesis

The T-box proteins are conserved regulators of heart development from Drosophila to human and other vertebrates. Drosophila Dorsocross complex (Doc), the three linked T-box genes (equivalent to mammalian Tbx4/Tbx5/Tbx6) are critical for Drosophila cardiac specification. Doc is required for the Drosophila cardiac specification (Reim and Frasch, 2005). Drosophila T-box genes midline and H15 pannier (i.e. equivalent to mammalian Tbx20) participate in cardiac specification in determining cardioblast differentiation, polarity and patterning (Miskolczi-McCallum et al., 2005; Reim et al., 2005). Midline represses Doc in a subset of cardioblasts,

and this is crucial for establishing the morphological and functional characteristics of the distinct classes of muscle cells, one functioning in the propulsion of haemolymph and another acting as valves (Reim *et al.*, 2005).

#### 1.15.2 Mechanisms of action of T-Box transcription factors

T-box proteins perform their functions by combinatorial and cooperative interaction with other transcription factors. Tbx1 and Tbx2 (Bruneau *et al.*, 2001; Habets *et al.*, 2002; Xu *et al.*, 2004a) and Tbx5 (Hiroi *et al.*, 2001) can cooperatively interact with Nkx2.5. Tbx5 can also cooperatively interact with GATA-4 (Garg *et al.*, 2003). In the developing hearts, Tbx20 directly interacts with Nkx2.5, GATA-4 and GATA-5 in regulation of gene expression in the developing heart (Stennard *et al.*, 2003). Drosophila Doc, like the mammalian T-box proteins, also functions in association with two Drosophila transcription factors, tinman (i.e. equivalent to mammalian Nkx2.5) and pannier (i.e. equivalent to mammalian SATA-4), during cardiac specification (Reim and Frasch, 2005).

# 1.15.3 Roles of Tbx2 and Tbx5 in heart chamber development

Mouse Tbx5 is expressed in the cardiac crescent, indicating its involvement in the earliest stages of cardiac induction, then in a graded fashion along the heart tube with the highest levels in the sinoatrial region (Bruneau *et al.*, 1999). In Tbx5 null embryos, the left ventricular and sinoatrial regions are severely hypoplastic (Bruneau *et al.*, 2001). However the formation of the

linear heart tube suggests that Tbx5 may be essential for regulation of chamber-specific gene expression and morphogenesis of the early heart but is not essential for cardiac lineage specification.

Tbx2 functions as a transcriptional repressor in non-chamber myocardium zones (i.e. outflow tract, inflow tract and AVC) of the tube heart in order to restrict chamber myocardium formation. Tbx2 cooperatively interacts with Nkx2.5. Their combinatorial interaction represses the expression of chamber myocardium specific genes in non-chamber myocardium zones (i.e. outflow tract, inflow tract and AVC) of the tube heart. ANF gene (first marker for chamber formation ) is specifically repressed by Tbx2/Nkx2.5 interaction (Charron *et al.*, 1992; Habets *et al.*, 2002; Harrelson *et al.*, 2004; Cai *et al.*, 2005).

In contrast to Tbx2, Tbx5 combinatorial interaction with Nkx2.5 activates the expression of chamber myocardium specific genes in chamber myocardium zones (i.e. left ventricle and left atrium) of the tube heart (Bruneau *et al.*, 2001; Habets *et al.*, 2002). ANF gene is specifically induced by Tbx5/Nkx2.5 interaction in chamber myocardium forming zones (left ventricle and left atrium) of the tube heart (Bruneau *et al.*, 2001). The opposing actions of Tbx2 and Tbx5 are required at various cardiac developmental stages for the correct formation of the heart.

# EXTRACELLULAR SIGNALLING PROTEINS AND

# **STEM DIFFERENTIATION INTO**

# CARDIOMYOCYTES

# 1.16 ROLES OF EXTRACELLULAR SIGNALLING PROTEINS IN STEM DIFFERENTIATION INTO CARDIOMYOCYTES

Extracellular signalling factors are central for stem cell differentiation or for remaining in the undifferentiated state. Stem cell cardiomyocyte lineage commitment involves complex networks of extracellular signalling molecules that activate various signalling pathways. These activated signal transduction pathways in turn exert complex positive or negative controls on this process.

Well known peptide growth factors or extracellular signalling molecules that have been identified as essential regulators during the development of vertebrate heart are also involved in stem cell differentiation into cardiomyocytes. These include the bone morphogenetic proteins (BMP) (Zhang and Bradley, 1996; Schlange *et al.*, 2000), sonic hedgehog (SHH) (Yamagishi *et al.*, 2003; Hu *et al.*, 2004), fibroblast growth factor (FGF) (Kelly *et al.*, 2001; Vitelli *et al.*, 2002), Wnt (Ku and Melton, 1993; Schneider and Mercola, 2001; Garriock *et al.*, 2005) and Notch proteins (Loomes *et al.*, 2002; Timmerman *et al.*, 2004). The role of BMP and Wnts in cardiogenesis and stem cell differentiation into cardiomyocytes are discussed as examples of the essential roles of these extracellular signalling factors in cardiomyocyte lineage commitment

# **1.17 BONE MORPHOGENETIC PROTEINS**

Bone morphogenetic proteins (BMPs) belong to the transforming growth factor- $\beta$  (TGF- $\beta$ ) superfamily of structurally related secreted proteins that are expressed in lateral endoderm and ectoderm and play central roles in the induction of heart formation in vertebrate embryos. The other members of this superfamily are TGF- $\beta$ , activins, nodal, anti-Mullerian hormone (AMH) and myostatins. BMPs were originally identified as proteins that induce ectopic bone and cartilage formation *in vivo* (Wozney *et al.*, 1988). BMP and its family members play crucial roles during embryonic development, cell division or proliferation, differentiation, organisation migration, adhesion, extracellular matrix production and apoptosis.

BMPs exert their biological effects by binding to heteromeric complexes of two types of receptors (types I and II receptors) with intrinsic serine/threonine kinase activity and activation of specific downstream intracellular effectors known as Smad proteins (Wrana *et al.*, 1994). Smads relay the signal from the cell membrane to the nucleus, where they affect the transcription of target genes.

BMP type II and I receptors are cell surface receptors. BMP type II receptors are single transmembrane proteins with glycosylated cysteine-rich extracellular ligand binding domains and cytoplasmic domains with intrinsic serine/threonine kinase activity(Lin *et al.*, 1992). Type I receptors share a glycine/serine residue-rich stretch (GS-domain) in the juxtamembrane region, which is essential for type I receptor activation (Wrana *et al.*, 1994). BMP bind

weakly to type I or type II receptors alone. BMPs bind to heteromeric type I/type II receptor complex with high affinity (ten Dijke *et al.*, 1994; Liu *et al.*, 1995; Rosenzweig *et al.*, 1995). BMP signalling occurs via at least two pathways: MAPK and Smad pathways.

# 1.17.1 BMP receptor Intracellular signalling and Smad proteins

The Smad (Similar To Mothers Against Decapentaplegic) transcription factors are the major intracellular signalling molecules or effectors of BMP-induced signalling pathway (Canalis et al 2003; Shi and Massague 2003; Kawabata et al 1998). BMP, TGF- $\beta$  and other family members induce their membrane receptors to directly activate the Smads which then form transcriptional complexes to control the expression of target genes. Eight different Smad proteins (Smads 1 through 8) have been identified in mammals. Five of the mammalian Smads (Smad 1, 2, 3, 5 and 8) act as substrates for the TGF- $\beta$ receptor family. These five Smads are known as receptor-regulated Smads (R-Smads). R-Smads 1, 5 and 8 serve mainly as specific substrates for activated BMP and Muellerian type I receptors while R-Smads 2 and 3 serves as specific substrates for activated TGF- $\beta$ , activin, and Nodal type I receptors. Smad 4 serves as a type I receptors common mediator or common partner Smad (Co-Smad) for all R-Smads. R-Smads activation by BMP receptors leads to their heterodimerization with the Co-Smad (i.e. Smad 4). Smad 6 and 7 are inhibitory Smads (I-Smad) that serve as decoys interfering with Smadreceptor or Smad-Smad interactions. In addition to this intracellular regulation by I-Smads, BMP signally is also regulated at the receptor level by

extracellular proteins (e.g. Noggin) that directly bind BMP molecules and inhibit their interaction with the BMP receptors.

Upon BMP-induced heteromeric complex formation, the constitutively active serine/threonine kinase of type II receptor phosphorylates type I receptor in its intracellular domain. Phosphorylation activate the kinase activity of type I receptor. The activated type I receptor phosphorylates and activates target proteins of the R-Smads (Wrana *et al.*, 1994; Chen and Weinberg, 1995). The phosphorylation of R-Smad causes a conformational change, which promotes heterodimerization with Smad4. The R-Smad/Smad4 complex translocates to the nucleus. Within the nucleus R-Smad/Smad4 bind to specific sequences in the promoters of BMP target genes and control their expression in a cell-type specific manner (Figure 1.14).



**Figure 1.14: BMP signalling pathway**. BMP bind and form a heteromeric complex with type I and type II BMP receptors. The type II receptor phosphorylates the type I receptor. This leads to the phosphorylation of receptor-regulated smads (SMAD1,5 or 8). The phosphorylated SMAD forms a complex with the common mediator or common partner Smad (SMAD4). This complex is then and is transported into the nucleus leading to the activation of target genes. The intracellular inhibitor of BMP signal transduction is SMAD6. BMP= bone morphogenetic protein, R-Smads= receptor-regulated Smads, Co-SMAD = common mediator or common partner Smad, I-SMAD = inhibitory Smads and P= phosphate group.

#### 1.17.2 Role of BMPs in stem cell differentiation into cardiomyocytes

As already stated, BMPs are expressed in lateral endoderm and ectoderm of vertebrate embryos. Stimulatory factors from the anterior endoderm in combination with BMP signals in the anterior lateral mesoderm promote heart formation (Schultheiss *et al.*, 1997; Andrée *et al.*, 1998; Schlange *et al.*, 2000; Tzahor and Lassar, 2001). It has been shown that BMP-2 soaked beads can induce the ectopic expression of Nkx2.5 and GATA-4 in chick embryos (Schultheiss *et al.*, 1997). BMP-2 can induce the expression of cardiac specific proteins in chick embryos and the addition of Noggin to explants resulted in loss of the expression of Nkx2.5, GATA-4, eHAND, MEF2A and MHC expression (Schlange *et al.*, 2000). This suggests that BMP-2 is upstream of Nkx2.5 and GATA-4, and as already mentioned, eHAND, MEF2A and MHC are under the control of GATA-4 and Nkx2.5.

DMSO induces the cardiac differentiation of P19CL6 (a derivative of P19 stem cells). Over-expression of noggin (a natural inhibitor of BMP) in P19CL6 cells (P19CL6 [noggin] cells) prevented DMSO-induced differentiation of P19 cells into cardiomyocytes (Monzen *et al.*, 1999). Spontaneously beating cells did not result and cardiac transcription factor and contractile proteins were not present in P19Cl6[noggin] cells. This cardiac differentiation arrest could be rescued by over-expression of BMP-2 or by directly adding BMP proteins to the culture medium (Monzen *et al.*, 1999). The data provide strong evidence for the involvement of BMPs in stem cell differentiation into cardiomyocytes.

#### 1.17.3 Role of BMP and MAPK signalling in cardiomyocyte differentiation

In addition to the Smad pathways discussed above, BMP signalling also occurs via MAPK signalling pathways. The MAPK TGF-β-activated kinase 1 (TAK1) is expressed in P19CL6 cells and therefore could be involved in the DMSO-induced differentiation of these cells into cardiomyocytes. Overexpression of dominant negative TAK1 led to an inhibition P19CL6 cell differentiation into cardiomyocytes while over-expression of wild-type TAK1 or constitutively active TAK1 rescued cardiac differentiation in P19CL6[noggin] cells (Monzen et al., 1999). The above data suggests that BMP-induce activation MAPK signalling can lead to cardiac lineage specification in ES cells.

# 1.17.4 Role of BMP and Smad Signalling in Cardiomyocyte Differentiation

P19CL6noggin cells, which constitutively over-express the BMP antagonist noggin, do not differentiate into cardiomyocytes when treated with DMSO. Over-expression of both Smad1 and Smad4 has been found to be sufficient to rescue cardiac differentiation in DMSO-treated P19Cl6[noggin] cells (Monzen *et al.*, 2001). The same studies demonstrated that over-expression of the inhibitory Smad6 completely blocked differentiation of P19CL6 (Monzen *et al.*, 2001), thus suggesting that like the BMP-mediated activation of the MAPK signalling pathway, the BMP-activated Smad pathway is also necessary for ES cell differentiation into cardiomyocytes

#### 1.17.5 Crosstalk between Smad and MAPK signalling

Experimental evidence suggests that ATF-2 (activating transcription factor 2) may be a common target for both Smad and TAK1 pathways in TGF- $\beta$  signalling (Makino *et al.*, 1999). ATF-2 activity is enhanced after phosphorylation by stress-activated protein kinases such as c-Jun N-terminal kinase and p38. ATF-2 has also been found to be bound by Smad3 and Smad4 upon TGF- $\beta$  stimulation (Makino *et al.*, 1999). It has also been found that the over-expression of dominant negative forms of ATF-2 inhibited the P19CL6 cell differentiation into cardiomyocytes, while wild type ATF-2 enhanced the ability of Smad 1&4 and TAK1 to rescue cardiac differentiation in P19CL6noggin cells (Monzen *et al.*, 2001).

ATF-2 was also found to stimulate  $\beta$ -MHC promoter activity in a synergistic manner with Smad1&4 and TAK1 and promoted terminal cardiomyocyte differentiation of P19CL6noggin, whereas over-expression of the dominant negative form of ATF-2 reduced the promoter activities of several cardiac-specific genes including GATA-4, MEF2C, MLC2v  $\beta$ -MHC, ANP and BNP and inhibited differentiation of P19CL6 (Monzen *et al.*, 2001). These results suggest that Smads, TAK1, and their common target ATF-2 cooperatively play a critical role during the differentiation of stem cells into cardiomyocytes.

#### 1.17.6 Summary of bone morphogenetic proteins

BMP have important roles in both *in vivo* cardiogenesis and the differentiation of stem cells into cardiomyocytes. BMP are expressed very early in the endoderm and ectoderm and play central roles in the induction of heart

formation in vertebrate embryos. In addition to the Smad pathways BMP signalling also occurs via MAPK signalling pathways. This is very important since phosphorylation is a key post-transcriptional mode of gene control. The MAPK signalling when activated could control many other targets including transcription factors and other kinases.

# 1.18 Wnt Proteins

Wnt proteins belong to a family of cysteine-rich and lipid-modified signalling proteins (Willert *et al.*, 2003). The Wnt signalling pathways are highly regulated and play major roles in embryological development, tissue morphogenesis, homeostasis and regeneration, cell proliferation (van de Wetering *et al.*, 2002) and differentiation, cell fate decisions, apoptosis, axial polarity and axonal guidance.

Wnt signalling cascade is initiated by Wnt binding to the cysteine-rich, extracellular domain of a frizzled (Fz) receptor. For the effective Wnt binding, in addition to the Fz receptor, the presence of single-pass transmembrane molecule of the low density lipoprotein receptor-related proteins (LRP) (also known as the arrow protein in Drosophila) (Wehrli *et al.*, 2000) or LRP5/6 (i.e. vertebrate equivalent of Drosophila arrow protein) (Pinson *et al.*, 2000; Tamai *et al.*, 2000) is required. The LRPs are co-receptors to the Fz receptors. The Fz receptors, like the LRPs, are also transmembrane proteins with a long N-terminal extension called a cysteine-rich domain (CRD). Wnt proteins bind directly to the CRD region (Bhanot *et al.*, 1996; Hsieh *et al.*, 1999; Dann *et al.*, 2001).

# 1.18.1 Wnt Signalling pathways

Wnt binding to Fz receptors can result in the activation of one of three different Wnt regulated pathways: Wnt/β-catenin (canonical signalling

pathway) and two non-canonical Wnt signalling pathways (i.e. the Wnt/planar cell polarity (Wnt/PCP) and Wnt/calcium (Wnt/Ca<sup>2+</sup>) pathway)

#### 1.18.1.1 Wnt canonical (Wnt/β-catenin) pathway

In the absence of Wnt/receptor interactions, the level of  $\beta$ -catenin is kept low by degradation induced by  $\beta$ -catenin association with a large multiprotein  $\beta$ catenin destruction complex that includes two scaffolding proteins (i.e. axin and adenomatous polyposis coli (APC)), two serine/threonine kinases  $\beta$  (i.e. glycogen synthase k inase 3 $\beta$  (GSK3 $\beta$ ) and casein kinase 1 (CK1) (Giles *et al.*, 2003; Logan and Nusse, 2004). In this complex,  $\beta$ -catenin is phosphorylated by CK1 and GSK3 $\beta$ . This phosphorylation triggers the ubiquitination of  $\beta$ -catenin resulting in its subsequent degradation in the proteosome.

Wnt binding to Fz and LRP receptor complex induces the phosphorylation of LRP (Tamai et al., 2004; Davidson et al., 2005). A cytoplasmic protein known as Dishevelled (Dsh/Dv1) is also phosphorylated and activated (Figure 1.15). The activation of Dsh leads to the recruitment of Axin to the membrane, where it interacts with the cytoslic tail of LRP. Axin recruitment to the membrane from the multiprotein **β**-catenin destruction complex (i.e. Axin/APC/CK1/GSK3B) leads to the subsequent disruption or inhibition of the destruction complex. GSK3 $\beta$  is therefore not able to phosphorylate  $\beta$ -catenin. The net effect of GSK3β inhibition is the stabilisation and the accumulation of the non-phosphorylated form of  $\beta$ -catenin in the cytosol. The non-

phosphorylated cytosolic β-catenin then translocates into the nucleus, where it competes with members of the groucho family of co-factors for interactions with the T-cell factor/lymphoid-enhancing factor 1 (TCF/LEF) family of transcription factors (Molenaar *et al.*, 1996; van de Wetering *et al.*, 1997; Cavallo *et al.*, 1998; Eastman and Grosschedl, 1999). Once an active (LEF/TCF)-β-catenin complex is formed the transcription of Wnt target genes is activated (Yamaguchi *et al.*, 1999; Ai *et al.*, 2000). TCF/LEFs bind to a conserved DNA sequence known as the Wnt-response element (WRE) (i.e. C/T-C-T-T-G-A/T-A/T) via their HMG domain (van de Wetering *et al.*, 1997). In the absence of Wnt, TCF form a complex with Groucho. This complex acts as transcriptional repressor of Wnt responsive genes (Cavallo *et al.*, 1998; Eastman and Grosschedl, 1999).



**Figure 1.15:** Activation of canonical (Wnt/  $\beta$ -catenin) pathway. (A) In the absence of Wnt  $\beta$ -catenin associates with the Axin-APC- GSK3 $\beta$  complex.  $\beta$ -catenin is phosphorylated !and then targeted for degradation. At the same time, Wnt target genes are repressed by the association of TCF with Groucho. (B) Wnt binding to the Frizzled and LRP receptors induces phosphorylation of LRP and recruitment of Axin. Dsh is also phosphorylated, and the Axin·APC·GSK-3 complex is inhibited, leading to accumulation of cytosolic  $\beta$ -catenin and its subsequent translocation into the nucleus. In the nucleus  $\beta$ -catenin replaces Groucho from the T-cell factor/lymphoid-enhancing factor 1 (TCF/LEF) and then activates target genes.

## 1.18.1.2 Non-canonical Wnt/Ca2+ signalling pathway

In addition to the canonical pathway described above, a non-canonical pathway known as the Wnt/Ca<sup>2+</sup> signalling pathway can also be activated by binding of Wnt proteins to Fz receptors. However unlike canonical pathway, the downstream pathways activated by this binding do not involve GSK-3ß or  $\beta$ -catenin. The activation of the Wnt/Ca2+ signalling pathway is thought to involved that activation of G-proteins and phospholipase C (PLC) leading to the release of intracellular Ca<sup>2+</sup> ions, The increased intracellular calcium levels in turn activate key enzymes such protein kinase C (PKC) and Ca2+/calmodulin-dependent protein kinase II (CaMKII) (Sheldahl et al., 1999; Pandur et al., 2002; Sheldahl et al., 2003) (Figure 1.16). It has been shown in zebrafish blastulae, that the over-expression of Wnt5a or rat Fz2 (i.e. Frizzled receptor) stimulates the frequency of calcium fluxes in the enveloping layer (EVL) cells (Slusarski et al., 1997). Similarly in Xenopus embryos, overexpression of Wnt5a or Wnt1 activated calcium-sensitive kinase protein kinase C (Sheldahl et al., 1999) and calcium/calmodulin-dependent kinase II (CamKII) (Kuhl et al., 2000). In contrast blocking the over-expressed Wnt signalling using antibodies to Frizzled-5 (wnt5a the receptor) was found to inhibit PKC activity (Weeraratna et al., 2002).



*Figure 1.16: Activation of Wnt/ca*<sup>2+</sup> *non-canonical pathway.* Wnt binding to the receptor stimulates G-proteins which in turn activate phospholipase C. This leads to the release of calcium ions and the activation of protein kinase C and Ca2+/calmodulin-dependent protein kinase II (CaMKII). Signalling events downstream from this stage may lead to the inhibition of the canonical wnt signalling pathway.

## 1.18.1.3 Non-canonical Wnt/PCP signalling pathway

The second non-canonical signalling is the Wnt/PCP pathway. The Wnt/PCP signalling pathway regulates the generation of planar cell polarity (PCP). Like in the other Wnt pathways, in this non-canonical pathway, Wnt binding to the Fz receptor activates Dsh but in this case, Dsh activation does not lead to either a stabilisation of  $\beta$ -catenin or to an influx or release of intracellular calcium ions. Instead Dsh is localised to the cell membrane where it mediates the activation of small GTPases (RhoA and Rac) which then activate Rho kinase and c-Jun N-terminal kinase (JNK) (Endo *et al.*, 2008) (Figure 1.17).



*Figure 1.17: Activation of Wnt/PCP non-canonical pathway.* Wnt binding to the receptor stimulates Dsh protein which in turn RhoA and Rac. RhoA and Rac respective activation of Rho kinase and JNK is thought to control cell migration and polarity and the cytoskeleton.

#### 1.18.2 Classes of Wnt proteins

Based on the preference or ability to signal through the canonical or noncanonical pathway, Wnt proteins are divided into two groups: Wnt1 group (e.g. Wnt1, Wnt3a, and Wnt8) and wnt5a group (e.g. Wnt4, Wnt5a, and Wnt11). Wnt1 group are thought to signal exclusively via the canonical Wnt/ $\beta$ -catenin pathway while the Wnt5a group normally signal through the non-canonical Wnt signalling (Tada and Smith, 2000; Pandur *et al.*, 2002; Maurus *et al.*, 2005). The signals generated by the Wnt5a group can suppress  $\beta$ -catenin-mediated signalling (Torres *et al.*, 1996; Topol *et al.*, 2003; Westfall *et al.*, 2003; Maye *et al.*, 2004). This suggests that wnt5a group of Wnt proteins function as negative regulators of Wnt1 class proteins.

## 1.18.3 Wnt inhibitors

#### 1.18.3.1 Endogenous Wnt inhibitors: secreted frizzled-related proteins

Secreted frizzled-related proteins (sFRP) are distinct but related to the Fz cell membrane receptor proteins (Rattner *et al.*, 1997). The Wnt binding domain is conserved among sFRP and Fz proteins. The sFRP exert their biological activity acting as competitive inhibitor for Wnts binding to their cell membrane receptors (Bafico *et al.*, 1999; Golan *et al.*, 2004; Roman-Roman *et al.*, 2004).

#### 1.18.3.2 Endogenous Wnt inhibitors: Dickkopf (Dkk) proteins

Dickkopf (Dkk) like the sFRP also inhibit the canonical Wnt signalling (Glinka *et al.*, 1998). However, unlike the sFRP which bind to Fz receptors, the Dkk exert their inhibitory activities on canonical Wnt by competitively binding to the extracellular domains of the co-receptors (LRP5 and LRP6) (Mao *et al.*, 2001; Semenov *et al.*, 2001). Beside their competitive inhibition of the canonical pathway at the LRP5/6 receptor level, Dkk signalling may also regulate the activity of canonical Wnt proteins (Bafico *et al.*, 2001; Davidson *et al.*, 2002; Mao *et al.*, 2002).

#### 1.18.4 Frizzled receptor family in the mesoderm

In addition to the enormous number of Wnt ligands, many different Wnt receptors of the Frizzled family have been detected in the mesoderm of the heart-forming fields, cardiac neural crest cells or the adult heart, including Fz-4 (DeRossi *et al.*, 2000), Fz-7 and Fz-8 (Wheeler and Hoppler, 1999; Saitoh *et al.*, 2001), Fz-10a and -10b (Moriwaki *et al.*, 2000), Fz-1 and Fz-2 (van Gijn *et al.*, 2001) and Fz-9 (Sagara *et al.*, 1998; Wang *et al.*, 1999). The sFRP-1 is expressed in the heart and in the visceral yolk sac during mouse development, and that sFRP-1 and mWnt-8 display overlapping expression patterns during heart morphogenesis (Jaspard *et al.*, 2000). The extensive expression of different Wnt proteins and receptors in the developing hearts is strong indication that Wnt proteins have a role to play in the different stages of cardiogenesis.

#### 1.18.5 Roles of Wnt roles in cardiogenesis

Vertebrate heart development starts with specification of myocardial precursor cells within the mesodermal germ layer of the early embryo. Several Wnt ligands and Frizzled receptors have been shown to be expressed in a special and temporal manner regulated during early heart development. Wnt-11 expression is found in the heart-forming fields in chicken, mouse and *Xenopus* (Ku and Melton, 1993). It has also been shown that Wnt11-R, a protein closely related to mammalian Wnt11, is required for heart morphogenesis in *Xenopus* (Garriock *et al.*, 2005). It has also be shown that Wnt (Wnt-11) gene is expressed by newly gastrulated mesoderm cells within chicken embryos in a pattern which suggests that it may be involved in formation of the cardiogenic fields (Eisenberg *et al.*, 1997). In mouse embryos, by 7.5-8.5 dpc , the expression of Wnt2 in the early heart field overlaps with that of Wnt-11 before it becomes restricted to the pericardium (Monkley *et al.*, 1996).

# 1.18.6 Roles of the canonical Wnt/β-catenin signalling in ES cell differentiation into cardiomyocyte

The current model of the role of Wnt in development of the heart is that inhibition of canonical Wnt/ $\beta$ -catenin signalling pathway is essential for the induction cardiac lineage commitment (Marvin *et al.*, 2001; Schneider and Mercola, 2001; Tzahor and Lassar, 2001; Pandur *et al.*, 2002). However this model is not universally supported by published data. Data from some studies suggest that canonical Wnt/ $\beta$ -catenin signalling inhibits cardiogenesis while

data from other studies support positive regulation of ES cell differentiation into cardiomyocytes by the Wnt/ $\beta$ -catenin signalling (Nakamura *et al.*, 2003; Naito *et al.*, 2006; Kwon *et al.*, 2007; Liu *et al.*, 2007).

It has been shown in *Xenopus* laevis that the administration of Dkk-1 and Crescent (Wnt inhibitors) and also the ectopic expression of GSK-3 $\beta$  can initiate cardiogenesis in ventral marginal zone mesoderm (Schneider and Mercola, 2001). Similarly in chick embryos Crescent and Dkk-1 were found to induce heart muscle formation in the posterior lateral plate mesoderm (Marvin *et al.*, 2001). These studies suggest that the Wnt/ $\beta$ -catenin signalling negatively regulates cardiogenesis and the activation of pathways that inhibit Wnt/ $\beta$ -catenin signalling is required for the induction of cardiogenesis.

In contrast to the inhibitory roles suggested by the above studies, data from other studies suggest that Wnt/β-catenin signalling positively regulates ES cell cardiac differentiation during the early stages. Using mouse embryos, it was found that Wnt8a was also specifically expressed in heart precursors at E8.5 and also expressed initially throughout the developing heart tube (Kwon *et al.*, 2007). It was also found that the expression of the early cardiac genes Nkx2.5 and Tbx5 in EBs was up-regulated by Wnt3a and down-regulated by Dkk-1.The addition of Wnt3a to EBs before the induction of early cardiac genes increased the number of beating EBs to more than 50% while the addition of Fz8/Fc or Dkk-1 and Fz8/Fc, at this stage resulted in a complete absence of beating EBs (Naito *et al.*, 2006; Kwon *et al.*, 2007). The second heart (SHF) field marker, Islet1 was similarly affected by Wnt3a and Dkk-1

(Kwon *et al.*, 2007). Constitutively active GSK-3 $\beta$  (i.e. degradation of  $\beta$ catenin) also completely blocked Tbx5 (a mesoderm marker) and GATA4 (Nakamura *et al.*, 2003; Kwon *et al.*, 2007) and also inhibited the expression of BMP2, BMP4, and FGF8 (Nakamura *et al.*, 2003). In contrast, LiCI (a GSK-3 $\beta$  inhibitor) significantly increased the expression of Nkx2.5, GATA4, Tbx5, BMP2, and BMP4 (Nakamura *et al.*, 2003). These results strongly suggest a positive rather than a negative regulation of the stem cell differentiation into cardiomyocytes.

In addition to functional data mentioned above, the wide expression of canonical Wnts in the developing heart may suggest positive roles for canonical Wnts during the myocardial fate specification. For example canonical Wnts including Wnt2 (Monkley *et al.*, 1996; Karasawa *et al.*, 2002), Wnt9a (Karasawa *et al.*, 2002; Person *et al.*, 2005) and Wnt6 (Rodriguez-Niedenfuhr *et al.*, 2003) are also expressed in the developing heart.

In addition to a positive role for Wnt/ $\beta$ -catenin signalling during ES cell cardiac differentiation, it was shown that this pathway exhibited biphasic effects. While Wnt/ $\beta$ -catenin signalling positively regulated ES cell differentiation into cardiomyocytes during the early phases, it negatively regulated ES cell cardiac differentiation during the late stages of cardiomyocyte differentiation (Nakamura *et al.*, 2003; Naito *et al.*, 2006; Kwon *et al.*, 2007). In support of a negative role of Wnt/ $\beta$ -catenin signalling during the late stage cardiomyocyte differentiation, it was shown that the treatment of EBs with Dkk-1 and Fz8/Fc, by day 5 (i.e. late cardiomyocyte differentiation

stage in those experiments) increased the number of cardiomyocytes (Naito *et al.*, 2006; Kwon *et al.*, 2007) thus suggesting a negative role of Wnt at this stage.

Support of a positive role for Wnt during the early phase of cardiogenesis comes from the high expression of Wnt3 and Wnt8a in the pre-streak and early gastrula-stage embryo (Kemp *et al.*, 2005). In ES cells, Wnt3 and Wnt8a are transiently induced between day 2 and 3 of differentiation. It was also shown that Dkk-1 addition at day 0, 1 or 1.5, completely inhibited the generation of cells expressing Flk1 (a lateral mesoderm early marker). Lineage commitment was reduced by 80%. In contrast Dkk1 added at day 2 and later had negligible effects (Lindsley *et al.*, 2006).

All the above data when considered together, suggest that Wnt/ $\beta$ -catenin signalling is required for commitment of mesodermal cells into cardiomyocytes during the early phase of the of ES differentiation into cardiomyocytes. The down-regulation of of cardiac gene expression by Dkk-1, Fz8/Fc and constitutively active GSK-3 $\beta$  suggest that Wnt/ $\beta$ -catenin signalling played positive roles during the early phases of ES cell cardiac differentiation. The inhibition of the expression of Nkx2.5, GATA-4, Tbx5, BMP2, and BMP4 by constitutively active GSK-3 $\beta$  and the induction of these genes by LiCl (a GSK-3 $\beta$  inhibitor) also suggest that Wnt/ $\beta$ -catenin signalling is required for commitment of mesodermal cells into a cardiomyocyte lineage and the differentiation of the committed cells into cardiomyocytes. The inhibition of the committed cells into cardiomyocytes.
expression of BMP2, BMP4, and FGF8 also suggest that Wnt pathway is upstream of the BMP and FGF signalling pathways.

#### 1.18.7 Summary of Wnt Proteins

There is extensive expression of different Wnt proteins and their receptors in the developing hearts. Wnt and their receptors are expressed in the heart forming fields, cardiac crest and also in the adult heart. Wnt signalling is very complex. Some Wnts activate the canonical pathway, which signals through stabilisation of  $\beta$ -catenin. Wnts in Wnt1 group signal mainly through  $\beta$ -catenin. Some Wnts in wnt5a group signal through Wnt/PCP pathway leading to the regulation of cell migration and polarity and the cytoskeleton. Other Wnts in wnt5a group signal through activation of phospholipase C leading to of Ca<sup>2+</sup> ion release, the activation of protein kinase C and Ca<sup>2+</sup>/calmodulin-dependent protein kinase II (CaMKII) and this has been suggested to cause to the inhibition of the canonical Wnt/ $\beta$ -catenin signalling.

While data from previous studies suggest that the Wnt/ $\beta$ -catenin signalling inhibit cardiogenesis, more recent data suggest that the canonical Wnt/ $\beta$ -catenin signalling play positive role during the early phases of cell lineage specification *in vivo* and in ES cell cardiac differentiation. The fact that data from some reports support a positive role for the Wnt/ $\beta$ -catenin signalling in the commitment of mesodermal cells into a cardiomyocyte lineage and the differentiation of the committed cells into cardiomyocytes, while data from other reports suggest a negative role, may be a reflection of the existence

large number of Wnts and also our lack of clear understanding of the roles of these proteins in cardiomyocyte differentiation.

#### 1.19 Overview of extracellular signalling molecules and transcription factors in cardiogenesis and stem cell differentiation into cardiomyocytes

There is no doubt that stem cells offer many important potential benefits for treating many diseases including cardiovascular diseases. The key problem is how to derive a particular cell type from stem cells. There are data on the process of stem cell differentiation into cardiomyocytes. Many signal molecules and signal transduction pathways have been implicated in stem cell differentiation but there is no complete account of the mechanisms that keep stem cells in the undifferentiated state. Equally, there is no comprehensive account of the mechanisms that can drive stem cell differentiation into a particular cell type.

Investigations of the molecular mechanisms of many transcription factors including Nkx2.5, and GATA transcription factors have helped clarify understanding of this area but these studies are still at an early stage and exact details of their modes of action are not fully understood. As it has been seen in other systems, it is more likely that it is the actual interactions (or combinatorial actions) rather than the role of each factor alone that may hold the keys to our understanding of their roles in cardiogenesis.

At the receptor level many molecules including Wnt, BMP have been shown to play important roles in stem cell and stem cell differentiation (Figure 1.18). Signalling downstream of various classical signalling pathways including

PI3K, PKC, MAPK and others have been implicated to have roles to play in stem cell differentiation into cardiomyocytes but the exact molecular mechanisms by which these may mediate the stem cell cardiomyocyte lineage commitment and differentiation is currently not fully understood.



Figure 1.18: Extracellular signalling molecules and transcription factor position in cardiogenesis. In response external conditions or factors signal molecules including, BMP, whits and other molecules are released. A range of specific signalling pathways are activated. Downstream proteins including kinases are activated. Cardiac specific transcription factors are expressed and these factors then initiate cardiac differentiation or cardiac lineage commitment. BMP= bone morphogenetic proteins, Oct-4= Octamer-binding protein-4, SHH= sonic hedgehog, FgF= fibroblast growth factor, TGF- $\beta$  = transforming growth factor- $\beta$  and DMSO= dimethylsulfoxide

## MECHANISMS OF HEART DEVELOPMENT

#### 1.20 Mechanisms of heart development

A clear understanding of the basic molecular mechanisms of heart development and the signalling pathways of several transcription factors that promote cardiac differentiation is relevant for stem cell differentiation into cardiomyocytes. Identification of soluble growth factors, transcription factors and signalling cascades capable of priming cardiogenesis is a crucial issue for the *in vitro* development of cardiomyocytes from ES cells. Many key transcription factors including, GATA factor, Nkx2.5 and T-box factors are expressed in the developing heart. In addition to these key transcription factors, many extracellular molecules including BMP and Wnt proteins that are upstream of these transcriptions factors are also expressed in the developing heart.

Early development of the heart is composed of two major stages: cardiac myocyte specification/differentiation and heart morphogenesis (i.e. the organisation of tissue into the specific anatomical structure). A great deal of research information about heart morphogenesis has been collected using *in vivo* and *in vitro* genetic manipulation, biochemical, and structural analyses. On the other hand there is less research information about the factors or mechanisms that regulate the cardiomyocyte differentiation.

The vertebrate heart formation involves cell commitment to cardiac lineage, proliferation, differentiation, migration and integrated multicellular morphogenesis. This complex process is regulated by various signalling molecules, including transcription factors. Vertebrate heart formation is highly

susceptible to errors that can result to congenital heart diseases (CHD). Congenital heart diseases with incidence of nearly 1% of live birth, are the most common forms of genetic diseases (Hoffman and Kaplan, 2002; Garg, 2006; Thom *et al.*, 2006).

#### 1.20.1 Origin of cardiac progenitor cells

#### 1.20.1.1 First heart field (FHF)

The heart is the first embryonic organ to become functional in the developing embryos. Heart formation is a process that is based on a series of progressive lineage restrictions. The whole heart is likely to be derived from a single mesodermal cardiac progenitor cell (Meilhac *et al.*, 2004).

However, an early lineage restriction that occurs prior to heart formation creates two profoundly distinct pools of cardiac progenitor cells. The myocytes of vertebrate heart are derived from (at least) these progenitor cell populations. The cardiogenic committed progenitor cells found in two bilaterally symmetrical regions of anterior lateral mesoderm constitute the first lineage known as the first heart field (FHF) (Buckingham *et al.*, 2005). The FHF forms the cardiac crescent (Meilhac *et al.*, 2004). The FHF cells begin differentiation at the crescent stage at approximately embryonic day 7.5 to 8.0 (E7.5-8.0) in the mouse embryo (corresponds to week 2-3 of human gestation). The two lateral halves of this cell population migrate, fuse at the anterior midline undergo early differentiation at the crescent stage.

These cell then form a simple linear heart tube (Franco *et al.*, 1998; Buckingham *et al.*, 2005).

From the very beginning this tubular heart exhibits slow autonomous peristaltic contractions (Garg, 2006; Srivastava, 2006) and consists of an interior layer of endocardial cells and an exterior layer of myocardial cells. The linear heart tube consists of cardiac progenitors for parts of the ventricles (Buckingham *et al.*, 2005). The other parts or compartments are formed by the addition of cells to its cranial and caudal poles (De La Cruz *et al.*, 1977; Noden, 1991; Redkar *et al.*, 2001). Subsequently, the linear heart tube undergoes rightward looping and is remodelled into a mature four-chambered heart i.e. ventricular and atrial chambers with inflow and outflow regions and valves to control the flow of blood. In mammals and also in birds, the four-chambered heart is separated into two ventricles and two atria by a septum that separates oxygenated blood from deoxygenated blood.

#### 1.20.1.2 Second heart field (SHF)

A progenitor cell lineage occupies a position caudally and medially to the first lineage at the cardiac crescent stage. This is the second heart field (SHF). The SHF is a population of undifferentiated multipotent cardiac progenitor cells that contribute to the heart tube growth at both inflow and outflow poles. A small number of myocytes in left ventricle (LV) and atria are derived from SHF. However The SHF provides the majority of myocytes in the right

ventricle (RV) and outflow tract (OFT) (Kelly *et al.*, 2001; Mjaatvedt *et al.*, 2001; Cai *et al.*, 2003).

The FHF and SHF comprises the ventral and dorsal-medial aspect of the cardiogenic plate respectively (Cai *et al.*, 2003). The FHF differentiates at the cardiac crescent stage while the SHF differentiates relatively late as the second lineage migrates in to join the already differentiated cells of the first lineage. Both lineages appear to be regulated by complex positive and negative signalling networks involving members of the bone morphogenetic protein (BMP), sonic hedgehog (SHH), fibroblast growth factor (FGF), Wnt, and Notch proteins.

## 1.20.2 Parallel between heart formation and ES cell differentiation into cardiomyocytes

Cardiogenesis is very complex process that involves many signalling molecules and signal transduction pathways. This process is regulated by various extracellular signalling molecules and transcription factors. A clear understanding of the basic molecular mechanisms of heart development is essential for the selective differentiation of ES cells to generate pure cardiac cell lineages. Some examples of the parallel between ES differentiation into cardiomyocytes and the vertebrate heart formation mechanisms are discussed to highlight the similarities between *in vivo* and *in vitro* cardiomyogenesis.

Mesodermal lineage in the FHF gives rise to cardiomyocytes. P19 stem cells differentiating into cardiomyocytes express brachyury T (mesodermal lineage commitment marker) (Grepin *et al.*, 1997). Brachyury T is a T-box transcription factor (Kispert *et al.*, 1995) and is necessary for the establishment of the mesoderm in vertebrates (Schulte-Merker *et al.*, 1992). ES cells undergoing cardiac differentiation brachyury T and cardiac specific transcription factors and structural proteins including Nkx2.5, GATA-4, MLC-2a MLC-2v, ANF, MHC and troponin T (Schwanke *et al.*, 2006). This suggests that vertebrate stem cells in culture follow the same cardiac differentiation programme as *in vivo*.

Nkx2.5 is one of the early markers during stem cell differentiation into cardiomyocytes (Nakamura *et al.*, 2003; Naito *et al.*, 2006; Kwon *et al.*, 2007). Nkx2.5 is also one of the earliest markers of heart field mesoderm. In the mouse, the expression of Nkx2.5 starts early in heart progenitor cells in the FHF and SHF continues at a high level in the heart through adulthood (Komuro and Izumo, 1993; Lints *et al.*, 1993; Lee *et al.*, 2004).

In *Xenopus* embryos, the expression of inhibitory or dominant negative mutants forms of Nkx2.5 (Nkx2.3 or XNkx2.5) resulted to complete lack of cardiac gene expression and complete absence of heart formation (Fu *et al.*, 1998; Grow and Krieg, 1998). Similarly in P19 stem cells the expression of a dominant-negative mutant of Nkx2.5 blocked the cell differentiation into cardiomyocytes (Jamali *et al.*, 2001).

GATA-6 is expressed in cardiac mesoderm before gastrulation (Peterkin *et al.*, 2003). In both *Xenopus* and zebrafish embryos, the inhibition of the GATA-6 mRNA translation into protein by anti-sense oligonucleotides led to markedly reduced levels of cardiac proteins in the embryos (Peterkin *et al.*, 2003). Zebrafish embryos depleted of both GATA-5 and GATA-6 can not develop hearts (Holtzinger and Evans, 2007). Mouse embryos null for both GATA-4 and GATA-6 had no hearts (Zhao *et al.*, 2008). Similarly, the inhibition of GATA-4 using GATA-4 antisense transcripts blocked P19 stem cell cardiac differentiation at the premyocardial (cardioblast) (Kispert *et al.*, 1995).

BMPs are expressed in lateral endoderm and ectoderm of vertebrate embryos. BMP signals in the anterior lateral mesoderm promote heart formation (Schultheiss *et al.*, 1997; Andrée *et al.*, 1998; Schlange *et al.*, 2000; Tzahor and Lassar, 2001). It has been shown that BMP soaked beads can induce the ectopic expression of Nkx2.5 and GATA-4 in chick embryos (Schultheiss *et al.*, 1997). Over-expression of noggin (a natural inhibitor of BMP) in P19 stem cells prevent their differentiation into cardiomyocytes while the over-expression of BMP-2 or direct addition of BMP proteins to the culture medium rescued the cell cardiac differentiation (Monzen *et al.*, 1999).

Wnt proteins, Wnts receptors, Wnt receptor antagonists and Wnt antagonists are important for the signalling events that control the development of the heart. These proteins have been detected in the mesoderm of the heart-

forming fields in vertebrates (DeRossi *et al.*, 2000) (Wang *et al.*, 1999; Wheeler and Hoppler, 1999; van Gijn *et al.*, 2001).

In ES cells differentiating into cardiomyocytes it was found that the addition of Wnt3a to EBs increased th expression of cardiac genes and differentiation into beating cardiomyocytes while the addition of Fz8/Fc or Dkk-1 and Fz8/Fc (wnt antagonists) resulted in a complete absence of beating EBs (Naito *et al.*, 2006; Kwon *et al.*, 2007).The SHF marker, Islet1 was similarly affected by Wnt3a and Dkk-1 (Kwon *et al.*, 2007).

#### 1.20.3 Summary of mechanisms of heart development

In summary the few selected examples discussed above, show the remarkable similarities between heart formation in the vertebrates and ES cell differentiation. This similarities show the importance or the relevance of understanding the basic mechanisms of heart development to the differentiation of ES cells into cardiomyocytes.

# STEM CELL RESEARCH AND CARDIOVASCULAR DISEASES

#### 1.21 Stem cell research and cardiovascular diseases

Cardiovascular diseases (e.g. heart failure) are among the leading causes of mortality and morbidity worldwide (Murray and Lopez, 1997) and particularly in the developed world (Rosamond et al., 1998). The high incidence of cardiovascular diseases is a combination of non-modifiable (non-preventable) and modifiable (preventable) risk factors. The non-modifiable risk factors include increasing age (Jousilahti et al.. 1999: Asia Pacific Cohort Studies Collaboration, 2006), gender (being male) (Jousilahti et al., 1999; Lloyd-Jones et al., 1999) and heredity. The modifiable risk factors include tobacco smoke (Kawachi et al., 1993; Raftopoulos et al., 1999; Van Berkel et al., 1999; Kaprio et al., 2000), high blood total cholesterol and LDL (Manolio et al., 1992; Stamler et al., 1993; Woodward et al., 2008), high systolic blood pressure (Rutan et al., 1989; Navas-Nacher et al., 2001; Sundaram et al., 2005; Prugger et al., 2008) and physical inactivity (Kaprio et al., 2000; Kruger et al., 2003; Sundaram et al., 2005; Jenum et al., 2006), obesity (Hubert et al., 1983; Li et al., 2002; Sundaram et al., 2005) and diabetes mellitus (Sundaram et al., 2005; Prugger et al., 2008). The high incidence of cardiovascular diseases makes the search for new therapeutic approaches for improving the functions of the damaged heart is a critical endeavour. Myocardial infarction is associated with irreversible loss of cardiomyocytes. Cardiomyocytes undergo apoptosis during the acute and chronic phases of myocardial infarction. This has been shown in human (Itoh et al., 1995; Saraste et al., 1997; Baldi et al., 2002; Akasaka et al., 2006), rat (Fliss and Gattinger, 1996; Palojoki et al., 2001; Chandrashekhar, 2005; Louhelainen et al., 2007) and the mouse (Bialik et al., 1997; Engel et al.,

2004). Studies have shown that the inhibition of cardiomyocyte apoptosis can slow down post-infarction cardiac remodelling and also improve cardiac function in the mouse (Diwan, 2007), rat (Chandrashekhar, 2005; Chen *et al.*, 2007) and rabbit (Qin *et al.*, 2007).

Myocardial infarction can lead to heart failure. The loss of cardiomyocytes poses a major difficulty or challenge in the treatment of the damaged heart since cardiomyocytes have very limited regeneration potential. This means that, currently, the only effective treatment for severe heart failure is heart transplantation (Anyanwu *et al.*, 2002; Hertz *et al.*, 2002). The worldwide heart transplant data from the International Society for Heart and Lung Transplantation (ISHLT) shows that transplant half-life (i.e. the time at which 50% of those transplanted remain alive) for heart transplant patients is 10 years (Taylor *et al.*, 2007). This represents a good extension to life, especially for elderly patients (e.g. those aged 60 years and above).

Despite the good survival rates of heart transplant patients, heart transplantation is not a viable treatment option and has not delivered or met the high expectation that it generated among patients with heart diseases, when it was first introduced. This is due to the increasing number of patients on the heart transplant waiting list as the result of the continuous decrease in the availability of hearts for transplant. Heart transplantation as a treatment for heart failure is limited by the acute shortage of donor hearts (Anyanwu *et al.*, 2002). The ISHLT data showed that the number of annual reported heart transplants since 1982 peaked at 4428 in 1994 (Trulock *et al.*, 2003; Taylor *et* 

*al.*, 2007). This peak was then followed by a gradual but steady decline as the results of decreased number of heart transplant procedures in many countries (Taylor *et al.*, 2006; Taylor *et al.*, 2007). The latest annual figure for heart transplants worldwide stands at 3095 in 2005 (Taylor *et al.*, 2007). The worldwide number of patients on the waiting list far exceeds the number of hearts available for transplant. In the United Kingdom alone, there were 379 patients on the cardiothoracic waiting list in 2006-2007 while in the same period only 292 cardiothoracic transplants were carried out (UK Transplant, 2007). The high incidence of heart diseases and the acute scarcity of donor hearts underline the urgent need to find alternative therapeutic approaches to heart transplantation.

Stem cell research may lead to a viable alternative to heart transplant for the treatment of heart disease. Several animal and human studies have demonstrated that transplantation of cardiomyocytes (or cardiac progenitor cells) may be an alternative treatment to whole heart transplant for patients with severe heart failure. ES cells therefore represent a potential alternative source of functionally intact cardiomyocytes for the treatment of cardiovascular diseases. Research evidence on the regenerative potential of stem cell-derived cardiomyocytes and progenitor cells for the prevention or treatment of heart failure. In a rat model of myocardial infarction (created by the ligation of the left anterior descending coronary artery) it has been shown that implantation of cardiac progenitor cells rescued scarred infarcted myocardium and improved cardiac function in rats (Bonaros *et al.*, 2006;

Gonzalez *et al.*, 2008; Rota *et al.*, 2008). Similarly, in a mouse myocardial infarction (also created by the ligation of the left anterior descending coronary artery), transplantation of human ES cell-derived cardiomyocytes improved left ventricular function (Cao *et al.*, 2008). The above data show that differentiation of stem cells into functional cardiomyocytes or cardiac progenitor cells follows by their engraftment into the damaged heart to regenerate healthy myocardial tissues to replace the damaged tissues may be used to slow the deterioration of the acutely or chronically damaged heart after myocardial infraction. This means that stem cell-based therapy holds a great potential as a very attractive alternative to heart transplant for the treating heart failure and other cardiovascular diseases.

#### 1.21.1 Stem cell therapy for infarcted hearts

One of the key issues still to be resolved is whether undifferentiated ES cells transplanted into normal or infarcted heart can be guided by the local environment in the heart to differentiate and contribute to cardiac functions. It has been shown that undifferentiated hES are not directed to specifically differentiate into cardiomyocytes in normal or infarcted heart of the nude mouse (Leor *et al.*, 2007). Undifferentiated ES cells transplanted into normal or infarcted heart undergo non-specific lineage differentiation (Leor *et al.*, 2007; Nussbaum *et al.*, 2007). In contrast to the above other groups have shown that transplantation of undifferentiated ES cells into infracted myocardium can induce the cardiac differentiation of the transplanted ES and improved cardiac functions. This has been shown in the rat (Behfar *et al.*,

2002; Hodgson *et al.*, 2004) and in the mouse (Nelson *et al.*, 2006; Singla *et al.*, 2006). Being pluripotent, ES cells may be difficult to differentiate into cardiac specific cells, if undifferentiated ES cells are directly transplanted into infarcted hearts.

As already mentioned *in vitro* differentiation of ES cells into functional cardiomyocytes has been achieved. Under appropriate experimental conditions, ES cells can faithfully follow the same course of cardiomyocyte differentiation as *in vivo* (Wobus *et al.*, 1991; Xu *et al.*, 1998; Kehat *et al.*, 2001; He *et al.*, 2003; Mummery *et al.*, 2003) There is also extensive evidence in the scientific literature that adult stem cells (e.g. MSC) can differentiate into cardiomyocytes (Makino *et al.*, 1999; Fukuda, 2001; Toma *et al.*, 2002; Xu *et al.*, 2004b; Antonitsis *et al.*, 2007; Bartunek *et al.*, 2007). These adult stem cell-derived cardiomyocytes, like ES cell-derived cardiomyocytes, express cardiac specific transcriptions factors (including GATA-4, and Nkx2.5, TEF1, MEF2C, MEF2A and MEF2D) during the course of their differentiation (Makino *et al.*, 1999; Fukuda, 2001). These cardiomyocytes also express  $\beta$ -MHC, desmin, and cardiac  $\alpha$ -actin and cardiac troponin T (Xu *et al.*, 2004b).

Beyond the fundamental question of whether of ES cells can differentiate into functional cardiomyocytes, the next question is: can ES cell-derived cardiomyocytes successfully engraft into adult animal hearts and improve post MI heart? Functional cardiomyocytes have been derived from cultures of human H1, H7, H9, and H14 embryonic stem cells (He *et al.*, 2003). These ES cells were originally derived from human blastocysts (Thomson *et al.*,

1998; Lavon and Benvenisty, 2003). Cardiomyocytes derived from these ES cells displayed expected cardiomyocyte morphology such Z bands and intercalated disks and they express numerous cardiac myocyte specific proteins including  $\alpha$ -cardiac actin, atrial myosin light chain, ventricular myosin light chain,  $\alpha$ -myosin heavy chain, atrial natriuretic peptide, and cardiac troponin T and I (He *et al.*, 2003). They also showed rhythmic contractions and long duration of action potentials (He *et al.*, 2003). Interestingly, similar results have been report with cardiomyocytes derived from MSC (Makino *et al.*, 1999; Fukuda, 2001; Toma *et al.*, 2002; Antonitsis *et al.*, 2007).

Despite the close similarities between stem cell-derived cardiomyocytes and normal cardiomyocytes, the long-term effects of cardiac cell transplantation on cardiac functions are unknown at the moment. However several animal studies demonstrate that successful engraftment of stem-cell derived cardiomyocytes into the adult heart is possible. ES-derived cardiomyocytes can successfully engraft into adult animal hearts and improve post myocardial infarction heart. This has been shown in pig complete atrioventricular block model [created by ablating the His bundle (the major electrical conduction pathway, connecting the atria with the ventricles)] (Kehat *et al.*, 2004) and rat myocardial infarction model (created by the ligation of the left anterior descending coronary artery) (Etzion *et al.*, 2001; Behfar *et al.*, 2002; Muller-Ehmsen *et al.*, 2002; Laflamme *et al.*, 2005; Caspi *et al.*, 2007b; Leor *et al.*, 2007). In the rat model of myocardial infarction, cardiomyocytes differentiated from human ES cells showed *in vivo* functional integration in the animal infarcted hearts (Laflamme *et al.*, 2007). The transplanted cardiomyocytes thickened the wall of the left

ventricle and enhanced ejection fraction and reduced paradoxical systolic bulging of the infarct in post myocardial infarction rats (Müller-Ehmsen *et al.*, 2002; Caspi *et al.*, 2007a). Interestingly results similar to those reported with ES cells have also been obtained using MSC. It has been shown, in the mouse (Kudo *et al.*, 2003; Grauss *et al.*, 2007) and the rat (Hahn *et al.*, 2008; Pasha *et al.*, 2008) models of myocardial infarction (both created by the ligation of the left anterior descending coronary artery), that transplantation of MSC led to reduction in infarct size, improved LV function and survival. The rat myocardial infarction model also showed that the transplanted MSC promoted neovascularisation and myogenesis (Pasha *et al.*, 2008) and a regain of 80–90% of the lost myocardial volume and completely normalised systolic and diastolic cardiac functions (Mangi *et al.*, 2003).

Like the need for prior differentiation of ES cells into cardiomyocytes before transplantation into infarcted hearts, MSC may need specific pre-treatment before transplantation. For example it was also shown that mice transplanted with MSC that were pre-treated with growth factors (i.e. BMP-2, FGF-2, IGF-1) before transplantation had fewer apoptotic cells in infarcted hearts, smaller infarct size and better cardiac function than mice transplanted with untreated MSC (Hahn *et al.*, 2008). Almost identical results were obtained with dogs that were treated MSC that were induce to differentiate by BMP-2, FGF-2 and IGF-1) (Bartunek *et al.*, 2007). The treated MSC expressed muscle-specific (MEF2) and cardiac-specific (GATA-4 and Nkx-2.5) transcription factors (Bartunek *et al.*, 2007), a strong evidence for their differentiation along the cardiac lineage. Myocardial injection of these cardiomyocytes differentiated

from autologous MSC into dogs with chronic myocardial infarction resulted in cardiac regeneration of the chronically infarcted myocardium (Bartunek *et al.*, 2007).

The role of cardiac endothelial cells in the use of stem cells to treat an infarcted heart is also important. It has also been demonstrated that transplantation of human MSC (Grauss *et al.*, 2007) or mouse MSC (Kudo *et al.*, 2003) into ischemic hearts of mice caused the transplanted MSC to differentiate into myocytes and endothelial cells and contribute to reduced infarct size, reduced fibrosis, significant preservation of LV ejection fraction, increased vascularity of the infarct scar and a marked reduction in the thinning of the infarcted wall in the mice (Kudo *et al.*, 2003; Grauss *et al.*, 2007). While it was not determined whether the cardiac improvements reported in these studies were the results of cell fusion (Terada *et al.*, 2002) or MSC actually differentiating into cardiac cells, these results suggest that both cardiomyocytes and cardiac endothelial cells are important for cardiac regeneration in infarcted hearts.

Collectively, the all the above studies with both ES cell-derived and adult stem cell-derived cardiomyocytes have demonstrated that stem cell based therapy can be used to improve the function of the damaged heart and also improve survival after myocardial injury. They also show that stem cell therapy may become an alternative therapeutic approach to the current use of human heart transplants for the treatment of heart failure in patients. However there are still

many critical hurdles that must be overcome before the therapeutic potential of stem cells can be realised.

#### 1.21.2 Challenges facing stem cell-based therapy

The potentials and challenges of stem therapy for treating human cardiovascular diseases are presented below. The challenges are by no means limited to cardiovascular diseases but are equally the concerns of stem cell-based therapies that may be targeting other human diseases. Progress has been made in the identification, isolation characterization, and maintenance of stem cells in culture (Evans and Kaufman, 1981; Martin, 1981; Shamblott *et al.*, 1998; Thomson *et al.*, 1998) but the specific differentiation of ES cell into cardiomyocytes (or any other particular cell type) remains a problem.

#### 1.21.2.1 Challenge of cardiac-specific differentiation of stem cells

Cardiac-specific differentiation of stem cells or the selection of lineages consisting of pure embryonic cardiomyocytes is essential and represents a scientific challenge. The search for growth factors and signalling molecules that are involved in cardiac-specific differentiation is therefore an active area of research. There is need for the design and establishment of suitable and reliable differentiation techniques that can overcome the problem of inefficient and also spontaneous stem cell differentiation. The homogeneous differentiation of stem cells into cardiomyocytes with sufficient yield is a critical requirement for the clinical use of stem cells in cardiovascular medicine. A serious limiting factor of stem cell therapy for the treatment of cardiovascular diseases is the current difficulty in generation sufficient number of stem-cell derived cardiac cells. Across many laboratories, differentiation methods are not only inefficient but sometime difficult to reproduce. The problems of variable reproducibility may be due to the variations in the conditions under which ES cells are routinely maintained and cultured or could be due to variation in differentiation conditions or both. There is a need for common stem cell culture conditions for the maintenance and for differentiation. This will not only simplify maintenance of multiple lines in single research but may help to improve the poor reproducibility of specific differentiation protocols among various research teams. Using two independently derived human embryonic stem cell lines (BG01 and HUES-7), it has been demonstrated in principle that it is possible to develop protocols that can be applied to multiple stem cell lines (Denning et al., 2006). Reproducible differentiation methods at the research levels may eventually develop into technologies that lead to generation of a large number of homogeneous stem-cell derived cell types. Such future technologies may in turn lead to easy availability of needed stem-cell-derived cell types, which can be applied in sufficient doses in order to achieve the maximum desired therapeutic benefits.

#### 1.21.2.2 Challenge of post-transplantation tumour formation

Indefinite self-renewal capacity and pluripotency are the key attractive features of stem cells. However these features also carry the potential risks for tumour formation after transplants. Undifferentiated or partially

differentiated ES cells can subsequently form post-transplantation teratomas (i.e. germline tumours consisting of various cell types). Therefore the development of robust and reproducible methods to derive a homogeneous population of cardiomyocytes from ES cells is essential for therapeutic application of ES cells. A homogeneous cell type not only ensures that a sufficient numbers of the desired cell type for a maximum therapeutic benefit are applied but most importantly it ensures the removal of unwanted cell types that could lead to harmful side effects.

Tumorigenesis following transplantation of undifferentiated ES cells or cell types differentiated from ES has been described in many reports (Thomson et al., 1998; Swijnenburg et al., 2005; Caspi et al., 2007a; Nussbaum et al., 2007). A test of five different human embryonic cell lines found that all produced teratomas after injection into severe combined immunodeficient (SCID)-beige mice (Thomson et al., 1998). Mostly important was the finding that each injected mouse formed teratomas that included cell types of the three germ layers - i.e. gut epithelium (endoderm), cartilage, striated muscle, bone, and smooth muscle and (mesoderm); and neural epithelium, embryonic ganglia and stratified squamous epithelium (ectoderm) (Thomson et al., 1998). The generation of all cell types from the three germ layers is an indication that uncontrolled differentiation occurred. Efforts to remove undifferentiated ES cell or enrich differentiated cell types is therefore an active area of research. It is also possible, in principle, to deplete undifferentiated cells from a heterogeneous cell population using molecular methods. A heterogeneous cell population (obtained from the differentiation of stem cells)

can be treated with a vector that puts an effector gene under control of a gene element that allows the specific and higher expression of transgene in the undifferentiated subpopulation. The transgene product can then be used to separate the undifferentiated cell population from the differentiated cells using specific antibodies. This has the potential to produce a cell population that is relatively enriched for differentiated cell types. For example, it has been shown that cardiomyocytes differentiated from ES cells that were transfected with a fusion gene consisting of alpha-cardiac myosin heavy chain promoter driving the aminoglycoside phosphotransferase (neomycin resistance gene) led to higher harvest of cardiomyocytes with undetectable levels of Oct-4 (Zandstra et al., 2003). The undetectable level of Oct-4 suggests that undifferentiated ES cells were depleted from the ES cell-derived cardiomyocyte population. Similarly, it has been shown that, when a construct of murine  $\alpha$ -MHC promoter driving the neomycin-resistance gene, was introduced into hES3 cells, a differentiated cell population with more than 99% cardiomyocytes resulted, after cardiac differentiation that was followed by G418 selection (Xu et al., 2008). These studies demonstrated not only that ES cell-derived cardiomyocytes can be enriched but also that the elimination of undifferentiated cells is possible. Another potential procedure to remove undifferentiated ES cells is by selection against undifferentiated ES cells using monoclonal antibodies against well-characterised human ES cell surface antigens (e.g.SSEA-4) (Shibata et al., 2006). Flow-cytometric cell sorting (i.e. fluorescence-activated cell sorting; FACS) and monoclonal antibodies can also be used to enrich differentiated cell types from

undifferentiated stem using antibodies against the specific marker of the differentiated cell types (Fukuda *et al.*, 2006).

#### 1.21.2.3 Summary of Stem cell research and cardiovascular diseases

The search for new therapeutic approaches for improving the function of the damaged heart is urgent. Cardiomyocytes have very limited regeneration potential so their loss, one of the underlying causes of myocardial infarction and eventual heart failure, is a major problem for treating heart failure. Currently, heart transplantation is the only cure for heart failure but this is not a viable treatment option due to the acute shortage of donor hearts.

There is sufficient evidence to conclude that the therapeutic application of stem cell-derived cardiomyocytes can be a realised. Various animal models have shown that stem cell-derived cardiomyocytes can successfully engraft into adult hearts and can also improve the function of infarcted hearts. However, realisation of the full potential of stem cells for the treatment of various heart diseases will be possible only after the resolution of key issues including the limited availability of homogenous stem-cell derived cardiomyocytes, potential immune rejection, post transplant teratoma formation and others. While there have been advances in the differentiation of stem cells into cardiomyocytes and better understanding of potential post transplantation risks (e.g. teratoma formation), a fully understanding about the above problems and how to avoid them is needed. The various reports of post-transplantation tumour formation in various models of stem cell-based

cell therapy and the various attempts to resolve this issue is a reflection of the seriousness of all the critical hurdles that must be overcome before stem cell therapy can become a clinical reality.

#### **1.22 Aim of the PhD project**

As already discussed above, ES cell-derived cardiomyocytes are potential or promising alternative to heart transplant in the treatment of heart diseases for cardiac repair. The realisation of stem cell therapy is been currently hampered by the critical lack of reliable, selective and readily reproducible differentiation methods. This difficulty is in turn due to the lack of a clear understanding of mechanisms of signal transduction pathways that mediate stem cell differentiation into cardiomyocytes. Without a clear understanding of the mechanism of stem cell cardiac lineage commitment and differentiation into cardiomyocytes, it would be difficult to obtain homogenous and sufficient number of cardiomyocytes for use in stem cell base therapy. The specific, efficient and selective differentiation of stem cells into cardiomyocytes is absolutely necessary for realisation of stem cell therapy for treating the damaged heart and this remains a tough scientific challenge.

The aim of this PhD thesis is therefore to investigate the cell signalling pathways that are responsible for mediating the differentiation of stem cells into cardiomyocytes. This PhD project, besides being a scientific challenge for the reason stated above, it has another challenging dimension, in that, it is the first of its kind in the host laboratory. The project is the first attempt by our

laboratory to direct our strong background in cell signalling research to the

studies of stem cell differentiation into cardiomyocytes.

The project specific aims/objectives are as follows:

- 1. The first objective is to establish cardiomyocyte differentiation models using H9c2 and P19 stem cells and the use of these models to accomplish the other specific objectives, which are, to:
- 2. Investigate the role of nitric oxide (NO) in the differentiation of stem cells into cardiomyocytes
- 3. Investigate the role of phosphoinositide 3-kinase (PI3K) signalling pathway in the differentiation of stem cells into cardiomyocytes
- 4. Investigate the role of protein kinase C (PKC) signalling pathway in the differentiation of stem cells into cardiomyocytes
- 5. Investigate the role of nuclear factor-kappa B (NF-kB) signalling pathway in the differentiation of stem cells into cardiomyocytes
- 6. Investigate the role of p38 mitogen-activated protein kinase (p38 MAPK) signalling pathway in the differentiation of stem cells into cardiomyocytes

Possible alternative research approaches, to the above that we took, include investigating the roles of traditional molecules (e.g. BMP, TGF- $\beta$ , Fgf, Wnt and growth factors) that have been implicated in stem cell differentiation into cardiomyocytes The signal transduction pathways that we investigated may be downstream of these molecules. Therefore an understanding of the roles of these downstream signal transduction pathways, in the differentiation of stem cells into cardiomyocytes, may give us a better insights or understanding of the roles of the roles of these traditional molecules mentioned above. This project is using two relevant cardiomyocyte differentiation models (H9c2 cell and P19

stem cell models) to investigate the roles of the signalling pathways (listed above) in stem cell differentiation into cardiomyocytes.

## **CHAPTER 2.0**

### MATERIALS AND METHODS

#### 2 Materials and Methods

#### 2.1 Resuscitation and culture of H9c2 cells from frozen stock

The H9c2 (2-1) line (H9c2 cells), at passage 8, was purchased from the European Collection of Cell Cultures [ECACC (No: 88092904), Salisbury, Wiltshire, UK]. On day 1, a 1ml vial of frozen H9c2 stem cells was quickly thawed in a water bath at 37°C (Katayama *et al.*, 1997). The cells were then resuspended in 9ml of fresh complete culture medium (i.e. Dulbecco's Modified Eagle's Medium (DMEM) supplemented with 10% foetal bovine serum (FBS), 100units/ml penicillin/100µg/ml streptomycin) in a 15ml centrifuge tube. All the components of the full medium were purchased from Invitrogen, UK. The tube was spun in a centrifuge at 1000 RPM at 4°C for 5 minutes. The medium was then aspirated and the pellet was kept. Next, 5 ml of new fresh complete medium was added to the tube and the clumped cells were dissociated by trituration (i.e. pipetting up and down). The dissociated cells were transferred into a new T25 (25 cm<sup>2</sup> of cell growth area) culture flask in a total volume of 5ml of fresh culture medium and cultured in a cell culture incubator at 37°C, 95% air and 5% CO<sub>2</sub>.

On day 2, the cells in the T25 flask were observed under a microscope to ensure that they were growing and also healthy. The spent medium was removed at day 3 and replaced with fresh complete growth medium. The cells were subsequently trypsinised and subcultured on day 4, as described below (Section 2.4), when they were approximately 60-70 % confluent. The cell

suspension was then placed in a T75 (75 cm<sup>2</sup> of cell growth area) flask and placed back into the incubator at 37°C, 95% air and 5% CO<sub>2</sub>. The cells were left to become 60-70 % confluent and then brought into suspension again. The cell suspension was split into five parts (1:5 splitting). One part (1ml) of the cell suspension was subcultured in a new T75 flask. The cells from this flask were used to generate more cells for routine cell culture (i.e. cell generation for experiments and storage). The rest of the cell suspension (i.e. 4 parts or 4ml) were frozen and stored in liquid nitrogen for future use as described below (Section 2.5).

# 2.2 Resuscitation and culture of P19 stem cell from frozen stock

P19 embryonal carcinoma stem cell (P19 stem cells) (passage 18) were purchased from the European Collection of Cell Cultures [ECACC (No: 95102107), Salisbury, Wiltshire, UK]. The resuscitation and culture of P19 stem cell from frozen stock was carried out as already described above for H9c2 cells (Section 2.1). The only modification made to the above was that the frozen vial of P19 stem cells was directly cultured in a T75 flask (i.e. without the initial culture in T25 as was the case with H9c2 cells). P19 stem cells do not need the initial culture in a small growth area (i.e. T25 flask) as they have high growth rates. The frozen vial of P19 stem cells was resuspended in fresh complete culture medium (i.e. alpha Minimum Essential Medium ( $\alpha$ -MEM) supplemented with 10% FBS and 100units/ml penicillin/100µg/ml streptomycin) in a 15 ml centrifuge tube. All the

components for the full medium were purchased from Invitrogen, UK. The cells were processed exactly as detailed for H9c2 cells.

#### 2.3 Routine cell culture

#### 2.3.1 Routine cell culture of H9c2 cells

H9c2 cells were routinely cultured in full culture medium (i.e. DMEM supplemented with 10% FBS, 100units/ml penicillin/100µg/ml streptomycin). The cells were maintained in a tissue culture incubator at 37°C, 95% air and 5% CO<sub>2</sub>. Cells were routinely subcultured when 60-70% confluence. This precaution was to prevent the loss of differentiation potential should the cells reach confluency.

#### 2.3.2 Routine cell culture of P19 stem cells

P19 stem cells were routinely cultured in full culture medium (i.e. α-MEM supplemented with 10% FBS, 100units/ml penicillin/100µg/ml streptomycin). The cells were maintained in a tissue culture incubator at 37°C, 95% air and 5% CO<sub>2</sub>. The cells were routinely subcultured when 60-70% confluent. This precaution of not allowing the cells to become confluent was to prevent their spontaneous or non-directed differentiation.

#### 2.4 Subculture of cells

P19 stem cells or H9c2 cells were subcultured when 60-70 % confluent. This was carried out by removing the culture medium from the flask, washing cells 3 times with warm (37°C) phosphate buffered saline (PBS) (Invitrogen, UK) and then adding 3ml of warm (37°C) 0.05% trypsin/EDTA solution (Invitrogen, UK) cells in T75 flasks (i.e. 120µl trypsin/cm<sup>2</sup>). Flasks were then placed in an incubator at 37°C, 95% air and 5% CO<sub>2</sub> for 3-5 minutes. The trypsin was subsequently inactivated when all the cells had detached by adding 7ml of full growth medium. The serum in the full growth medium contains inhibitors that inhibit or neutralise the trypsin. If the cells were not required for studies, the cell suspension was placed in a T75 flask and cultured to 60-70% confluent. If required, the cells were used for experiments.

#### 2.5 Cryopreservation of cells

Excess cells or cells not required for immediate experimentation, were routinely cryopreserved and stored for later use. To freeze the cells, they were brought into suspension as before. The cell suspension was transferred into a 15ml centrifuge tube and spun at 1000 RPM for 5 minutes at 4°C. The medium was then aspirated and the pellet was kept. The H9c2 cell pellet was re-suspended in freezing medium (i.e. 90% FBS and 10% dimethylsulfoxide (DMSO) (Invitrogen, UK)). The P19 stem cell pellet was re-suspended in freezing medium (i.e. 90% FBS or 10% glycerol (Invitrogen, UK)). The cell suspension was gently but thoroughly triturated to break apart

any cell clumps. The cell suspension was then aliquoted in 1ml volumes in cryovials and placed in a Thermo Scientific NALGENE® Mr. Frosty (i.e. a controlled rate freezing container) (Fisher Scientific, UK) for freezing at -80°C overnight before transferring into liquid nitrogen for long term storage. Normally one T75 flask of cells was frozen into two cryovials, each containing 1ml of resuspended cells (i.e. splitting ratio of 1:2)

#### 2.6 Quantification of cells

In order to plate cells at the required densities, the total number of cells from each flask was determined using the Trypan Blue exclusion assay. The flask of cells was first trypsinised as described (Section 2.4). The cell suspension was transferred into a 15ml centrifuge tube and spun at 1000 RPM for 5 minutes at 4°C. The medium was then aspirated and the pellet was kept. The pellet resuspension was carried out in a small known volume (usually, 1-2 ml) of complete cell culture medium for each T75 flask of cells. Then 100 $\mu$ l of the cell suspension was mixed with an equal volume of 0.4% w/v) Trypan blue in an eppendorff tube. A moistened cover-slip was slipped over the chamber of a clean Neubauer haemocytometer. A 10 $\mu$ l aliquot of the cell/Trypan blue suspension was then added to each chamber on the haemocytometer and viewed under a light microscope using a 10X magnification on the eyepiece lens and a 20X objective lens.

Cells were counted in the two chambers of haemocytometer. All the cells in the 1mm center square (E) and four 1mm corner squares (A, B, C and D)
were counted in the first chamber (Figure 2.10). Separate counts of viable (seen as translucent with no blue staining) and non-viable (stained blue) cells were recorded. This procedure was repeated for the second chamber. In total, the cells in ten squares (five from each chamber) were counted.



*Figure 2.10: A schematic diagram (cartoon) of a haemocytometer* a. Diagram shows a chamber of a haemocytometer and the areas (A, B, C, D and E) from which the cells were counted.

The cell concentration in the cell suspension was determined as follows:

Concentration of viable cells (cells/ml) = <b>A x C x D</b>		
Where:		
A is the mean number of viable cells counted i.e.	Total number of viable cells counted	
<b>C</b> is the dilution factor and	Number of squares used	
<b>D</b> is the correction factor ( <b>x10</b> <sup>4</sup> )		
Total number of viable cells per flask =		
concentration of viable cells X volume of cell suspension		

#### 2.7 Cell viability (MTT) assay

In order to eliminate any non selective cytotoxic actions of the drug used, cell viability was determined after each treatment by monitoring the cellular metabolism of (3-[4, 5-dimethylthiazol-2-yl] 2, 5-diphenyl-tetrazolium bromide (MTT) to formazan. This is a widely used cytotoxicity assay in which the water soluble tetrazolium salt is converted into an insoluble purple compound. formazan, by cleavage of the tetrazolium ring by mitochondrial dehydrogenases in viable cells (Mosmann, 1983). This assay is therefore largely a measure of mitochondrial activity. The MTT can be used to determine cytotoxicity of compounds. The water insoluble formazan dissolves in acidic isopropanol. The absorbance of formazan in the solution is proportional to the concentration of MTT metabolised and is believed to be a good indicator of cell viability. Toxic concentrations of a compound would lead to a reduction in the mitochondrial dehydrogenase ability to metabolise (MTT) to formazan. Toxic concentrations of a compound would therefore produce a reduced absorbance in the MTT assay. In contrast to the MTT assay, the Trypan Blue exclusion assay used to determine the number of cells for experiments, is based on the ability of viable (live) cells to exclude Trypan Blue and appear translucent (with no blue staining) under the microscope. Dead (non-viable) cells, on the other hand, cannot exclude Trypan Blue and are therefore stained blue, as the result of Trypan blue uptake, and appear blue under the microscope.

H9c2 cells  $(2x10^3 \text{ cells/cm}^2)$  were seeded in p60 dishes or 6-well plates in full culture medium and allowed to grow to 60-70% confluence. The full growth medium was removed and the cells were washed twice with warm  $(37^{\circ}\text{C})$  PBS. The differentiation medium (i.e. DMEM supplemented with 1% FBS and 100units/ml penicillin/100µg/ml streptomycin) containing the appropriate concentration of the selected drug was added to the cells. The cells were incubated at  $37^{\circ}$ C, 95% air and 5% CO<sub>2</sub> for defined time periods. Control cells were incubated in the differentiation medium without the selected drug.

At the end of the incubation period, 200µl of MTT (Sigma, UK) stock (5mg/ml, made in PBS) was added to 2ml of culture medium in each p60 dish, giving a final concentration of 0.5mg/ml. The cells were incubated at 37 °C, 95% air and 5% CO<sub>2</sub> for one hour. At the end of the incubation period, the cells were removed from the incubator and the culture medium was removed. The cells were incubated for 10 minutes with 1ml of acidic isopropanol (0.04M HCl in absolute isopropanol) to dissolve the formazan crystals. The resulting formazan solution was thoroughly vortexed to dissolve precipitates and the absorbance was measured at 540 nm using a microplate reader (Labsystems Multiskan Ascent, UK).

The cell viability in the control cells (untreated cells) was considered as 100%. The drug concentration-dependent (or time-dependent) viability in the treated cells was calculated a percentage of control viability. The cell viability at each drug concentration (or time point) was calculated as a percentage of control viability at that drug concentration (or time point). The viability values

were calculated using the formula shown below and reported as percentage of control cells.

% Cell Viability =  $\frac{A_{540} \text{ Drug Treated Cells}}{A_{540} \text{ Control Cells}} \times 100\%$ 

### 2.8 Differentiation of H9c2 cells into cardiomyocytes

#### 2.8.1 Initiation of H9c2 Cell differentiation into cardiomyocytes

H9c2 cells  $(2x10^3 \text{ cells/cm}^2)$  (passage number not more than 25) were seeded in p60 dishes or 6-well plates and allowed to become 60-70% confluent in full culture medium. The monolayers of cells were washed twice with warm (37°C) PBS and re-incubated in differentiation medium (DMEM supplemented with 1% FBS. 100units/ml penicillin/100µg/ml streptomycin) to initiate differentiation (Kageyama et al., 2002; Pagano et al., 2004; Hunter et al., 2007). The total proteins in a parallel dish were extracted and used as the zero time point when analysing changes in expression of targeted cell differentiation markers. Other dishes were incubated in the differentiation medium for defined time periods. Changes in cell morphology were recorded for up to 12 or 14 days by taking photos every two days, using a standard microscope and a digital eyepiece. Cell lysates were generated in parallel at the end of each time point and stored at -20°C until analysed by western blotting, using specific monoclonal antibodies against cardiac myosin heavy chain (MHC), cardiac troponin I (cTnI) or cardiac ventricular myosin light chain (MLC-1v), as described below (Section 2.14)

## 2.8.2 Effects of drugs (signalling pathway inhibitors) on the differentiation of H9c2 cells into cardiomyocytes

To determine whether the activation of a particular signalling pathway is required during the differentiation of H9c2 cells into cardiomyocytes, a drug (i.e. selective and specific inhibitor of that signalling pathway) was used to block the pathway. The effects of the pathway inhibition were then observed or measured. For the determination of the effects of drugs on the differentiation of H9c2 cells into cardiomyocytes, both the designated control cells and the cells to be treated with the selected drug (i.e. drug treated cells) were separately seeded in p60 dishes or 6-well plates at a density of  $2 \times 10^3$ cells/cm<sup>2</sup>. The cells were placed in the incubator at 37°C, 95 % air and 5% CO<sub>2</sub> and allowed to become 60-70% confluent in full culture medium in the absence of the selected drug. The culture medium was removed and the monolayers of cells were washed twice with warm (37°C) PBS. The control cells were re-incubated in fresh full growth medium. The drug treated cells were re-incubated in the fresh full growth medium supplemented with the appropriate concentration of the selected drug. After one hour of incubation the culture medium was removed from both the control and the treated cells. The monolayers of cells were washed twice with warm (37°C) PBS. The control cells were re-incubated in fresh differentiation medium (i.e. DMEM supplemented with 1% FBS, 100 units/ml penicillin/100 µg/ml streptomycin) to initiate differentiation. The fresh differentiation medium added to the control H9c2 cells did not contain any drug. The drug treated cells were re-incubated in the fresh differentiation medium supplemented with the appropriate concentration of the selected drug. The total proteins in a parallel dish were

extracted and used as the zero time point when analysing changes in expression of targeted cell differentiation markers. Other dishes were incubated in the differentiation medium in the presence or absence of selected drug as appropriate for defined time periods. Changes in cell morphology were recorded for up to 12 or 14 days by taking photos every two days, using a standard microscope and a digital eyepiece. Cell lysates were generated in parallel at the end of each time point and stored at -20°C until analysed by western blotting, using specific monoclonal antibodies against cardiac myosin heavy chain (MHC) or ventricular cardiac myosin light chain 1(MLC-1v) as described below (Section 2.14).

# 2.9 Differentiation of P19 Stem cells into beating cardiomyocytes

#### 2.9.1 Initiation of P19 cell differentiation into cardiomyocytes

To initiate differentiation, P19 stem cells  $(1 \times 10^{6} \text{ cells})$  (passage not more than 30) were seeded in P100 Petri (i.e. microbiological) dishes in the differentiation medium (i.e.  $\alpha$ -MEM supplemented with 10% FBS, 100 units/ml penicillin/100 µg/ml streptomycin and 0.8% DMSO). P19 cells grown in Petri dishes (i.e. non-adherent conditions) do not attach but instead form cell aggregates known as embryoid bodies (EB) which subsequently differentiate into cardiomyocytes (McBurney *et al.*, 1982; Habara-Ohkubo, 1996; Brewer *et al.*, 2005). Embryoid bodies were allowed to form over a period of 4 days before transferring these into cell culture grade p60 dishes containing 4ml of full growth medium without DMSO. The day on which the embryoid bodies

were transferred to cell culture grade dishes was labelled or designated as day 0. According to this labelling system, day 2 and day 4, for example, refer to the sixth and eighth day respectively since the P19 stem cell differentiation experiment was started with the setup of embryoid bodies. The growth medium was changed every other day and the cell monolayers were monitored daily for up to 12 or 14 days for the appearance of beating cardiomyocytes. Photographs were taken and where beating cardiomyocytes were present, videos were also recorded. Cell lysates were generated in parallel for western blot analysis using specific monoclonal antibody against cardiac troponin I (TnI) as described below (Section 2.14).

## 2.9.2 Effects of drugs (signalling pathway inhibitors) on P19 stem cell differentiation into cardiomyocytes

To determine whether the activation of a particular signalling pathway is required during the differentiation of P19 stem cells into cardiomyocytes, a selective and specific inhibitor of that signalling pathway was used to block the pathway. The effects of the pathway inhibition were then observed or measured. For the determination of the effects of drugs on the differentiation of P19 stem cells into cardiomyocytes, both control and drug-treated P19 stem cells ( $1 \times 10^6$  cells) were separately seeded in p100 Petri dishes. Control P19 stem cells were seeded in differentiation medium (i.e.  $\alpha$ -MEM supplemented with 10% FBS, 100units/ml penicillin/100µg/ml streptomycin and 0.8% DMSO). The drug-treated cells were initially seeded in culture medium as above but in the absence DMSO. The cells were placed in the incubator at 37°C, 95% air and 5% CO<sub>2</sub>. After one hour of incubation, the cell

suspensions in the Petri dishes of the drug-treated cells were supplemented with 0.8% DMSO and the cells were returned to the incubator. The control cells were left as before.

Four days after DMSO treatment, the EBs were transferred to cell culture grade p60 dishes or 6-well plates in full growth medium (i.e. without DMSO). Hereafter, the growth medium was changed every other day. The cell monolayers were monitored daily for up to 12 or 14 days for the appearance of beating cardiomyocytes. Cell photographs were taken and where beating cardiomyocytes were present, videos were also recorded. Cell lysates were generated in parallel for western blot analysis using appropriate selective antibodies as described below (Section 2.14).

### 2.9.3 Effects of delayed inhibition of signalling pathways on P19 stem cell differentiation into beating cardiomyocytes

The activation of a signalling pathway may be required only at a particular stage of a biological process. To determine the stage at which a particular signalling pathway is activated or required during the differentiation of P19 stem cells into cardiomyocytes, a selective and specific inhibitor of that signalling pathway was separately added to one batch of cells during DMSO-induced differentiation at the EB formation stage. The inhibitor was also separately added to another batch of cells four days after DMSO-initiated differentiation (i.e. inhibitor was only added after the formation of EB in DMSO). EB formation can be carried out over several days but a four day period is commonly used. We therefore decided to test the effects of the

addition of inhibitor after four days of DMSO-induced differentiation of P19 stem cells.

The control P19 cells were treated with differentiation medium (i.e. α-MEM supplemented with 10% FBS, 100units/ml penicillin/100µg/ml streptomycin and 0.8% DMSO) containing an appropriate concentration of the selected inhibitor on the day the EBs were set up in Petri dishes. Control cells were exposed to the inhibitor only during the EB formation stage (i.e. no inhibitor exposure after the EB formation stage).

The <u>delay-inhibition P19 cells</u> were not initially treated with the inhibitor on the day the EBs were setup and were also not treated with the inhibitor throughout the EB formation stage, in Petri dishes. The delay-inhibition P19 cells were initially treated with differentiation medium (as above) in the absence of the selected inhibitor on the day the EBs were set up in Petri dishes. The delay-inhibition P19 stem cells were only treated with the differentiation medium containing the inhibitor from the day the EBs were transferred from Petri dishes to cell culture dishes (i.e. only after the four-day EB formation stage).

For differentiation experiments, both control P19 cells and delay-inhibition P19 cells  $(1 \times 10^6 \text{ cells})$  were separately seeded in p100 Petri dishes. Control P19 cells were seeded in DMSO-free medium containing the appropriate concentration of the selected inhibitor. The delay-inhibition P19 cells were in seeded in differentiation medium (i.e.  $\alpha$ -MEM supplemented with 10% FBS,

100units/ml penicillin/100µg/ml streptomycin and 0.8% DMSO) in the absence of the selected inhibitor. All the cells were placed in the incubator at  $37^{\circ}$ C, 95 % air and 5% CO<sub>2</sub>. After one hour of incubation, the cell suspensions in the Petri dishes of the control P19 cells were supplemented with 0.8% DMSO (i.e. became differentiation medium plus drug) and the cells were returned to the incubator. The delay-inhibition P19 cells were left as initially set (i.e. differentiation medium only).

Four days after DMSO treatment, the EB of the control P19 cells were transferred to cell culture grade p60 dishes or 6-well plates in full growth medium (i.e. without DMSO). For the EB of the delay-inhibition P19 cells, after EB transfer to cell culture grade p60 dishes or 6-well plates, the differentiation medium was supplemented with the appropriate concentration of the selected inhibitor (i.e. became differentiation medium plus drug) for upto day 12.

The monolayers of both groups of cells were monitored daily for up to 12 days for the appearance of beating cardiomyocytes. Cell photos were taken and where beating cardiomyocytes were present, videos were also recorded. Cell lysates were generated in parallel for western blot analysis using a specific monoclonal antibody against cardiac troponin I as described below (Section 2.14).

# 2.10 Cross-talks between PI3K signalling pathway and other signalling pathways (PKC and p38 MAPK)

## 2.10.1 Time course of Akt phosphorylation at serine 473 residue in H9c2 cells differentiating into cardiomyocytes

To examine possible crosstalk(s) between the PI3K signalling pathway and the other signalling pathways (PKC and p38 MAPK), cross talk studies were carried out. For determination of the time course of Akt phosphorylation at serine 473 residue, H9c2 cells were seeded in p60 dishes at a density of 2x10<sup>3</sup> cells/cm<sup>2</sup> in full growth medium (i.e. DMEM supplemented with 10% FBS and 100units/ml penicillin/100µg/ml streptomycin). The cells were placed in the incubator at 37°C, 95 % air and 5% CO<sub>2</sub> and allowed to become 60-70% confluent. The culture medium was removed and the monolayers of cells were washed twice with warm (37°C) PBS. All the cells were incubated in fresh serum-free culture medium (i.e. DMEM supplemented with 100units/ml penicillin/100µg/ml streptomycin but no FBS) for 24 hours. After this period, the serum-free culture medium was removed and the monolayers of cells were washed twice with warm (37°C) PBS. Control cells were re-incubated in fresh serum-free medium and 1% FBS-treated H9c2 cells (i.e. cells treated with low serum or e differentiation medium) were re-incubated in fresh differentiation medium (i.e. DMEM supplemented with 1% FBS and 100units/ml penicillin/100µg/ml streptomycin). All the cells were placed in the incubator at 37°C, 95 % air and 5% CO<sub>2</sub> and left for 0, 5, 10, 20, 30 minutes, 1, 2, 4, 6, 48 and 72 hours. Cell lysates were generated at the end of each time point and stored at -20°C until analysed by western blotting, using a

specific monoclonal antibody against phospho-Akt (Ser473) (Cell Signaling Technology via New England Biolabs, UK) as described below (Section 2.14).

#### 2.10.2 Effects of drugs (signalling pathway inhibitors) on Akt phosphorylation at serine 473 residue in H9c2 cells differentiating into cardiomyocytes

To examine the effects of drugs (i.e. selective and specific inhibitors of signalling pathways) on Akt phosphorylation at serine 473 residue, H9c2 cells were seeded in p60 dishes at a density of 2x10<sup>3</sup> cells/cm<sup>2</sup> in full growth medium (i.e. DMEM supplemented with 10% FBS and 100units/ml penicillin/100µg/ml streptomycin). The cells were placed in the incubator at 37°C, 95 % air and 5% CO<sub>2</sub> and allowed to become 60-70% confluent. The culture medium was removed and the monolayers of cells were washed twice with warm (37°C) PBS. The cells were incubated in fresh serum-free culture medium (i.e. DMEM supplemented with 100units/ml penicillin/100µg/ml streptomycin but no FBS) for 24 hours. After this period, the serum-free culture medium was removed from the dishes and the monolayers of cells were washed twice with warm (37°C) PBS and the following cell treatment were carried out. Three experimental groups were then designated according how each particular group of cells was treated.

**Control H9c2 cells:** Control H9c2 cells (labelled as control) were treated with only serum-free medium throughout the Akt phosphorylation procedure (including the initial 24 hour treatment with serum-free medium). These cells

were not treated with any inhibitor or drug and were also not exposed to 1% FBS medium.

**1 % FBS-treated H9c2 cells:** 1 % FBS-treated H9c2 cells (labelled as 1% FBS) were treated only with differentiation medium (i.e. DMEM supplemented with 1% FBS and 100units/ml penicillin/100µg/ml streptomycin) (i.e. after the initial 24 hour treatment with serum-free medium). These cells were also not treated with any inhibitor or drug.

**Inhibitor-treated H9c2 cells** were labelled with the name of the inhibitor. After the initial 24 hour treatment with serum-free medium, the Inhibitortreated H9c2 cells were first treated in the appropriate concentration of the selected drug in fresh serum-free medium. After one hour incubation at 37°C, 95 % air and 5% CO<sub>2</sub>, the serum-free medium, containing the drug, was removed and the differentiation medium (i.e. DMEM supplemented with only with 1% FBS and 100units/ml penicillin/100µg/ml streptomycin) containing a fresh preparation the appropriate concentration of the selected drug was added to the cells.

After the appropriate treatment, cells were placed in the incubator at  $37^{\circ}$ C, 95 % air and 5% CO<sub>2</sub> and left for 15 minute incubation. After this period, cell lysates were generated and stored at -20°C until analysed by western blotting, using specific monoclonal antibodies against Akt phosphorylated at serine 473 residue (phospho-Akt-Ser473) as described below (Section 2.14).

### 2.11 Total protein extraction for western blotting

When required, lysates were generated from cells using a lysis buffer consisting of 20mM Tris-HCl, pH7.4, 1% SDS and 150mM NaCl. Cells were washed twice with ice-cold PBS and up to 500µl of hot (heated to 95°C) lysis buffer was added to cells in wells or dishes. The plates/dishes were immediately placed on ice and then 2.5µl of 200mM PMSF (i.e. 1mM PMSF final concentration) was added to cells in the dish. The cells were scraped off the dishes. The dishes with the scraped cells were left on ice for 15 minutes. During this time the cell lyates were regularly agitated by gentle trituration to obtain a homogenous lysate. The cell lysate or extract was transferred into a microcentrifuge tube and heated at 95°C for 5 minutes. The sample was then sonicated at 5-10 second interval for 1 minute. The homogenate or clear cell extract was transferred into a microcentrifuge tube and centrifuged at 1400 RPM at 4°C for 30 minutes to pellet insoluble materials. The pellet was discarded. The supernatant was transferred to a new microcentrifuge tube. The total proteins in the lysates were determined (see Section 2.12). The cell lysates or protein samples were prepared for loading onto gels as described below (section 2.13.1). Western blotting was usually carried out immediately or samples were stored at -20°C for later western blot analysis (Section 2.14).

# 2.12 Total protein determination in cell lysates using the BCA assay

The Pierce bicinchoninic acid (BCA) protein assay (Perbio Science, UK Ltd) was used to determine the total protein concentration in the cell lysates. This is a detergent-compatible formulation for the colorimetric detection and quantitation of total proteins. This method is a combination of the reduction of cupric ions [Cu (II)] to cuprous ions [Cu (I)] by proteins in an alkaline medium (i.e. biuret reaction) and a selective and sensitive colorimetric detection of the cuprous ions. The purple-colour reaction product of BCA assay is formed by the chelating of one Cu (I) ion by two molecules of BCA (Smith *et al.*, 1985; Wiechelman *et al.*, 1988). The purple complex formed has a maximum absorbance at 562nm. The absorbance is directly proportional to protein concentration (Smith *et al.*, 1985).

The colour produced from this reaction is stable and increases in a proportional manner over a broad range of increasing protein concentrations (Smith *et al.*, 1985). This colour complex exhibits a strong absorbance at 562 nm. The BCA method is not a true end-point method (i.e. colour development does not stop at the time of absorbance measurement). However, following incubation, the rate of continued colour development is sufficiently insignificant to allow large numbers of samples to be assayed together.

## 2.12.1 Preparation of bovine serum albumin (BSA) standards for the BCA assay

The microplate version of the BCA assay was carried out on the cell lysates using bovine serum albumin (BSA) (Sigma, UK) standard concentrations (0.0 to  $0.5\mu g/\mu I$ ). The microplate version of the BCA is linear between 0.02 to  $2\mu g/\mu I$  of BSA or proteins. The BSA standard concentrations were prepared as shown in Table 1 from a 1mg/mI stock solution of BSA. All dilutions were carried out in the same lysis buffer used to lyse cells.

Volume (µl) of	Volume (µl) of	Final BSA
1mg/ml solution	lysis buffer	concentration (µg/µl)
0	1000	0
25	975	0.025
50	950	0.050
100	900	0.100
200	800	0.200
300	700	0.300
400	600	0.400
500	500	0.500

Table 1: Preparation of Diluted Albumin (BSA) Standards

#### 2.12.2 Procedure of the BCA assay

Pierce BCA assay reagent A contains sodium carbonate, sodium bicarbonate, bicinchoninic acid and sodium tartrate in 0.1M sodium hydroxide. Reagent B contains 4 % (w/v) Copper (II) sulfate pentahydrate (CuSO4  $\cdot$  5H<sub>2</sub>O). A 1:50 dilution of reagent B in reagent A was prepared. A 10 µl aliquot of either standards or cell lysate was added in triplicate to a 96-well plate. To each well containing standards or cell samples, 100µl of the BCA reagent was added.

The lid or cover was put on the plate. The plate was incubated for 45-60 minutes at room temperature and the absorbance measured at 620nm on a multiscan plate reader (Labsystems Multiskan Ascent, UK). The average absorbance measured for the blanks (i.e. lysate buffer alone) was subtracted from the absorbance measurements of all other values. A protein standard curve was generated by plotting the average corrected absorbance (i.e. measured absorbance – blank absorbance) for each BSA standard against its concentration in  $\mu g/\mu l$ , using Microsoft Excel 2003 software. The equation for the best fit of standard curve was used to determine the protein concentration of each unknown sample. Alternative to the standard curve generation, using Microsoft Excel, the linear regression routine in GraphPad Prism software (version 4) was used to work out the protein concentration in lysates. In Microsoft Excel and GraphPad Prism software, only standard curves with r<sup>2</sup> value (a measure of linearity of BSA standard concentrations) of 97% or better were used. A dilution factor of 11 was applied to the calculations to determine the protein concentration in the cell lysates. A representative BCA protein standard curve is shown in Figure 2.11.



*Figure 2.11: A Representative BCA Standard Curve.* The protein standard curve was set up as described above. The average absorbance measurement of the blank was subtracted from the absorbance values of each standard used. The mean of triplicate readings was then taken and plotted against the BSA concentration used. This standard curve is representative of several BCA assays carried out during the course of the studies.

Using the equation generated from the standard curve (Figure 2.11), for a sample with an absorbance of 0.039 at 620nm; the corresponding protein concentration was calculated as follows:

y = 0.0837x - 0.0003 x =  $\frac{y + 0.0003}{0.0837} = \frac{0.039 + 0.0003}{0.0837} = 0.4735 \ \mu g/\mu l$ 

Applying a dilution factor of 11 (10µl of sample in 110µl total volume) gave:

Protein Concentration =  $0.4735 \times 11 = 5.21 \, \mu g/\mu l$ 

Volume of sample containing the selected amount of total protein was calculated and this was used to load gels for western blotting as required.

#### 2.13 Western blot analysis

The Western blotting procedure carried out consisted of the following steps: cell lysate preparation, separation or resolution of the proteins in the lysate by gel electrophoresis, transfer of the resolved proteins from the gel to a membrane, immunoblotting (i.e. incubation antibody of the transferred proteins on the membrane with antibodies), detection of the proteins bands that have been recognised or bound by the antibodies on x-ray films and analysis of the bands captured on the films.

#### 2.13.1 Sample preparation for western blot analysis

A volume of protein extract containing an appropriate amount (e.g. 40, 60 or  $80\mu g$ ) of total proteins was added to an equal volume of 2X concentrated sample buffer containing Tris-HCI, pH6.8 (120mM), SDS (4%), glycerol (10%),  $\beta$ -mercaptoethanol (2%) and bromophenol blue 0.006%).

## 2.13.2 Sodium dodecyl sulphate polyacrylamide gel electrophoresis (SDS PAGE)

A resolving and stacking gel were prepared on a Bio-Rad Mini-PROTEAN II casting stand (Bio-Rad, UK). For the casting of 5ml of an 8.0% resolving gel (enough for one mini blot), the following components were added in the order listed: double distilled water (2.34ml), 30% acrylamide/bisacrylamide 37.5:1 (1.30ml), 1.5M Tris-HCl, pH8.8 (1.25ml), 10% sodium dodecyl sulphate (SDS) (0.050ml), 10% ammonium persulfate (APS) (0.025ml) and N,N,N',N'-

tetramethylethylenediamine (TEMED) (0.005ml). For a 10% resolving gel, the water and acrylamide/bisacrylamide volumes were adjusted to 2.0ml and 1.67ml respectively and everything else stayed the same. Similarly for a 12% resolving gel, the water and acrylamide/bisacrylamide volumes were adjusted to 2.0ml and 1.67ml respectively and everything else stayed the same.

After polymerisation of gels was complete, 5% stacking gel was prepared by adding the following components: water (2.84ml), 30% acrylamide/bisacrylamide 37.5:1 (0.83ml), 0.5M Tris-HCl pH6.8 (1.25ml), 10% SDS (0.050ml), 10% APS (0.025ml) and TEMED (0.005ml). The stacking gel was layered on top of the resolving gel and allowed to polymerise.

The gel was transferred to a Bio-Rad Mini-PROTEAN II (Bio-Rad, UK) electrophoresis tank. A 1x electrophoresis tank buffer was prepared from a 10X stock consisting of: 30.28 g/L of Tris base (0.25M, pH 8.3), 144.0 g/L of glycine (1.92M), 100ml of 10% SDS (1%). Enough 1x diluted electrophoresis buffer to cover the top of the gel, was added to the gel tank. The prepared lysates along with a biotinylated protein ladder (Cell Signaling Technology via NEB, UK) were heated at 95°C for 5 minutes prior to loading onto the gel. The gel was run at a constant voltage of 100V for 40-60 minutes or until the bromophenol blue front had migrated to the bottom of the gel.

#### 2.13.3 Transfer of proteins from gel to PVDF Membrane

The semi-dry transfer with a single buffer system was used to transfer the protein proteins from the gel to a 0.45µm pore size polyvinylidene fluoride (PVDF) membrane (Amersham, UK). Following SDS-PAGE, a 10X transfer buffer (1litre) was prepared using the following: 58.2 g/L of Tris base (0.48M, pH 8.3), 29.3 g/L of glycine (0.39M), 3.75ml of 10% SDS (0.038%). This was then stored at 4<sup>o</sup>C and diluted 1:10 in double distilled water to give a 1x working solution containing 20% methanol. The PVDF membrane was cut to the same size as the gel, soaked in methanol for 1 minute, rinsed in excess distilled water for 1 minute and then equilibrated in the transfer buffer for at least 5 minutes.

Six filter papers, each cut to the same size as the gel, were soaked in the diluted transfer buffer. When available, two pieces of extra thick blot papers (Bio-Rad, UK) (each used in place of three filter papers) were used. Three of the six buffer-soaked filters were placed on the electroblotter (semi-dry transfer systems) (Bio-Rad, UK). The PVDF membrane was then placed on top of the stacked soaked filter papers on the electroblotter. A reference of the protein side on the PVDF membrane was made by cutting a small notch in one corner of the membrane. The gel was removed from the gel cast and placed on top of the PVDF membrane (Figure 2.12). The other three buffer-soaked filter papers were positioned over the gel and air bubbles between the gel and the filter papers were removed by gently rolling a clean 25ml pipette over the top filter or blot paper. A constant current of 0.8mA/cm<sup>2</sup> was applied to the gel and the protein transfer was carried out for 2-3 hours.



*Figure 2.12: Semi-dry transfer of protein from gel to PVDF.* This schematic diagram illustrates the arrangement of components in the semi-dry transfer assembly.

#### 2.13.4 Blocking of the membrane

The membranes were blocked to prevent non-specific binding of antibodies to charged and other molecules on the membrane, thus preventing or reducing background noise. One litre of 10X Tris-buffered saline (TBS) solution containing 24.2g/L of Tris base (20mM; pH 7.5), 87.7g/L of NaCl (150mM) was prepared and normally stored at 4°C. The 10x TBS was used to prepare a 100ml of blocking buffer as follows: 10ml of 10x TBS, 5g (5% w/v) dry virtually fat-free milk (Marvel Brand) and 100 $\mu$ l (0.1% v/v) of Tween 20 (Sigma, UK) in double distilled water. The membrane was incubated in the blocking buffer for 1 hour at room temperature or overnight at 4°C.

#### 2.13.5 Western blot antibody staining

After blocking, the membrane was incubated, using gentle agitation on a shaker, with the primary antibody in a sealed plastic bag. The incubation was carried out at room temperature for one hour. The primary antibody used to probe the blots was mouse monoclonal antibody to cardiac myosin heavy chain (1:2500) or mouse monoclonal antibody to cardiac troponin I (1:2500) or mouse monoclonal antibody to cardiac troponin I (1:2500) or mouse monoclonal antibody to troponin I (1:2500) or mouse monoclonal antibody to troponin I (1:1000) in blocking buffer. These primary antibodies were purchased from Abcam (Cambridge, UK). For the lysates generated from cross-talk studies, the primary antibody used in the blocking buffer to probe the blot, was the mouse monoclonal antibody to phospho-Akt (Ser473) (i.e. Akt phosphorylated at serine 473) (1:1000) (Cell Signaling Technology via New England Biolabs (NEB), UK).

After incubation with the primary antibody, the blot was washed three times (each wash lasting for 5- 10 minutes on a shaker) in wash buffer made up with 1x TBS containing 0.1% Tween 20. The incubation reactions and post incubation washes as described for the primary antibodies were repeated using goat anti-mouse IgG conjugated to horseradish peroxidase (1:5000) (Cell Signaling Technology via NEB, UK) as the secondary antibody. An antibiotin antibody (1:2000) (Cell Signaling Technology via NEB, UK) as the secondary antibody. An antibiotin antibody (1:2000) (Cell Signaling Technology via NEB, UK) also conjugated to horseradish peroxidase was added in order to detect the protein ladder. Anti- $\beta$ -actin antibody (1:10000) (Sigma, UK) conjugated to horseradish peroxidase was added in order to detect as a loading control.

#### 2.13.6 Detection of Protein Bands Using ECL Reagents

Detection or visualization of targeted protein bands (i.e. bound antibodies) was carried out in a dark room using enhanced chemiluminescence (ECL). The ECL reagent was prepared following the manufacturer's instructions by mixing equal volumes of reagent 1 and 2 supplied in the ECL Western Blotting Detection kit (Amersham, UK). The solution was immediately added to the membrane and incubated for 1 minute at room temperature. Excess detection reagent was drained from the membrane using a tissue paper. The blot was covered with cling film, ensuring there were no trapped air bubbles on the membrane. A sheet of autoradiography film, Hyperfilm ECL (Amersham, UK) was placed on top of the wrapped membrane in a film cassette and exposed for 5 minutes. The membrane was then developed,

fixed, rinsed under running tap water and then dried in air at room temperature.

#### 2.13.7 Quantification of protein bands

The developed protein bands on the film were scanned using Epson (Perfection 2480 Photo) Scanner and the intensities of the protein bands on the scanned image were measured using densitometery software, Syngene Gene Tools (version 3.00).

### 2.14 Data Analysis

Data obtained from at least 3 independent experiments, (performed in single or duplicates, where possible), were used for statistical analysis. The one-way analysis of variance (ANOVA) test, followed by post-hoc Bonferroni's multiple comparison test or Dunnett's multiple comparison test, was carried out using GraphPad Prism version 4. A difference between the treatments was considered significant when P< 0.05. Data presented are means ± SD from at least 3 independent experiments.

### **CHAPTER 3.0**

### ESTABLISHMENT OF CARDIAC DIFFERENTIATION MODELS

#### 3.1 INTRODUCTION

In order to be able to conduct the studies described in this thesis, it was essential to develop an *in vitro* model of cardiomyocyte differentiation. This was achieved by exploiting the ability of H9c2 cells to differentiate into cardiomyocytes when cultured in growth medium containing a reduced concentration of serum (Hescheler *et al.*, 1991) and also by using the mouse pluripotent P19 embryonal carcinoma stem cell line which can be maintained and differentiated into cardiomyocytes under standard cell culture conditions (e.g. 10% FBS and without the need for LIF or feeder cells) (McBurney *et al.*, 1982; Wobus *et al.*, 1991; Angello *et al.*, 2006).

#### 3.1.1 H9c2 cell line model of cardiomyocyte differentiation

The H9c2 cell line is a well established cardiomyocyte differentiation model. These cells were established from embryonic rat ventricles by Kimes and Brandt (1976) and are now widely used by researchers as an *in vitro* model for generating cardiomyocytes (Kim *et al.*, 1999; Menard *et al.*, 1999; Hong *et al.*, 2001; Kageyama *et al.*, 2002; Giusti *et al.*, 2004; Pagano *et al.*, 2004).

In addition to differentiation studies, H9c2 cells have been used as a simple and relevant model to study various mechanisms in cardiomyocytes including apoptosis (Ekhterae *et al.*, 1999; Chae *et al.*, 2001; Hong *et al.*, 2001; Kageyama *et al.*, 2002; Bonavita *et al.*, 2003; Kim *et al.*, 2005; Pesant *et al.*, 2006; Qin *et al.*, 2006), glucose transport (Armoni *et al.*, 2005), opioid-induced acute cardioprotection (Gross *et al.*, 2006) and reactive oxygen species (ROS)-induced cardiotoxicity in cardiomyocytes (Salvatorelli *et al.*, 2006).

The differentiation of these cells into cardiomyocytes is a highly ordered and regulated multi-step process that is initiated in response to a number of environmental signals including exposure to low serum content (Tamai *et al.*, 2000). The differentiation is initiated by mononucleated cell alignment, elongation and fusion leading to the formation of multinucleated myotubes. This process is accompanied by an increase in the expression of muscle specific proteins (Ludolph and Konieczny, 1995). Manipulations such the effect of drugs (e.g. inhibitors of signalling pathways) can be easily monitored during the cardiac differentiation of these cells. The H9c2 cell line is therefore a good model for investigating the signalling events that control cardiomyocyte differentiation.

#### 3.1.2 P19 cell line model of cardiomyocyte differentiation

The P19 embryonal carcinoma stem cell line (P19 stem cells) is pluripotent and has been used as an *in vitro* model of stem cell differentiation into cardiomyocytes (Wobus *et al.*, 1991; Angello *et al.*, 2006). The cardiomyocyte differentiation of P19 stem cells is initiated in the presence of DMSO (McBurney *et al.*, 1982). The cardiac specific transcription factors, GATA-4 and Nkx2-5, are expressed in during the differentiation of P19 cells into cardiomyocytes (Skerjanc *et al.*, 1998).

Using these two cell types, we have developed systems which generate cardiomyocytes and thus provide models which we could exploit to investigate the mechanisms that regulate the differentiation of stem cells into cardiomyocytes

#### 3.2 MATERIALS AND METHODS

#### 3.2.1 Culture and differentiation of H9c2 Cells

The routine culture of H9c2 cells was carried out as described in section 2.3.1 (Chapter 2). H9c2 cells were trypsinised and the appropriate cell density for the differentiation experiments was calculated as described in section 2.6 (Chapter 2). H9c2 cells  $(2x10^3 \text{ cells/cm}^2)$  were seeded in p60 dishes or 6-well plates in full culture medium (i.e. DMEM supplemented with 10% FBS, 100units/ml penicillin/100µg/ml streptomycin), placed in a tissue culture incubator at  $37^{\circ}$ C, 95% air and 5% CO<sub>2</sub> and allowed to become 60-70% confluent. The differentiation of H9c2 cells into cardiomyocytes was carried out in 1% serum as described in section 2.8.1 (Chapter 2).

#### 3.2.2 Culture and differentiation of P19 stem cells

The routine culture of P19 stem cells was carried out as described in section 2.3.2 (Chapter 2). To initiate differentiation, P19 stem cells  $(1 \times 10^6 \text{ cells})$  were seeded in P100 Petri (i.e. microbiological) dishes in the differentiation medium (i.e.  $\alpha$ -MEM supplemented with 10% FBS, 100 units/ml penicillin/100 µg/ml streptomycin and 0.8% DMSO) and the differentiation of P19 stem cells into beating cardiomyocytes was carried out as described in section 2.9.1 (Chapter 2).

#### 3.2.3 Western blot analysis

In order to perform the western blot analysis, the following procedures were carried out as described in Chapter 2 under the sections indicated in parenthesis: cell lysate generation (section 2.11), preparation of BSA standards (section 2.12.1) and BCA assay (section 2.12.2). The western blotting was carried out as described in section 2.13.

#### 3.2.4 Data analysis

Data obtained from independent experiments were used for statistical analysis as described in section 2.14 (Chapter 2).

### 3.3 RESULTS

## 3.3.1 Establishment and characterization of H9c2 Cell differentiation model

Freshly cultured undifferentiated H9c2 cells in full growth medium containing 10 % FBS initially were rounded but subsequently attached and grew as a monolayer. These undifferentiated H9c2 cells grew as single cells with clearly defined boundaries and looked almost spindly-shaped in morphology (Figure 3.10). This morphology is maintained throughout culture if the cells are not induced to undergo differentiation. The cells became highly confluent but failed to initiate cell alignment, elongation and fusion to form myotubes (Figure 3.11). In agreement with the lack of morphological changes in H9c2 cultured in full growth medium cardiac myosin heavy chain (alpha or beta isoform) ( $\alpha$ / $\beta$ -MHC) was not expressed as determined by western blotting (Figure 3.12).



*Figure 3.10: Morphology of H9c2 cells (undifferentiated) in routine culture.* H9c2 cells cultured in full growth medium consisting of DMEM supplemented with 10% FBS and 100 units/ml penicillin/ 100  $\mu$ g/ml streptomycin grew as single elongated cells (white arrows). The photograph at day 4 was taken under 200X magnification and is representative of several microscope fields from several cultured flasks.




Figure 3.11: Morphology of undifferentiated H9c2 cells in full culture medium H9c2 cells cultured in full growth medium consisting of DMEM supplemented with 10% FBS and 100 units/ml penicillin/100  $\mu$ g/ml streptomycin grew as single elongated cells becoming very tightly packed and undefined after 10-14 days in culture. The photographs (200X) are representative microscope fields for the indicated days.





#### 3.3.2 Differentiation of H9c2 cells into cardiomyocytes

In contrast to the growth pattern of H9c2 cells observed in the full growth medium, when these cells were cultured in 1% serum, they initially grew as flat, elongated mononucleated myoblasts similar to the morphology seen in the 10% FBS culture medium. However after four days in low serum culture, neighbouring cells began to fuse with complete differentiation becoming evident at day 12 to day 14. The differentiated cells acquired a myotube-like morphology (Figure 3.13). Spontaneous contraction, a distinguishing feature of cardiomyocytes, was however not observed in these cells.

The morphological changes seen in H9c2 cells undergoing differentiation into cardiomyocytes were accompanied by parallel increases in the expression of cardiac specific markers, including myosin heavy chain (MHC) (Figure 3.14) and ventricular myosin light chain (MLC-1v) (Figure 3.15). The expression of MHC was not detected in differentiated cells between days 0 to 2 in cells induced to differentiate. Myosin heavy chain expression was however evident at day 4 and continued to increase up to day 10, reaching a peak from day 12 to day 14. By comparison, MLC-1v expression, although absent in lysates from cells at day 0, was quite evident from day 2, reaching a peak between day 10 to 12.





*Figure 3.13: Morphology of differentiated H9c2 cells in culture.* H9c2 cells cultured in differentiation medium consisting of DMEM supplemented with 1% FBS and 100 units/ml penicillin/100  $\mu$ g/ml streptomycin grew initially grew as single elongated cells (white arrows on day 0 and day 2) which by day 4 began to fuse, forming myotubes that were more evident at days 12 to 14 (black arrows on day 4 to 14). The photographs were taken under 200X magnification are representative microscope fields for the indicated days.



*Figure 3.14: Changes in the expression of myosin heavy chain in differentiated H9c2 cells.* H9c2 cells were cultured in differentiation medium consisting of DMEM supplemented with 1% FBS and 100 units/ml penicillin/100  $\mu$ g/ml streptomycin. Cell lysates were prepared at the end of each time point specified on the graph and subjected to western blotting using a specific monoclonal anti-MHC antibody. A blot of lysates from cells cultured in is shown (Panel A). The bar graph is the densitometric data (mean ± SD) from three independent experiments (Panel B).





Figure 3.15: Changes in the expression of ventricular myosin light chain in differentiated H9c2 cells. H9c2 cells were cultured in differentiation medium consisting of DMEM supplemented with 1% foetal bovine serum (FBS) and 100 units/ml penicillin/100  $\mu$ g/ml streptomycin. Cell lysates were prepared at the end of each time point specified on the graph and subjected to western blotting using a specific monoclonal anti-MLC-1v antibody. A blot of lysates from cells cultured in is shown (Panel A). The bar graph is the densitometric data (mean ± SD) from three independent experiments (Panel B).

# 3.3.3 Establishment and characterization of P19 stem cell differentiation model

P19 embryonal carcinoma stem cells cultured in normal cell culture grade dishes in complete culture medium did not form embryoid bodies and did not differentiate into cardiomyocytes. Instead, they grew as a single cell monolayer but in clusters (Figure 3.16). In contrast, P19 cells cultured in microbiological Petri dishes in complete cell culture medium resulted in the cells aggregating and growing in suspension as embryoid bodies (EBs). This occurred within a few hours with the aggregates growing in size over time. The sizes and shapes of the formed EBs were however highly variable. The EBs increased in size over time (Figure 3.17).

# **Adherent P19 Stem Cells**



Figure 3.16: Normal growth pattern of P19 embryonal carcinoma stem cells in culture. P19 cells were seeded into standard cell culture dishes and allowed to grow in complete culture medium ( $\alpha$ -MEM supplemented with 10%. The cells grew as clusters (white arrows) in a monolayer FBS and 100 units/ml penicillin/100 µg/ml streptomycin) did not form embryoid bodies. The photograph was taken under 200X magnification and is representative of several microscope fields from more than three cultured flasks.

# P19 stem cell Embryoid Bodies at Day 1



Figure 3.17: Formation of embryoid bodies from P19 embryonal carcinoma stem cells in culture. P19 cells were seeded into Petri dishes and allowed to grow in complete culture medium ( $\alpha$ -MEM supplemented with 10% FBS and 100 units/ml penicillin/100 µg/ml streptomycin) and 0.8% DMSO to form embryoid bodies (white arrows). The photograph of the developing embryoid body at 24 hour was taken under 200X magnification and it is representative of several microscope fields from three independent experiments.

#### 3.3.4 Differentiation of P19 stem cells into beating cardiomyocytes

The transfer of EB from Petri dishes to standard tissue culture dishes resulted in cell adhesion. The cell growth pattern changed from the three dimensional EB to a two dimensional monolayer (Figure 3.18). Twenty fours later, EB adhesion was followed by cell migration and proliferation. A rapid cell growth was observed in these early stages with monolayers becoming confluent by day 4. The first beating clusters of cardiomyocytes were routinely detected by day 6 or 7. When present, the size and number of beating clusters of cardiomyocytes usually increased up to day 14. There was a high degree of variability both in term of the clusters of beating cells and in the frequency at which they appeared to beat. Sometimes, isolated clusters of beating cells within monolayers (day 6 and day 14) were seen. At other times, uniform beating monolayers (day 8, day 10 and day 12) were seen. Some colonies demonstrated a much more rhythmic and rapid rate of contraction (day 8, day 10 and day 12) while others showed an intermediate or much slower contraction rates or uncoordinated contractions (day 6, and day 14). The slow or uncoordinated contractions appeared to occur more in isolated clusters of cells within monolayers while the much more rhythmic and rapid rate of contraction was evident in uniform monolayers.



Figure 3.18: DMSO-induced differentiation of P19 stem cells into beating cardiomyocytes. P19 stem cell EBs were formed in differentiation medium consisting of  $\alpha$ -MEM supplemented with 10% FBS, 100 units/ml penicillin/100 µg/ml streptomycin and 0.8% DMSO for four days and then transferred into standard cell culture dishes in DMSO-free medium. DMSO caused the differentiation of P19 stem cells into beating cardiomyocytes. The beating areas of cardiomyocytes are indicated by the dotted white circles. The beating focus of each beating area is shown by a black arrow. Photographs and movie clips of beating cells were taken under 200X magnification on the days indicated and are representative of microscope fields (for their respective days) from at least three independent differentiation experiments. *NBC = no beating cardiomyocytes*.

#### 3.4 DISCUSSION

This project is the first of its kind in the laboratory in which it was carried out. The host laboratory has strong background in the research field of cell signalling in cardiovascular biology but had carried out no research work on stem cell differentiation at the beginning of this project. This project is therefore a new direction in the research activity in our laboratory.

As discussed in the background to this project, cardiomyocyte lineage specification during cardiogenesis is a complex event. Therefore in the design of a study to explore cardiomyocyte differentiation *in vitro* the choice of a model has to be given careful consideration. Thus the first challenge was to find a model to investigate the cell signalling events that mediates cardiomyocyte differentiation. It was decided to use a model that is not only relevant but also simple so that it can be easily manipulated to address key questions about signalling events during cardiomyocyte differentiation. The P19 stem cell and H9c2 cell models meet this condition. Therefore given that this is the first stem cell project in our laboratory, the P19 stem and H9c2 cell lines were attractive models to seek key answers or cues to signalling pathway that mediate cardiac differentiation.

The H9c2 cell line was chosen due to its relevance in cardiac regeneration, simplicity of use as well as the fact that this cell line has been extensively used by other researchers as a cardiomyocyte differentiation model and also for the investigation of other molecular or biological process in cardiomyocytes (Kim *et al.*, 1999; Menard *et al.*, 1999; Hong *et al.*, 2001;

Kageyama *et al.*, 2002; Giusti *et al.*, 2004; Pagano *et al.*, 2004). In addition to the differentiation studies, H9c2 cells have been used as a simple and relevant model to study various mechanisms in cardiac muscle cells including apoptosis (Ekhterae *et al.*, 1999; Chae *et al.*, 2001; Hong *et al.*, 2001; Kageyama *et al.*, 2002; Bonavita *et al.*, 2003; Pesant *et al.*, 2006; Qin *et al.*, 2006), glucose transport (Armoni *et al.*, 2005), opioid-induced acute cardioprotection (Gross *et al.*, 2006) and reactive oxygen species (ROS) cardiotoxicity (Salvatorelli *et al.*, 2006). Differentiated H9c2 cells preserve the receptor systems, signalling pathways and cardiac proteins (e.g. cardiac Ltype Ca2+ channels) found in primary cardiomyocytes (Hescheler *et al.*, 1991; Sipido and Marban, 1991). The H9c2 cell line is therefore an appropriate model for our studies aimed at investigating the signalling mechanisms that regulate differentiation into cardiomyocytes.

Freshly cultured H9c2 cells in a suspension of complete growth medium looked rounded. Once attached, the undifferentiated cells grew as spindle-shaped mononucleated cells and did not express cardiac specific markers. The cells maintained this morphology throughout their period in culture (14 days). On the other hand, differentiated cells looked elongated and often fuse to form multinucleated myotubes. The expression of both MHC and MLC-1v was also evident in the cells differentiated in culture medium containing low serum (1% FBS).

The mechanism by which low serum induces the differentiation of H9c2 cells into cardiomyocytes is not clear; however there are possible and/or plausible

explanations. A critical event in the activation of the canonical Wnt/β-catenin pathway is level of cytosolic β-catenin. The phosphorylation of β-catenin by Glycogen synthase kinase 3β (GSK-3β) mediates the ubiquitination of βcatenin resulting in the subsequent degradation of β-catenin by proteosomes (Giles *et al.*, 2003; Logan and Nusse, 2004). While phosphorylated β-catenin is targeted for degradation, the non-phosphorylated form of β-catenin is degradation-resistant. A decreased in phosphorylation of β-catenin causes an increase in the levels of β-catenin in the cytoplasm and this leads to its subsequent translocation into the nucleus. In the nucleus, β-catenin, in a complex with LEF/TCF, activates the transcription of Wnt target genes (Yamaguchi *et al.*, 1999; Ai *et al.*, 2000).

Lithium (a GSK-3 $\beta$  inhibitor) treatment mimics the activation of the canonical Wnt/ $\beta$ -catenin signalling pathway by inducing the stabilization and nuclear translocation of  $\beta$ -catenin via inhibition of  $\beta$ -catenin phosphorylation by GSK-3 $\beta$  (Klein and Melton, 1996; Hedgepeth *et al.*, 1997; Rao *et al.*, 2005). Our data showed that H9c2 cells in 10% serum do not differentiate into cardiomyocytes but do so in low serum medium. It has also been shown that the treatment of H9c2 cells with lithium can cause H9c2 cell to differentiate into cardiomyocytes (Kashour *et al.*, 2003). This suggests that lithium treatment mimics low serum condition and also that low serum, like lithium, may activate some elements of the canonical Wnt/ $\beta$ -catenin signalling pathway which may play roles in the initiation of H9c2 cell differentiation into cardiomyocytes. Whether low serum, like lithium, can induce the activation of the canonical Wnt/ $\beta$ -catenin signalling pathway warrants further investigation.

H9c2 cells generally grow very slowly. The splitting ratio or cell density is therefore a critical consideration and once determined for optimal growth, the ratio should be adhered to in order to avoid inconsistent growth rates which, from our observations, can influence differentiation. Another factor worth mentioning is the fact that when brought from long term storage (in liquid nitrogen) the cells should first be cultured at high density in a smaller culture flask than that in which they were originally frozen from. This was found to be critical in helping the cells to regain their growth potential after a long stay in the liquid nitrogen. Where this is not carried out, the cells will either not recover from the freezing or may show sluggish growth and may never fully recover in future passages. Routine splitting ratios of 1:3-1:5 of 60-70% confluent H9c2 were found to be appropriate for maintaining the cells in a viable healthy state.

While H9c2 cells are easy to manipulate and differentiate into cardiomyocytes under standard laboratory conditions, the myoblast population in the cell line can easily be lost if proper care is not taken during routine cell culture. Therefore when H9c2 cells are used for cardiac differentiation experiments, the prevention of the depletion of the myoblast population is essential. Prolonged or continual culture of H9c2 cells without subculturing, during routine cell culture, will lead to a rapid depletion of the myoblast population in the cell line. This was avoided in our studies by making sure that at all times while in culture the cells were not allowed to become more than 60-70 % confluent. If the cells became fully confluent for any reason they were

discarded and a new batch was resuscitated from the liquid nitrogen cell stock.

The H9c2 cardiac differentiation model, like most models, is not a perfect substitute for the *in vivo* processes. Despite the many attractive features of working with these cells, one of the serious limitations of the H9c2 model for cardiac differentiation is that the cell line was derived from an embryonic heart. This means the cardiac lineage commitment may have already taken place. However cardiac specific genes such as those which code for cardiac specific myosin light chain and ventricular myosin light chain-1 are not constitutively expressed in undifferentiated H9c2 cells but are strongly expressed when the cells are induced to differentiate into cardiomyocytes. This makes these cells relevant for exploring the signalling transduction events that drives their differentiation into cardiomyocytes.

The use of P19 stem cell line as a model for stem cell differentiation into cardiomyocytes was also developed in parallel with the H9c2 cell model. The P19 stem cell line, like H9c2 cells, is relatively simple and can be easily manipulated in experiments. With proper care for the cells during routine cell culture and also during differentiation experiments, P19 stem cell differentiation into cardiomyocytes can be highly regulated and reproducible. An essential element in ensuring differentiation of ES cells is the formation of three-dimensional (3D) aggregates known as embryoid bodies (EB). P19 cells when cultured in the suspension form EB. The EB consists of ectodermal, mesodermal and endodermal cell types and the 3D environment in the EB helps in the recapitulation of the early stages of embryonic development thus

initiating the cell differentiation into cell types of all the three germ layers including cardiomyocytes (Wobus *et al.*, 1991; Maltsev *et al.*, 1993; Denning *et al.*, 2006), endothelial cells (Zhang *et al.*, 2005), neuronal cells (Bain *et al.*, 1995), adipocytes (Dani *et al.*, 1997) and blood cell precursors (Schmitt *et al.*, 1991). In P19 stem cells, EB formation in the presence of DMSO induces the cells to specifically differentiate into cardiomyocytes.

Between 6 to 8 days after EB transfer into cell culture grade dishes, beating cardiomyocytes were regularly observed in DMSO-induced P19 stem cells. While beating cardiomyocytes can be readily obtained, the population is highly heterogeneous. The beating rates from one cluster to another varied considerably even within the same culture dish. Where the main research focus is the unravelling of the signalling pathways that are mediating the differentiation of P19 stem cells into cardiomyocytes, this lack of homogenous P19 stem-derived beating cardiomyocytes may not be a major problem.

P19 stem cells have several good features. The cells are pluripotent and can be maintained continuously in serum-supplemented growth medium without the need for LIF or mitotically inactive feeder cells to keep them in the undifferentiated state. The fact that these cells can be maintained under standard cell culture conditions means that key experiments can be performed at relatively very low cost in comparison to the same experiments being performed using other ES cells that may need special conditions such as knockout serum, LIF or feeder cells. Another strong point for the use of P19 stem cells is that the cell line is a well-established model of

cardiomyocyte differentiation (Wobus *et al.*, 1994; Zhang *et al* 2005; Angello *et al.*, 2006). Moreover P19 cells can be induced to differentiate into cardiomyocytes fairly readily following exposure to DMSO (McBurney *et al.*, 1982; Monzen *et al.*, 1999; Naito *et al.*, 2003; Brown *et al.*, 2004; Brewer *et al.*, 2005).

In culture medium containing DMSO, P19 cells differentiating into cardiomyocytes recapitulate the principal biological features or stages of embryonic cardiogenesis. The expression of cardiac transcription factors including Nkx-2.5 (Skerjanc *et al.*, 1998; Monzen *et al.*, 1999; Wen *et al.*, 2007) GATA-4 (Monzen *et al.*, 1999; Uchida *et al.*, 2007; Wen *et al.*, 2007) and MEF2C (Skerjanc *et al.*, 1998; Wen *et al.*, 2007) have been well documented in P19 stem cells differentiating into cardiomyocytes. Cardiac specific proteins, including MHC (Habara-Ohkubo, 1996) and MLC (Habara-Ohkubo, 1996; Moore *et al.*, 2004), are expressed by cardiomyocytes derived from P19 stem cells. This means that P19 stem cells differentiating into cardiomyocytes follow the same mechanisms as normal embryonic stem cells.

It is not entirely clear how DMSO induces P19 stem cells to differentiate along the cardiac lineage. DMSO actions are not specific for P19 stem cells and neither are its actions limited to induction of cell differentiation. Beside the induction of cell differentiation, many other biological effects including membrane transport, vasodilation, muscle relaxation and antagonism of

platelet aggregation, have been attributed to DMSO (Jacob and Herschler, 1986).

Some available evidence suggests a possible role of DMSO in the Wnt/βcatenin signalling pathway for the commitment of P19 stem cell to differentiate into cardiomyocytes. Evidence for a regulatory role of DMSO on canonical Wnt/β-catenin signalling was shown by the ability of DMSO to induce a decrease in levels of phosphorylated  $\beta$ -catenin and the elevation of the nonphosphorylated form of  $\beta$ -catenin in the cytoplasm (Nakamura *et al.*, 2003). It was also shown that the levels of Wnt3a and Wnt8a were up-regulated by DMSO before the expression of cardiac specific transcription factors such as Nkx2.5, GATA-4 and Tbx5 (Nakamura et al., 2003). Wnt3a and Wnt8a are believed to signal exclusively via the canonical Wnt/β-catenin pathway (Tada and Smith, 2000; Pandur et al., 2002; Maurus et al., 2005). The DMSO induced elevation of cytoplasmic  $\beta$ -catenin levels was suppressed by Frizzled-8/Fc (Fz-8/Fc) (Nakamura et al., 2003), a protein that belongs to the secreted frizzled-related proteins. Secreted frizzled-related proteins are in vivo Wnt antagonists (Rattner et al., 1997; Bafico et al., 1999; Pera and De Robertis, 2000; Shibata et al., 2000; Marvin et al., 2001). The treatment of P19 stem cells with Fz-8/Fc also inhibited the DMSO-induced expression of cardiac transcription factors, cardiogenic growth factors (i.e BMP2, BMP4 and and cardiac  $\alpha$ -MHC (Nakamura *et al.*, 2003). This is a strong FGF8) evidence that DMSO activates or modulates the the Wnt/β-catenin signalling pathway.

Glycogen synthase kinase  $3\beta$ , as mentioned above, is responsible for keeping the canonical Wnt/ $\beta$ -catenin pathway in the "off" or inactive state through the phosphorylation of  $\beta$ -catenin canonical wnt/ $\beta$ -catenin signalling. Constitutively active GSK- $3\beta$  blocks DMSO-induced differentiation of P19 stem cells while lithium and Wnt3A-conditioned medium up-regulated early cardiac markers and the proportion of differentiated P19 stem cells in the presence of DMSO (Nakamura *et al.*, 2003). The inhibition of DMSO-induced cardiac differentiation of P19 stem cells by constitutively active GSK- $3\beta$  also suggests that DMSO may act through the Wnt/ $\beta$ -catenin signaling pathway. All the above data when taken together form a strong evidence to suggest that Wnt/ $\beta$ -catenin signalling is activated during the initiation of cardiac differentiation of P19 stem cell and that DMSO mode of actions may be the activation or modulation of the canonical wnt/ $\beta$ -catenin signalling pathway.

DMSO is not an endogenous signalling molecule but a clear understanding of The mode of action of DMSO in the induction of P19 stem cell differentiation into cardiomyocytes may increase our understanding of stem cell differentiation into cardiomyocytes. For example the suppression of DMSOinduced cardiac specific transcription factors (such as Nkx2.5 and GATA-4) in P19 cells by Wnt antagonists and GSK-3β inhibitors (Nakamura *et al.*, 2003) is also seen in other stem cell cardiac differentiation models (e.g. AB2.2 ES cells) where DMSO is not the cardiac inducing agent (Liu *et al.*, 2007). Thus a clearer understanding of DMSO roles in P19 stem cell differentiation into cardiomyocytes may lead to improved protocols for the differentiation of stem cells into cardiomyocytes.

There are several potential cardiac markers for monitoring the differentiation of stem cells into cardiomyocytes. These include Nkx2.5, GATA-4 (Kehat *et al.*, 2001; Nakamura *et al.*, 2003; Anderson *et al.*, 2007), myocyte enhancer factor (MEF2C) (Denning *et al.*, 2006; Anderson *et al.*, 2007), tropomyosin (Moore *et al.*, 2004),  $\alpha$ -actinin (Denning *et al.*, 2006), cardiac troponin I (cTnI) (Kehat *et al.*, 2001), cardiac troponin T (cTnT) (Kehat *et al.*, 2001), atrial natriuretic peptide (ANP) (Denning *et al.*, 2006), cardiac myosin light chain isoforms (MLC-2v and MLC-2a) (Kehat *et al.*, 2001; Moore *et al.*, 2004), MLC-1v & MLC-1a and cardiac myosin heavy chain isoforms ( $\alpha$ -MHC and  $\beta$ -MHC) (Kehat *et al.*, 2001; Denning *et al.*, 2006; Anderson *et al.*, 2007). For this project we monitored cardiac differentiation by the expression of cardiac  $\alpha$ -MHC/ $\beta$ -MHC, MLC-1v (in H9c2 cells) and cTnI (in P19stem cells). Ventricular myosin light chain 1 (MLC-1v) along with MLC-2v is present in both ventricles of the heart (Chuva de Sousa Lopes *et al.*, 2006).

### 3.5 Summary

Freshly cultured undifferentiated H9c2 cells in full growth medium containing 10 % FBS attached and grew as a monolayer with spindly-shaped morphology. In agreement with the lack of morphological changes, undifferentiated cells did not expressed cardiac myosin heavy chain as determined by western blotting. In contrast to their growth pattern in full medium, when H9c2 cells were cultured in 1% serum, the cells became aligned, elongated, fused and formed myotubes. The morphological changes seen in H9c2 cells undergoing differentiation into cardiomyocytes were accompanied by parallel increases in the expression of cardiac specific markers, including cardiac myosin heavy chain (MHC) and cardiac ventricular myosin light chain (MLC-1v).

P19 embryonal carcinoma stem cells cultured in normal cell culture grade dishes in complete culture medium (with or without 0.8% DMSO) did not form embryoid bodies (EB) but grew as clusters, in a single cell monolayer, which did not differentiate into cardiomyocytes. In contrast, P19 cells cultured in microbiological Petri dishes in complete culture medium, supplemented with 0.8% DMSO, formed EBs. The transfer of EB from Petri dishes to standard tissue culture dishes resulted in cell adhesion, the formation of a two-dimensional monolayer, cell migration and proliferation. The first beating clusters of cardiomyocytes were routinely detected by day 6 or 7. The size and number of beating clusters of cardiomyocytes usually increased up to day 14. There was a high degree of variability both in term of the clusters of beating cells and in the frequency at which they appeared to beat.

In conclusion, models of cardiomyocyte differentiation using P19 cells and H9c2 cells were successfully established in this project. These models were subsequently used in further studies exploring the mechanisms that may regulate stem cell differentiation into cardiomyocytes. Both P19 stem cells and H9c2 cells are relevant models for cardiomyocyte differentiation studies.

# CHAPTER 4.0

# **ROLE OF NITRIC OXIDE IN CARDIOMYOCYTE**

# DIFFERENTIATION

## 4.1 INTRODUCTION

#### 4.1.1 Physiological functions of nitric oxide

Nitric oxide (NO) is a gas that plays many important roles as signalling molecule in the cardiovascular system. A key feature of this molecule is its high but time limited reactivity. The roles of nitric oxide in the cardiovascular system include the maintenance of vascular integrity through the inhibition of platelet aggregation (Freedman *et al.*, 1999), leukocyte–endothelium adhesion (Kubes *et al.*, 1991; Lefer *et al.*, 1999) and the regulation of cardiac smooth muscle contraction (Skarsgard *et al.*, 2000) to cause vasodilatation. NO relaxes the smooth muscles to cause vasodilation. Nitric oxide also functions as neurotransmitter in the nervous system.

#### 4.1.2 Biosynthesis of nitric oxide

The generation of nitric oxide is controlled by nitric oxide synthases (NOS) of which three isoforms exist: cytokine-inducible NOS (iNOS), endothelial NOS (eNOS) and neuronal NOS (nNOS). These enzymes display tissue-specific expression and their activation is also related to the cell status and specific stages during development. Neuronal NOS is predominantly expressed in certain neurons and in skeletal muscle, eNOS is predominantly expressed in endothelial cells and iNOS is expressed by macrophages and monocytes. Monocytes are phagocytes circulating in the blood. When a monocyte leaves the blood and enters into tissues, it undergoes a series of cellular changes to become a macrophage.

#### 4.1.3 Molecular or physiological targets of nitric oxide

Nitric oxide mediates its second messenger or signalling roles through soluble guanylate cyclase (sGC), its main physiological target (Gold *et al.*, 1990; Trigo-Rocha *et al.*, 1993; Phung *et al.*, 1999). Nitric oxide activates guanylate cyclase by binding to the heme group. The activation of sGC by NO results in increased levels of cGMP. Nitric oxide, through cGMP, regulates various downstream protein kinases (e.g. cGMP-dependent kinase) (Archer *et al.*, 1994; Pineda *et al.*, 1996), ion channels (i.e. cGMP-dependent ion channels) (D'Ascenzo *et al.*, 2002) and cGMP-dependent phosphodiesterases (Wexler *et al.*, 1998; Zima *et al.*, 2000). In the vasculature NO relaxation of vascular smooth muscle is also mediated through cGMP. The role of nitric oxide as a neurotransmitter in the nervous system is also mediated by increase in cGMP level.

In contrast to the beneficial cGMP-dependent signalling roles of NO, cGMPindependent actions may be toxic to the cells and tissues. For example, NO reactions with superoxide anion to form peroxynitrite anion(Brealey *et al.*, 2002), the inhibition of mitochondrial complexes I and IV (Takehara *et al.*, 1995; Okada *et al.*, 1996; Brealey *et al.*, 2002) and the activation of poly-ADP ribose polymerase (PARP) (Endres *et al.*, 1998) leading to depletion of NAD+ (a key glycolytic substrate) and the eventual depletion of cellular energy stores are all general mechanisms that underlie some of the toxic effects of NO.

#### 4.1.4 Nitric oxide and stem cell differentiation

Some available data suggest that NO may have positive roles during stem cell commitment to cardiac lineage. On the hand, there are other reports with data to suggest that NO may negatively regulate stem cell differentiation into cardiomyocytes. Therefore currently the exact role of NO during stem cell commitment to cardiac lineage is not clear.

Reports that NO may be involved in stem differentiation into cardiomyocytes have been published (Kanno et al., 2004). NO treatment of ES D3 cells using NO donors and also iNOS transfected in the ES cell were found to increase the number and the size of beating foci in EB outgrowths (Kanno et al., 2004). It has also been demonstrated that incubation of EBs with the NOS inhibitor N (G)-methyl-L-arginine acetate (L-NMMA) markedly attenuates their differentiation into cardiomyocytes with the inhibition being reversed by the application of the NO-donor spermine-NONOate (Bloch et al., 1999). Taken together, these findings indicate a positive role for NO in the differentiation of stem cells into cardiomyocytes. In addition to stem cells, NO may also regulate differentiation of numerous other cell types. For example, NO has been reported to promote the differentiation of mouse bone marrow clonal stromal cells (ST2 cells) to osteoblast-like cells (Hikiji et al., 1997; Koyama et al., 2000), human neuroblastoma cells (Obregon et al., 1997; Ghigo et al., 1998), human trophoblast (Lyall et al., 1998; Sanyal et al., 2000), as well as regulate differentiation of NG108-15 (Cote et al., 1998) and PC12 cells (Phung et al., 1999; Nakagawa et al., 2000) into neurones.

The role of NOS and NO in heart muscle during embryonic development has been extensively investigated. On day 7.5, no NOS isoform was detected in mouse and rat embryos (Bloch *et al.*, 1999). However from E9.5 both rat and mouse embryos displayed prominent expression of iNOS and eNOS, which correlated with high expression of soluble guanylate cyclase (sGC) and also with elevated levels of cyclic guanosine monophosphate (cGMP) (Bloch *et al.*, 1999).

It was found that after E14.5, the expression of both iNOS and eNOS was down-regulated and also shortly after birth, eNOS expression was detectable while that of iNOS was barely detectable (Bloch *et al.*, 1999). In ES D3 cells, the NOS expression pattern in seen *in vivo* was also seen in these cells (Bloch *et al.*, 1999). Similarly, the expression of eNOS and nNOS but not iNOS in undifferentiated ES cells was also reported in another study (Krumenacker *et al.*, 2006). The expression of all three NOS isoforms were detected at different phases after the initiation of differentiation of ES cells into cardiomyocytes (Krumenacker *et al.*, 2006). In these studies, nNOS expression decreased within one day while iNOS mRNA became detectable after several days, and eNOS increased after 7-10 days (Krumenacker *et al.*, 2006). The differential expression of the various NOS isoforms suggest that NO may be involved in early stages of cardiogenesis and also that there may be change in NO requirement as differentiation progresses.

Similarly it was shown that oxytocin-induced differentiation of P19 stem cells into cardiomyocytes was inhibited by the NOS inhibitor N (G)-nitro-L- arginine

methyl ester (L-NAME). The NO donor SNAP was able to reverse L-NAME mediated inhibition of P19 cell differentiation into cardiomyocytes (Danalache *et al.*, 2007). These studies clearly indicate a role for NO and the NOS enzymes in stem cell differentiation but what is evident is that this may be a complex process. This complexity is highlighted by the fact that suppression of NOS activity by L-NAME has also been shown to increase the number of stem and progenitor cells in the bone marrow of irradiated mice (Michurina *et al.*, 2004), indicating a negative regulatory role for NO in the generation of, at least, bone marrow derived stem and progenitor cells. The role of NO therefore clearly warrants further investigation, especially in determining the precise role of the different NOS isoforms in the differentiation process and indeed the levels and profile of NO release that may be required for promoting or regulating differentiation.

The aim of these studies was therefore to confirm the role of NO in the generation of stem cell derived cardiomyocytes and to further investigate the potential mechanisms that may mediate the process. To achieve these aims NO donors with varying profiles of NO release were used in these investigations. The following nitric oxide donors were used:

NOC-5 (3-[2-Hydroxy-1-(1-methylethyl)-2-nitrosohydrazino]-1-propanamine) is a stable NO donor that can spontaneously release two equivalents of nitric oxide in solution under physiological conditions without any cofactor. NOC-5 is fast releasing NO donor with a half-life of 93 minutes. *SIN-1 (i.e. 3-Morpholinosydnonimine)* is also a fast releasing NO donor with a half-life of 40 minutes (Lomonosova *et al.*, 1998). It spontaneously decomposes to yield

nitric oxide and superoxide anion radicals. NOC-18 (*i.e. 2, 2'-* (*Hydroxynitrosohydrazino*)*bis-ethanamine*) is a slow releasing NO donor halflife with a half-life of 20 hours (Shami *et al.*, 1998). SNAP (i.e. ( $\pm$ )-S-Nitroso-N*acetylpenicillamine*) is also a slow releasing NO donor with a half-life of 37 hours (Ferrero *et al.*, 1999).

In further studies examining the role of NO in cardiac differentiation, guanosine 8-Bromo-,3',5'-cyclic monophosphate (8-Bromo-cGMP) was also used to determine whether NO may act through the cGMP pathway as indicated indirectly in other studies. 8-Bromo-cGMP is a cell-permeable cGMP analog that referentially activates protein kinase G. It is more resistant than cGMP to cleavage by phosphodiesterases.

The rationale for taking this approach is because NO generally acts by elevating cGMP levels within vascular cells in exerting its actions. Thus if NO was activating the guanylate cyclase pathway and enhancing cGMP levels in H9c2 cells then we would anticipate that treatment of cells with 8-Bromo-cGMP would mimic the effects seen with the NO donors. These studies were carried out using the H9c2 cells initially and it was hoped that the studies will be extended to the P19 stem cell line. The latter was however not used because of the negative data obtained with the H9c2 cells and time constraints.

## 4.2 MATERIALS AND METHODS

#### 4.2.1 Culture and differentiation of H9c2 Cells

The routine culture of H9c2 cells was carried out as described in section 2.3.1 (Chapter 2). H9c2 cells were trypsinised and the appropriate cell density for the differentiation experiments was calculated as described in section 2.6 (Chapter 2). H9c2 cells  $(2x10^3 \text{ cells/cm}^2)$  were seeded in p60 dishes or 6-well plates in full culture medium (i.e. DMEM supplemented with 10% FBS, 100units/ml penicillin/100µg/ml streptomycin), placed in a tissue culture incubator at 37°C, 95% air and 5% CO<sub>2</sub> and allowed to become 60-70% confluent. The differentiation of H9c2 cells into cardiomyocytes was carried out in 1% serum as described in section 2.8.1 (Chapter 2). The effects of the nitric oxide donors (NOC-5, NOC-18, SIN-1 or SNAP, each at a concentration of 100µM) on the differentiation of H9c2 cells into cardiomyocytes were investigated as described in section 2.8.2 (Chapter 2). The role of 8-BromocGMP (100µM) on the differentiation of H9c2 cells into cardiomyocytes was also investigated as described in section 2.8.2 (Chapter 2). The concentrations of NOC-5, NOC-18, SIN-1, SNAP and 8-Bromo-cGMP, used in experiments, were prepared by diluting the stock concentrations. The preparation of stock concentration, for each of the above compounds, was carried out as explained (Appendix III).

#### 4.2.2 Cell viability assay

To examine the effects of various NO donors on H9c2 cell viability, the MTT assay was carried out as described in section 2.7 (Chapter 2).

#### 4.2.3 Western blot analysis

In order to perform the western blot analysis, the following procedures were carried out as described in the Chapter 2 under the sections indicated in parenthesis: cell lysate generation (section 2.11), preparation of BSA standards (section 2.12.1) and BCA assay (section 2.12.2). The western blotting was carried out as described in section 2.13.

### 4.2.4 Data analysis

Data obtained from independent experiments were used for statistical analysis as described in section 2.14 (Chapter 2).

## 4.3 RESULTS

# 4.3.1 Effects of the NO donor NOC-5 on H9c2 cell differentiation into cardiomyocytes

Differentiation of H9c2 cells into cardiomyocytes, as determined by elongation and cell fusion, was significantly delayed in cells treated with NOC-5 when compared to controls. By day 4, differentiation was visible in control H9c2 cells. In contrast, NOC-5 treated cells maintained their normal single cell and spindle shaped morphology. However, by day 8, both controls and NOC-5 treated cells appear differentiated and this was clearly evident at day 12 (Figure 4.10). Thus although NOC-5 did not inhibit the long term differentiation of H9c2 cells into cardiomyocytes, it appeared to delay the onset of the process.

In parallel with the morphological changes, exposure to NOC-5 also caused a significant reduction in the expression of MLC-1v in treated cells. This was more evident at day 4 where MLC-1v expression was less than 50% of the levels seen in the control cells (p < 0.01) (Figure 4.11). Although MLC-1v levels increased at day 8 (p > 0.05) and at day 12 (p > 0.05), the levels detected were still only about 80% of those detected in control cells.



Figure 4.10: Effects of the NO donor NOC-5 on H9c2 cell differentiation into cardiomyocytes: morphological changes. Control H9c2 cells were cultured in differentiation medium consisting of DMEM supplemented with 1% FBS and 100 units/ml penicillin/100 µg/ml streptomycin. The differentiation medium of treated cells was further supplemented with 100µM of the nitric oxide donor, NOC-5. Cell alignment, elongation and myotube formation in H9c2 cells treated with 100µM of NOC-5 was slow in onset when compared to the control cells. While the control cells showed visible differentiation by day 4 (shown by black arrows), NOC-5 treated cells appeared to maintain their normal single cell and spindle shaped morphology. Morphologically, both control and NOC-5 treated H9c2 cells appeared to show marked differentiation by day 12 (shown by black arrows). Both control cells and NOC-5-treated cells showed the normal single cell morphology (shown by white arrows) at day 0. The photographs were taken under 200X magnification and are representative of at least three differentiation experiments.



Figure 4.11: Effects of the NO donor NOC-5 on H9c2 cell differentiation into cardiomyocytes: cardiac ventricular myosin light chain 1 expression. Western blot analysis of MLC-1v revealed that expression of this protein was suppressed by NOC-5 in comparison to the control expression. The blot is representative of at least three differentiation experiments (panel A). The densitometric data (mean  $\pm$  SD), from three independent experiments, show the time dependent changes in MLC-1v expression in NOC-5 treated H9c2 cells (panel B).

# 4.3.2 Effects of the NO donor SIN-1 on H9c2 cell differentiation into cardiomyocytes

SIN-1, like NOC-5, inhibited the early phase (day 4) of H9c2 cell differentiation into cardiomyocytes but NOC-5 appeared to have more effect at this stage. Cell elongation and cell fusion was slow in the H9c2 cells that were treated with SIN-1 in comparison to the control cells. By day 4, the usual differentiation was visible in the control H9c2 cells but not in the SIN-1 treated H9c2 cells. SIN-1 treated cells maintained their normal single cell and spindle shaped morphology (Figure 4.12). H9c2 cells in both control medium and SIN-1 supplemented medium showed differentiation at day 8 and day 12. Differentiation in both control and SIN-1 treated H9c2 cells was extensive by day 12.

The expression of MLC-1v in SIN-1 treated H9c2 cells was reduced at day 4, 8 and 12 in comparison to control cells (p > 0.05). MLC-1v levels in SIN-1 treated were just a little above 50% the control levels by day 4 (p > 0.05) (Figure 4.13). By day 8 MLC-1v levels in treated cells had recovered to more than 75% of control levels (p > 0.05). But by day 12, MLC-1v levels had only recovered to a little over 90% of MLC-1v levels in the control cells. MLC-1v in levels SIN-1 treated H9c2 cells never reached the levels seen in the control cells (p > 0.05). MLC-1v levels in both control and SIN-1 treated H9c2 cells were in agreement with the differentiation or morphological changes observed in these cells (Figure 4.13).



Figure 4.12: Effects of the NO donor SIN-1 on H9c2 cell Differentiation into Cardiomyocytes: Morphological changes. Control H9c2 cells were cultured in differentiation medium consisting of DMEM supplemented with 1% 1% FBS and 100 units/ml penicillin/100  $\mu$ g/ml streptomycin. The differentiation medium of treated cells was further supplemented with 100 $\mu$ M of the NO donor, SIN-1. Cell alignment, elongation and myotube formation in H9c2 cells treated with 100 $\mu$ M of the SIN-1 was slow in onset when compared to the control cells. Control cells showed visible differentiation by day 4, but NOC-5 treated cells appeared to maintain their normal single cell and spindle shaped morphology (shown by black arrows). Morphologically, both control and SIN-1 treated H9c2 cells appeared to show marked differentiation by day 12. Both control cells and SIN-1-treated cells showed the normal single cell morphology (shown by white arrows) at day 0. The photographs were taken under 200X magnification and are representative of at least three differentiation experiments.


Figure 4.13: Effects of the NO donor SIN-1 on H9c2 cell Differentiation into Cardiomyocytes: cardiac ventricular myosin light chain 1 expression. Western blot analysis of MLC-1v revealed that expression of this protein was slightly suppressed by SIN-1 in comparison to the control expression. The blot is representative of at least three independent differentiation experiments (panel A). The densitometric data (mean ± SD), from three independent experiments, show the time dependent changes in MLC-1v expression in SIN-1 treated H9c2 cells (panel B).

# 4.3.3 Effects of the NO donor NOC-18 on H9c2 cell differentiation into cardiomyocytes

Treatment of H9c2 cells with NOC-18 delayed their differentiation into cardiomyocytes as determined by the lack of differentiation associated morphological changes. By day 4, the usual differentiation was visible in the control H9c2 cells but not in the NOC-18 treated H9c2 cells. Like the SIN-1 treated cells, cell elongation and fusion was slow in the H9c2 cells that were treated with NOC-18 in comparison to the control cells. NOC-18 treated cells maintained their normal single cell and spindle shaped morphology (Figure 4.14). Both control and NOC-18 treated cells showed increased differentiation from day 8 to day 12. Differentiation or morphological changes in both control and NOC-18 treated H9c2 cells were extensive by day 12.

By day 4, MLC-1v levels in NOC-18 treated H9c2 cells were lower than the levels determined in the control cells (p > 0.05). The delay in MLC-1v expression, although not statistically significant, was in agreement with the lack morphological changes seen in these cells. By day 8, MLC-1v levels in NOC-18 cells had recovered to more than 95% of control levels p > 0.05)(Figure 4.15). By day 12, MLC-1v levels in NOC-18 treated slightly fell to about 80% of control levels p > 0.05). The levels of MLC-1v determined in the NOC-18 treated H9c2 cells never reached the levels seen in the control cells. In both control cells and NOC-18 treated H9c2 cells the levels of MLC-1v determined in the agreement with the differentiation or morphological changes observed in these cells.



Figure 4.14: Effects of the NO donor NOC-18 on H9c2 cell Differentiation into Cardiomyocytes: Morphological changes. Control H9c2 cells were cultured in differentiation medium consisting of DMEM supplemented with 1% 1% FBS and 100 units/ml penicillin and 100 µg/ml streptomycin. The differentiation medium of treated cells was further supplemented with 100µM of the nitric oxide donor, NOC-18. Differentiation (i.e. cell alignment, elongation and myotube formation) in H9c2 cells treated with NOC-18 was slow in onset when compared to the control cells. By day 4, differentiation (shown by black arrows) was visible in the control cells but not in NOC-18 treated cells which appear to maintain their normal single cell and spindle shaped morphology (shown by black arrows). Morphologically, both control and NOC-18 treated H9c2 cells appeared to show marked differentiation by day 12. Both control cells and NOC-18-treated cells showed the normal single cell morphology (shown by white arrows) at day 0. The photographs were taken under 200X magnification and are representative of at least three differentiation experiments.





# 4.3.4 Effects of the NO donor SNAP on H9c2 cell differentiation into cardiomyocytes

H9c2 cells treated with SNAP appeared to have differentiated to the same extent as the control cells. By day 4, the usual differentiation was visible in both control and SNAP treated H9c2 cells. The degree of cell elongation and fusion in both H9c2 and SNAP treated H9c2 cells were very similar. Morphological changes in both control and SNAP treated H9c2 cells increased from day 4 and by day 12, these changes were extensive as usual (Figure 4.16).

In agreement with the morphological seen in the control cells and SNAP treated cells, MLC-1v levels in SNAP treated H9c2 cells were almost equal to those seen in the control cells. SNAP treatment slightly reduced the expression of MLC-1v (p > 0.05) (Figure 4.17). By day 8, SNAP treatment increased the expression of MLC-1v by more than 100% of control levels (Figure 4.17). However the SNAP mediated increase in MLC-1v levels was not statistically significant (p > 0.05). By day 12, both control and SNAP treated cells had similar levels of MLC-1v (p > 0.05). MLC-1v levels in both control and SNAP-treated H9c2 cells were in agreement with the differentiation or morphological changes observed in these cells.



Figure 4.16: Effects of the NO donor SNAP on H9c2 cell Differentiation into Cardiomyocytes: Morphological changes. Control H9c2 cells were cultured in differentiation medium consisting of DMEM supplemented with 1% 1% FBS and 100 units/ml penicillin and 100  $\mu$ g/ml streptomycin. The differentiation medium of treated cells was further supplemented with 100 $\mu$ M of the nitric oxide donor, SNAP. The differentiation (i.e. cell alignment, elongation and myotube formation) (shown by black arrows) of H9c2 cells treated with SNAP was almost as much as the control cells. The degrees of cell elongation and fusion in both H9c2 and SNAP treated H9c2 cells were very similar. Morphological changes in both control and SNAP treated H9c2 cells increased from day 4 and by day 12, these changes were extensive as usual. Both control cells and SNAP-treated cells showed the normal single cell morphology (shown by white arrows) at day 0. The photographs were taken under 200X magnification and are representative of at least three independent experiments.



Figure 4.17: Effects of the NO donor SNAP on H9c2 cell Differentiation into Cardiomyocytes: cardiac ventricular myosin light chain 1 expression. MLC-1v levels in control and SNAP treated H9c2 cells were almost equal. SNAP treatment did not inhibit the expression of MLC-1v in H9c2 cells differentiating into cardiomyocytes. SNAP treatment increased MLC-1v levels in comparison to the control cells. The blot is representative of at least three independent experiments (panel A). The densitometric data (mean  $\pm$  SD), from three independent experiments, show the time dependent changes in MLC-1v expression in SNAP treated H9c2 cells (panel B).

# 4.3.5 Effects of 8-Bromo-cGMP on H9c2 cell differentiation into cardiomyocytes

To further explore the actions of NO in regulating the differentiation of H9c2 cell into cardiomyocytes, we examined whether 8-Bromo-cGMP, an analogue of the second messenger cGMP would mimic the effects seen with the NO donors. As shown in Figure 4.18, H9c2 cells treated with 8-Bromo-cGMP appeared to have differentiated almost as much as the control cells. Similar to the NO donor treated cells, the degree of cell elongation and fusion in both controls and 8-Bromo-cGMP treated H9c2 cells appeared at day 8 and at day 12. In parallel, the expression of MLC-1v in 8-Bromo-cGMP-treated H9c2 cells was reduced at day 4 (p > 0.05) but indistinguishable from the levels detected in the control cells at day 8 (p > 0.05) and at day 12 (p > 0.05) (Figure 4.19). Thus, like the NO donors, 8-Bromo-cGMP regulated H9c2 cell differentiation into cardiomyocytes by delaying the induction of this process.



*Figure 4.18: Effects of 8-Bromo-cGMP on H9c2 cell Differentiation into Cardiomyocytes: morphological changes*. Control H9c2 cells were cultured in differentiation medium consisting of DMEM supplemented with 1% FBS and 100 units/ml penicillin and 100 µg/ml streptomycin. The differentiation medium of treated cells was further supplemented with 100µM of 8-Bromo-cGMP. The differentiation (i.e. cell alignment, elongation and myotube formation) (shown by black arrows) of H9c2 cells treated with 8-Bromo-cGMP appeared to be almost as much as the control cells. The degrees of cell elongation and fusion in both H9c2 and 8-Bromo-cGMP treated H9c2 cells were very similar. Morphological changes in both control and 8-Bromo-cGMP treated H9c2 cells were extensive by day 12. Both control cells and SNAP-treated cells showed the normal single cell morphology (shown by white arrows) at day 0. The photographs were taken under 200X magnification and are representative of at least three independent experiments.



Figure 4.19: Effects of 8-Bromo-cGMP on H9c2 cell Differentiation into Cardiomyocytes: ventricular myosin light 1 (MLC-1v) expression. Western blot analysis of MLC-1v revealed that expression of this protein was initially delayed in 8-Bromo-cGMP treated H9c2 cells in comparison to the control expression (A). Thereafter MLC-1v levels increased to more than those seen in the control cells. The blot is representative of at least three independent experiments. The densitometric data (mean  $\pm$  SD), from three independent experiments, show the time dependent changes in MLC-1v expression in 8-Bromo-cGMP treated H9c2 cells (B).

## 4.3.6 Viability of H9c2 cells in various concentration of NOC-5

There was no difference between the viability of control H9c2 cells and cells treated with NOC-5 at concentrations ranging from  $3\mu$ M to  $30\mu$ M (p>0.05) (Figure 4.20). There was however a significant difference between the viability of control H9c2 cells and those cell treated NOC-5 concentrations above  $30\mu$ M (p<0.01). At these high concentrations, NOC-5 reduced cell viability to 66% at  $100\mu$ M and 9.5 % at  $1000\mu$ M. As a result  $100\mu$ M was selected for investigating the effects of NOC-5 on H9c2 cell differentiation into cardiomyocytes.



*Figure 4.20: Viability of H9c2 cells in the presence of various concentration of NOC-5.* H9c2 cells were seeded in 6-well plates and treated with NOC-5 as described in the methods section above (Section 4.3.1). The MTT assay was carried out at the end of the treatment. The data were analysed by one-way analysis of variance (ANOVA) test. The control viability was compared to the viability at each concentration using post-hoc Dunnett's multiple comparison test. There was no significant difference in cell viability between the control and each NOC-5 concentrations from 3 to  $30\mu$ M (p>0.05). There was a significant difference in cell viability between the control and each NOC-5 concentrations form 100 to  $1000\mu$ M (p<0.01). The data presented in the graph are the mean  $\pm$  SD from three independent experiments.

### 4.3.7 H9c2 Cell Viability in Various concentration of SIN-1

As seen with NOC-5, SIN-1 had no significant effect on cell viability at concentrations of  $3\mu$ M to 30 (Figure 4.21; p>0.05). There was a significant difference between the viability of control H9c2 cells and the viability of the H9c2 cells that were treated with each SIN-1 concentration above  $30\mu$ M (p<0.01). At 100uM SIN-1 reduced viability to 71.7% while the viability at with 1000 $\mu$ M SIN-1 was 10% that of control cells.



Figure 4.21: Viability of H9c2 cells in the presence of various concentration of SIN-1. H9c2 cells were seeded in 6-well plates and treated with SIN-1 as described in the methods section above (Section 4.3.1). The MTT assay was carried out at the end of the treatment. The data were analysed by one-way analysis of variance (ANOVA) test. The control viability was compared to the viability at each concentration using post-hoc Dunnett's multiple comparison test. There was no significant difference in cell viability between the control and each SIN-1 concentrations from 3 to  $30\mu$ M (p>0.05). There was a significant difference in cell viability between the control and each SIN-1 concentrations from 100 to  $1000\mu$ M (p<0.01). The data presented in the graph are the mean ± SD from three independent experiments.

### 4.3.8 H9c2 Cell Viability in Various concentration of NOC-18

Consistent with the other NO donors, NOC-18 did not cause any distinct cytotoxicity when used at concentrations of up  $30\mu$ M (Figure 4.22; p>0.05). Concentrations of  $100\mu$ M and above however caused marked cytotoxicity of the same magnitude seen with the other compounds. At  $100\mu$ M NOC-5 the cell viability was reduced to 67.5% while the viability of H9c2 cells treated with  $1000\mu$ M NOC-18 was only 9.8% of control cell viability.



Figure 4.22: Viability of H9c2 cells in the presence of various concentration of NOC-18. H9c2 cells were seeded in 6-well plates and treated with NOC-18 as described in the methods section above (Section 4.3.1). The MTT assay was carried out at the end of the treatment. The data were analysed by one-way analysis of variance (ANOVA) test. The control viability was compared to the viability at each concentration using post-hoc Dunnett's multiple comparison test. There was no significant difference in cell viability between the control and each NOC-18 concentrations from 3 to  $30\mu$ M (p>0.05). There was a significant difference in cell viability between the control and each NOC-18 concentrations from 100 to  $1000\mu$ M (p<0.01). The data presented in the graph are the mean  $\pm$  SD from three independent experiments.

# 4.3.9 H9c2 Cell Viability in Various concentration of SNAP

Of all the NO donors used, SNAP appears to be the least toxic and best tolerated since incubation of cells with this compound produced only marginal toxicity at concentrations of up to 200 $\mu$ M (Figure 4.23; p>0.05). SNAP at concentration at 300 $\mu$ M reduced viability to 63 % that of control H9c2 cells. SNAP concentration of 100 $\mu$ M, with cell viability of 92% was selected for investigating the effects of SNAP on H9c2 cell differentiation into cardiomyocytes.



Figure 4.23: Viability of H9c2 cells in the presence of various concentration of SNAP. H9c2 cells were seeded in 6-well plates and treated with SNAP as described in the methods section above (Section 4.3.1). The MTT assay was carried out at the end of the treatment. The data were analysed by one-way analysis of variance (ANOVA) test. The control viability was compared to the viability at each concentration using post-hoc Dunnett's multiple comparison test. There was no significant difference in cell viability between the control and each SNAP concentrations from 10 to 200 $\mu$ M (p>0.05). There was a significant difference in cell viability between the control and each SNAP concentrations from 10 to 200 $\mu$ M (p>0.05). There was a significant difference in cell viability between the control H9c2 cells an H9c2 cell treated with 300 $\mu$ M SNAP (p<0.05). The data presented in the graph are the mean ± SD from three independent experiments.

# 4.4 DISCUSSION

Nitric oxide has been suggested to play a role in stem cell differentiation into different lineages including cardiomyocytes. In this regard, NO has been reported to facilitate mouse embryonic stem cell differentiation into cardiomyocytes (Kanno et al., 2004). In NO donor (SNAP, PAPA/NO and DEA/NO) treated ES D3 cells and also in ES cells transfected with iNOS, the number of EBs with beating foci and the size of beating foci in the EB outgrowths was increased. In support of the cardiac differentiation of these cells being mediated by NO, the number of EBs with beating foci and also the size of the foci were decreased by L-NAME (Kanno et al., 2004). In similar study in ES D3 stem cells, it was shown that exposure of cells to arginine vasopressin (AVP) increased the number of beating embryoid bodies and also caused an increase in the expression of GATA-4. These AVP effects on the cells were also found to be antagonised by L-NAME (Gassanov et al., 2007) thus again suggesting a positive role for nitric oxide in stem cell differentiation into cardiomyocytes. As already highlighted, expression of eNOS, iNOS and nNOS have also been found in undifferentiated ES cells (Bloch et al., 1999; Krumenacker et al., 2006) with the profile of expression changing during the differentiation process (Krumenacker et al., 2006). The above mentioned data suggest a positive role of NO in the differentiation of stem cells into cardiomyocytes.

In addition to reported positive role of NO in the differentiation of stem cells into cardiomyocytes, NO has been reported to play positive role in the differentiation of other cell types, For example, over-expression of nNOS or

exogenous nitric oxide has also been shown to increase neuronal differentiation of neuroblastoma cells (Ciani *et al.*, 2004). A similar report has also shown that nNOS and nitric oxide are required for neuronal differentiation of PC12 cells. In the latter, transfection of PC12 cells with a nNOS dominant negative mutant block nerve cell growth factor-induced neuronal differentiation (Phung *et al.*, 1999), thus suggesting that nitric oxide has a positive role to play in PC12 cell neuronal differentiation. NO has also been shown to increased differentiation of cultured rat preadipocytes into adipocytes (Yan *et al.*, 2002).

In contrast to the positive roles of nitric oxide in cell differentiation, it has recently been reported that the differentiation of preadipocytes (3T3-L1 cells) into adipocytes is suppressed by NO (Kawachi *et al.*, 2007). In addition, inhibition of NO synthesis using L-NAME increased the number of stem and progenitor cells in the bone marrow of irradiated mice (Michurina *et al.*, 2004). Thus, it is evident that the actions of NO in regulating stem cell production and/or differentiation may be a complex process that requires further investigation. The conflicting reports about the role of NO in cell differentiation or variation in experimental procedures used by different laboratories. Nitric oxide is a molecule with a transient half-life and a diverse array of biological functions in humans. Thus when examining the actions of NO, the experimental conditions may be critical in dictating the results obtained.

We investigated the role of NO in the differentiation of H9c2 cells into cardiomyocytes using both fast NO releasing donors, NOC-5 and SIN-1 with respective half-lives of 40 and 93 minutes and the slow NO releasing donors NOC-18 and SNAP with respective half-lives of 20 and 37 hours. Our data showed that NO donors with different profiles of NO release can have different effects on H9c2 cell differentiation into cardiomyocytes. Both SIN-1 and NOC-5 delayed the onset of differentiation of H9c2 cells into cardiomyocytes. However SNAP and NOC-18 were not as effective in delaying the onset of differentiation or long term differentiation of H9c2 cells into cardiomyocytes. By day 8, H9c2 cell differentiation, as determined by MLC-1v western blot, in the presence of SNAP or NOC-18 had recovered to levels that were almost the same as those seen in the control cells. In fact by day 8, the expression of MLC-1v in SNAP treated was greater than that seen in the control cells. The delayed effects of fast-releasing NO donors on the onset of H9c2 cell differentiation can be interpreted as evidence that NO (at least from that fast-releasing NO donors) negatively regulates H9c2 cell differentiation into cardiomyocytes. However it should be noted, that fastreleasing NO donors release large amounts of their NO content in a matter of minutes and this may be detrimental to cells independent of direct and specific negative effects on the signalling events mediating the H9c2 cells differentiation into cardiomyocytes.

NO produced in small or regulated amount (as produced by constitutively expressed nNOS and eNOS), is involved in cell signalling leading to the regulation of many downstream targets. Soluble guanylyl cyclase, an enzyme

that generates cGMP, is the main physiological target for NO (Gold et al., 1990; Trigo-Rocha et al., 1993; Phung et al., 1999). Nitric oxide, by regulating the production of cGMP, also regulates various downstream protein kinases (e.g. cGMP-dependent kinase) (Archer et al., 1994; Pineda et al., 1996), ion channels (i.e. cGMP-dependent ion channels) (D'Ascenzo et al., 2002) and cGMP-dependent phosphodiesterases (NO can cause stimulation or inhibition) (Wexler et al., 1998; Zima et al., 2000). However, excessive formation of NO may be detrimental to the cells and tissues. This is evident from several reports that implicate NO in the pathogenesis of many diseases including stroke (Archer et al., 1994; ladecola et al., 1994; Ai et al., 2000; Li and Forstermann, 2000) and sepsis (Brealey et al., 2002). Excessive NO can disrupt cellular functions by binding to the heme group in cytochrome c oxidase and inhibit the enzyme leading to increased production of superoxide in the mitochondrial electron transport chain. The increased superoxide can further react with NO to yield peroxynitrite which can directly damage the enzymes in the mitochondrial electron transport chain (Brealey et al., 2002). This can cause major disturbances in the cells ability to produce ATP that is required for the maintenance of many vital cellular functions. Peroxynitrite is also an unstable compound which can breakdown into hydroxyl (OH<sup>-</sup>) and nitrite (NO<sub>2</sub>) radicals. The hydroxyl radical causes the most damage to cells and tissues and may be responsible for many cellular damages that are attributed to excess production of NO. The nitrite radical can covalently modify proteins by nitration and this can also lead to detrimental effects on cells.

Apoptosis is a vital biological process but abnormal apoptosis (i.e. too little or excessive apoptosis) as the results of breakdown of the balance between

anti-apoptotic and pro-apoptotic cellular signals, is detrimental to cells. Nitric oxide can mediate either apoptosis or cell survival. For example, it has been shown that over-expression of nNOS or the slow-releasing NO donor, DETA NONOate (i.e. NOC-18) protects human neuroblastoma SK-N-BE cells from serum withdrawal-induced apoptosis (Ciani et al., 2002b). Normal levels of NO can prevent apoptosis by inhibiting the activities of caspases including caspase-3, caspase-8 and 9 (Kim et al., 1997; Li et al., 1999; Zhou et al., 2005). Nitric oxide may also prevents apoptosis by regulating signalling from the fas receptor, another apoptotic mechanism (Hebestreit et al., 1998). On the other hand higher levels of NO may overcome the cell protective mechanisms against oxidative stress leading to apoptosis or other cytotoxic effects. The expression of key pro-apoptotic proteins, such as bax and caspases have been shown to be repressed by endogenous NO while the inhibition of NO production induces the expression of these pro-apoptotic proteins (Thippeswamy et al., 2001). Similarly the inhibition of NO production has also been shown to decrease the mRNA and protein levels of Bcl-2, an anti-apoptotic gene (Ciani et al., 2002a). It has been shown that inhibition of NO production by L-NAME, resulted in progressive apoptosis in cochlear ganglion cells (CGC) cells while slow-releasing NO donors or a cGMP analogue rescued the L-NAME induces apoptosis (Ciani et al., 2002a). From the above it can be appreciated that normal levels of NO can promote cell survival while excess NO can contribute to cell damage and death. Our MTT data, with all the NO donors used, showed that NO is toxic at high concentrations. Therefore, the balance between the transient but very high NO concentrations and the sustained production of NO at low concentrations

may determine whether NO plays a beneficial or detrimental role in a biological process. In cells undergoing differentiation, shifting this balance in favour of cell survival may play critical or positive roles, since cell differentiation is a very stressful process and this means that normal level of NO as generated by nNOS and eNOS or control released by slow-releasing NO donor may have positive effects on stem cell differentiation into cardiomyocytes. The fact that our data showed that fast NO releasing donors markedly delayed the onset of differentiation of H9c2 cells into cardiomyocytes while slow NO releasing donors were not as effective in delaying the onset of differentiation or long term differentiation of H9c2 cells into cardiomyocytes shows the need for a fine balance between the transient but very high NO concentrations and the sustained production of NO at low levels or concentrations.

To further explore the role of NO on the differentiation process, the effects of 8-Bromo-cGMP on H9c2 cell differentiation was investigated. As already mentioned, NO exerts most of its physiological actions via activation of soluble guanylate cyclase, which leads to an increase in intracellular concentration of the second messenger, cGMP (Gold *et al.*, 1990; Trigo-Rocha *et al.*, 1993; Phung *et al.*, 1999). The analog of cGMP, 8-Bromo-cGMP, is more resistant to phosphodiesterases than endogenous cGMP.

Our 8-Bromo-cGMP data are in agreement with the data obtained with some of the NO donors used in that treatment of cells with 8-Bromo-cGMP also delayed the onset of differentiation and reduced the expression of MLC-1v

significantly at the earlier stages of differentiation. However after prolonged incubations, MLC-1v expression was virtually indistinguishable in both controls and drug treated cells. What is not clear is whether this recovery is due to a loss in the actions of NO and/or 8-Bromo-cGMP over time. In our experiments, cells were exposed to the compounds for one hour prior to and then 48 hours after initiation of differentiation. After this period, cells were cultured in differentiation medium alone and it is possible that as a consequence of omitting the compounds from the incubation medium, their actions gradually diminished over time. What is clear however is the fact that NO (at least NO from fast-releasing NO donors) and indeed 8-Bromo-cGMP may inhibit rather than enhance the differentiation process, at least over the initial stages of the process. This is clearly contradictory to the other studies highlighted above which show a positive regulatory role of NO in the differentiation of stem cells not only into cardiomyocytes but also into other lineages. Our studies, as they stand, give little indication as to conflicting findings and further experiments are clearly needed especially in real stem cells or stem cell lines. These studies although planned were not carried out because of time constraints but are clearly needed. It would also be very useful in future studies to use NOS inhibitors and also combine NO donors with cGMP analogs (e.g. Br-cGMP) to complement the findings with the NO donors.

One possible but perhaps less convincing reason as to why the NO donors suppressed differentiation in our studies may be because of their ability to suppress cellular metabolic activity as indicated by the decreases in MTT

metabolism observed in the cytotoxicity assays carried out. Even if this did not result in cell death the reduction in metabolic activity within cells could regulate various cellular processes, including differentiation. The concentrations of the different compounds used were however well within the ranges used in several other studies. In fact NOC-5 has been used at concentrations ranging from 0.1-3000µM (Hotta et al., 1999; Okuyama et al., 2000; Kim et al., 2005) while NOC-18 and SNAP have both been used in other studies at concentrations of up to 500µM (Palmer et al., 2000; Sasaki et *al.*, 2000; Bal-Price and Brown, 2001; Niedbala *et al.*, 2002; Kim *et al.*, 2005). SIN-1 has been used at up to 2000µM (Mistry and Garland, 1998; Mathy-Hartert et al., 2000; Eligini et al., 2001; Kim et al., 2005). These are all much higher than the 100µM used in our studies. The 100µM concentration was chosen because the cells appeared to recover, without adversely detrimental effects, after the NO donors compound were removed from the incubation medium. In the future research, dose-response studies to clearly establish effective and non-toxic concentrations of these compounds, would very useful

# 4.5 Summary

NOC-5 and SIN-1 appeared to delay the onset of H9c2 cell differentiation into cardiomyocytes, as determined by cell elongation and fusion to form myotubes. In parallel with the morphological changes, NOC-5 and SIN-1 also caused a reduction in the expression of MLC-1v in differentiating H9c2 cells. NOC-18 also delayed the onset of H9c2 cell differentiation into cardiomyocytes but to a lesser extent than NOC-5 and SIN-1. Differentiation

associated morphological changes in both control and SNAP-treated H9c2 cells were similar. In agreement with the morphological seen in the control cells and SNAP treated cells, MLC-1v levels in SNAP-treated and control H9c2 cells were almost equal. 8-Bromo-cGMP, an analogue of cGMP, appeared to delay the onset of H9c2 cell differentiation into cardiomyocytes. However, like NOC-5 and SIN-1, 8-Bromo-cGMP failed to block the long term differentiation of H9c2 cells into cardiomyocytes. In parallel, expression of MLC-1v was reduced at day 4 but thereafter MLC-1v expression in controls and 8-Bromo-cGMP-treated cells were indistinguishable.

In conclusion, our data suggest that generation of cardiomyocytes from H9c2 cells may be negatively regulated by NO and this may occur through generation of cGMP. Our data are however preliminary and further studies are needed to confirm these findings. Moreover, a detailed study of the full signalling pathway activated by NO may prove important in understanding how this molecule regulates differentiation.

# CHAPTER 5.0

# **ROLE OF PROTEIN KINASE C IN CARDIOMYOCYTE**

# DIFFERENTIATION

# 5.1 INTRODUCTION

#### 5.1.1 Structural features of protein kinase C isoforms

Protein kinases Cs (PKC) are members of the serine-threonine second messenger-dependent protein kinase family. They play pivotal roles in mediating cellular responses to extracellular stimuli that regulate cell proliferation, differentiation and apoptosis. PKC isoforms are single polypeptides with an N-terminal regulatory region and a C-terminal catalytic region. A key regulatory and structural feature of the PKC isoforms is the presence of four conserved domains (i.e. C1, C2, C3 and C4) (Coussens *et al.*, 1986) (Figure 5.10).

The C1 domain contains a cysteine-rich motif (Parker *et al.*, 1986) that forms the 1, 2-diacyglycerol (DAG) (endogenous ligand) binding site. Diacylglycerol and phorbol esters (PKC activators) are hydrophobic anchors that recruit PKC enzyme to cell membranes (Orr *et al.*, 1992; Mosior and Epand, 1993). Phorbol esters, which are non-hydrolysable analogues of DAG, also bind to the C1 domain (Johnson *et al.*, 2000). The C1 domain also contains the autoinhibitory pseudosubstrate sequence upstream of the cysteine-rich motif (House and Kemp, 1987; House and Kemp, 1990). The C2 domain contains the recognition site for acidic lipids and in conventional PKC isoforms. It also contains the Ca<sup>2+</sup> ion binding site. The C3 domains form the ATP binding site and the C4 domain form the substrate binding site.



Figure 5.10: A schematic diagram showing the structure of conventional, novel and atypical PKCs. The pseudosubstrate domain (gold), C1 domain motifs (blue), C2 domain (green), C3 the ATP-binding lobe (pink) and C4 the substrate-binding lobe (gray) are shown. The C2 domain (i.e. novel C2) of novel nPKCs lacks  $Ca^{2+}$  binding capacity. The C1 (atypical C1) of atypical protein kinase Cs have only one Cys-rich motif which does not bind DAG or phorbol ester. PSS = pseudosubstrate, PHS = phosphatidyl-L-serine, PBE = phorbol esters and ATP= adenosine triphosphate

#### 5.1.2 Generation of PKC second messengers

Stimulus (e.g. growth factor) mediated activation of membrane bound phospholipase C (PLC) plays a central role in the activation of both conventional PKCs and novel PKCs. The stimulation of many G-protein coupled receptors can lead to the activation of PLC. Activated PLC converts phosphatidylinositol (1, 4) bisphosphate (PIP2) into inositol (1, 4, 5) trisphosphate (IP3) and DAG.

#### 5.1.3 Differential requirement of second messengers by PKC isoforms

All PKC isoforms via their C1 domains bind phosphatidyl-L-serine (phosphatidylserine) (Hurley *et al.*, 1997), a membrane acidic phospholipids that is located only on the cytoplasmic side of cell membranes. Phosphatidylserine binding is necessary for the high-affinity interaction of PKC with DAG containing membranes (Orr and Newton, 1994). The ten PKC isoforms are classified into three subgroups (conventional novel, and atypical PKCs) based on the composition of their regulatory domains moiety (Nishizuka, 1995; Musashi, 2000; Newton, 2003) (Figure 5.10). The particular configuration or domains in the regulatory region of each isoform determines the cofactor requirement of that isoform

#### 5.1.3.1 Conventional PKC (cPKC) isoforms

The conventional PKC alpha, beta<sub>1</sub>, beta<sub>2</sub> and gamma isoforms ( $\alpha$ ,  $\beta_I$ ,  $\beta_{II}$ ,  $\gamma$ ) contain functional C1 and C2 domains (Figure 5.10). They therefore require DAG and Ca<sup>2+</sup> ions for their full activities and functions.

### 5.1.3.2 Novel PKC (nPKC) isoforms

The novel PKC delta, epsilon, eta and theta isoforms ( $\delta$ ,  $\varepsilon$ ,  $\eta$ ,  $\theta$ ) contain a functional C1 domain and a non-ligand-binding (i.e. non-Ca<sup>2+</sup>/membranebinding) novel C2 domain (Figure 5.10). They are therefore Ca<sup>2+</sup> independent but require DAG for their full activation and functions. The C1 domain of novel PKCs has an intrinsic affinity for DAG-containing membranes that is 2 fold higher than that of the C1 domain of conventional PKCs (Giorgione *et al.*, 2006). This allows the novel PKCs to respond to agonists that cause the production of DAG but not Ca<sup>2+</sup> ion release. In contrast, cPKCs have to be targeted to membranes by their Ca<sup>2+</sup>-binding C2 domain, in response to elevated of cytosolic Ca<sup>2+</sup> ons, before they can to respond to DAG (Nalefski and Newton, 2001).

### 5.1.3.3 Atypical PKC (aPKC) isoforms

Atypical PKC zeta and tau/lamda isoforms ( $\zeta$ , i/ $\lambda$ ) contain a single non-DAGbinding ("atypical") C1 domain but no C2 domain (Figure 5.10). Therefore they do not respond to Ca<sup>2+</sup> ion or DAG or phorbol esters.

### 5.1.4 PKC activation by phosphorylation,

All PKC isoyzmes have to be primed by a series of ordered phosphorylations before they can become structurally competent for catalysis. These kinases share three conserved phosphorylation motifs which must be primed by phosphorylations at critical residues or else kinases will remain catalytically inactive (Newton, 2003).

The processing of conventional  $PKC_{BII}$  serves a model for the processing of PKC isoforms. The first phosphorylation of cPKC<sub>BII</sub> is catalyzed on a critical threonine residue (threonine 500), in the activation loop, by PDK-1 (i.e. the same kinase that activates Akt at threonine 308 in Akt activation loop). The importance of this phosphorylation event was demonstrated when it was found that mutation of the critical threonine residue to a neutral (i.e. nonphosphorylatable residue) inhibited PKC activity (Cazaubon et al., 1994). PDK-1 is responsible for the phosphorylation-driven priming of all PKC isoforms. PDK-1 phosphorylates conventional PKCs (Dutil et al., 1998), novel PKCs (Le Good et al., 1998; Cenni et al., 2002) and atypical PKCs (Chou et al., 1998; Le Good et al., 1998) at the critical threonine residues. In ES cells lacking PDK1 (PDK-1<sup>-/-</sup> cells), the intracellular levels of PKC $\alpha$ , PKC $\beta$ I, PKC $\gamma$ , PKC $\delta$  and PKC $\epsilon$ , were markedly reduced (Balendran *et al.*, 2000). While PDK-1 activity directed at Akt is phosphoinositide-dependent (i.e. PIP3 must be bound to Akt PH domain), PDK-1-driven phosphorylation of PKCs is PIP3 independent (Sonnenburg et al., 2001). This PDK-1-driven phosphorylation in activation followed by the loop is two further intramolecular autophosphorylations on two conserved residues; one on threonine 450 in the

turn motif in the C-terminal region and another on serine 660 in the hydrophobic motif (Newton, 2003). These phosphorylation events produce a mature (i.e. fully phosphorylated) PKC that is catalytically competent but not activated.

# 5.1.5 PKC activation by second messengers

The mature PKCs are localized to the cytosol and maintained in an inactive conformation by an auto-inhibitory pseudosubstrate sequence that blocks the active site. This auto-inhibition is relieved by allosteric interactions with Diacylglycerol binds both cPKC second messengers. and nPKC. Diacylglycerol binding recruits the enzymes to the membranes through their C1 and C2 domains. This induces a conformational change in the enzyme leading to the exposure of the auto-inhibitory pseudosubstrate domain and the removal of the pseudosubstrate from the substrate binding site thus allowing the substrate to bind and be phosphorylated (Orr et al., 1992; Orr and Newton, 1994; Sakai et al., 1997; Oancea and Meyer, 1998). In addition, IP3 mediates the release of Ca<sup>2+</sup> ions from intracellular stores and the Ca<sup>2+</sup> ions in turn potentiate the activation of cPKC. The activity of fully activated PKC activity is terminated by metabolism of DAG. As atypical PKCs do not respond to DAG or Ca<sup>2+</sup>, the only clear regulation is their requirement for PDK-1 for the phosphorylation step that primes these isoforms (Chou et al., 1998; Le Good et al., 1998). These PKC isoforms have a pseudosubstrate sequence that allosterically regulates the enzyme. Their activation may also

depend on PI3K and increase in PIP3 (Standaert *et al.*, 2001) but the exact mechanism of this is not clear.

# 5.1.6 PKC isoforms and cell differentiation

PKC isoforms have been implicated in the differentiation of various cell types (Balazovich *et al.*, 1987; Melloni *et al.*, 1987; Kanakaraj *et al.*, 1998; Kim *et al.*, 2005; Marchisio *et al.*, 2005a). Differential regulation of PKC isoforms has been shown during the differentiation of various cell types. For example, PKC  $\alpha$ ,  $\zeta$ , and  $\delta$  are strongly and preferentially expressed and activated by phosphorylation during DMSO-induced erythroid differentiation of Friend erythroleukemia cells (Marchisio *et al.*, 2005a). Similarly it has also been shown that enhanced protein levels and activities of PKC  $\beta$ I and  $\beta$ II isoforms are required for the monocytic differentiation of HL-60 cells (Kim *et al.*, 2005). It has also been reported that the expressions of PKC  $\alpha$  and  $\zeta$  did not change significantly during the differentiation of immature mononuclear satellite cells to polynuclear myotubes (Boczan *et al.*, 2000). In contrast, the expression of PKC  $\gamma$  and  $\eta$  increased with differentiation while PKC  $\theta$  showed high expression during the early phases of differentiation, but a decreased expression in the differentiated myotubes (Boczan *et al.*, 2000).

#### 5.1.7 PKC isoforms and stem cell differentiation

Like other cell types, differential regulation of PKC isoforms has also been shown during the differentiation of stem cells. Studies have shown that the expression of specific PKC isoforms changed significantly as ES cells differentiate into cardiomyocytes. It has been shown that the differentiation of ES cells into beating cardiomyocytes requires the down-regulation of PKC  $\beta$ and  $\zeta$  in conjunction with the up-regulation of PKC  $\epsilon$ (Zhou *et al.*, 2003).

From the above, it can be seen that while PKC isoforms have been implicated in the differentiation of stem cells and other cell types, their exact roles are currently poorly understood. Therefore the aim of this study is to investigate the role of the PKC pathway in the cardiac differentiation of H9c2 cell and P19 stem cells. H9c2 and P19 stem cells are different cell types but both can differentiate in cardiomyocytes. Investigation of the role of PKC in the cardiac differentiate in cardiomyocytes. Investigation of the role of PKC in the cardiac differentiation of these cells may yield information about the cell specific effects or possible cell-type independent effects of PKC during cardiac differentiation. To achieve the aim a PKC inhibitor, bisindolylmaleimide I (BIM-I) (Appendix VI (a)) was used for this study. BIM-I is a potent inhibitor for PKC (IC50=10nM) (Toullec *et al.*, 1991). It is highly selective for  $\alpha$  (IC50=20nM),  $\beta_{I}$ (IC50=17nM)  $\beta_{II}$ , (IC50=16nM),  $\gamma$  (IC50=20nM) (Toullec *et al.*, 1991). BIM-I also inhibit novel PKC<sub> $\delta$ </sub> (IC50=210nM), PKC<sub> $\epsilon$ </sub> (IC50=132nM) and atypical PKC<sub> $\zeta$ </sub> (IC50=5800nM)(Martiny-Baron *et al.*, 1993). BIM-I is a PKC competitive inhibitor and competes for the ATP-binding site of PKC (Toullec *et al.*, 1991).

# 5.2 MATERIALS AND METHODS

#### 5.2.1 Culture and differentiation of H9c2 Cells

The routine culture of H9c2 cells was carried out as described in section 2.3.1 (Chapter 2). H9c2 cells were trypsinised and the appropriate cell density for the differentiation experiments was calculated as described in section 2.6 (Chapter 2). H9c2 cells  $(2x10^3 \text{ cells/cm}^2)$  were seeded in p60 dishes or 6-well plates in full culture medium (i.e. DMEM supplemented with 10% FBS, 100units/ml penicillin/100µg/ml streptomycin), placed in a tissue culture incubator at  $37^{\circ}$ C, 95% air and 5% CO<sub>2</sub> and allowed to become 60-70% confluent. The differentiation of H9c2 cells into cardiomyocytes was carried out in 1% serum as described in section 2.8.1 (Chapter 2). The role of PKC signalling pathway in the differentiation of H9c2 cells into cardiomyocytes was investigated. The effects of the inhibition of the PKC signalling pathway were investigated using 10µM of BIM-I as described in section 2.8.2 (Chapter 2).

A BIM-I stock (10 mM solution) was prepared by dissolving 1mg of BIM-I in 242 $\mu$ I of DMSO. The stock solution was distributed into 10 or 20 $\mu$ I aliquots and stored in the freezer at -20°C. The aliquots were protected from light during storage and use. The differentiation medium was used to dilute (1:1000) the frozen stock BIM-I solutions (10mM) to the BIM-I concentration (10 $\mu$ M) used in the differentiation experiments. The preparation of stock BIM-I concentration was carried out as explained in appendix VIII. More details about BIM-I are located in appendix VI (a).

#### 5.2.2 Culture and differentiation of P19 stem cells

The routine culture of P19 stem cells was carried out as described in section 2.3.2 (Chapter 2). To initiate differentiation, P19 stem cells  $(1\times10^{6} \text{ cells})$  were seeded in P100 Petri (i.e. microbiological) dishes in the differentiation medium (i.e.  $\alpha$ -MEM supplemented with 10% FBS, 100 units/ml penicillin/100 µg/ml streptomycin and 0.8% DMSO) and the differentiation of P19 stem cells into beating cardiomyocytes carried out as described in section 2.9.1 (Chapter 2). The role of PKC signalling pathway in the DMSO-induced differentiation of P19 stem cells into beating cardiomyocytes was investigated. The effect of the inhibition of PKC signalling pathway by 10µM of BIM-I on the DMSO-induced P19 stem cell differentiation into beating cardiomyocytes was investigated as described in section 2.9.2 (Chapter 2). Furthermore the effects of delayed inhibition of PKC on P19 stem cell differentiation into cardiomyocytes were also investigated as described in section 2.9.3 (Chapter 2).

#### 5.2.3 Cell viability assay

To examine the effects of the PKC inhibition on H9c2 cell viability, the MTT assay was carried out as described in section 2.7 (Chapter 2).

# 5.2.4 Western blot analysis

In order to perform the western blot analysis, the following procedures were carried out as described in the Chapter 2 under the sections indicated in

parenthesis: cell lysate generation (section 2.11), preparation of BSA standards (section 2.12.1) and BCA assay (section 2.12.2). The western blotting was carried out as described in section 2.13.

#### 5.2.5 Data analysis

Data obtained from independent experiments were used for statistical analysis as described in section 2.14 (Chapter 3).

# 5.3 RESULTS

# 5.3.1 Effects of protein kinase C inhibition on H9c2 cell differentiation into cardiomyocytes

The inhibition of the PKC signalling pathway, using BIM-I ( $10\mu$ M, caused marked inhibition of cell elongation, fusion and myotube formations that usually accompany H9c2 cell differentiation into cardiomyocytes. Control H9c2 cells cultured in the absence of BIM-I underwent differentiation and formed myotubes as seen previously. In contrast, cells exposed to BIM-I ( $10\mu$ M) maintained their undifferentiated single cell and spindle shaped morphology at day 4 and 8. By day 12 there were only marginal signs of cell elongation, fusion and myotube formation in the in BIM-I treated cells (Figure 5.11).

In agreement with the lack of morphological change observed in BIM-I treated H9c2 cells, the expression of MLC-1v protein in these cells was completely blocked up to day 4. Expression of MLC-1v in BIM-I treated cells was however detectable at day 8, increasing marginally by day 12. MLC-1v levels in BIM-I treated cells, by day 12, were just slightly above 50% of the levels seen in the control cells (p<0.001) (Figure 5.12).


Figure 5.11: Effects of the protein kinase C inhibition on H9c2 cell differentiation into cardiomyocytes: morphological changes. Control H9c2 cells were cultured in differentiation medium consisting of DMEM supplemented with 1% FBS and 100units/ml penicillin/100µg/ml streptomycin. The above differentiation medium was further supplemented with 10µM of BIM-I for the treated cells. BIM-I caused significant inhibition of H9c2 cell differentiation (i.e. cell alignment, elongation and myotube formation). Cells cultured in the absence of this compound followed the normal differentiation pattern (shown by black arrows in day 4 to day12 control cells). In contrast, cells exposed to BIM-I maintained their normal single cell and spindle shaped morphology (shown by black arrows in day 4 to day12 BIM-treated cells). Morphologically at day 4, 8 and 12 there were only marginal signs of differentiation in BIM-I treated H9c2 cells. Both control cells and BIM-I-treated cells showed the normal single cell morphology (shown by white arrows) at day 0. The photographs were taken under 200X magnification and are representative of at least three independent differentiation experiments.





Figure 5.12: Effects of the protein kinase C inhibition on H9c2 cell differentiation into cardiomyocytes: cardiac ventricular myosin light 1(MLC-1v) expression. Western blot analysis of MLC-1v revealed that the expression of this protein in H9c2 cells was blocked by treating the cells with 10µM of BIM-I. A a marginal expression started at day 8 and by day 12, MLC-1v levels in BIM-I treated cells were just slightly above 50% of the levels seen in the control cells. At day 4, day 8 and day 12, there was significant difference between control cells and BIM-treated cells (p<0.001). This is indicated by a horizontal line (drawn from day 4 to day 12) with the indicated p-value. The blot is representative of at least three differentiation experiments (panel A). The densitometric data (mean  $\pm$  SD), from three independent differentiation experiments, show the time dependent changes in MLC-1v expression in BIM-I treated H9c2 cells (panel B).

# 5.3.2 Viability of H9c2 cells in 10 $\mu M$ of the protein kinase C inhibitor, BIM-I

The widely used concentration of 10 $\mu$ M for the PKC inhibitor BIM-I was selected for use in the experiments designed to investigate the effects of PKC inhibition on H9c2 differentiation into cardiomyocytes. BIM-I at the concentration of 10 $\mu$ M, like the other widely used concentrations of the other inhibitors used in this project, was not overtly toxic to the cells (Figure 5.13). The average viability was about 73% of the viability determined in the control cells. The control viability was significantly different than that in the treated cells at 6, 12, 24 and 48 hours (p< 0.01). At 48 hours (longest time that the control viability and this was comparable or similar to the viability seen in the cells treated with other drugs.



*Figure 5.13: Viability of H9c2 Cells in the presence of BIM-I.* H9c2 cells were seeded in 6-well plates and treated with 10 $\mu$ M BIM-I as described in the methods section above (section 5.3.1). The MTT assay was carried out at the end of each time point after the treatment. BIM-I at the concentration of 10 $\mu$ M was not toxic to the cells. The average viability was about 73% of the viability determined in the control cells. The control viability (control bar in the graph) was significantly different than the viability, calculated in BIM-I treated cells, at 6, 12, 24 and 48 hours (p< 0.01). At 48 hours (longest time that the cells were in the drug), the viability of treated cells was about 75% of the control viability and this was comparable or similar to the viability seen in the cells treated with other drugs. The MTT data presented in the graph are the mean  $\pm$  SD from three independent experiments

# 5.3.3 Effects of the protein kinase C inhibition on P19 stem cell differentiation into beating cardiomyocytes

The control and BIM-treated P19 cell embryoid bodies followed their normal pattern of growth as already described in the establishment of the P19 cardiomyocyte differentiation model (Chapter 3). Upto day 5, beating cardiomyocytes were not usually observed in control or treated cells. Between day 6 and 7, P19 stem cells treated with DMSO in the absence of BIM-I usually began to show signs of beating cardiomyocytes (Figure 5.15). When present, the number of beating clusters of cardiomyocytes usually continued to increase up to day 14. In contrast, P19 stem cells treated with DMSO in the presence of 10 $\mu$ M BIM-I did not show any signs of beating cardiomyocytes were observed in any of the dishes containing BIM-I treated P19 stem cells over the whole course of the experiment (Figure 5.14).

P19 stem cell-derived cardiomyocytes were highly variable. Beating rates amongst the clusters of cardiomyocytes were highly variable. The patterns of cardiomyocyte beating clusters (including number and beating rates), from P19 stem cells treated with DMSO in the absence of BIM-I (i.e. no inhibition of PKC signalling pathway), were as already described in Chapter 3 (see Figure 5.15).

In agreement with the lack of appearance of beating cardiomyocytes in BIM-I treated P19 stem cells, the long term expression of cardiac restricted protein troponin I was inhibited at day 8 and day 12. (Figure 5.16)



Figure 5.14: A schematic diagram of the effects of protein kinase C Inhibition on DMSO-induced differentiation of P19 stem cells into cardiomyocytes. PKC inhibition with BIM-I ( $10\mu$ M) blocked the differentiation of P19 stem cells into beating cardiomyocytes, when the inhibitors was added to the cells during the EB formation stage in the absence of BIM-I beating cardiomycytes were seen from day 6 to day 14 after the EB were transferred to cell culture dishes.



Figure 5.15: Effects of protein kinase C Inhibition on DMSO-induced differentiation of P19 stem cells into cardiomyocytes. Control P19 stem cell EBs were formed in differentiation medium consisting of  $\alpha$ -MEM supplemented with 10% FBS and 100 units/ml penicillin/100 µg/ml streptomycin and 0.8% DMSO. The above differentiation medium was supplemented with 10µM of BIM-I for the treated P19 stem cells. DMSO-induced differentiation of P19 stem cells into beating cardiomyocytes was inhibited by BIM-I. The areas of beating cardiomyocytes are shown within the closed figures drawn with white dotted lines. The beating focus of each beating area is shown by a black arrow. Photographs and movie clips of beating cells were taken under 200X magnification on the days indicated and are representative of microscope fields (for their respective days) from at least three independent differentiation experiments. D10 to D14 = day 10 to day 14.



Figure 5.16: Effects of the protein kinase C inhibition on P19 stem cell differentiation into cardiomyocytes: Troponin I expression. Control P19 stem cells were cultured in differentiation medium consisting of  $\alpha$ -MEM supplemented with 10% FBS, 100 units/ml penicillin and 100µg/ml streptomycin as described in section 5.3.2. The above differentiation medium was further supplemented with 10µM of BIM-I for the treated cells. In BIM-treated P19 stem cell, the long term expression of cardiac restricted protein troponin I was inhibited at day 8 and day 12. While beating cardiomyocytes were observed in some dishes of the control cells, beating cardiomyocytes were not seen in any of the BIM-I treated P19 stem cells. The blot is representative of 3 differentiation experiments.

# 5.3.4 Effects of delayed inhibition of protein kinase C on P19 stem cell differentiation into beating cardiomyocytes

To determine the stage at which PKC may regulate the differentiation of P19 stem cells into cardiomyocytes, BIM-I was added to cell culture four days after exposure to DMSO (Figure 5.17). Control P19 stem cells were treated with DMSO and BIM-I at the EB formation stage and for comparison, parallel dishes of P19 cells (*delay-inhibition P19 cells*) were treated with DMSO only at EB formation stage and then only treated with BIM-I after the EB were transferred to culture dishes (section 5.3.2).

Twenty four hours after the setup of cells in Petri dishes, cell aggregates (EBs) were observed in the control cells and also in the <u>delay-inhibition P19</u> <u>cells</u>. The EBs in both groups of cells underwent normal adhesion followed by proliferation and migration and both control and experimental cells were confluent by day 4. As already described, no beating cardiomyocytes were observed in both control and delay-treatment cells before day 6. By day 6 or 7, the first beating appeared in the <u>delay-inhibition P19 cells</u> (Figure 5.18). As expected, DMSO failed to induce P19 stem cell differentiation into beating cardiomyocytes, when 10µM of BIM-I was added at the EB formation stage.

However when DMSO was allowed to initiate P19 stem cell differentiation before the inhibition of the PKC signalling pathway (using 10µM of BIM-I), BIM-I failed to block the DMSO-induced differentiation of P19 stem cells into beating cardiomyocytes (Figure 5.17). The patterns of beating cardiomyocyte clusters (including number and beating rates) from P19 stem cells, in which

DMSO was allowed to initiate their differentiation before the inhibition of the PKC signalling pathway, were as already described in Chapter 3 (see Figure 5.18).



Figure 5.17: Effects of delayed inhibition of protein kinase C on P19 stem cell differentiation into beating cardiomyocytes. In P19 stem treated with BIM-I (10 $\mu$ M) four days after the initiation of DMSO-induced cardiac differentiation, BIM-I failed to block the cell differentiation into beating cardiomyocytes.



Figure 5.18: Effects of delayed inhibition of protein kinase C on P19 stem cell differentiation into beating cardiomyocytes. Control EBs were formed in differentiation medium consisting of  $\alpha$ -MEM supplemented with 10% FBS, 100units/ml penicillin/100µg/ml streptomycin, 0.8% DMSO and 10µM of BIM-I. For the delay inhibition of PKC, BIM-I was added four days after DMSO. Delayed addition of BIM-I has no effects on DMSO-induced differentiation of P19 stem cells into beating cardiomyocytes. The beating areas of cardiomyocytes are indicated by the dotted white circles. The beating focus of each beating area is shown by a black arrow. Photographs and movie clips of beating cells were taken under 200X magnification on the days indicated and are representative of microscope fields (for their respective days) from at least three independent differentiation experiments. D10 to D14 = day 10 to day 14.

## 5.4 DISCUSSION

At both morphological and biochemical levels, the data showed that the inhibition of PKC by BIM-I markedly inhibited H9c2 cell differentiation into cardiomyocytes. Control cells cultured in the absence of BIM-I underwent the usual differentiation associated morphological changes (i.e. cell alignment, elongation and fusion to form myotues). In contrast BIM-treated H9c2 cells maintained their normal single cell and spindle shaped morphology with only marginal signs of morphological changes. This strongly suggests that the activation of PKC signalling may have key roles to play in the events that are responsible for causing these morphological changes during H9c2 cardiac differentiation. In agreement with the lack of morphological changes observed in BIM-I treated H9c2 cells, the BIM-I suppression of the expression of MLC-1v also suggest roles for PKC activation during H9c2 cell differentiation into cardiomyocytes.

In agreement with the PKC pathway having a role to play during H9c2 cell differentiation into beating cardiomyocytes, PKC inhibition also blocked the differentiation of P19 stem cells into cardiomyocytes. Control P19 stem cells differentiated into beating cardiomyocytes and expressed cardiac specific troponin I while BIM-I treated cells failed to differentiate into beating cardiomyocytes. Furthermore P19 stem cell cardiac differentiation as determined by the expression of cardiac specific troponin I was also inhibited. Our data are significant in that, it is suggestive of a need for PKC activation in both H9c2 and P19 stem cells. H9c2 and P19 stem cells are different cell types but both can differentiate into cardiomyocytes. The need for PKC

activation for the differentiation of both cell types (i.e. H9c2 and P19 cells) therefore suggests that the activation of PKC signalling pathway may be a fundamental or an essential signalling event during the differentiation of stem cells into cardiomyocytes.

The human embryonal carcinoma cell line NT2/D1 can differentiate into neuronal cells in the presence of retinoic acid. An increased PKC activity has been shown to accompany neuronal differentiation of these cells (McCarthy *et al.*, 1995). All-*trans*retinoic acid (ATRA) can induce the differentiation of the human promyelocytic leukaemia cell line (HL-60) into mature granulocyte-like cells. It has also been found that the activation of PKC is necessary for ATRA-induced differentiation of HL-60 cells (Kanakaraj *et al.*, 1998; Kim *et al.*, 2005).

PKC translocation from the cytoplasm to the plasma membrane is a central mechanism for the activation and activity of PKC isoforms, however selective nuclear accumulation of PKC  $\zeta$  have been reported in HL-60 cells induced to differentiate along the granulocytic lineage by ATRA (Zauli *et al.*, 1996; Bertolaso *et al.*, 1998; Neri *et al.*, 1999). PKC  $\zeta$  also accumulated in the nuclei of U937 cells (human promonocytic leukemia cells) that have undergone monocyte differentiation(Kiley and Parker, 1995). More importantly, it has been reported that PKC  $\zeta$  nuclear accumulation, was blocked by the inhibition of PI3K using Wortmannin (Neri *et al.*, 1999).The inhibition of PKC  $\zeta$  nuclear translocation, by wortmannin , has also been reported during ischemia in rat hearts (Mizukami *et al.*, 1997). Increased expression and phosphorylation of

PKC  $\alpha$ ,  $\varepsilon$ ,  $\zeta$  and  $\delta$  has been shown following DMSO-induced differentiation in Friend erythroleukemia cells also suggest a possible role of this PKC in the differentiation of these cells (Marchisio *et al.*, 2005b). All the above results suggest that PKC may be involved in cell differentiation. These studies also suggest that the localisation of PKC isoforms to different subcellular compartments may be a mode of regulation or an indication of unique functions for individual PKC isoforms.

In agreement with suggestions from the previous studies that PKC isoforms have role to play in cell differentiation, our data suggest that PKC have may a role in the differentiation of both P19 stem cells and H9c2 cells. BIM-I blocked the differentiation of both cells into cardiomyocytes. While previous data suggest important roles for PKC during cell differentiation and our data specifically suggest a role for PKC in the differentiation of stem cells into cardiomyocytes, the exact mechanism by which PKC can mediate its positive effects on cell differentiation is not clearly defined. PKC isoforms are second messenger dependent kinases. PKC activation depends on either DAG or Ca<sup>2+</sup> or both depending on the isoform. The increase in intracellular DAG or Ca<sup>2+</sup> is mediated by phospholipase C (PLC). The elevation of intracellular Ca<sup>2+</sup> ions is essential for the activation of conventional PKC isoforms. It is therefore very interesting to note that that some members of the Wnt-5a group that signal through the non-canonical Wnt/Ca<sup>2+</sup> signalling pathway caused a PLC mediated increased in intracellular Ca<sup>2+</sup> ion levels (Sheldahl et al., 1999; Pandur et al., 2002; Sheldahl et al., 2003) that go on to activate PKC. It is/was generally thought that signalling events downstream of Whts

belonging to Wnt-5a group, via the Wnt/Ca<sup>2+</sup> signalling pathway, cause the inhibition of the canonical Wnt/ $\beta$ -catenin signalling pathway (Torres *et al.*, 1996; Topol *et al.*, 2003; Westfall *et al.*, 2003; Maye *et al.*, 2004). However it is becoming clear that the assigned sole function of the non-canonical Wnt/Ca2+ signalling as an inhibitor of the canonical Wnt/ $\beta$ -catenin signalling pathway may not be as simple as reported.

As discussed in Chapter 3, the available evidence suggests that DMSOinduced differentiation of P19 stem cells into cardiomyocytes may occur through the Wnt/ $\beta$ -catenin signalling pathway. Calcium ion release from the Wnt/Ca<sup>2+</sup> signalling would be expected to activate PKCs. If the Wnt/Ca<sup>2+</sup> signalling pathway function is the inhibition of the Wnt/ $\beta$ -catenin signalling pathway, then PKC (at least the conventional PKC isoforms) would be expected to play negative roles in the differentiation of P19 stem cell into cardiomyocytes. Clearly our data are in direct contrast to that expectation. Our data showed that inhibition of PKC led to the inhibition of the cardiac differentiation of P19 stem cells and H9c2 cells.

The activation of the Wnt/Ca<sup>2+</sup> signalling pathway has been shown to cause calcium ion influx and the activation of PKC (Sheldahl *et al.*, 2003). Interestingly it has also been shown that the activation PKC inactivated GSK- $3\beta$  and increased the accumulation of cytosolic  $\beta$ -catenin (Garrido *et al.*, 2002). In the same study, it was also found that the inhibition of Ca<sup>2+-</sup> dependent PKC isoforms activated GSK- $3\beta$ . Wnt-3a and lithium mimicked PKC activation (Garrido *et al.*, 2002; Sanai *et al.*, 2004). Clearly the

inactivation of GSK-3 $\beta$  by PKC and the accumulation of cytosolic  $\beta$ -catenin will enhance rather than inhibit the Wnt/ $\beta$ -catenin signalling pathway. Whether PKC in our experimental system can phosphorylate and inactivate GSK-3 $\beta$  remains to be examined. Whether PKC in our experimental system was activated by Ca<sup>2+</sup> ions from the Wnt/Ca<sup>2+</sup> signalling pathway also remains to be examined. It is possible that the PKC in our experimental system may be activated in a mechanism that is entirely independent of the Wnt/Ca<sup>2+</sup> signalling pathway. However if PKC in our system was activated by Wnt/Ca<sup>2+</sup> signalling pathway then our data suggest that Wnt/Ca<sup>2+</sup> signalling pathway may also contribute to Wnt/ $\beta$ -catenin mediated differentiation of stem cells in cardiomyocytes in contrast to the inhibitory role assigned to this pathway during cardiogenesis.

# 5.5 Summary

The inhibition of the PKC signalling pathway, using BIM-I (10µM), caused marked inhibition of cell elongation, fusion and myotube formation that usually accompany H9c2 cell differentiation into cardiomyocytes. Control H9c2 cells cultured in the absence of BIM-I underwent differentiation and formed myotubes. In agreement with the lack of morphological change observed in BIM-I-treated H9c2 cells, the expression of MLC-1v protein in these cells was completely blocked up to day 4 and then significantly inhibited thereafter.

P19 stem cells cultured in 0.8% DMSO in the absence of BIM-I differentiated into beating cardiomyocytes. In contrast, P19 stem cells treated with 0.8%

DMSO in the presence BIM-I (10µM) did not differentiate into beating cardiomyocytes. However when DMSO was allowed to initiate P19 stem cell differentiation before the addition of BIM-I, the cells differentidate into beating cardiomyocytes (i.e. BIM-I failed to block their DMSO-induced cardiac differentiation). In agreement with the lack of appearance of beating cardiomyocytes in BIM-I treated P19 stem cells, the long term expression of cardiac restricted protein troponin I (TnI) was inhibited.

In conclusion, our data therefore suggest that the activation of PKC signalling pathway may be required for the differentiation of H9c2 cell and P19 stem cell differentiation into cardiomyocytes. The data also suggest that the activation of PKC signalling pathway may be required during the early stages of P19 stem cell differentiation into cardiomyocytes.

# CHAPTER 6.0

# ROLE OF p38 MAP KINASE IN H9C2 AND P19 STEM CELL DIFFERENTIATION INTO CARDIOMYOCYTES

### 6.1 INTRODUCTION

#### 6.1.1 Mitogen-activated protein kinase

The p38 kinases are serine/threonine kinases that belong to a very large protein kinase family known as the mitogen-activated protein kinases (MAPKs). The MAPKs control several cell functions including gene expression (Wu et al., 2004; Lawrence et al., 2007; Mochizuki et al., 2007), differentiation (Zetser et al., 1999; Jadlowiec et al., 2004; Bokui et al., 2008; Kook et al., 2008), proliferation (Souza et al., 2004; Erlich et al., 2007; Jia et al., 2008), migration (Goetze et al., 1999; Rousseau et al., 2006; Jia et al., 2008) and cell survival or apoptosis (Chuang et al., 2000; Pan et al., 2002; Chen et al., 2008; Cho et al., 2008). They are activated by a range of extracellular stimuli (mitogens) including growth factors (Moriuchi et al., 2001; McFarland and Pesall, 2008), hypotonic stress (Niisato et al., 2007) and oxidative stress (Kurata, 2000; Gaitanaki et al., 2003; Kim et al., 2004; Frossi et al., 2007). The effects of antioxidant on various MAPKs may depend on the MAPK and/or the activating stimulus. For example the antioxidants, ascorbic acid, catalase and superoxide dismutase, have been reported to suppress p38-MAPK signalling pathway in the perfused frog hearts (Gaitanaki et al., 2006). On the other hand, in cultured rat aortic smooth muscle cells, diphenyleneiodonium chloride (DPI), ascorbic acid, N-acetyl cysteine (NAC), diethyldithiocarbamic acid 6-Hydroxy-2,5,7,8-(DETC) and tetramethylchromane-2-carboxylic acid (Trolox C) have no effects on angiotensin II-induced activation of ERK 1/2 activation (Kyaw et al., 2001). In contrast, angiotensin II-induced activation of p38 MAPK was inhibited by DPI

and ascorbic acid, NAC (at high concentration) but not DETC or Trolox C (Kyaw *et al.*, 2001).

The MAPK signalling pathway architecture consists of an arrangement of three sequentially acting protein kinases (Figure 6.10). The pathway is activated by a small guanosine 5'-triphosphate (GTP)-binding protein (GTP) protein (of the Ras or Rho family) or by an adaptor protein. The adaptor protein transmits the signal directly to MAP kinase kinase kinase (MAPKKK) (i.e. the kinase at the top of this architecture). Alternatively, the adaptor protein can also transmit the signal through a mediator kinase (i.e. MAP kinase kinase kinase kinase kinase (MAPKKK)) which phosphorylates and activates MAPKKK. In the second step, the activated MAPKKK phosphorylate and activate MAP kinase kinase kinase (MAPKK) which in turn activates MAPK. Activated MAPK then phosphorylate a wide range of substrates.

Specificity is achieved among this complex arrangement of competing MAPK pathways because the kinase activities at each sequential phosphorylation step are directed toward specific downstream kinases by specific protein-protein interactions. Specific scaffold proteins that interact with substrates and their corresponding kinases along with the presence of specific docking sites or sequences in the MAPKs ensure specific recognition and activation by appropriate upstream kinases (Bardwell *et al.*, 2003; Tatebayashi *et al.*, 2003; Sharma and Mondal, 2006). Termination of MAPK activation (i.e. inactivation) is regulated by dephosphorylation that is carried out by MAPK phosphatases (MKPs) (Bardwell *et al.*, 2003; Noordman *et al.*, 2006).



Figure 6.10: MAPK signalling pathway architecture. A schematic diagram showing the three sequential levels for the activation of downstream MAPKs by upstream MAPKs in the architecture of the MAPK signalling pathways. Some examples of targets for terminal kinases (i.e. p38, ERK and cjun/SAPK) and downstream biological effects are shown in the diagram. NFAT = Nuclear Factor of Activated T cells, STAT= Signal Transducers And Activators Of Transcription, MEF2C = Myocyte Enhancer Factor 2 C, ATF2= Activating Transcription factor 2, MSK= MAPK/SAPK-activated kinase, MNK1= MAPK signal-activating kinase 1, Elk-1 = E26-like protein 1, Transforming growth factor- $\beta$ -activated kinase (TAK), Mixed lineage kinase 3, MEKK1/4 = Mitogen ERK kinase kinase, 1/4, ASK= apoptosis stimulating kinase, DLK= dual leucine zipper bearing kinase, ERK= extracellular signal-regulated kinase JNK= c-Jun N-terminal kinase, SAPK= stress-activated protein kinase, MAPKKK=MAP kinase kinase kinase. MAPKK=MAP kinase kinase. MAPK = Mitogen activated protein kinase, MEK = Mitogen-activated Protein/Extracellular Signal-regulated Kinase Kinase (MEK).

#### 6.1.2 Mitogen-activated protein kinase family

The MAPK family consists of three subgroups: p38 MAPK, extracellular signal-regulated kinase (ERK) and c-Jun N-terminal kinase/stress-activated protein kinase (JNK/SAPK).

## 6.1.2.1 P38 MAPK Pathway

Of the four p38 isoforms (p38 $\alpha$  p38 $\beta$ , 38 $\gamma$  and p38 $\delta$ ), p38 $\alpha$  is expressed in all tissues. The p38 MAPK pathway is activated by serum (Beier and LuValle, 1999), osmotic stress (Bell et al., 2000; Volonte et al., 2001), UV radiation (Bulavin et al., 1999; Bodero et al., 2003; Sethi and Sodhi, 2004) and heat shock (Dorion et al., 2002; Zhou et al., 2005; Venkatakrishnan et al., 2006). The p38 MAPK pathway is also activated by proinflammatory cytokines (Nick et al., 1999; Suzuki et al., 2001). The p38 MAPK pathway is involved in apoptosis (Chuang et al., 2000; Cho et al., 2008), stress (Frossi et al., 2007; Niisato et al., 2007), proliferation (Souza et al., 2004), cell differentiation (Morooka and Nishida, 1998; Cuenda and Cohen, 1999; Zetser et al., 1999; Davidson and Morange, 2000; Kook et al., 2008) and cell survival. In the p38 pathway, stimulus-activated G proteins activate MAPKKKs. The MAPKKKs in turn phosphorylate and activate the MAPKKs (e.g. MKK3 and MKK6). These MAPKK then activate p38. The substrates for activated p38 kinases include cytosolic phospholipase A2 (Kramer et al., 1996; Hazan-Halevy and Levy, 2000; Degousee et al., 2001; You et al., 2005), the microtubule-associated protein Tau (Zhu et al., 2000; Li et al., 2003), transcription factors (MEF2C

(Yang *et al.*, 1999; Zhao *et al.*, 1999; de Angelis *et al.*, 2005)) and other protein kinases.

#### 6.1.2.2 Extracellular signal-regulated kinase (ERK)

The ERK are activated by a wide range of extracellular stimuli including growth factors (Bottazzi et al., 1999; Ouwens et al., 2002; Bobick et al., 2007), serum (Frost et al., 1994; Beier and LuValle, 1999), phorbol esters, cytokines and osmotic stress (Chiri et al., 2004). These stimuli activate the small GTPbinding protein Ras. Activated Ras recruit Raf to the cell membrane where it becomes activated. The activated Raf (i.e. the MAPKKK) then phosphorylates and activates the MAPKK (MEK1 and 2) which in turn phosphorylate ERK1/2. The substrates for activated ERK1/2 include other downstream kinases such as ribosomal S6 kinases (RSKs) (Smith et al., 1999), MAPK/SAPK-activated kinase (MSK) (Deak et al., 1998) and MAPK signal-activating kinase 1 (MNK1) (Fukunaga and Hunter, 1997; Waskiewicz et al., 1997). Activated ERK1/2 also phosphorylate transcription factors such as NF-AT (Ali et al., 2000; Sanna et al., 2005), Elk-1(Muller et al., 1997; Vanhoutte et al., 1999; Thiels et al., 2002), c-Fos (Vanhoutte et al., 1999; Monje et al., 2003; Burch et al., 2004; Pellegrino and Stork, 2006), c-Myc, (Pintus et al., 2002; Ramljak et al., 2003; Serra et al., 2008), STAT3 (Kanai et al., 2003) and apoptosisrelated proteins (Jan et al., 1999; Scheid et al., 1999; Ellert-Miklaszewska et al., 2005; Sawatzky et al., 2006; Caraglia et al., 2007). The ERK1/2 signaling pathway is involved in cell proliferation (Souza et al., 2004), differentiation

(Kim *et al.*, 2005), transformation and apoptosis (Jan *et al.*, 1999; Chuang *et al.*, 2000; Ellert-Miklaszewska *et al.*, 2005; Caraglia *et al.*, 2007).

#### 6.1.2.3 C-Jun N-terminal kinase/stress-activated protein kinase

The c-Jun N-terminal kinase/stress-activated protein kinases (JNK/SAPK) pathway are activated by growth factors, cytokines and stress related responses (e.g. UV radiation and heat). JNK/SAPK signalling pathway is mainly involved in responses to stress. Like ERK, JNK/SAPK are also involved in cell proliferation (Sabapathy *et al.*, 2001; Zhang *et al.*, 2005; Ding *et al.*, 2007), differentiation (Cuenda and Cohen, 1999; Lemonnier *et al.*, 2004) and apoptosis (Chuang *et al.*, 2000; Cho *et al.*, 2008).

In the JNK/SAPK pathway, activated G proteins (Cdc42, Rac, or Ras) activate several MAPKKK (e.g. apoptosis stimulating kinase (ASK) and other molecules including transforming growth factor-β-activated kinase (TAK)). These MAPKKKs, in turn, activate the MAPKKs (e.g. MKK4 and MKK7). The activated MAPKKs in turn directly phosphorylate and activate JNK/SAPK. The substrate for activated JNK/SAPK includes several transcription factors such as c-jun (Hatzoglou *et al.*, 2000; Zhong *et al.*, 2007), ATF2 (Hayakawa *et al.*, 2003; Bailey and Europe-Finner, 2005), STAT3 (Turkson *et al.*, 1999) and Elk-1 (Hatzoglou *et al.*, 2000; Zhang *et al.*, 2007; Zhong *et al.*, 2007).

#### 6.1.2.4 MAPK and stem cell differentiation

The MAPK have been implicated in the differentiation of various cells including the differentiation of stem cells into cardiomyocytes and other cell types (Zhang and Bradley, 1996; Schlange et al., 2000). However the exact mechanism by which MAPK can regulate cell differentiation is not currently clear. There is evidence that GATA-4 actions may require the activation of MAPK (i.e. p38) dependent pathway (Charron et al., 2001; Liang et al., 2001). Given their implication in the differentiation of various cells and the lack of clear understanding of their role in differentiation, the MAPK are therefore good candidates for studies aimed at understanding the cell signalling pathways that are responsible for mediating stem cell differentiation into cardiomyocytes. P38 MAPK is downstream of PI3K and our data have already suggested that when PI3K is inhibited, the differentiation of H9c2 and P19 stem cell differentiation into cardiomyocytes, is also inhibited. We therefore extended our studies to determine whether the signalling events that control differentiation of these cells into cardiomyocytes are linked to the activation of the p38 MAPK.

The aim of this study was to investigate the role of the p38 MAPK pathway in the cardiac differentiation of H9c2 and P19 stem cells. Like p38 MAPK, ERK and JNK have also been implicated in differentiation of various cell types. We initially started by examining the role of p38 MAPK in the hope that the study could be extended to examination of the roles of ERK and JNK. The roles of JNK and ERK cardiac differentiation were not examined in this project due to time constraints and to other factors. The design of this study focused on the

use of a specific competitive inhibitor of p38 MAP kinase known as SB203580 (4-(4-Fluorophenyl)-2-(4-methylsulfinylphenyl)-5-(4-pyridyl)1H-imidazole) (IC50=0.6µM) (Cuenda *et al.*, 1995). SB203580 (100µM) (i.e. ten times the widely used concentration of 10µM), does not inhibit JNK/SAP kinase or p42 MAP kinase (Cuenda *et al.*, 1995). SB20380 competes for enzyme ATP binding site of p38 MAPK (Cuenda *et al.*, 1995; Young *et al.*, 1997; Gum *et al.*, 1998).

### 6.2 MATERIALS AND METHODS

#### 6.2.1 Culture and differentiation of H9c2 Cells

The routine culture of H9c2 cells was carried out as described in section 2.3.1 (Chapter 2). H9c2 cells were trypsinised and the appropriate cell density for the differentiation experiments was calculated as described in section 2.6 (Chapter 2). H9c2 cells  $(2x10^3 \text{ cells/cm}^2)$  were seeded into p60 dishes or 6-well plates in full culture medium (i.e. DMEM supplemented with 10% FBS, 100units/ml penicillin/100µg/ml streptomycin), placed in a tissue culture incubator at  $37^{\circ}$ C, 95% air and 5% CO<sub>2</sub> and allowed to become 60-70% confluent. The differentiation of H9c2 cells into cardiomyocytes was carried out in 1% serum as described in section 2.8.1 (Chapter 2). Role of the p38 MAPK signalling pathway in the differentiation of H9c2 cells into cardiomyocytes was investigated using SB203580 (10µM) as described in section 2.8.2 (Chapter 2).

#### 6.2.2 Culture and differentiation of P19 stem cells

The routine culture of P19 stem cells was carried out as described in section 2.3.2 (Chapter 2). To initiate differentiation, P19 stem cells  $(1\times10^{6} \text{ cells})$  were seeded in P100 Petri (i.e. microbiological) dishes in the differentiation medium (i.e.  $\alpha$ -MEM supplemented with 10% FBS, 100 units/ml penicillin/100 µg/ml streptomycin and 0.8% DMSO) and the differentiation of P19 stem cells into beating cardiomyocytes carried out as described in section 2.9.1 (Chapter 2). The role of p38 MAPK signalling pathway in the DMSO-induced differentiation

of P19 stem cells into beating cardiomyocytes was investigated using SB203580 (10 $\mu$ M) as described in section 2.9.2 (Chapter 2). An SB203580 stock (10 mM solution) was prepared by dissolving 1mg of SB203580 in 264.97 $\mu$ l of DMSO. The stock solution was distributed into 10 or 20 $\mu$ l aliquots and stored in the freezer at -20°C. The aliquots were protected from light during storage and use. The differentiation medium was used to dilute (1:1000) the frozen stock SB203580 solutions (10mM) to the SB203580 concentration (10 $\mu$ M) used in the differentiation experiments. The preparation of stock SB203580 concentration was carried out as explained in appendix VIII. More details about SB203580 are located in appendix VI (c). Furthermore the effects of delayed inhibition of p38 MAPK on P19 stem cell differentiation into cardiomyocytes were also investigated as described in section 2.9.3 (Chapter 2).

#### 6.2.3 Cell viability assay

To examine the effects of the inhibition of p38 MAPK, with SB203580, on H9c2 cell viability, the MTT assay was carried out as described in section 2.7 (Chapter 2).

#### 6.2.4 Western blot analysis

In order to perform the western blot analysis, the following procedures were carried out as described in the Chapter 2 under the sections indicated in parenthesis: cell lysate generation (section 2.11), preparation of BSA

standards (section 2.12.1) and BCA assay (section 2.12.2). The western blotting was carried out as described in section 2.13.

# 6.2.5 Data analysis

Data obtained from three independent experiments were used for statistical analysis as described in section 2.14 (Chapter 2).

### 6.3 RESULTS

# 6.3.1 Effects of the p38 MAPK inhibition on H9c2 cell differentiation into cardiomyocytes

As described in previous Chapters, morphological changes and marker expression were observed or determined during the course of the differentiation experiments. Morphologically, treatment of H9c2 cells with SB203580 (10µM) caused little or no inhibition of their differentiation into cardiomyocytes. In both control cell and cells treated with SB203580, cell elongation was visible from day 4. By day 8 to day 12, cell fusion and myotube formation were extensive and to the same degree in both groups (Figure 6.11).

In contrast to the apparent lack of inhibition of the usual morphological changes that accompanied H9c2 cell differentiation into cardiomyocytes, SB203580 treated cells decreased the expression of MLC-1v by day 4 (p<0.05) (Figure 6.12). However, MLC-1v I expression in SB203580 treated cell had reached the same level seen in the control cells by day 8 (p>0.05). By day 12, MLC-1v expression in SB203580 treated cells fell back to nearly the levels seen in these cells at day 4 (p<0.05) (Figure 6.12).



Figure 6.11: Effects of p38 MAPK inhibition on H9c2 cell differentiation into Cardiomyocytes: Morphological changes. Control H9c2 cells were cultured in differentiation medium consisting of DMEM supplemented with 1% FBS and 100 units/ml penicillin and 100 µg/ml streptomycin. The differentiation medium of treated cells was further supplemented with 10µM of SB203580. Morphologically, SB203580 treatment of H9c2 cells shows very little signs of inhibition of their differentiation (i.e. cell alignment, elongation and myotube formation) into cardiomyocytes inhibited. Both control and SB203580 treated cells differentiated (shown by black arrows in day 4 to day12 control and SB20350-treated cells) to almost the same degree. By day 8 to day 12, cell fusion and myotube formation were extensive and to the same degree in both groups. Both control cells and SB20359-treated cells showed the normal single cell morphology (shown by white arrows) at day 0. were taken under 200X magnification and are The photographs representative of at least three independent differentiation experiments.





Figure 6.12: Effects of p38 MAPK inhibition on H9c2 cell differentiation into Cardiomyocytes: cardiac ventricular myosin light 1 expression. Western blot analysis of MLC-1v revealed that expression of this protein was initially delayed cells (by day 4, p<0.05) in comparison to the control expression (panel A). By day 8, MLC-1v expression in treated cell reached levels seen in control cells (p>0.05) but by day 12, MLC-1v level fell back to nearly the levels seen at day 4 (p<0.05). The blot is representative of at least three independent experiments. The densitometric data (mean  $\pm$  SD), from three independent differentiation experiments, show the time dependent changes in MLC-1v expression in SB203580 treated H9c2 cells (panel B).

#### 6.3.2 Viability of H9c2 cells in 10µM of the p38 MAPK inhibitor, SB203580

The widely reported concentration of 10µM for the p38 MPAK inhibitor SB203580 was selected for use in the experiments designed to investigate the effects of p38 MAPK inhibition on H9c2 differentiation into cardiomyocytes. The MTT cell viability test was carried out to determine if this concentration was toxic to the cells over the period that the cells were exposed to the drug. SB203580 at the concentration of 10µM was not toxic to the cells at any of the incubation time points. The average viability was about 89% of the viability determined in the control cells. There was a slight reduction in SB203580 treated cells at 12 and 24 hours (p < 0.05) but at 48 hours, the cell viability recovered to a value that was not significantly different from the control viability (p > 0.05) (Figure 6.12). At 48 hours the viability was about 91% of the viability determined in the control cells (Figure 6.13).



Figure 6.13: Viability of H9c2 Cells in the presence of the p38 MAPK inhibitor, SB203580. H9c2 cells were seeded in 6-well plates and treated with 10µM SB203580 as described in the methods section above (Section 6.3.3). The MTT assay was carried out at the end of the treatment. At 48 hours the viability of SB203580-treated cels was not significantly different from the control viability (p > 0.05). The data presented in the graph are the mean ± SD from three independent experiments.

### 6.3.3 Effects of p38 MAPK inhibition on P19 stem cell differentiation into cardiomyocytes

Given the lack of effects of the p38 MAPK inhibitor on H9c2 cell differentiation into cardiomyocytes, we next investigated how p38 MAPK inhibition may affect P19 cell differentiation into cardiomyocytes. In contrast to the lack of inhibition H9c2 cell differentiation into cardiomyocytes, SB203580 markedly inhibited P19 Stem cells differentiation into cardiomyocytes as determined by the expression of the cardiac restricted protein troponin (TnI). (Figure 6.16) Control P19 stem showed a small level of Tnl expression at the time of EB transfer from Petri dishes to cell culture plates. In control P19 cells, there was a marked increase on day 4 and this was followed further increased on day 8. The levels of TnI in SB203580 treated P19 stem cells were markedly reduced at every time point of the differentiation experiment as determined by western blot for TnI. The inhibited or low levels of TnI detected in SB203580 treated cells are in agreement with the lack of appearance of beating cardiomyocyte clusters in these cells. While beating cardiomyocytes were observed in the control cells as usual, no beating cardiomyocytes were seen in the SB203580 treated P19 stem cells. As was observed in the cases of PKC inhibition (Chapter 5) and PI3K inhibition (Chapter 7), it was also observed that P19 stem treated with SB203580 four days after the initiation of DMSO-induced differentiation of P19 stem cells failed to block the cell differentiation into beating cardiomyocytes (Figure 6.15). However, SB203580 blocked P19 stem cell differentiation into beating cardiomyocytes, when added to the cells during the EB formation stage (Figure 6.14).



Figure 6.14; A schematic diagram of the effects of p38 MAPK inhibition on P19 stem cell differentiation into cardiomyocytes P38 MAPK inhibition with SB203580 (10 $\mu$ M) blocked the differentiation of P19 stem cells into beating cardiomyocytes, when the inhibitors was added to the cells during the EB formation stage in the absence of SB203580 beating cardiomycytes were seen from day 6 to day 14 after the EB were transferred to cell culture dishes.



Figure 6.15; A schematic diagram of the effects of delayed inhibition of p38 MAPK on P19 stem cell differentiation into cardiomyocytes. In P19 stem treated with SB203580 ( $10\mu$ M) four days after the initiation of DMSO-induced cardiac differentiation, SB2093580 failed to block the cell differentiation into beating cardiomyocytes.


Figure 6.16: Effects of p38 MAPK inhibition on P19 stem cell differentiation into cardiomyocytes: troponin I expression. Control P19 stem cells were cultured in differentiation medium consisting of  $\alpha$ -MEM supplemented with 10% FBS, 100 units/ml penicillin and 100µg/ml streptomycin as described in section 6.3.2. The above differentiation medium was further supplemented with 10µM of SB203580 for the treated cells. SB203580 treatment markedly inhibited P19 Stem cells differentiation into cardiomyocytes as determined by the expression of cardiac restricted troponin I (TnI). While beating cardiomyocytes were observed in the some dishes of the control cells, beating cardiomyocytes were not seen in any of the SB203580 treated P19 stem cells. The blot is representative of three independent differentiation experiments.

#### 6.4 DISCUSSION

The MAPKs are among the signalling pathways that have been well studied. The MAPKs are involved in the control of gene expression (Wu *et al.*, 2004; Lawrence *et al.*, 2007; Mochizuki *et al.*, 2007), proliferation (Souza *et al.*, 2004; Erlich *et al.*, 2007; Jia *et al.*, 2008), migration (Goetze *et al.*, 1999; Rousseau *et al.*, 2006; Jia *et al.*, 2008) and cell survival or apoptosis (Chuang *et al.*, 2000; Pan *et al.*, 2002; Chen *et al.*, 2008; Cho *et al.*, 2008).

The activation of the p38 MAPK signalling has been implicated in the differentiation of various cell types (Gallea *et al.*, 2001; Tuli *et al.*, 2003; Chang *et al.*, 2007). including stem cell differentiation into cardiomyocytes (Monzen *et al.*, 1999; Davidson and Morange, 2000; Aouadi *et al.*, 2006). The precise role of these kinases and how they relate to the complex network of signaling events that lead to differentiation is still not clearly understood.

One of the key MAPKs, the p38 MAPK, is downstream of PI3K (Salh *et al.*, 2002). However, the p38 MAPK pathways can also be activated through PI3K-independent mechanisms. As our data have already implicated PI3K in stem cell differentiation into cardiomyocyte (see Chapter 7) we therefore extended our studies to determine whether the signalling events that control H9c2 and/or P19 stem differentiation cells into cardiomyocytes are linked to the activation of the p38 MAPK. A specific inhibitor of the p38 MAPK signalling pathway, SB203580 (Cuenda *et al.*, 1995; Young *et al.*, 1997; Gum *et al.*, 1998) was used in this study. The 10µM of SB203580 used in this study has been reported in other studies (Nick *et al.*, 1999; Barancik *et al.*, 2001;

Kwiecinska *et al.*, 2005). Our data also showed that this concentration is not toxic to H9c2 cells. The effect of SB203580 on stem cell differentiation into cardiomyocytes reported here may therefore be due to the specific pharmacological inhibition of the p38 MAPK signalling pathway by this compound.

Our data suggest that p38 MAPK signalling may have a role in the differentiation of P19 stem cells and H9c2 cells into cardiomyocytes. SB203580 (10µM) completely blocked the differentiation of P19 stem cells into beating cardiomyocytes. SB203580 also markedly reduced the expression of cardiac restricted troponin I in P19 stem cells. Similarly, during the differentiation of H9c2 cells into cardiomyocytes, SB203580 delayed the expression of MLC-1v. The expression of MLC-1v subsequently recovered as time progressed during the 14 day incubation period. Perhaps of interest is the fact that SB203580 did not significantly alter the morphological changes (i.e. cell elongation, cell fusion and formation of myotubes) that accompany the initiation of H9c2 differentiation into cardiomyocytes. This may however be due to the fact that the morphological changes generally become more apparent at the later stages of differentiation by which time the cells have started to recover from the inhibition caused by SB203580. Whether the lack of a significant effect, by SB203580, on the morphological changes that accompanied H9c2 differentiation into cardiomyocytes, is due to p38 MAPK having a lesser role in these changes (than in the regulation of MLC-1v expression) was not investigated. This may need further investigation in future studies. As our data stand, it would appear that the role of the p38 MAPK in

stem cell differentiation is more prominent in P19 when compared to the H9c2 cells. A role for the p38 MAPKs in stem cell differentiation indicated in our studies would be consistent with reports in other systems (Gallea *et al.*, 2001; Tuli *et al.*, 2003; Chang *et al.*, 2007); (Monzen *et al.*, 1999; Davidson and Morange, 2000; Aouadi *et al.*, 2006).

Our data have further suggested that the p38 MAPK mediated mechanisms or events that control P19 stem cells differentiation into cardiomyocytes are switched on very early following DMSO treatment. Since SB203580 (10µM) was able to block P19 stem differentiation into beating cardiomyocytes only when added at the time of initiation of P19 cell differentiation (i.e. at the EB formation stage). Addition of SB203580 to P19 cells at the EB formation stage completely blocked cell differentiation into beating cardiomyocytes. However when exposed to cells four day after DMSO treatment, SB203580 had no effect on the ability of the cells to differentiate into beating cardiomyocytes. This strongly suggests that activation of the p38 MAPK pathway occurs very early in the differentiation process. The data also suggest that the activation of p38 MAPK pathway may be transient since late application of SB2003580 failed to inhibit the differentiation process. Whether p38 activation is indeed transient or whether it is sustained but become irreverent after the initiation of the differentiation process remains to be determined.

Bone morphogenetic proteins are well known extracellular signalling molecules that are involved in stem cell differentiation into cardiomyocytes (Zhang and Bradley, 1996; Schultheiss *et al.*, 1997; Schlange *et al.*, 2000).

The core signalling mechanism of BMP is through the activation of SMAD proteins (Wrana et al., 1994; Monzen et al., 2001). However BMP signalling can also be channelled through MAPK via SMAD-independent mechanisms (Gallea et al., 2001). Also GATA-4, a key cardiac transcription factor may be a direct downstream target of several kinases including the MAPK (Liang et al., 2001) as there are many potential phosphorylation sites within this transcription factor. It has been reported that GATA-4, is a nuclear mediator of RhoA signalling. The convergence of RhoA on GATA signalling was suggested by the finding that RhoA potentiates the activity of GATA-4 via a p38 MAPK-dependent pathway that phosphorylates GATA-4 activation domains. It was also found that GATA binding sites mediate RhoA activation of target cardiac promoters. Thus, it is possible that p38 MAPK may regulate stem cell differentiation through its activation by upstream signaling events associated with the Rho pathway and subsequent downstream regulation of the transcriptional activities of GATA-4 (Charron et al., 2001). Whether this is part of a sequence of events that also requires PKC and/or PI3K in our cell systems remain to be determined.

Besides showing that p38 MAPK is required for cardiomyocyte differentiation, the project has gone further to investigate if the p38 MAPK affects cardiac differentiation in conjunction with other leading pathways that were investigated in this project. This is explained in more detail in Chapter 7 where the 'cross-talk' studies are discussed.

#### 6.5 Summary

The differentiation of H9c2 cells into cardiomyocytes was blocked by the inhibition of p38 MAPK signalling using SB203580 (10µM). However SB203580 had little or no effect on the differentiation associated morphological changes in H9c2 differentiating into cardiomyocytes. In both control cell and SB203580-treated cells, morphological changes were visible. In contrast to the apparent lack of inhibition of morphological changes in SB203580-treated H9c2 cells, the expression of MLC-1v was decreased in these cells.

In contrast H9c2 cells, SB203580 markedly inhibited the expression of cardiac restricted protein troponin (TnI) and also blocked the cell differentiation into beating cardiomyocytes. The small expression of TnI seen in control P19 stem cells, at the time of EB transfer to cell culture plates, increased upto day 8. In comparison to control cells, TnI levels in SB203580-treated P19 stem cells were reduced. The inhibited or low levels of TnI detected in SB203580-treated cells were in agreement with the lack of appearance of beating cardiomyocytes. It was also observed that SB203580 blocked P19 stem cell differentiation into beating cardiomyocytes, when added to the cells during the EB formation stage. In contrast, SB203580 failed to block the differentiation of P19 stem cells into beating cardiomyocytes, when the cells were treated, with this compound, four days after the initiation of DMSO-induced cardiac differentiation.

In conclusion, our data therefore suggest that the activation of p38 MAPK signalling pathway may be an essential signalling event during the differentiation of stem cells into cardiomyocytes. The data also suggest that the p38 MAPK exerts its effects at the early stages of P19 stem cell differentiation into cardiomyocytes.

### CHAPTER 7.0

# ROLE OF PHOSPHOINOSITIDE 3-KINASE IN H9c2 AND P19 STEM CELL DIFFERENTIATION INTO CARDIOMYOCYTES

#### 7.1 INTRODUCTION

#### 7.1.1 Phosphoinositide 3-kinases

The phosphoinositide 3-kinase (PI3K) enzymes are present in all cell types and have been shown to be crucial regulators for many cell functions including cell cycle control (Rosenzweig *et al.*, 1997), proliferation (Erlich *et al.*, 2007; Nakao *et al.*, 2007; Ramos-Nino *et al.*, 2008; Rosseland *et al.*, 2008), survival (Li *et al.*, 2001; Hambardzumyan *et al.*, 2008; Lim *et al.*, 2008) and migration (Nakao *et al.*, 2007).

Class I PI3K (IA and IB) are the most extensively studied isoforms. Class 1A PI3Ks are heterodimers of regulatory (p85) and a catalytic (p110) subunits. Class II PI3Ks are monomeric proteins that lack adapter subunits. The preferential substrates of class II PI3Ks are phosphatidylinositol and phosphatidylinositol phosphate. Class III PI3Ks, like class I PI3Ks, are also heterodimeric enzymes consisting of regulatory and catalytic subunits.

The catalytic subunits of class I PI3K are activated by signals originating from upstream receptor tyrosine kinases (for Class IA PI3K) or G-protein-coupled receptors (for Class IB PI3K).

Phosphoinositide 3-kinase catalyses the transfer of the gamma-phosphate from ATP to the 3'-OH position of the inositol ring in the membrane bound phosphatidylinositol (4, 5)-bisphosphate (PIP<sub>2</sub>) to phosphatidylinositol (3, 4,

5)-trisphosphate (PIP<sub>3</sub>). The reverse reaction to convert PIP<sub>3</sub> back to PIP<sub>2</sub> is catalysed by PTEN (phosphatase and tensin homolog deleted on chromosome 10), a phosphatase (Maehama and Dixon, 1998).

Downstream events following the formation of PIP<sub>3</sub> lead to the activation of Akt or protein kinase B (PKB). Akt is a serine/threonine protein kinase (Jones *et al.*, 1991) and a critical component of the PI3K signalling. For example, PI3K transcriptional regulation of genes (e.g. induction of c-myc and Bcl-2) (Ahmed *et al.*, 1997) and stimulation of protein synthesis by controlling the translation initiation factor eIF-4E via phosphorylation of 4E-BP1 (Gingras *et al.*, 1998) are mediated via Akt. Akt-dependent regulation of growth factor and anti-apoptotic signals, by PI3K, have been shown in a variety of cell types including BAF/3 cells (Ahmed *et al.*, 1997), cerebellar neurons (Dudek *et al.*, 1997), H19-7 cells (Eves *et al.*, 1998) and human embryonic kidney 293-EBNA cells (Kauffmann-Zeh *et al.*, 1997).

The PIP<sub>3</sub> produced by the PI3K reaction activates the phosphatidylinositoldependent protein kinases (PDK-1 and PDK-2) by recruiting them to the plasma membrane. PIP<sub>3</sub> also binds to the pleckstrin homology (PH) domain of Akt and anchors it to the plasma membrane (Klippel *et al.*, 1997) where Akt is activated. The activation of Akt is regulated through two distinct phosphorylation events which have been reported to occur in a sequential manner. The PDK1-mediated phosphorylation of Akt at threonine 308 (Thr308) in the domain activation loop precedes the PDK2-mediated phosphorylation at serine 473 (Ser473) (Alessi *et al.*, 1996; Alessi *et al.*,

1997a; Alessi *et al.*, 1997b). The phosphorylation of Akt at Thr308 by PDK-1 is believed to render the Ser473 site at Akt C-terminal tail accessible to PDK-2. Akt is fully activated only if it is phosphorylated at both Thr308 and ser473 sites (Alessi *et al.*, 1997a). Other downstream signalling molecules with PH domains that are target by PIP3 include protein kinase C and MAPK signalling pathway.

In addition to Akt activation, PDK1 is also a converging point for a number of different pathways including the phosphorylation conventional PKCs (Dutil *et al.*, 1998), novel PKCs (Le Good *et al.*, 1998; Cenni *et al.*, 2002) and atypical PKCs (Chou *et al.*, 1998; Le Good *et al.*, 1998; Dong *et al.*, 1999), p70 S6 kinase (Pullen *et al.*, 1998; Flynn *et al.*, 2000; Kuemmerle, 2003) and glucocorticoid-regulated kinase (SGK) (Kobayashi and Cohen, 1999; Kobayashi *et al.*, 1999).

PI3K signalling is terminated by the degradation of PIP3 by at least two different phosphatases known as SHIP (Src homology 2-containing inositol 5-phosphatase) and PTEN (tensin homolog deleted on chromosome 10 protein (PTEN) (Koyasu, 2003). PTEN and SHIP remove the 3-phosphate of PIP3 to re-generate PI(4,5)P<sub>2</sub>. This means that PTEN activity blocks all the main downstream effects of PI3K. In contrast, SHIP removes the 5-phosphate from the inositol ring of PIP3 to generate PI(3,4)P<sub>2</sub>. This means that dephosphorylation of PIP3 by SHIP block some of the main downstream effects of PI3K.

#### 7.1.2 PI3K and stem cell differentiation into cardiomyocytes

The PI3K pathway is one of the well studied signalling pathways whose signalling mechanisms are well established. PI3K signalling has been implicated in the differentiation of stem cells into cardiomyocytes (Klinz et al., 1999; Sauer et al., 2000; Naito et al., 2003). Cardiogenesis or stem cell differentiation into cardiomyocytes is a very complex process. It is therefore very unlikely that this process will be controlled by a single signalling pathway. Interaction or cross-talk between pathways is more likely to be key mode of regulation of stem cells differentiation into cardiomyocytes. Our studies therefore take the approach of not only examining the role of PI3K but also investigated if there is any cross-talk between PI3K and PKC and also between PI3K and p38 MAPK during cardiac cardiomyocyte differentiation. Therefore the aim of this study is to investigate the role of the PI3K pathway and possible cross-talk between the PI3K and other well established pathways (i.e. PKC and p38 MAPK) that have been shown to have role to play in cardiac differentiation. To achieve the aims a specific PI3K inhibitor LY294002 [2-(4-Morpholinyl)-8-phenyl-4H-1-benzopyran-4-one] (IC<sub>50</sub>=1.4µM) was used for this study (Vlahos et al., 1994). LY294002 is cell-permeable, potent and specific PI3K inhibitor that acts on the enzyme ATP-binding site (Vlahos et al., 1994). LY294002 (50µM) completely inhibited PI3K but failed to inhibit other kinases including serine/ threonine kinases (e.g. PKC, PKA, MAP kinase and S6 kinase), protein tyrosine kinases (e.g. EGF receptor tyrosine kinase and c-src kinase), lipid kinases (e.g. PI 4-kinase and diacylglycerol kinase) and rabbit kidney ATPase (Vlahos et al., 1994).

#### 7.2 MATERIALS AND METHODS

#### 7.2.1 Culture and differentiation of H9c2 Cells

The routine culture of H9c2 cells was carried out as described in section 2.3.1 (Chapter 2). H9c2 cells were trypsinised and the appropriate cell density for the differentiation experiments was calculated as described in section 2.6 (Chapter 2). H9c2 cells  $(2x10^3 \text{ cells/cm}^2)$  were seeded in p60 dishes or 6-well plates in full culture medium (i.e. DMEM supplemented with 10% FBS, 100units/ml penicillin/100µg/ml streptomycin), placed in a tissue culture incubator at 37°C, 95% air and 5% CO<sub>2</sub> and allowed to become 60-70% confluent. The differentiation of H9c2 cells into cardiomyocytes was carried out in 1% serum as described in section 2.8.1 (Chapter 2). The role of PI3K signalling pathway in the differentiation of H9c2 cells into cardiomyocytes was investigated. The effects of the inhibition of the PI3K signalling pathway were investigated using 20µM of LY294002 as described in section 2.8.2 (Chapter 2). A LY294002 stock (10mM solution) was prepared by dissolving 1mg of LY294002 in 325.3µl of DMSO. The stock solution was distributed into 10 or 20µl aliquots and stored in the freezer at -20°C. The aliquots were protected from light during storage and use. The differentiation medium was used to dilute (1:500) the frozen stock LY294002 solutions (10mM) to the LY294002 concentration (20µM) used in the differentiation experiments. The preparation of stock LY294002 concentration was carried out as explained in appendix VIII. More details about LY294002 are located in appendix VI (b).

#### 7.2.2 Cross-talk studies

To examine possible PI3K/PKC and PI3K/p38 MAPK cross-talk, cross-talk studies were designed and carried out. In order to carried out the cross-talk investigation, the following procedures were carried out as described in the Chapter 2 under the sections indicated in parenthesis: time course of Akt phosphorylation at serine 473 (2.10.1) and the effects of drugs (inhibitors) on PI3K activation as determined by Akt phosphorylation at serine 473 (Section 2.10.2).

#### 7.2.3 Culture and differentiation of P19 stem cells

The routine culture of P19 stem cells was carried out as described in section 2.3.2 (Chapter 2). To initiate differentiation, P19 stem cells  $(1\times10^{6} \text{ cells})$  were seeded in P100 Petri (i.e. microbiological) dishes in the differentiation medium (i.e.  $\alpha$ -MEM supplemented with 10% FBS, 100 units/ml penicillin/100 µg/ml streptomycin and 0.8% DMSO) and the differentiation of P19 stem cells into beating cardiomyocytes carried out as described in section 2.9.1 (Chapter 2). The role of PI3K signalling pathway in the DMSO-induced differentiation of P19 stem cells into beating cardiomyocytes was investigated. The effect of the inhibition of PI3K signalling pathway by 20µM of LY294002 on the DMSO-induced P19 stem cell differentiation into beating cardiomyocytes was investigated as described in section 2.9.2 (Chapter 2). Furthermore the effects of delayed inhibition of PI3K on P19 stem cell differentiation into cardiomyocytes were also investigated as described in section 2.9.3 (Chapter 2).

#### 7.2.4 Cell viability assay

To examine the effects of the PI3K inhibition on H9c2 cell viability, the MTT assay was carried out as described in section 2.7 (Chapter 2).

#### 7.2.5 Western blot analysis

In order to perform the western blot analysis, the following procedures were carried out as described in Chapter 2 under the sections indicated in parenthesis: cell lysate generation (section 2.11), preparation of BSA standards (section 2.12.1) and BCA assay (section 2.12.2). The western blotting was carried out as described in section 2.13.

#### 7.2.6 Data analysis

Data obtained from independent experiments were used for statistical analysis as described in section 2.14 (Chapter 2).

#### 7.3 RESULTS

#### 7.3.1 Effects of the PI3K Inhibition on H9c2 Cell Differentiation into Cardiomyocytes

The differentiation of H9c2 cells in culture medium supplemented with the PI3K inhibitor, LY294002, was completely blocked throughout the differentiation period. Control H9c2 cells differentiated into cardiomyocytes as determined by the usual morphological changes of cell elongation, fusion and myotube formation. In contrast LY294002 treated H9c2 cells maintained their normal single cell and spindle shaped morphology (Figure 7.10). By Day 4, the usual morphological changes that accompany H9c2 cell differentiation were visible in the control cells. In contrast, these changes only began at day 8 in LY294002 treated H9c2 cells and were never as extensive as those seen in the control cells by day 12.

In agreement with the lack of morphological changes observed in LY294002 treated H9c2 cells; the expression of MLC-1v protein was blocked at day 4 (p<0.01), day 8 (p< 0.05) and at day 12 (p>0.05) (Figure 7.11). The MLC-1v expression never reached the same levels as in control cells. MLC-1v levels in both control and LY294002 treated H9c2 cells were in agreement with the degree of differentiation or morphological changes observed in these cells. MLC-1v expression in LY294002 treated H9c2 cells was less than 25% of control by day 4 and the by day 8 the expression was still less than 50 % of the levels seen in the control cells.



**Figure 7.10: Effects of the PI3K inhibition on H9c2 cell differentiation into cardiomyocytes: morphological changes.** Control H9c2 cells were cultured in differentiation medium consisting of DMEM supplemented with 1% FBS and 100units/ml penicillin/100µg/ml streptomycin. The above differentiation medium was further supplemented with 20µM of LY294002 for the treated cells. Morphologically, LY294002 treatment significantly inhibited cell alignment, elongation and myotube formation. LY294002 treated H9c2 cells largely maintained their normal single cell and spindle shaped morphology which is in contrast to control H9c2 cells which showed the usual morphological changes. The photographs were taken under 200X magnification and are representative of at least 3 differentiation experiments.





Figure 7.11: Effects of the PI3K inhibition on H9c2 cell differentiation into cardiomyocytes: cardiac ventricular myosin light 1(MLC-1v)expression. In agreement with the lack of morphological changes observed in LY294002 treated H9c2 cells, western blot analysis of MLC-1v revealed that LY294002 treatment significantly inhibited this protein expression up to day 8, in comparison to the control expression (A). Marginal expression was seen after day 8. The blot is representative of at least 3 differentiation experiments (panel A). The densitometric data (mean  $\pm$  SD), from three independent differentiation experiments, show the time dependent changes in MLC-1v expression in LY294002 treated H9c2 cells (panel B)

#### 7.3.2 Effects of LY294002 (20µM) on the viability of H9c2 cells

The widely reported concentration of 20 $\mu$ M for the PI3K inhibitor LY294002 was selected for use in the experiments designed to investigate the effects of PI3K inhibition on H9c2 differentiation into cardiomyocytes. The cell viability test (MTT assay) was carried out to determine if this concentration was toxic to the cells over the period that the cells were left in this drug. LY294002 at the concentration of 20 $\mu$ M was not toxic to the cells. There was no statistical difference between the viability of control cells and the cells cultured in LY294002 (p > 0.05) (Figure 7.12). The average viability was about 85% of the viability determined in the control cells. By 48 hours the viability was about 77% of the viability determined in the control cells.



Figure 7.12: The viability of H9c2 cells in the presence of the PI3K inhibitor, LY294002. Control H9c2 cells were seeded in 6-well plates in differentiation medium consisting of DMEM supplemented with 1% FBS and 100units/ml penicillin/100µg/ml streptomycin. The above differentiation medium was further supplemented with  $20\mu$ M of LY294002 for the treated cells as described in the method section above (Section 7.2.4). The MTT assay was carried out at the end of the treatment. Cell viability of LY29402 treated cells at each time point was calculated as a percentage of the cell viability in the control cells at that time point. There was no statistical difference between the viability of control H9c2 cells and LY294002 treated H9c2 cells (p > 0.05). The data presented in the graph are the mean ± SD from three independent experiments.

# 7.3.3 Effects of PI3K inhibition on P19 stem cell differentiation into beating cardiomyocytes

To investigate the effects of PI3K inhibition on the differentiation of P19 stem cells into cardiomyocytes, LY294002 was added to cell culture of P19 stem cells during DMSO induced differentiation. Control P19 stem cells were treated with DMSO only while the differentiation medium of the LY294002treated P19 cells was supplemented with 20µM of LY294002 at the EB formation stage. Control and LY2940002 treated EBs followed their normal pattern of growth as already described in the establishment of the P19 cardiomyocyte differentiation model (Chapter 3). The EBs seen on day 1 grew in size and number over time. Transfer of EBs from Petri dishes to standard tissue culture dishes resulted in cell adhesion and the subsequent growth pattern into a monolayer. This was then followed by cell migration and proliferation. By day four, both treated and control cells were confluent. By day 7, the first beating clusters of cardiomyocytes appeared in the DMSO treated but not in the cells exposed to LY294002 (Figure 7.14). No beating cardiomyocytes were observed in any of the dishes containing LY294002 treated P19 cells over the whole course of the experiment (Figure 7.13 & When present, the number of beating clusters of Figure 7.14). cardiomyocytes usually continues to increase up to day 14. The patterns of cardiomyocyte beating clusters (including number and beating rates) from P19 stem cells treated with DMSO in the absence of LY294002 (i.e. no inhibition of PI3K signalling pathway) were as already described in Chapter 3 (see Figure 7.14).

In agreement with the lack of appearance of beating cardiomyocytes LY294002 treated P19 stem cell, the expression of cardiac restricted protein troponin I was inhibited (Figure 7.15). Troponin levels in the LY294002 treated P19 stem cells were lower than those in the control cells at all time points between day 4 and day 12.



Figure 7.13: A schematic diagram of the effects of the PI3K inhibition on DMSO-induced differentiation of P19 stem cells into cardiomyocytes. PI3K inhibition with LY294002 ( $20\mu$ M) blocked the differentiation of P19 stem cells into beating cardiomyocytes, when the inhibitors was added to the cells during the EB formation stage in the absence of LY294002 beating cardiomycytes were seen from day 6 to day 14 after the EB were transferred to cell culture dishes.



Figure 7.14: Effects of the PI3K inhibition on DMSO-induced differentiation of P19 stem cells into beating cardiomyocytes. Control P19 stem cell embryoid bodies were formed in differentiation medium consisting of  $\alpha$ -MEM supplemented with 10% FBS, 100 units/ml penicillin/100 µg/ml streptomycin and 0.8% DMSO. The above differentiation medium was further supplemented with 20µM of LY294002 for the treated P19 stem cells during the EB formation stage. DMSO caused the differentiation of P19 stem cells into beating cardiomyocytes. This DMSO-induced cardiac differentiation of the P19 stem cells was inhibited by LY294002. The areas of beating cardiomyocytes are shown within the closed figures drawn with white dotted lines. The beating focus of each beating area is shown by a black arrow. Photographs and movie clips of beating cells were taken under 200X magnification on the days indicated and are representative of microscope fields (for their respective days) from at least 3 differentiation experiments.



Figure 7.15: Effects of the PI3K inhibition on P19 stem cell differentiation into cardiomyocytes: cardiac troponin I expression. Control P19 stem cells were cultured in differentiation medium consisting of  $\alpha$ -MEM supplemented with 10% FBS, 100 units/ml penicillin/100µg/ml streptomycin. The above differentiation medium was further supplemented with 20µM of LY294002 for the treated cells. In agreement with the lack of appearance of beating cardiomyocytes in LY294002-treated P19 stem cell, the expression of the cardiac restricted protein, troponin I was inhibited. Troponin levels in the LY294002-treated P19 stem cells were lower than those in the control cells at all time points between day 4 and day 12. The blot is representative of three differentiation experiments.

## 7.3.4 Effects of delayed inhibition of PI3K on P19 stem cell differentiation into beating Cardiomyocytes

To determine the stage at which PI3K may regulate the differentiation of P19 stem cells into cardiomyocytes, LY29402 was added to cell culture four days after exposure to DMSO. Control P19 stem cells were treated with DMSO and LY294002 at the EB formation stage and for comparison, parallel dishes of P19 cells (delay-inhibition P19 cells) were treated with DMSO only at EB formation stage and then only treated with LY294002 after the EB were transferred to culture dishes (see section 2.9.3). Four days later, after the transfer of EBs to cell normal culture dishes, the EBs in both groups of cells underwent normal adhesion, proliferation and migration were confluent by day 4. No beating cardiomyocytes were observed in both control P19 cell and *delay-inhibition P19 cells* before day 6. By day 6 or 7, the first beating cells appeared in the *delay-inhibition P19 cells* (Figure 7.17). As expected, DMSO failed to induce P19 stem cell differentiation into beating cardiomyocytes, when LY294002 was added at the EB formation stage. However when DMSO was allowed to initiate P19 stem cell differentiation before 20µM of LY294002 was added to the cells, LY294002 failed to block the DMSO-induced differentiation of P19 stem cells into beating cardiomyocytes (Figure 7.16 & Figure 7.17). The patterns of cardiomyocyte beating clusters from P19 cells, in which DMSO was allowed to initiate their differentiation before the inhibition of the PI3K signalling pathway, were as already described in Chapter 3 (see Figure 7.17).



Figure 16: A schematic diagram of the effects of delayed inhibition of PI3K on P19 stem cell differentiation into beating Cardiomyocytes. In P19 stem treated with LY294002 ( $20\mu$ M) four days after the initiation of DMSO-induced cardiac differentiation, LY294002 failed to block the cell differentiation into beating cardiomyocytes



Figure 7.17: Effects of delayed inhibition of PI3K on P19 stem cell differentiation into beating cardiomyocytes. Control EBs were formed in differentiation medium (i.e.  $\alpha$ -MEM supplemented with 10% FBS, 100 units/ml penicillin/100 µg/ml streptomycin, 0.8% DMSO and 20µM of LY294002). For the delay inhibition of PI3K, EBs were initially formed in the above differentiation medium but in the absence of LY294002. DMSO failed to induce P19 stem cell differentiation into beating cardiomyocytes when differentiation was initiated in the presence of LY294002. However when DMSO was allowed to initiate P19 stem cell differentiation before the addition of LY294002, the cell differentiated into beating cardiomyocytes. The areas of beating cardiomyocytes are shown within the closed figures drawn with white dotted lines. The beating focus of each beating area is shown by a black arrow. Photographs and movie clips of beating cells were taken under 200X magnification on the days indicated and are representative of microscope fields (for their respective days) from at least 3 differentiation experiments.

### 7.4 CROSS-TALK STUDIES

# 7.4.1 Time course of Akt phosphorylation following the initiation of H9c2 cell differentiation into cardiomyocytes by 1% serum

Undifferentiating H9c2 cells maintained a baseline level of phosphorylated Akt The initiation of H9c2 cell differentiation into cardiomyocytes induced a more than 2-fold rapid increase in Akt phosphorylation at residue serine 473 within 5 minutes. The increased phosphorylated Akt levels were sustained for 6 hours before declining. The levels of phosphorylated Akt gradually fell back to baseline levels after 24 hours and by 48-72 hours, phosphorylated Akt level fell below the baseline values (Figure 7.18)



Figure 7.18: Time course of Akt phosphorylation following the initiation of H9c2 cell differentiation into cardiomyocytes by 1% serum. Control H9c2 cells were cultured in serum-free medium consisting of DMEM supplemented with 100units/ml penicillin/100µg/ml streptomycin in the absence of FBS. Serum-treated (1% FBS) control H9c2 cells were cultured in above supplemented with 1% FBS. A more than 2-fold increase in phosphorylated was induced by 1% FBS treatment and this was sustained for 6 hours before declining. Fold increase was calculated as the ratio of induced phosphorylated Akt level to baseline phosphorylated Akt. The blot is representative of at least 3 individual experiments (panel A). The densitometric data (mean  $\pm$  SD), from three independent experiments, show the time dependent fold change in phosphorylated Akt in treated H9c2 cells (panel B).

#### 7.4.2 Effects of PI3K inhibition on Akt phosphorylation at serine 473 in H9c2 cells differentiating into cardiomyocytes

To examine the effects of inhibiting PI3K signalling on the activation of this pathway, H9c2 cells <u>(LY294002-treated H9c2 cells)</u> were treated with 20 $\mu$ M of LY294002. For parallel comparison, cells exposed only to serum-free medium in the absence of LY294002 <u>(control H9c2 cells)</u> and another batch of cell exposed only to 1% FBS medium in the absence of LY294002 (<u>1 % FBS-treated H9c2 cells</u>) were included (see method section 2.10.2).

The initiation of H9c2 cell differentiation into cardiomyocytes, by 1% serum (i.e. 1% FBS), induced a marginal increase in Akt phosphorylation at serine 473 (p>0.05) (Figure 7.20). This differentiation associated increase in Akt phosphorylation (in the 1 % FBS treated cells) was completely abolished by 20 $\mu$ M LY294002 (p<0.01) (PI3K inhibition) (Figure 7.19). There was a significant fold reduction in Akt phosphorylation at serine 473, between control cells and 20 $\mu$ M of LY294002-treated cells (p<0.05) (Figure 7.19).



60 kDa



Figure 7.19: Effects of PI3K inhibitor LY29402 on Akt-ser473 phosphorylation in H9c2 cells differentiating into cardiomyocytes. Treatment of cells with 1% FBS induced a rapid increase in phosphorylated Akt levels. This increased Akt phosphorylation was completely abolished by 20µM of LY294002. There was no significant fold increase, in Akt phosphorylation at serine 473, between control cells and 1% FBS-treated cells (p>0.05). There was a significant fold reduction, in Akt phosphorylation at serine 473, between 1% FBS-treated cells and 20µM of LY294002-treated cells (p<0.01). There was also a significant fold reduction, in Akt phosphorylation at serine 473, between control cells and 20µM of LY294002treated cells (p<0.01). Fold increase was calculated as the ratio of induced phosphorylated Akt level to baseline phosphorylated Akt. The comparisons between control vs 1% FBS (p>0.05), 1% FBS vs 20µM LY294002 (p<0.01) and control vs 20µM LY294002 (p<0.05) are indicated by lines connecting the respective pairs (i.e. treatments). The blot is representative of three independent experiments (panel A). The densitometric data (mean ± SD), from the three independent experiments, show the time dependent fold change in phosphorylated Akt in respective treatment shown on the graph (panel B).

# 7.4.3 Effects of protein kinase C inhibition on Akt phosphorylation at serine 473 in H9c2 cell differentiating into cardiomyocytes

To examine the effects of inhibiting PKC signalling on the activation of the PI3K signalling pathway, H9c2 cells <u>(*BIM-I-treated H9c2 cells*)</u> were treated with 10µM of BIM-I. For parallel comparison, cells exposed only to serum-free medium in the absence of BIM-I <u>(control H9c2 cells)</u> and another batch of cell exposed only to 1% FBS medium in the absence of BIM-I (<u>1 % FBS-treated</u> <u>H9c2 cells</u>) were included (see method section 2.10.2).

The initiation of H9c2 cell differentiation into cardiomyocytes, by 1% serum (i.e. 1% FBS), induced a marginal increase in Akt phosphorylation at serine 473 (p>0.05) (Figure 7.20). This differentiation associated increase in Akt phosphorylation (in the 1 % FBS treated cells) was significantly enhanced by the 10 $\mu$ M BIM-I (i.e. PKC inhibition) (p<0.01) There was a significant fold increase, in Akt phosphorylation at serine 473, between control cells and 10 $\mu$ M of BIM-I-treated cells (p<0.01) (Figure 7.20).





Figure 7.20: Effects of PKC inhibition on Akt phosphorylation at serine 473 residue in H9c2 cells differentiating into cardiomyocytes: PI3K and **PKC cross-talk.** Treatment of cells with 1% FBS induced a rapid increase in phosphorylated Akt levels. This increased Akt phosphorylation was further enhanced by 10µM of BIM-I. There was no significant fold increase, in Akt phosphorylation at serine 473, between control cells and 1% FBS-treated cells (p>0.05). There was a significant fold increase, in Akt phosphorylation at serine 473, between 1% FBS-treated cells and 10µM of BIM-I-treated cells (p<0.01). There was a significant fold increase, in Akt phosphorylation at serine 473, between control cells and 10µM of BIM-I-treated cells (p<0.01). Fold increase was calculated as the ratio of induced phosphorylated Akt level to baseline phosphorylated Akt. The comparisons between control vs 1% FBS (p>0.05), 1% FBS vs 10µM BIM-I (p<0.01) and control vs 10µM BIM-I (p<0.01) are indicated by lines connecting the respective pairs (i.e. treatments). The blot is representative of three independent experiments (panel A). The densitometric data (mean  $\pm$  SD), from the three independent experiments, show the time dependent fold change in phosphorylated Akt in respective treatment shown on the graph (panel B).

## 7.4.4 Effects of p38 MAPK inhibition on Akt phosphorylation at serine 473 in H9c2 cell differentiating into cardiomyocytes

To examine the effects of inhibiting PKC signalling on the activation of the PI3K signalling pathway, H9c2 cells <u>(SB203580-treated H9c2 cells)</u> were treated with 10 $\mu$ M of SB203580. For parallel comparison, cells exposed only to serum-free medium in the absence of SB20580 <u>(control H9c2 cells)</u> and another batch of cell exposed only to 1% FBS medium in the absence of SB203580 (<u>1 % FBS-treated H9c2 cells</u>) were included (see method section 2.10.2).

Initiation of H9c2 cell differentiation into cardiomyocytes, by 1% serum, led to a marginal increase in Akt phosphorylation at serine 473 (p>0.05) (Figure 7.21).This differentiation associated increase in Akt phosphorylation (in the 1 % FBS treated cells) was significantly reduced by the 10 $\mu$ M SB203580 (i.e. p38 MAPK inhibition) (p<0.01). There was also a significant fold reduction, in Akt phosphorylation at serine 473, between control cells and 10 $\mu$ M SB203580-treated cells (p<0.01) (Figure 7.21).



Figure 7.21: Effects of protein p38 MAPK inhibition on Akt phosphorylation at serine 473 residue in H9c2 cells differentiating into cardiomyocytes: PI3K and p38 MAPK cross-talk. Treatment of cells with 1% FBS induced a rapid increase in phosphorylated Akt levels. This increased Akt phosphorylation was significantly reduced by 10µM SB203580. There was no significant fold increase, in Akt phosphorylation at serine 473, between control cells and 1% FBS-treated cells (p>0.05). There was a significant fold reduction, in Akt phosphorylation at serine 473, between 1% FBS-treated cells and 10µM SB203580-treated cells (p<0.01). There was also a significant fold reduction, in Akt phosphorylation at serine 473, between control cells and 10µM SB203580-treated cells (p<0.01). Fold increase was calculated as the ratio of induced phosphorylated Akt level to baseline phosphorylated Akt. The comparisons between control vs 1% FBS (p>0.05), 1% FBS vs 10µM SB203580-treated cells (p<0.01) and control vs 10µM SB203580-treated cells (p<0.01) are indicated by lines connecting the respective pairs (i.e. treatments). The blot is representative of three independent experiments (panel A). The densitometric data (mean  $\pm$  SD), from the three independent experiments, show the time dependent fold change in phosphorylated Akt in respective treatment shown on the graph (panel B).

#### 7.5 DISCUSSION

Published data suggest important roles for PI3K signalling during the differentiation of ES cells into cardiomyocytes (Klinz *et al.*, 1999; Sauer *et al.*, 2000). Treatment of differentiating ES cell cultures with the PI3K inhibitor LY294002 was shown to reduce the number of  $\alpha$ -actinin positive cardiomyocytes and the number of embryoid bodies containing beating foci (Klinz *et al.*, 1999).

Our data showed that the activation of the PI3K signalling pathway may be essential for differentiation of H9c2 cells into cardiomyocytes. Cell elongation and fusion forming myotubes, the main morphological changes or features associated with H9c2 cell differentiation into cardiomyocytes, were significantly inhibited by LY294002. The expression of cardiac specific ventricular myosin light chain 1 was also completely blocked in these cells. These results suggest that activation of the PI3K pathway may be essential for differentiation of H9c2 cells into cardiomyocytes.

Consistent with the H9c2 cell data, the inhibition of PI3K signalling also completely blocked the differentiation of P19 stem cells into beating cardiomyocytes and also inhibited the expression of cardiac restricted troponin I in these cells. Moreover, our data suggest that not only is activation of PI3K signalling critical for P19 stem cell differentiation into cardiomyocytes but that the early timing of this pathway may be critical. The inhibition of PI3K signalling, by LY294002 (20µM), four days after the initiation of DMSO induced differentiation failed to blocked P19 stem cell differentiation into
beating cardiomyocytes. These results suggest that activation of the PI3K pathway may be essential for differentiation of P19 stem cells into cardiomyocytes. The failure of LY294002 to block the differentiation of P19 stem cells into cardiomyocyte when added four days after DMSO also suggests that that PI3K signalling pathway is activated during the very early stage of P19 stem cell differentiation into cardiomyocytes. The data also suggest that the activation of PI3K pathway may be transient since late inhibition of PI3K by LY294002 failed to inhibit the differentiation process. Whether PI3K activation is indeed transient or whether it is sustained but become irreverent after the initiation of the differentiation process remains to be determined. Our data are in agreement with similar observations in which late LY294002 treatment failed to suppress the expression of Nkx-2.5 and GATA-4 in P19CL6 cells (a clonal derivative of P19 stem cells) (Naito *et al.*, 2003).

The 20µM concentration of LY294002 used in the present project was selected from published literature and appears to be a commonly used concentration for blocking PI3K activity (Qi *et al.*, 1999; Tamir and Bengal, 2000; Du *et al.*, 2001; Naito *et al.*, 2003; Fukuda *et al.*, 2005; Shaik *et al.*, 2008; Smyth *et al.*, 2008). LY294002 is not only potent but is also a specific inhibitor for blocking PI3K signalling (Vlahos *et al.*, 1994). LY294002, even at a higher concentration (50µM), specifically inhibited PI3K (Vlahos *et al.*, 1994). LY294002 (50µM) completely inhibited PI3K but failed to inhibit the activities of a wide range of other kinases including, PKC, PKA, MAP kinase, S6 kinase, EGF receptor tyrosine kinase, c-src kinase, PI 4-kinase and

diacylglycerol kinase) and ATPase (Vlahos *et al.*, 1994). Furthermore our cytotoxicity studies using MTT showed that LY294002, at 20uM concentration, had no cytotoxic effects. Thus, the inhibition of H9c2 and P19 stem cell differentiation into cardiomyocytes by LY294002 (20µM), as shown by our data, is due to LY294002 inhibition of the PI3K signalling pathway. Our data therefore suggest that the activation of the PI3K signalling is required for stem cell differentiation into cardiomyocytes. The data also suggest an early requirement for the activation of the PI3K signalling during the early stages of stem cell differentiation into cardiomyocytes.

Our data have also shed some light on whether activation of Akt by phosphorylation is required to drive cardiomyocyte differentiation. Akt is activated by sequential phosphorylation, first at threonine 308 by PDK1, and then, finally at serine 473 by PDK2 (Alessi et al., 1996; Alessi et al., 1997a; Alessi et al., 1997b). Our data clearly showed that PI3K signalling is required for stem cell differentiation into cardiomyocytes. However the data obtained on Akt are less convincing. Indeed exposure of cells to LY294002 did result in inhibition of Akt phosphorylation at serine 473 which correlates well with the ability of the drug to modulate differentiation. However, our cross-talk studies using BIM-I, a specific and selective PKC inhibitor, to block the activation of the PKC signalling pathway, revealed an unexpected increase in Akt phosphorylation at serine 473, even though BIM-I blocked cardiac differentiation of both H9c2 cells and P19 stem cells. This would strongly suggest that phosphorylation of Akt in differentiating stem cells may have no direct relevance to the process itself.

The enhanced Akt activation (phosphorylation at serine 473) by PKC inhibition cannot be explained by non-specificity of the concentration ( $10\mu$ M) of BIM-I used in these experiments. It has been reported, in haematopoietic progenitor cells, that PKC inhibition, using another PKC inhibitor, Ro-31-8220 (bisindolylmaleimide IX) ( $5\mu$ M), enhanced Akt activation (increased phosphorylation at serine 473 and threonine 308), in response to c-Kit stimulation (Edling *et al.*, 2007). It has also been shown that Ro 31-8220 ( $3\mu$ M), bisindolylmaleimide VIII ( $3\mu$ M), and 30nM of LY 379196 (another bisindolylmaleimide) increase Akt phosphorylation at Serine 473 in A549 cells and HEK293 cells (Wen *et al.*, 2003). Increased Akt phosphorylation in HEK293 cells over-expressing Akt was diminished by phorbol 12-myristate-13-acetate (PMA) a potent PKC activator (Wen *et al.*, 2003). Our data therefore suggest that PI3K may be activating alternative downstream targets independent of Akt (Figure 7.22).

The question of whether Akt is required for stem cell differentiation into cardiomyocytes can be investigated by knocking out Akt, in stem cells. This could be done using pharmacological Akt specific inhibitors. Alternatively, Akt can be knocked out using molecular methods. For example, Akt siRNA can be used to block Akt mRNA translation into Akt protein. An Akt dominant negative stem cells, through the mutation of the critical amino acid residues (i.e. serine 473 or both threonine 308 and serine 473) can also be created. The mutant stem cells (either from Akt siRNA or Akt mutation) can then be induced to differentiate into cardiomyocytes. If Akt is not required, for stem cell differentiation into cardiomyocytes, then the mutant stem cells will retain

their ability to differentiate into cardiomyocytes. However, if Akt is required, for stem cell differentiation into cardiomyocytes, then the mutant stem cells will have reduced ability to differentiate into cardiomyocytes.

The Wnt/β-catenin signalling pathway has been reported to be involved in the cardiac lineage specification or differentiation. Wnt ligands and receptors are expressed in a spatially and temporally manner and regulated during early heart development. (Ku and Melton, 1993; Monkley et al., 1996; Eisenberg et al., 1997; Garriock et al., 2005). Moreover, the canonical Wnt/β-catenin signalling positively regulates ES cell differentiation. In the mouse, the expression of the early cardiac genes Nkx2.5 and Tbx5 (mesoderm marker) in EBs can be up-regulated by Wnt3a and down-regulated by Dkk-1(Wnt antagonist). The addition of Wnt3a to ES cells during the early phase of EB formation increased the number of beating EBs by more than 50% while the addition of Fz8/Fc (secreted or soluble Wnt antagonist) or Dkk-1 and Fz8/Fc at this stage leads to a complete absence of beating EBs (Naito et al., 2006; Kwon et al., 2007). Constitutively active GSK-3β (i.e. degradation of βcatenin) also completely blocked the expression of Tbx5 and GATA4 (Nakamura et al., 2003; Kwon et al., 2007). In contrast, lithium chloride (a GSK-3<sup>β</sup> inhibitor) significantly increased the expression of Nkx2.5, GATA4, Tbx5 (Nakamura et al., 2003). The down-regulation of Nkx2.5, GATA4 and Tbx5 expression by Dkk-1, Fz8/Fc and constitutively active GSK-3β and the induction of these genes by lithium suggests that Wnt/ $\beta$ -catenin signalling play positive roles in driving ES cell differentiation into cardiomyocytes.

Previous studies have suggested that PI3K signalling could involve activation of the Wnt/β-catenin pathway and this may be independent of Akt. Lithium treatment mimics Wnt/β-catenin signalling activation. Lithium induces the stabilisation and nuclear accumulation of β-catenin by inhibiting the phosphorylation of GSK-3β (Klein and Melton, 1996; Hedgepeth et al., 1997; Rao et al., 2005). Our data showed that H9c2 cells in 10% serum do not differentiate into cardiomyocytes. But it has been shown that treatment of H9c2 cells with lithium can overcome the inhibitory effects of serum and drive the cell differentiation into cardiomyocytes (Kashour et al., 2003). LY294002 blocked lithium-induced differentiation of H9c2 cells (Kashour et al., 2003) thus suggesting a link between the PI3K and the Wnt/ $\beta$ -catenin pathway. In contrast, it was also found that over-expression of Akt in H9c2 did not increase stabilisation or nuclear accumulation of β-catenin or increased H9c2 cell differentiation. Lithium also induced differentiation in H9c2 expressing dominant-negative Akt (Kashour et al., 2003). These results suggest that PI3K signalling and Wnt/β-catenin signalling interacted in an Akt-independent manner or mechanism.

Our data suggest that PI3K regulates cardiomyocyte differentiation in an Aktindependent mechanism. Our data are in agreement with the previous data mentioned above. The new downstream target, independent of Akt, through which PI3K may cross-talk with Wnt/β-catenin signalling to regulate cardiomyocyte differentiation, is not known. Considering all of the above data, we are tempted to speculate that activation of PI3K in our cells models could drive differentiation mainly through activation of the Wnt/β-catenin signalling

pathway and not through phosphorylation of Akt. However further investigations (using Akt inhibitors, or Akt siRNA, or dominant negative Akt stem cells, along with studies to correlate the activation (or inhibition) of the Wnt/ $\beta$ -catenin signalling to stem cell cardiomyocyte differentiation) are required to confirm our data and determine the exact role of PI3K activation and Akt during the differentiation of stem cell into cardiomyocytes.

Further support of our suggestion that PI3K may regulate Wnt/ $\beta$ -catenin signalling through an Akt-independent mechanism comes from studies which showed that the activation of the Wnt/ $\beta$ -catenin signalling did not lead to GSK-3 $\beta$  phosphorylation (Ding *et al.*, 2000; McManus *et al.*, 2005). While Akt can inactivate GSK-3 $\beta$  by phosphorylation at serine 9 in response to the activation of other pathways, for example in response to insulin-like growth factor (IGF) and other factors (Desbois-Mouthon *et al.*, 2002), GSK-3 $\beta$  did not become phosphorylated at serine 9 in response to Wnt signalling (Ding *et al.*, 2000; McManus *et al.*, 2005).

As already discussed, GSK-3 $\beta$  inactivation is necessary for ES cell commitment to cardiac lineage differentiation. GSK-3 $\beta$  is constitutively active and downstream of Akt. Since Akt negatively regulates the activity of GSK-3 $\beta$ then activated Akt, following the activation of the PI3K signalling pathway, would be expected to inactivate GSK-3 $\beta$  and block its negative effects on cytoplasmic  $\beta$ -catenin levels thus leading to enhanced activation of the Wnt signalling and subsequent increase in ES cell differentiation into cardiomyocytes. However Akt-mediated inactivation of GSK-3 $\beta$  may not be a

relevant or a major event for the Wnt/ $\beta$ -catenin signalling pathway. In both wild type and knockin cells homozygous for both GSK-3 $\beta$  and GSK-3 $\alpha$ , Wnt3a was found to inactivate GSK-3, stabilised  $\beta$ -catenin and stimulated Wnt-dependent gene transcription but failed to induce a detectable increase in the phosphorylation of GSK-3 $\alpha$  (at seine 21) and GSK-3 $\beta$  (at serine 9) in the wild-type cells (McManus *et al.*, 2005). The above data show that Wnt-mediated inactivation of GSK-3 $\beta$  is not dependent on GSK-3 $\beta$  phosphorylation at serine 9. Thus the inactivation of GSK-3 $\beta$  by the Wnt signalling pathway may be achieved through a mechanism that is independent of Akt inactivation of GSK-3 $\beta$ . The above is in line with our suggestion the Akt phosphorylation may not be relevant for PI3K mediated regulatory effects on stem cell differentiation into cardiomyocytes.

PI3K inhibition in our experiments inhibited the differentiation of P19 stem cells in beating cardiomyocytes and the expression of troponin I. PI3K inhibition also inhibited the differentiation of H9c2 cells as determined by the lack of differentiation associated morphological changes and the inhibition of MLC-1v expression. While PI3K inhibition blocked the differentiation of both P19 stem cells and H9c2 cells into cardiomyocytes, our cross-talk data showed that PKC inhibition enhanced Akt activation (i.e. phosphorylation at serine 473). However as already shown (Chapter 5) PKC inhibition also blocked P19 stem cells and H9c2 cells into cardiomyocytes. Therefore our data suggest that PI3K positively regulates stem cell differentiation into cardiomyocytes not through Akt activation, but through a yet unknown signalling mechanism that is independent of Akt activation (Figure 7.22).

The data from the cross-talk studies suggested possible interactions between PI3K and PKC and also between PI3K and p38 MAPK signalling during cardiomyocyte differentiation. Our data suggest that activated PKC or p38 MAPK pathways may modulate the activation of the PI3K pathways as determined by Akt phosphorylation at serine 473. The inhibition of PKC pathway led to an elevation of Akt phosphorylation at serine 473, thus suggesting a possible negative role for PKC in the regulation of PI3K at the Akt level. In contrast the inhibition of p38 MAPK pathway led to a reduction in Akt phosphorylation at serine 473, thus suggesting a possible negative role for PKC in the regulation of PI3K at the PI3K phosphorylation at serine 473, thus suggesting a possible positive role for p38 MAPK in the regulation of PI3K also at the level of Akt. It not clear how PKC or p38 MAPK may regulate PI3K pathway activation by modulating Akt phosphorylation at serine 473.

The PI3K signalling controls many biological processes. It is therefore more likely that this pathway may be regulated by cross-talk with different signalling pathways. The fact that the PKC specific inhibitor BIM-I caused increased Akt phosphorylation suggests that PKC may play a negative role in the regulation of PI3K pathway. PKC is downstream of both PI3K and PDK1 (Chou *et al.*, 1998; Dutil *et al.*, 1998; Balendran *et al.*, 2000) therefore the effects of any cross-talk between PI3K and PKC on cardiac cardiomyocyte differentiation cannot be attributed to direct effect on either PI3K or PDK1.

Our data give ground for possible speculation about how BIM-I (i.e. PKC inhibition) might caused increase phosphorylation of Akt at serine 473. First,

PKC could inhibit Akt activation by targeting PDK2 (i.e. an elusive and currently unknown kinase that is thought to be responsible for Akt Akt phosphorylation at serine 473). If PKC does indeed control PDK2 as suggested by our data (i.e. increased Akt phosphorylation that accompanied the inhibition of PKC) then PKC may be a negative regulator of PDK2 since the inhibition of PKC by BIM-I appeared to "increase" the ability of "PDk2" to phosphorylate Akt-ser473.

The action of the lipid phosphatase, PTEN on PIP<sub>3</sub> limits the activation of Akt (Maehama and Dixon, 1998). However since PTEN does not directly dephosphorylate Akt, it is currently not known how Akt signalling is terminated once it is activated. While the dephosphorylation mechanism to directly inactivate Akt is yet to be found, evidence suggests that a phosphatase known as PHLPP (PH domain leucine-rich repeat protein phosphatase) selectively and specifically dephosphorylates Akt at serine 473 (Gao *et al.*, 2005). The second speculation therefore is that PKC may be an activator (i.e. positive regulator) of PHLPP "a putative phosphatase" that may be responsible for dephosphorylating AKt at serine 473 back to the baseline level since BIM-I would reduce the ability of PKC to activate that "putative phosphatase", thus limiting the ability of this "phosphatase to dephosphorylate Akt, in presence of BIM-I.

If indeed as suggested by our data, PKC does negatively regulate PI3K via PDK2 or Akt-related "phosphatase" then the significance of such negative

regulation of Akt by PKC is worth exploring in order to get a better understanding of the interaction between PI3K and PKC signalling events. However according to our data, PKC regulation of PI3K either via PDK2 or Akt-related "phosphatase" may be of no or little relevance to the differentiation of stem cells into cardiomyocytes since our data also suggested that Akt activation is of no or little relevance in this process.

Our data also suggest a possible cross-talk between PI3K and p38 MAPK. PDK1 has been cloned and sequenced (Alessi *et al.*, 1997a) and it is well established that PDK1 phosphorylate Akt at threonine 308 (Alessi et al., 1997a; Alessi et al., 1997b). However the identity of the kinase that is responsible for Akt phosphorylation at serine 473 remains unknown. Our data showed reduced phosphorylation of Akt at serine 473 in the presence of SB203580 (10µM). We speculate that p38 MAPK may have an additional activity that can phosphorylate Akt at serine 473 in a phosphoinositidedependent manner. Alternatively, p38 MAPK may be the unidentified PDK2. It is interesting to note that p38 MAPK is among a range of possible PDK2 candidates. Other candidate kinases include mitogen-activated protein (MAP) kinase-activated protein kinase-2 (MK2) (Alessi et al., 1996), integrin-linked kinase (ILK) (Delcommenne et al., 1998; Persad et al., 2000), DNAdependent protein kinase (DNK-PK), cPKC<sub>a</sub> (Partovian and Simons, 2004) and  $cPKC_{\beta}$  (Kawakami *et al.*, 2004). The question of whether p38 MAPK has a PI3K-dependent and phosphoinositide-dependent activity that can phosphorylate Akt at serine 473 or whether p38 MAPK and PDK2 are one and the same kinase remains. While Akt phosphorylation at serine 473 may not be

of major relevance to the differentiation of stem cells into cardiomyocytes since both PI3K and p38 MAPK are implicated in the differentiation of stem into cardiomyocytes, then the resolution of this question may help in advancing our understanding of the relationship between PI3K and p38 MAPK during cell differentiation and this needs further investigation.



Figure 7.22: PI3K and PKC regulate stem cell differentiation into cardiomyocytes through possible Akt-independent mechanisms. PI3K activation causes Akt activation (phosphorylation at both threonine 308 and serine 473) but PI3K may regulate stem cell differentiation into cardiomyocytes through a possible Akt-independent mechanism. PKC may modulate the phosphorylated states of Akt, since PKC inhibition enhanced Akt phosphorylation at serine 473. PKC inhibition also blocked the differentiation of H9c2 cells and P19 stem cells into cardiomyocytes. Therefore PKC may regulate H9c2 cells and P19 stem cell differentiation into cardiomyocytes through possible Akt-independent mechanism(s). PKC may negatively regulate the activity of 3-Phosphoinositide-Dependent Protein Kinase 2 (PDK2) since the inhibition of PKC by BIM-I appeared to "increase" the ability of "PDK2" to phosphorylate Akt-ser473. Alternatively, PKC maybe an activator (i.e. positive regulator) of "a putative phosphatase" (PP) that may be responsible for dephosphorylating AKt at back to the baseline level. The possible Akt-independent mechanisms or downstream target(s), through which PI3K and PKC may regulate H9c2 cell and P19 stem cell differentiation into cardiomyocytes is/are currently unknown and warrants further investigation.

As stated above, the phosphorylation of Akt in differentiating stem cells may have no direct relevance to the cell differentiation into cardiomyocytes. Therefore, in addition to any possible modulation of Akt activity (as suggested by the inhibition of Akt activation by SB203580), the data also suggest that p38 MAPK may cause cardiac differentiation independent of Akt (Figure 7.23). Like in the case of PKC, the data also point to a possible target, downstream of/ independent of Akt, through which p38 MAPK might cause the differentiation of H9c2 cell and stem cell into cardiomyocytes. This possible key downstream target is currently unknown and warrants further investigation.



*Figure 7.23: PI3K and p38 MAPK regulate stem cell differentiation into cardiomyocytes through possible Akt-independent mechanisms.* PI3K activation causes the Akt activation (phosphorylation at both threonine 308 and serine 473) but PI3K may regulate stem cell differentiation into cardiomyocytes through a possible Akt-independent mechanism. P38 MAPK may modulate the phosphorylated states of Akt, since p38 MAPK inhibition inhibited Akt phosphorylation at serine 473. however, the inhibition of p38 MAPK also blocked the differentiation of H9c2 cells and P19 stem cells into cardiomyocytes. Therefore p38 MAPK may regulate H9c2 cells and P19 stem cell differentiation into cardiomyocytes through a possible Akt-independent mechanism (s). The possible Akt-independent mechanisms or downstream target, through which PI3K and p38 MAPK may regulate H9c2 cell and P19 stem cell differentiation into cardiomyocytes is/are currently unknown and warrants further investigation.

#### 7.6 SUMMARY

The inhibition of PI3K using 20µM of LY294002 completely blocked the differentiation of H9c2 cells into cardiomyocytes (as as determined by the usual morphological changes of cell elongation, fusion and myotube formation. LY294002 treated H9c2 cells largely maintained their normal single cell and spindle shaped morphology which was in contrast to control H9c2 cells which showed the usual morphological changes. In agreement with the lack of morphological changes observed in LY294002 treated H9c2 cells, the expression of MLC-1v protein was significantly inhibited. Our data therefore suggest that activation of the PI3K signalling pathway is required for H9c2 cell differentiation into cardiomyocytes.

Treatment of H9c2 cells with 1% FBS induced a rapid increase in phosphorylation (at serine 473) but this differentiation associated increase in Akt phosphorylation was completely abolished by  $20\mu$ M of LY294002. This differentiation associated increase in Akt phosphorylation was significantly enhanced by BIM-I ( $10\mu$ M) but significantly reduced by SB203580 ( $10\mu$ M). Our data therefore showed that PI3K is activated during 1% FBS-induced differentiation of H9c2 cells into cardiomyocytes and also that PKC and p38 MAPK may modulate Akt phosphorylation at serine 473.

P19 stem cultured in the presence of 0.8% DMSO but in the absence of LY294002 (absence of PI3K inhibition) differentiated into beating cardiomyocytes. However PI3K inhibition, with LY294002 (20µM), completely blocked the DMSO-induced differentiation of P19 stem cells into beating

cardiomyocytes. The expression of cardiac specific troponin I was also reduced in P19 stem cells treated with LY294002.

When DMSO was allowed to initiate P19 stem cell differentiation, for 4 days before the addition of LY294002 to the cells, LY294002 completely failed to block the cell differentiation into beating cardiomyocytes. Our data therefore suggest that activation of the PI3K signalling pathway is required for P19 stem cell differentiation into cardiomyocytes. In addition the data suggest that PI3K signalling is required during the early stages of P19 stem cell differentiation into cardiomyocytes.

In conclusion, the activation of the PI3K signalling pathway is required for P19 stem cells and H9c2 cell differentiation into cardiomyocytes. The activation of PI3K signalling is required during the early stages of stem cell differentiation into cardiomyocytes. Our data also suggest that PI3K positively regulates stem cell differentiation into cardiomyocytes not through Akt activation, but through a yet unknown signalling mechanism that is independent of Akt activation.

Our data also suggest that PKC may also be a negative regulator of Akt phosphorylation at serine 473 either via an unknown PDK2 or via positive regulation of a possible "phosphatase" that is responsible for the dephosphorylation of Akt at serine 473. It is also a suggestion from our data that p38 MAPK may have a PI3K-dependent and phosphoinositide-dependent

activity with the ability to phosphorylate Akt at serine 473 in or p38 MAPK and PDK2 may be one and the same kinase.

# CHAPTER 8.0

# ROLE OF NF-kB IN H9C2 CELL DIFFERENTIATION

# INTO CARDIOMYOCYTES

#### 8.1 INTRODUCTION

NF-kB signalling pathway consists of a family of structurally related and evolutionarily conserved proteins known as NF-κB proteins. There are five mammalian NF-kB proteins: RelA (p65), RelB, c-Rel, NF-κB1 (p50/105) (i.e. p50 and its precursor p105), and NF-κB2 (p52/p100) (i.e. p52 and its precursor p100). These proteins can form homodimers and heterodimers. These dimers have different DNA site specificities and also different DNA binding affinities.

A common structural feature of NF-kB proteins is the presence of a highly conserved 300 amino acid domain at their N-terminals. This conserved sequence is known as the Rel homology domain (RHD). Rel homology domain is responsible for DNA binding, dimerisation, and association with inhibitory proteins (Coleman et al., 1993; Toledano et al., 1993). In unstimulated cells, NF-KB proteins exist in their transcriptionally inactive forms bound to proteins that inhibit their activation. These inhibitory proteins are known as inhibitors of NF- $\kappa$ B (I $\kappa$ B). Seven mammalian I $\kappa$ B proteins (I $\kappa$ B $\alpha$ , IκBβ, IκBγ, IκBε, Bcl-3, p105 and p100) has been identified. This non-covalent interaction between the NF-kB dimers and the NF-kB inhibitors is thought to conceal the nuclear localisation signal (NLS) of NF-KB proteins thus preventing their translocation into the nucleus. Therefore inactive NF-kB proteins are retained in the cytoplasm. The NF-kB signalling can be activated either by the classical (canonical) pathway or by the alternative (noncanonical or novel) pathway.

In the classical pathway, IκB is phosphorylated (Li *et al.*, 1994; Naumann and Scheidereit, 1994; Traenckner *et al.*, 1995; Zandi *et al.*, 1998) by IκB kinase (IKK). This phosphorylation leads to polyubiquitination of IkB. This phosphorylation mediated polyubiquitination triggers IkB degradation by the ubiquitin-proteasome pathway. IkB degradation releases NF-κB of IkB inhibition and thereby activates NF-κB. The activated NF-κB then translocates into the nucleus (Naumann and Scheidereit, 1994). In the nucleus, NF-kB activate the expression of a various target genes including genes coding for other transcription factors, cytokines, growth factors and regulators of apoptosis.

NF-kB activation by the non-canonical pathway involves the activation of NFkB inducing kinase (NIK) which in turn activates IKK (Tse *et al.*, 2007). The activated IKKa then phosphorylates p100 and this in turn undergoes proteasomal processing to generate p52. The generated p52 form a dimer with RelB (i.e. RelB/p52). The RelB/p52 complex then translocate to the nucleus to activate genes distinct from those regulated by the canonical pathway.

To ensure that the activation of NF-kB signalling is transient and self-limiting, a series of mechanisms including a negative feedback loop regulates NF-κB activation. The negative feedback loop involves the induction of IkB genes by the activated NF-kB. This leads to the synthesis of new IkBα proteins. The newly synthesised IkB then enter the nucleus, bind to activated NF-KB and return it to the cytoplasm in the inactive form

#### 8.1.1 NF-κB signalling and stem cell differentiation

Transcription factors are among the key molecules for gene regulation at the level of transcription. Transcription factors are proteins that bind specific DNA or promoter sequences in promoters leading to the alteration of the transcription of downstream target genes under the promoter control. We have an understanding of the roles of cardiac transcription factors (e.g. GATA-4, Nkx.5) in cardiogenesis and ES cell differentiation into cardiomyocytes. However there is very little information on the role of other or non-cardiac transcription factors which may be upstream and may be activated much earlier than the cardiac transcription factors.

If more information about transcription factors upstream of cardiac transcription factors become available then coupling that information with what we already known about cardiac transcription factors may tremendously increase our understanding of cardiogenesis and ES cell differentiation into cardiomyocytes. This may lead to better and more cardiac specific differentiation of ES into cardiomyocytes. This in turn will help in the realization of the therapeutic potential of stem cell-derived cardiomyocytes for the treatment of cardiovascular diseases.

We decided to investigate the role of NF-kB, which is downstream of PI3K. As already shown by our data, PI3K has role to play in the differentiation of stem cells into cardiomyocytes. The aim of this study is to investigate the role of NF-kB pathway in the cardiomyocyte differentiation. To examine the specific roles of NF-kB action during cardiomyocyte differentiation, we chose CAPE

(caffeic acid phenethyl ester), an inhibitor of NF-kB activation (Natarajan *et al.*, 1996), in this study.

CAPE inhibits the activation of the classical NF-kB pathway by suppressing the degradation of IkB- $\alpha$  (Natarajan *et al.*, 1996; Abdel-Latif *et al.*, 2005; Song *et al.*, 2008). CAPE (IC<sub>50</sub>= 2 $\mu$ M) is a potent NF-kB activation inhibitor (Reddy *et al.*, 2004).

## 8.2 MATERIALS AND METHODS

#### 8.2.1 Culture and differentiation of H9c2 Cells

The routine culture of H9c2 cells was carried out as described in section 2.3.1 (Chapter 2). H9c2 cells were trypsinised and the appropriate cell density for the differentiation experiments was calculated as described in section 2.6 (Chapter 2). H9c2 cells  $(2x10^3 \text{ cells/cm}^2)$  were seeded in p60 dishes or 6-well plates in full culture medium (i.e. DMEM supplemented with 10% FBS, 100units/ml penicillin/100µg/ml streptomycin), placed in a tissue culture incubator at 37°C, 95% air and 5% CO<sub>2</sub> and allowed to become 60-70% confluent. The differentiation of H9c2 cells into cardiomyocytes was carried out in 1% serum as described in section 2.8.1 (Chapter 2). The role of NF-kB signalling pathway in the differentiation of H9c2 cells into cardiomyocytes was investigated. The effects of the inhibition of the activation of NF-kB signalling pathway were investigated using 30µM of CAPE as described in section 2.8.2 (Chapter 2). A CAPE stock (100mM solution) was prepared by dissolving 5mg of CAPE in 176µl of DMSO. The stock solution was distributed into 20 or 50µl aliquots and stored in the freezer at -20°C. The aliquots were protected from light during storage and use. The differentiation medium was used to, first dilute the frozen stock CAPE solutions to 100µM (1:1000). The 100µM solution was then finally diluted (1:3.3) to the CAPE concentration (30µM) used in the differentiation experiments. The preparation of stock CAPE concentration was carried out as explained in appendix VIII. More details about CAPE are located in appendix VI (d).

#### 8.2.2 Cell viability assay

To examine the effects of the NF-kB inhibition on H9c2 cell viability, the MTT assay was carried out as described in section 2.7 (Chapter 3).

### 8.2.3 Western blot analysis

In order to perform the western blot analysis, the following procedures were carried out as described in the Chapter 3 under the sections indicated in parenthesis: cell lysate generation (section 2.11), preparation of BSA standards (section 2.12.1) and BCA assay (section 2.12.2). The western blotting was carried out as described in section 2.13.

### 8.2.4 Data analysis

Data obtained from independent experiments were used for statistical analysis as described in section 2.14 (Chapter 3).

## 8.3 RESULTS

# 8.3.1 Effects of inhibition of NF-kB activation on H9c2 cell differentiation into cardiomyocytes

Consistent with previous data described above, culture of H9c2 cells in media with 1% serum resulted in a time dependent differentiation of the cells in to cardiomyocytes. More importantly, treatment with CAPE delayed the differentiation process with myotube formation becoming evident at day 8 to 12 (Figure 8.10). This effect is more evident when MLC-1v expression was monitored. As shown in Figure 8.11, levels of MLC-1v were significantly reduced at day 4 when compared to controls and even though levels recovered at day 8 and 12, expression was still marginally lower than in controls.



Figure 8.10: Effects of inhibition of NF-kB activation on H9c2 cell differentiation into cardiomyocytes: morphological changes. Control H9c2 cells were cultured in differentiation medium consisting of DMEM supplemented with 1% FBS and 100units/ml penicillin/100µg/ml streptomycin. The differentiation medium of treated cells was further supplemented with 30µM of CAPE. Incubation of H9c2 cells with CAPE, an inhibitor of the translocation of NF-kB into the nucleus, significantly inhibited H9c2 cell differentiation (i.e. cell alignment, elongation and myotube formation) into cardiomyocytes. CAPE treated H9c2 cells largely maintained their normal single cell and spindle shaped morphology (shown by black arrows in day 4 to day 12 NF-kB-treated cells). In contrast, control H9c2 cells showed the usual morphological changes (shown by black arrows in day 4 to day 12 control cells). Both control cells and NF-kB-treated cells showed the normal single cell morphology (shown by white arrows) at day 0. The photographs were taken under 200X magnification and are representative of at least three independent differentiation experiments.



Figure 8.11: Effects of inhibition of NF-kB activation on H9c2 cell differentiation into cardiomyocytes: cardiac ventricular myosin light 1 expression. In agreement with the lack of morphological changes observed in CAPE treated H9c2 cells, western blot analysis of MLC-1v revealed that expression of this protein was also suppressed by CAPE in comparison to the control expression (panel A). The blot is representative of at least three differentiation experiments. The densitometric data (mean  $\pm$  SD), from three independent experiments, show the time dependent changes in MLC-1v expression in CAPE treated H9c2 cells (panel B).

#### 8.3.2 Viability of H9c2 cells in 30µM of NF-kB activation inhibitor CAPE

The 30 $\mu$ M of CAPE, used our experiments that were designed to investigate the effects of the inhibition of NF-kB activation on H9c2 differentiation into cardiomyocytes was not toxic to the cells. The average viability was about 89% of the viability determined in the control cells. Twenty four hours after the introduction of CAPE to the cells, there was no statistically significant difference between the viability calculated in control H9c2 cells and CAPE treated cells (p > 0.05) (Figure 8.12). By 48 hours, the maximum length of time for which the cells were exposed to CAPE, there was a slight reduction in viability of CAPE treated cells (p < 0.05). The viability of CAPE cells at this time was about 73% of the viability determined in the control cells.



Figure 8.12: The viability of H9c2 cells in the presence of the NF-kB activation inhibitor, CAPE. H9c2 cells were seeded in 6-well plates in differentiation medium consisting of DMEM supplemented with 1% FBS and 100units/ml penicillin/100µg/ml streptomycin and 30µM of CAPE as described in the methods section above (Section 4.123). The MTT assay was carried out at the end of the treatment. Twenty four hours post CAPE, there was no difference in viability between control and CAPE treated cells (p > 0.05). A slight reduction in cell viability (p < 0.05) was seen at 48 hours. The data presented in the graph are the mean  $\pm$  SD from three independent experiments.

#### 8.4 DISCUSSION

There is a large amount of data on the role played by cardiac transcription factors in the differentiation process and this includes extended studies on the GATA factors (GATA 4, 5 and 6) (Laverriere et al., 1994; Molkentin et al., 1997; Murphy et al., 1997; Di Lisi et al., 1998; Charron et al., 1999; McBride and Nemer, 2001), Nkx2.5 (Bodmer, 1993; Biben et al., 1997; Tanaka et al., 1999; Bruneau et al., 2000; Jamali et al., 2001; Lee et al., 2004; Zaffran and Frasch, 2005) and T-box factors (Chapman et al., 1996; Li et al., 1997; Bruneau et al., 1999; Bruneau et al., 2001; Jerome and Papaioannou, 2001; Habets et al., 2002; Christoffels et al., 2004; Singh et al., 2005). However there is very little clear information on the role of NF-kB during the differentiation of stem cells into cardiomyocytes. NF-kB, a ubiquitous transcription factor, is present in most cells and plays key roles in many cellular processes including cell proliferation and growth (Guttridge et al., 1999; Hinz et al., 1999; Preciado et al., 2005), apoptosis (Wang et al., 1998; Kaltschmidt et al., 1999; Li et al., 2004; Tse et al., 2007) and in disease processes including inflammation (Tsao et al., 1997; Miagkov et al., 1998).

As far as we are aware, our data are the first to demonstrate a direct requirement of NF-kB activation for cardiac differentiation as demonstrated by the lack of differentiation of H9c2 cells into cardiomyocytes in the presence of the NF-kB inhibitor CAPE. CAPE is a potent and specific inhibitor of NF-kB activation (Natarajan *et al.*, 1996). CAPE was used at 30uM concentration and the MTT data showed that this compound was not markedly toxic at this concentration. Thus the effects of CAPE on the cell differentiation may be

mostly likely due to the inhibition of NF-kB signalling by blocking NF-kB translocation into the nucleus.

Our data showed that blocking NF-kB activation with CAPE delayed the onset of the differentiation of H9c2 cells into cardiomyocytes. Inhibition of H9c2 differentiation into cardiomyocytes as monitored by the morphological changes was evident upto day 8. While MLC-1v levels in CAPE treated cell recovered by day 8 and day 12 (the last day for this experiment), they were however never equal to the levels seen in the control cells. Our data therefore suggest that activation of NF-kB may support H9c2 cell differentiation into cardiomyocytes.

The suggestion by our data that NF-kB may have a role to play in H9c2 cell differentiation into cardiomyocytes is in agreement with recent emerging evidence that suggest positive roles for NF-kB in the differentiation of various other cells including osteoclast differentiation (Abu-Amer, 2001; Vaira *et al.*, 2008), the differentiation of HL-60 cells into monocytes (Tse *et al.*, 2007), neuronal differentiation of neuroblastoma cells (Feng and Porter, 1999), PC12 cells (Foehr *et al.*, 2000) and P19 stem cells (Liu *et al.*, 2004).

While our data and the reports mentioned above from other studies suggest a positive role for NF-kB in cell differentiation, other studies suggest that NF-kB may be a negative regulator of myogenesis due to its ability to promote cell proliferation or growth activity (Lehtinen *et al.*, 1996; Guttridge *et al.*, 1999; Hinz *et al.*, 1999; Guttridge *et al.*, 2000). NF-kB (p50 subunit and p65

subunit) activates the cyclin D1 promoter (Hinz et al., 1999) and inhibits C2C12 myogenesis by means of its growth-promoting activity through a direct transcriptional regulation of cyclin D1 (Guttridge et al., 1999). The inactivation of NF-kB caused a delayed and reduced expression of cyclin D1 in the G1 phase and also affected pRB phosphorylation and cell cycle progression from G1 to S-phase (Hinz et al., 1999). Cyclins are key regulatory proteins that are involved in the regulation of cell cycle progression. They regulate the cell cycle progression by controlling the activities of cyclin-dependent protein kinases (CDK). Mutated IkB- $\alpha$ , with serine residues (ser32 and ser36) changed to alanines, is not subject to phosphorylation and subsequent degradation following an NF-kB-activating stimulus (Brockman et al., 1995). C2C12 myoblasts containing this mutated IkB-a were found to proliferate slower than control cells and exit the cell cycle faster than control cells when (Guttridge et al., 1999). myogenesis was induced Wild-type C2C12 myoblasts were also found to down-regulate the expression of NF-kB before the induction of myogenin and the initiation of myotube formation (Lehtinen et al., 1996). In C2C12 myoblasts, the loss of p50 and p65 NF-kB DNA-binding activity, correlated with myogenic differentiation (Guttridge et al., 1999). The above therefore suggest that NF-kB may negatively regulate myogenesis.

As cell proliferation and cell differentiation appear to be mutually exclusive processes, the growth promoting activity of NF-kB and also its ability to inhibit myogenic transcription factors (Guttridge *et al.*, 2000) may consistent with its role as a negative regulator of myogenesis. If NF-kB is fundamentally a negative regulator of cell differentiation then, CAPE would be expected to

improve H9c2 differentiation into cardiomyocytes. However according to our data CAPE instead blocked H9c2 cell differentiation into cardiomyocytes. Our data may appear to contradict the negative role suggested for NF-kB during cell differentiation, by the other studies (Lehtinen *et al.*, 1996; Guttridge *et al.*, 1999; Hinz *et al.*, 1999; Guttridge *et al.*, 2000). It should however be noted that an exit from the cell cycle is an essential condition for initiating differentiation. Therefore cells that have been induced to differentiate may transiently block NF-kB activation (i.e. transiently remove the growth promoting activity of NF-KB) in order to exit the cell cycle and initiate differentiation. However in the long term, after this transient inactivation of NF-kB, the restored NF-kB activity may be beneficial for the proliferation. Nuclear factor-kappa B may indirectly promote H9c2 cell differentiation into cardiomyocytes by promoting the survival and proliferation of H9c2 cells that have already differentiated or committed to differentiation.

In an experimental condition, prolonged inhibition of NF-kB activation (for example in the presence of CAPE) may impair the ability of the cells, including those that have become committed to cardiomyocyte differentiation, to proliferate. In such experimental conditions, the inhibition of NF-kB activation may have a long term negative role on cell differentiation, due to its ability to impair cell proliferation, independent of any direct effect on the mechanisms mediating the differentiation. The grow-promoting activity of NF-kB, independent of any possible direct role on the induction of cell differentiation, is very important since cells undergoing differentiation are experiencing

tremendous stress in changing from cell type to another and need mechanisms that can enhance their survival. Thus the activation of NF-kB may be required during the differentiation of H9c2 cells into cardiomyocytes.

#### 8.5 Summary

The time dependent differentiation of the H9c2 cells into cardiomyocytes in 1% FBS culture medium was significantly delayed by the inhibition of NF-kB signalling by CAPE ( $30\mu$ M). CAPE inhibition of H9c2 differentiation was more evident when MLC-1v expression was monitored as marker of H9c2 cell cardiac differentiation.

In conclusion, our data suggest that NF-kB activation may be required for H9c2 cell differentiation into cardiomyocytes. Whether NF-kB has a direct active role in inducing the differentiation of H9c2 cells into cardiomyocytes or that NF-kB promotes the survival of the cells that have already differentiated into cardiomyocytes or committed to cardiac differentiation, need further investigation.

**CHAPTER 9.0** 

**GENERAL DISCUSSION** 

## 9.1 Establishment of cardiac differentiation models

The aim of this thesis was to investigate the signal transduction pathways that that are involved in the differentiation of stem cells into cardiomyocytes. The first objective was to establish cardiomyocyte differentiation models using H9c2 and P19 stem cells and to use these models to accomplish the specific objectives of the thesis. The specific objectives were the investigation of the roles of (i) nitric oxide, (ii) phosphoinositide 3-kinase (PI3K), (iii) protein kinase C (PKC), (vi) NF-kB and (vii) p38 MAP kinase signalling pathways in the differentiation of stem cells cardiomyocytes. Stem cell research is a new area in our laboratory and the models needed for the studies carried out in this thesis had to be established and validated.

We have established the H9c2 cell line as a model of cardiomyocyte differentiation and used it for the investigation of the roles of various cell signalling pathways in the differentiation of stem cells into cardiomyocytes. The criteria used were relevance of the model to *in vivo* cardiomyocyte differentiation and simplicity in term of model manipulation. Both models met these conditions in term of being relevant and simple.

The H9c2 cell line was chosen because of its relevance to cardiac regeneration and simplicity of use. The H9c2 cell model has also extensively been used by other researchers not only as a cardiomyocyte differentiation model (Kim *et al.*, 1999; Menard *et al.*, 1999; Hong *et al.*, 2001; Kageyama *et al.*, 2002; Giusti *et al.*, 2004; Pagano *et al.*, 2004) but also in the investigation of other biological processes in cardiomyocytes (Ekhterae *et al.*, 1999;
Kageyama *et al.*, 2002; Armoni *et al.*, 2005; Gross *et al.*, 2006; Qin *et al.*, 2006; Salvatorelli *et al.*, 2006). The H9c2 cell line is therefore an appropriate model for our studies aimed at investigating the signalling mechanisms that regulate differentiation into cardiomyocytes.

Freshly cultured H9c2 cells in a suspension of complete medium were rounded. When undifferentiated H9c2 cells attach to culture dishes, they grow as spindle-shaped mononucleated cells and did not express cardiac specific markers. In agreement with the lack of morphological changes, undifferentiated cells did not expressed cardiac myosin heavy chain as determined by western blotting. However H9c2 cells cultured in low serum medium differentiate, become elongated and fuse to form multinucleated myotubes and express cardiac specific markers (cardiac myosin heavy chain ( $\alpha/\beta$ -MHC) and cardiac ventricular myosin light chain (MLC-1v).

Despite the advantage of the simplicity of the H9c2 cardiac differentiation model, like most models, is not a perfect substitute for the *in vivo* processes. One of the limitations of the H9c2 model for cardiac differentiation is that the cell line was derived from an embryonic heart. This means that the cardiac lineage commitment may have already taken place. However cardiac specific genes such as that which code for cardiac specific myosin heavy chain and ventricular myosin light chain-1 are not constitutively expressed in undifferentiated H9c2 cells until the cells are induced to differentiate into cardiomyocytes. This makes these cells very relevant for exploring of the

signalling transduction events that drives their differentiation into cardiomyocytes.

The mouse P19 embryonal carcinoma cell line (P19 stem cells) was also developed as a model for stem cell differentiation into cardiomyocytes in parallel with the H9c2 cells. P19 embryonal carcinoma stem cells cultured in normal cell culture grade dishes in complete culture medium whether in the presence or absence of 0.8% DMSO did not form embryoid bodies (EB) but grew as clusters, in a single cell monolayers. These clusters did not differentiate into cardiomyocytes. In contrast, P19 cells cultured in microbiological Petri dishes in complete culture medium, supplemented with 0.8% DMSO, formed EBs. The transfer of EB from Petri dishes to standard tissue culture dishes resulted in cell adhesion, the formation of a two-dimensional monolayer, cell migration and proliferation. The first beating clusters of cardiomyocytes were routinely detected by day 6 or 7. The size and number of beating clusters of cardiomyocytes usually increased up to day 14. There was a high degree of variability both in term of the clusters of beating cells and in the frequency at which they appeared to beat.

The P19 stem cell model, like the H9c2 cell model is relatively simple and can be easily manipulated in experiments. P19 stem cells are pluripotent and can develop into cell types of the three germ layers. With proper care during their routine culture and also during experiments P19 stem cell differentiation into cardiomyocytes appears to be highly regulated and reproducible. Like other stem cell lines, P19 stem cells form embryoid bodies in suspension culture under non-adherent conditions. In the presence of DMSO, P19 stem cell

cardiomyocyte differentiation can be induced during the embryoid body formation.

The major advantage of the P19 cell models is that this pluripotent cell line can be maintained continuously in serum-supplemented cell growth medium without the need for LIF or feeder cells to keep them in the undifferentiated state. This allows stem cell research to be carried out at relatively very low cost in comparison to the use of other stem cells that may need special conditions such as knockout serum, LIF or feeder cells. Another strong point for the use of P19 stem cells is that the cell line is a well-established model of cardiomyocyte differentiation (McBurney et al., 1982; Wobus et al., 1991; Monzen et al., 1999; Naito et al., 2003; Brown et al., 2004; Brewer et al., 2005; Angello et al., 2006). The ability of P19 stem cells to differentiate into cardiomyocytes has been clearly established by the expression of cardiac transcription factors including Nkx2.5 (Skerjanc et al., 1998; Monzen et al., 1999; Wen et al., 2007) GATA-4 (Monzen et al., 1999; Uchida et al., 2007; Wen et al., 2007), MEF2C (Skerjanc et al., 1998; Wen et al., 2007) and other cardiac specific proteins including cardiac troponin I (Choi et al., 2004; Fathi et al., 2008) during the cardiac differentiation of these cells.

In conclusion, models of cardiomyocyte differentiation using P19 cells and H9c2 cells were established. The established cardiac differentiation models were subsequently used in further studies exploring the mechanisms that may regulate stem cell differentiation into cardiomyocytes. New or better models may be used in future experiments. However these models, due to their

simplicity, relevance to cardiac differentiation and relative low cost may still be a good a starting point for researchers that are new to the study of signalling pathways in stem cell differentiation into cardiomyocytes.

# 9.2 Effects of nitric oxide donors on H9c2 differentiation into cardiomyocytes

We examined the role of nitric oxide (NO) in the differentiation of H9c2 cells into cardiomyocytes using fast NO releasing donors (NOC-5 and SIN-1) and slow NO releasing donors (NOC-18 and SNAP). SIN-1 and NOC-5, markedly delayed the onset of differentiation of H9c2 cells into cardiomyocytes while SNAP and NOC-18, were not as effective in delaying the onset of differentiation or long term differentiation of H9c2 cells into cardiomyocytes.

In parallel with the morphological changes, NOC-5 and SIN-1 also caused a reduction in the expression of MLC-1v in differentiating H9c2 cells. NOC-18 also delayed the onset of H9c2 cell differentiation into cardiomyocytes but to a lesser extent than NOC-5 and SIN-1. Differentiation associated morphological changes in both control and SNAP-treated H9c2 cells were similar. In agreement with the morphological seen in the control cells and SNAP treated cells, MLC-1v levels in SNAP-treated and control H9c2 cells were almost equal. 8-Bromo-cGMP, an analogue of cGMP, appeared to delay the onset of H9c2 cell differentiation into cardiomyocytes. However, like NOC-5 and SIN-1, 8-Bromo-cGMP failed to block the long term differentiation of H9c2 cells into cardiomyocytes. In parallel, expression of MLC-1v was reduced at day 4 but thereafter MLC-1v expression in controls and 8-Bromo-cGMP-treated cells were indistinguishable.

Fast NO releasing donors including NOC-5 ( $t_{1/2}$  = 93 minutes) and SIN-1 ( $t_{1/2}$  = 40 minutes) release a large part of their NO content in a matter of minutes.

In contrast, slow NO releasing donors including NOC-18 ( $t_{1/2} = 20$  hours) and SNAP ( $t_{1/2} = 37$  hours) release their NO content slowly over time in manner that may resemble the manner that NO is produced by constitutively expressed nNOS and eNOS. The differential affects of fast and slow releasing NO donors on the differentiation of H9c2 cells into cardiomyocytes may be due to their differences in the release of NO.

Excessive NO can disrupt cellular functions and also induce oxidative stress via events that lead to formation of superoxide, peroxynitrite, hydroxyl radical and nitrite radicals, all of which can cause damage to cells and tissues. Nitric oxide released by NOC-18 and SNAP may protect the cells against apoptosis (Kim *et al.*, 1997; Hebestreit *et al.*, 1998) by the suppression of pro-apoptotic ghenes (Thippeswamy *et al.*, 2001) and induction of anti-apoptotic genes (Ciani *et al.*, 2002). NOC-18 and SNAP may therefore promote cell survival and contribute to differentiation of stem cells into cardiomyocytes. In contrast, NOC-5 and SIN-1, by their excessive production of NO, may induce apoptosis and therefore have negative effects on stem cells into cardiomyocytes.

Our findings with 8-Bromo-cGMP is in agreement with the data obtained with some of the NO donors used in that treatment of cells with 8-Bromo-cGMP also delayed the onset of differentiation. The 8-Bromo-cGMP data also suggest role of endogenous NO in H9c2 cell differentiation. The fact that NO (at least NO from fast-releasing NO donors) and also 8-Bromo-cGMP inhibited rather than enhance the differentiation process, at least over the

initial stages of the process, is clearly contradictory to the other studies which show a positive regulatory role of NO in the differentiation of stem cells into cardiomyocytes and also into other lineages. Our studies, as they stand, give little indication as to conflicting findings and further experiments are clearly needed especially in real stem cells or stem cell lines. These studies although planned were not carried out because of time constraints but are clearly needed.

In conclusion, our data would suggest that generation of cardiomyocytes from H9c2 cells may be negatively regulated by endogenous NO and this may occur through generation of cGMP. Our data are however preliminary and further studies are needed to confirm these findings. Moreover, a detailed study of the full signalling pathway activated by NO may prove important in understanding how this molecule regulates differentiation.

## 9.3 Effects of PKC inhibition on H9c2 and P19 stem cell differentiation into cardiomyocytes

The inhibition of the PKC signalling pathway, using BIM-I (10µM), caused marked inhibition of differentiation associated morphological changes (i.e. cell elongation, fusion and myotube formation) that usually accompany H9c2 cell differentiation into cardiomyocytes. Control H9c2 cells cultured in the absence of BIM-I underwent differentiation and formed myotubes. In agreement with the lack of morphological change observed in BIM-I-treated H9c2 cells, the expression of a cardiac specific maker (MLC-1v protein) in these cells was completely blocked up to day 4 and then significantly inhibited thereafter. This suggests that the activation of PKC may have fundamental roles in the events that are responsible for H9c2 cell differentiation into cardiomyocytes.

Similarly PKC inhibition, with BIM-I (10µM), blocked the differentiation of P19 stem into beating cardiomyocytes. P19 stem cells cultured in 0.8% DMSO in the absence of BIM-I differentiated into beating cardiomyocytes. In contrast, P19 stem cells treated with 0.8% DMSO in the presence BIM-I (10µM) did not differentiate into beating cardiomyocytes. In agreement with the lack of appearance of beating cardiomyocytes in BIM-I treated P19 stem cells, the long term expression of cardiac restricted protein troponin I (TnI) was inhibited.

Our data also suggested that the PKC signalling events that are responsible for mediating P19 stem cell differentiation into cardiomyocytes are switched on very early following DMSO treatment since the PKC inhibitor, BIM-I was

only able to block P19 stem differentiation into beating cardiomyocytes only if added at the EB formation stage (i.e. at the time of initiating of P19 cell cardiac differentiation). BIM-I application to P19 stem cells during the four days of EB formation completely blocked cell differentiation into beating cardiomyocytes. However the treatment of P19 stem cells with BIM-I, four days after DMSO induced cardiac differentiation had started, had no effect on the ability of the cells to differentiate into beating cardiomyocytes.

The cross-talk data, discussed below, also showed that PKC may regulate the PI3K signalling pathway, another key signalling pathway that has been widely reported to be required for stem cell differentiation into cardiomyocytes and other cell types. This may however have no direct relevance to the differentiation process since PKC enhanced the phosphorylation of Akt, a protein that is thought to be central to many of PI3K biological effects, but blocked differentiation. This information suggests that while it is desirable to target and investigate the role of specific pathways individually, studies aimed at identifying points of convergence or divergence and indeed cross-talk between pathways are needed in order to get a clear and better understanding of the processes that regulate differentiation.

In conclusion, our data therefore suggest that the activation of PKC signalling pathway may be required for the differentiation of H9c2 cell and P19 stem cell differentiation into cardiomyocytes. The data also suggest that the activation of PKC signalling pathway may be required during the early stages of P19 stem cell differentiation into cardiomyocytes.

# 9.4 Effects of p38 MAPK inhibition on H9c2 and P19 stem cell differentiation into cardiomyocytes

The differentiation of H9c2 cells into cardiomyocytes was blocked by the inhibition of p38 MAPK signalling using SB203580 (10µM). However SB203580 had little or no effect on the differentiation associated morphological changes in H9c2 differentiating into cardiomyocytes. In both control cell and SB203580-treated cells, morphological changes were visible. In contrast to the apparent lack of inhibition of morphological changes in SB203580-treated H9c2 cells, the expression of MLC-1v was decreased in these cells. Our data therefore suggest that p38 MAPK signalling pathway may have a role in the differentiation of H9c2 cells into cardiomyocytes.

In contrast H9c2 cells, SB203580 markedly inhibited the expression of cardiac restricted protein troponin (TnI) and also blocked the cell differentiation into beating cardiomyocytes. The small expression of TnI seen in control P19 stem cells, at the time of EB transfer to cell culture plates, increased upto day 8. In comparison to control cells, TnI levels in SB203580-treated P19 stem cells were reduced. The inhibited or low levels of TnI detected in SB203580-treated cells were in agreement with the lack of appearance of beating cardiomyocytes. It was also observed that SB203580 blocked P19 stem cell differentiation into beating cardiomyocytes, when added to the cells during the EB formation stage. In contrast, SB203580 failed to block the differentiation of P19 stem cells into beating cardiomyocytes, when the cells were treated, with this compound, four days after the initiation of DMSO-induced cardiac differentiation. Our data therefore suggest that p38 MAPK signalling pathway signalling has a role in the differentiation of P19 stem cells into

cardiomyocytes and also that this signalling pathway is switched on during the early stages of P19 stem cell differentiation into cardiomyocytes.

While our data suggest that p38 MAPK signalling pathway may have roles to play in the differentiation of both H9c2 cells and P19 stem cells into cardiomyocytes, the data also showed that p38 MAPK signalling may have a more prominent role in the differentiation of P19 stem cells into cardiomyocytes than in the differentiation of H9c2 cells into cardiomyocytes. SB203580 (10µM) completely block the differentiation of P19 stem cells into beating cardiomyocytes and also markedly reduced the expression of troponin I. In contrast SB203580 failed to block the differentiation-induced morphological changes (i.e. cell elongation, cell fusion and formation of myotubes) in H9c2 cells but did delay the expression of MLC-1v during the differentiation of H9c2 cells into cardiomyocytes. The findings that SB203580 blocked both the differentiation of P19 stem cells into beating cardiomyocytes and the expression of troponin I (in P19 cells), while in H9c2 cells, only blocked the expression of MLC-1v but not the differentiation-induced morphological changes of H9c2, may imply that p38 MAPK have a lesser role in these differentiation-induced morphological changes. This question of whether this is the case was not investigated and it may need further investigation in future studies. Nevertheless, our data suggest that p38 MAPK may be involved in the cascade that drives the differentiation of both cell types into cardiomyocytes. P19 stem cells are undifferentiated cells without any prior cardiac lineage specification. In contrast, H9c2 cells were derived from an embryonic rat heart. The apparent lesser effect of SB203580 on the

differentiation of H9c2 cells into cardiomyocytes and the inability of SB203580 to block P19 cell differentiation into cardiomyocytes, when added to P19 stem cells four after DMSO-induced differentiation, may imply that p38 MAPK activation may be more relevant at early stages of cardiac differentiation.

In conclusion, our data therefore suggest that the activation of p38 MAPK signalling pathway may be an essential signalling event during the differentiation of H9c2 cells and P19 stem cells into cardiomyocytes. The data also suggest that the p38 MAPK exerts its effects at the early stages of P19 stem cell differentiation into cardiomyocytes.

### 9.5 Effects of the PI3K Inhibition on P19 stem and H9c2 Cell differentiation into cardiomyocytes

Our data showed that the activation of the PI3K signalling pathway may be essential for differentiation of both H9c2 and P19 stem cells into cardiomyocytes. The differentiation of both P19 stem cells and H9c2 cells into cardiomyocytes were significantly inhibited by LY294002 ( $20\mu$ M) treatment of these cells.

The inhibition of PI3K, using LY294002 (20µM), completely blocked the differentiation of H9c2 cells into cardiomyocytes (as as determined by the usual morphological changes of cell elongation, fusion and myotube formation). LY294002 treated H9c2 cells largely maintained their normal single cell and spindle shaped morphology which was in contrast to control H9c2 cells which showed the usual morphological changes. In agreement with the lack of morphological changes observed in LY294002 treated H9c2 cells, the expression of cardiac specific ventricular myosin light chain 1 (MLC-1v protein) was significantly inhibited. Our data therefore suggest that activation of the PI3K signalling pathway is required for H9c2 cell differentiation into cardiomyocytes.

Treatment of H9c2 cells with 1% FBS induced a rapid increase in phosphorylation (at serine 473) but this differentiation associated increase in Akt phosphorylation was completely abolished by  $20\mu$ M of LY294002. This differentiation associated increase in Akt phosphorylation was significantly enhanced by BIM-I (10 $\mu$ M) but significantly reduced by SB203580 (10 $\mu$ M).

Our data therefore show that PI3K is activated during 1% FBS-induced differentiation of H9c2 cells into cardiomyocytes and also that PKC and p38 MAPK may modulate Akt phosphorylation at serine 473.

P19 stem cultured in the presence of 0.8% DMSO but in the absence of LY294002 (absence of PI3K inhibition) differentiated into beating cardiomyocytes. However PI3K inhibition, with LY294002 (20µM), completely blocked the DMSO-induced differentiation of P19 stem cells into beating cardiomyocytes. The expression of cardiac specific troponin I was also reduced in P19 stem cells treated with LY294002.

When DMSO was allowed to initiate P19 stem cell differentiation, for four days before the addition of LY294002 to the cells, LY294002 completely failed to block the cell differentiation into beating cardiomyocytes. Our data therefore suggest that activation of the PI3K signalling pathway is required for P19 stem cell differentiation into cardiomyocytes. In addition, like the PKC and p38 MAPK data, these results suggest that PI3K signalling is required during the early stages of P19 stem cell differentiation into cardiomyocytes.

Akt activation by sequential phosphorylation is a critical event in the PI3K signalling and is generally thought to be a central event in most of the effects mediated by PI3K activation (Ahmed *et al.*, 1997; Dudek *et al.*, 1997; Kauffmann-Zeh *et al.*, 1997; Eves *et al.*, 1998; Gingras *et al.*, 1998). Our data showed that while PI3K is required for stem cell differentiation into cardiomyocytes, Akt phosphorylation is not sufficient to induce cardiomyocyte

differentiation since enhanced Akt activation failed to induce H9c2 cell cardiac differentiation. Our cross-talk studies with PKC showed that PKC inhibition enhanced Akt phosphorylation at serine 473 (i.e. Akt activation). As already discussed (Chapter 7), the enhanced Akt activation by PKC inhibition cannot be explained by non-specificity BIM-I (10µM). Other studies using different PKC inhibitors Ro-31-8220 (bisindolyImaleimide including IX), bisindolylmaleimide VIII and LY 379196 (another bisindolylmaleimide) have shown that PKC inhibition increased Akt activation (Wen et al., 2003; Edling et al., 2007). Akt phosphorylation, in HEK293 cells over-expressing Akt (i.e. cells that show higher levels of Akt activation), was inhibited by phorbol 12myristate-13-acetate (PMA) a potent PKC activator (Wen et al., 2003).

The enhancement of Akt activation by PKC inhibition may therefore suggest that PKC modulates Akt activity. However, since PKC inhibition also inhibited the differentiation of both H9c2 cells and P19 stem cells into cardiomyocytes. These data suggest that the possible crucial role of PI3K in cardiac differentiation may be carried out via mechanism(s) that is/are independent of Akt activation since enhancement of Akt phosphorylation failed to enhance the differentiation of either P19 stem cell or H9c2 cells into cardiomyocytes. However, PI3K inhibition (i.e. prevention of Akt activation) by LY294002 also blocked H9c2 and P19 stem cell differentiation into cardiomyocytes. The data suggest that the differentiation of H9c2 and P19 stem may be dependent on PI3K signalling, but independent of Akt activation. The data further point to a possible target downstream of PI3K through which the effects of PI3K on cardiomyocyte differentiation may be channelled independently of Akt. This

possible key downstream target is currently unknown and warrants further investigation.

The fact that PKC inhibition enhanced the activation of Akt but at the same time also inhibited the cardiac differentiation of both H9c2 cells and P19 stem cells may suggest that, during cardiac differentiation, there is no cross-talk between PI3K and PKC, at least not at a level that involves Akt. The mechanism(s) by which BIM-I promotes Akt phosphorylation is not clear. The enhancement of Akt by the PKC specific inhibitor BIM-I suggests that PKC may play a negative role in the regulation of PI3K pathway. PKC acts downstream of both PI3K and PDK1 (Chou *et al.*, 1998; Dutil *et al.*, 1998; Balendran *et al.*, 2000). Therefore the potential negative regulation of the PI3K signalling pathway by PKC cannot be attributed to a direct effect on either PI3K or PDK1.

Our data give ground for possible speculation about how BIM-I (i.e. PKC inhibition) might have increased Akt phosphorylation at serine 473. PKC could inhibit Akt activation by targeting PDK2, an elusive and currently unknown kinase that is thought to be responsible for Akt phosphorylation at serine 473. Available evidence also suggests that a phosphatase known as PHLPP (PH domain leucine-rich repeat protein phosphatase) selectively and specifically Akt at Ser473 (Gao et al., 2005). Therefore another dephosphorylates possible mechanism by which PKC inhibition might lead to increase Akt phosphorylation may be that PKC is an activator (i.e. positive regulator) of PHLPP "a putative phosphatase" that may be responsible for

dephosphorylating AKt at serine 73 back to the baseline level, following a phosphorylation event. If this is the case then the ability of BIM-I to increase Akt phosphorylation could be explained by the fact that BIM-I reduces the ability of PKC to activate that "putative phosphatase", thus limiting the ability of this "phosphatase to dephosphorylate AKt-ser473, hence the increased AKt-ser473 levels in presence of BIM-I. However it is not known whether PKC can regulate PHPLP.

Our data suggest possible cross-talk between PI3K and p38 MAPK. While PDK1 has been cloned and sequenced (Alessi *et al.*, 1997a) and it is established that PDK1 phosphorylates Akt at threonine 308 (Alessi *et al.*, 1997a; Alessi *et al.*, 1997b), the identity of the kinase that is responsible for Akt phosphorylation at serine 473 remains unknown. Our data showed reduced phosphorylation of Akt at serine 473 when the p38 MAPK inhibitor SB203580 ( $10\mu$ M) was added to the cells. While there is growing list of possible PDK2 candidates including p38 MAPK, the identity of PDK2 remains elusive. The question of whether p38 MAPK has a PI3K-dependent and phosphoinositide-dependent activity that can phosphorylate Akt at serine 473 or whether p38 MAPK and PDK2 are one and the same kinase remains. This needs further investigation.

In conclusion, the activation of the PI3K signalling pathway is required for P19 stem cells and H9c2 cell differentiation into cardiomyocytes. The activation of PI3K signalling is required during the early stages of stem cell differentiation into cardiomyocytes. Our data also suggest that PI3K positively regulates

stem cell differentiation into cardiomyocytes not through Akt activation, but through a yet unknown signalling mechanism that is independent of Akt activation.

Our data suggest no possible cross-talk between PI3K and PKC, at least not at a level that involves Akt. The data also suggest that PKC may also be a negative regulator of Akt phosphorylation at serine 473 either via an unknown PDK2 or via positive regulation of a possible "phosphatase" that is responsible for the dephosphorylation of Akt at serine 473. It is also a suggestion from our data that there may be a possible cross-talk between PI3K and p38 MAPK and that p38 MAPK may also have a PI3K-dependent and phosphoinositidedependent activity with the ability to phosphorylate Akt at Serine 473 or p38 MAPK and PDK2 may be one and the same kinase.



*Figure 9.10: PI3K, PKC and p38 MAPK regulate stem cell differentiation into cardiomyocytes through possible Akt-independent mechanisms.* PI3K activation causes the Akt activation (phosphorylation at both threonine 308 and serine 473) but PI3K may regulate stem cell differentiation into cardiomyocytes through a possible Akt-independent mechanism. Both PKC and P38 MAPK may modulate the phosphorylated states of Akt, However, the inhibition of PKC and p38 MAPK also blocked the differentiation of H9c2 cells and P19 stem cells into cardiomyocytes. Therefore PI3K, PKC, p38 MAPK may regulate H9c2 cells and P19 stem cell differentiation into cardiomyocytes through a possible Akt-independent mechanism(s). The possible Akt-independent mechanisms or downstream target, through which PI3K, PKC and p38 MAPK may regulate H9c2 cell and P19 stem cell differentiation into cardiomyocytes is/are currently unknown

#### 9.6 Effects of the inhibition of the activation of NF-kB on H9c2 cell differentiation into cardiomyocytes

The differentiation of H9c2 cells into cardiomyocytes, in the presence of 1% FBS, was significantly delayed by the inhibition of NF-kB signalling by CAPE (30µM). CAPE inhibition of H9c2 differentiation was more evident when MLC-1v expression was monitored as marker of H9c2 cell cardiac differentiation.

There is a vast amount of data on the roles of cardiac transcription factors but very little information on other non cardiac transcription factors such as NF-κB on the differentiation of stem cells into cardiomyocytes. Our findings are the first to suggest a direct requirement for NF-kB activation for cardiac differentiation as shown by the lack of differentiation of H9c2 cells into cardiomyocytes in the presence of CAPE (30µM). Our data showed that blocking NF-kB activation using CAPE delayed the onset of the differentiation of H9c2 cells into cardiomyocytes.

NF-kB plays a major role in promoting cell survival. It promotes cell proliferation through a direct transcriptional regulation of cyclin D1 (Guttridge *et al.*, 1999). The inactivation of NF-B caused a delayed and reduced expression of cyclin D1 in the G1 phase and also affected pRB phosphorylation and cell cycle progression from G1 to S-phase (Hinz *et al.*, 1999).

As cell proliferation and cell differentiation appear to be mutually exclusive processes. NF-kB, due to its growth-promoting activity, may have negative

effect on cell differentiation (Lehtinen *et al.*, 1996). But in the long term, NFkB may indirectly promote H9c2 cell differentiation into cardiomyocytes by promoting the survival of H9c2 cells that have already differentiated or become committed to differentiation into cardiomyocytes. This role of NF-kB is very important since cells undergoing differentiation are experiencing the stress of changing from one cell type to another and need mechanisms that can enhance their survival. Our data constitute important preliminary findings that warrant further investigations.

In conclusion, our data suggest that NF-kB activation may be required for H9c2 cell differentiation into cardiomyocytes. Whether NF-kB has a direct active role in inducing the differentiation of H9c2 cells into cardiomyocytes or that NF-kB promotes the survival of the cells that have already differentiated into cardiomyocytes or committed to cardiac differentiation, need further investigation.

#### 9.7 CONCLUSION

H9c2 and P19 stem cell lines are relevant and complementary models for the investigation of the intracellular signalling mechanisms that mediate stem cell differentiation into cardiomyocytes. The PKC, PI3K, p38 MAPK and NF-kB signalling pathways are relevant and are involved in the differentiation of stem cells into cardiomyocytes as inhibition of these pathways was accompanied by inhibition of differentiation associated morphological changes (i.e. cell alignment, elongation and fusion to form myotubules in H9c2 cells), inhibition of functional or physiological changes (the appearance of beating cardiomyocytes in P19 stem cells) and inhibition of associated expression cardiac specific protein markers. The next logical stem step is to determine the exact role of these pathways in the initiation of stem cell cardiac differentiation. This would require the design of the experiments that will examine the direct effects of the inhibition or enhancement of these pathways on the expression of cardiac transcription factors (e.g. GATA-4, Nkx2.5). These further studies will establish whether they pathways are core key players or whether their role is the facilitation of stem cardiac differentiation programme once it has been initiated.

Our nitric oxide data suggested that NO delayed the onset of differentiation. This is clearly contradictory to the other studies cited which show a positive regulatory role of NO in the differentiation of stem cells not only into cardiomyocytes but also into other lineages. Our NO studies, as they stand, give little indication as to the reason for these conflicting findings and further

experiments, including concentration-effects curves, are clearly needed especially in real stem cells or stem cell lines.

### **FURTHER WORK**

#### **Further work**

This thesis produced important new data which in turn give rise to new questions (which in a year or so won't seem to be important) that should be resolved in further work.

- 1. Our data suggest that the PI3K, PKC and p38MAPK pathways may be required for stem cell differentiation into cardiomyocytes. These pathways were implicated by monitoring terminal cardiac differentiation (i.e. beating cells or cardiac structural proteins). They question of whether these pathways have direct active roles in inducing the differentiation of stem cells into cardiomyocytes or whether they just facilitate the differentiation of stem into cardiomyocytes needs to be resolved. For each of the signalling pathways implicated a direct connection between their activation and the specific induction of stem cell cardiac lineage commitment needs to be established by monitoring the specific induction of the expression of early cardiac genes and cardiac transcription factors.
- 2. Our data suggest a role for NF-kB in the differentiation of H9c2 cells into cardiomyocytes. These are new data showing the involvement of this ubiquitous but non-cardiac transcription factor in cardiac differentiation. These data should be reproduced in stem cells or stem cell lines using the same criteria suggested for PI3K, PKC and p38 MAPK.

- 3. If the results from 1 or 2 are positive for any of pathways implicated (i.e. PI3K, PKC, p38MAPK or NF-kB) then to get better insight on the role of the implicated pathway, these studies should be extended to the examination of the connection of that pathway to major cardiac differentiation related pathways such as the BMP and Wnt signalling pathways.
- 4. For each of the pathways implicated, specific and selective activation should be carried out, in stem cells or stem cell lines, to complement the inhibition data obtained.
- 5. Our nitric oxide studies were carried out in H9c2 cells and failed to resolve previous conflicting findings. Further experiments should be carried out in stem cells or stem cell lines to resolve these conflicting results by monitoring the specific induction of the expression of early cardiac genes and cardiac transcription factors.
- 6. While our data suggest that Akt phosphorylation at serine 473 may not be of major relevance to the differentiation of stem cells into cardiomyocytes since PI3K, p38 MAPK and PKC are implicated in the differentiation of stem into cardiomyocytes then it may be worthwhile to use other p38 MAPK and PKC inhibitors (preferably more specific and selective) to reproduce the inhibition of PI3K (inhibition of Akt

phosphorylation at serine 473) by the inhibition of p38 MAPK and the activation of PI3K by the inhibition of PKC.

- 7. More specific and selective inhibitors for the various PKC isoforms should be examined in future studies in order to get a better insight of the role of individual PKC isoforms in the differentiation of stem cells into cardiomyocytes.
- 8. The concentrations of the various inhibitors, used in our studies, are in agreement with published concentrations. However in future studies, concentration-effect experiments should be carried out to examine the specificity of drugs used in our studies.

### REFERENCES

Abdel-Latif, M.M., Windle, H.J., Homasany, B.S., Sabra, K. and Kelleher, D. (2005). Caffeic acid phenethyl ester modulates Helicobacter pylori-induced nuclear factor-kappa B and activator protein-1 expression in gastric epithelial cells. *Br J Pharmacol*, **146**, 1139-47.

Abu-Amer, Y. (2001). IL-4 abrogates osteoclastogenesis through STAT6dependent inhibition of NF-kappaB. *J Clin Invest*, **107**, 1375-85.

Ahmed, N.N., Grimes, H.L., Bellacosa, A., Chan, T.O. and Tsichlis, P.N. (1997). Transduction of interleukin-2 antiapoptotic and proliferative signals via Akt protein kinase. *Proc Natl Acad Sci U S A*, **94**, 3627-32.

Ai, Z., Fischer, A., Spray, D.C., Brown, A.M.C. and Fishman, G.I. (2000). Wnt-1 regulation of connexin43 in cardiac myocytes. *J. Clin. Invest.*, **105**, 161-171.

Aicher, A., Kollet, O., Heeschen, C., Liebner, S., Urbich, C., Ihling, C., Orlandi, A., Lapidot, T., Zeiher, A.M. and Dimmeler, S. (2008). The Wnt Antagonist Dickkopf-1 Mobilizes Vasculogenic Progenitor Cells via Activation of the Bone Marrow Endosteal Stem Cell Niche. pp. 796-803.

Akasaka, Y., Morimoto, N., Ishikawa, Y., Fujita, K., Ito, K., Kimura-Matsumoto, M., Ishiguro, S., Morita, H., Kobayashi, Y. and Ishii, T. (2006). Myocardial apoptosis associated with the expression of proinflammatory cytokines during the course of myocardial infarction. *Mod Pathol*, **19**, 588-598.

Alessi, D.R., Andjelkovic, M., Caudwell, B., Cron, P., Morrice, N., Cohen, P. and Hemmings, B.A. (1996). Mechanism of activation of protein kinase B by insulin and IGF-1. *Embo J*, **15**, 6541-51.

Alessi, D.R., Deak, M., Casamayor, A., Caudwell, F.B., Morrice, N., Norman, D.G., Gaffney, P., Reese, C.B., MacDougall, C.N., Harbison, D., Ashworth, A. and Bownes, M. (1997). 3-Phosphoinositide-dependent protein kinase-1 (PDK1): structural and functional homology with the Drosophila DSTPK61 kinase. *Curr Biol*, **7**, 776-89.

Alessi, D.R., James, S.R., Downes, C.P., Holmes, A.B., Gaffney, P.R.J., Reese, C.B. and Cohen, P. (1997). Characterization of a 3-phosphoinositidedependent protein kinase which phosphorylates and activates protein kinase B[alpha]. *Current Biology*, **7**, 261-269.

Alexson, T.O., Hitoshi, S., Coles, B.L., Bernstein, A. and van der Kooy, D. (2006). Notch signaling is required to maintain all neural stem cell populations--irrespective of spatial or temporal niche. *Dev Neurosci*, **28**, 34-48.

Ali, H., Ahamed, J., Hernandez-Munain, C., Baron, J.L., Krangel, M.S. and

Patel, D.D. (2000). Chemokine production by G protein-coupled receptor activation in a human mast cell line: roles of extracellular signal-regulated kinase and NFAT. *J Immunol*, **165**, 7215-23.

Ambrosetti, D.C., Basilico, C. and Dailey, L. (1997). Synergistic activation of the fibroblast growth factor 4 enhancer by Sox2 and Oct-3 depends on protein-protein interactions facilitated by a specific spatial arrangement of factor binding sites. *Mol. Cell. Biol.*, **17**, 6321-6329.

Ambrosetti, D.-C., Scholer, H.R., Dailey, L. and Basilico, C. (2000). Modulation of the Activity of Multiple Transcriptional Activation Domains by the DNA Binding Domains Mediates the Synergistic Action of Sox2 and Oct-3 on the Fibroblast Growth Factor-4 Enhancer. *J. Biol. Chem.*, **275**, 23387-23397.

Amit, M., Carpenter, M.K., Inokuma, M.S., Chiu, C.-P., Harris, C.P., Waknitz, M.A., Itskovitz-Eldor, J. and Thomson, J.A. (2000). Clonally Derived Human Embryonic Stem Cell Lines Maintain Pluripotency and Proliferative Potential for Prolonged Periods of Culture. *Developmental Biology*, **227**, 271-278.

Anderson, D., Self, T., Mellor, I.R., Goh, G., Hill, S.J. and Denning, C. (2007). Transgenic enrichment of cardiomyocytes from human embryonic stem cells. *Mol Ther*, **15**, 2027-36.

Andrée, B., Vorbusch, B., Arnold, H.H., Brand, T. and Duprez, D. (1998). BMP-2 induces ectopic expression of cardiac lineage markers and interferes with somite formation in chicken embryos. *Mechanisms of Development*, **70**, 119-131.

Angello, J.C., Kaestner, S., Welikson, R.E., Buskin, J.N. and Hauschka, S.D. (2006). BMP induction of cardiogenesis in P19 cells requires prior cell-cell interaction(s). *Developmental Dynamics*, **235**, 2122-2133.

Anhuf, D., Weissenbach, M., Schmitz, J., Sobota, R., Hermanns, H.M., Radtke, S., Linnemann, S., Behrmann, I., Heinrich, P.C. and Schaper, F. (2000). Signal Transduction of IL-6, Leukemia-Inhibitory Factor, and Oncostatin M: Structural Receptor Requirements for Signal Attenuation. *J Immunol*, **165**, 2535-2543.

Antonitsis, P., Ioannidou-Papagiannaki, E., Kaidoglou, A. and Papakonstantinou, C. (2007). In vitro cardiomyogenic differentiation of adult human bone marrow mesenchymal stem cells. The role of 5-azacytidine. *Interact CardioVasc Thorac Surg*, **6**, 593-597.

Anyanwu, A.C., Rogers, C.A. and Murday, A.J. (2002). Intrathoracic organ transplantation in the United Kingdom 1995-99: results from the UK cardiothoracic transplant audit. *Heart*, **87**, 449-454.

Aouadi, M., Bost, F., Caron, L., Laurent, K., Le Marchand Brustel, Y. and Binetruy, B. (2006). p38 Mitogen-Activated Protein Kinase Activity Commits

Embryonic Stem Cells to Either Neurogenesis or Cardiomyogenesis. *Stem Cells*, **24**, 1399-1406.

Arai, F., Hirao, A., Ohmura, M., Sato, H., Matsuoka, S., Takubo, K., Ito, K., Koh, G.Y. and Suda, T. (2004). Tie2/angiopoietin-1 signaling regulates hematopoietic stem cell quiescence in the bone marrow niche. *Cell*, **118**, 149-61.

Archer, S.L., Huang, J.M., Hampl, V., Nelson, D.P., Shultz, P.J. and Weir, E.K. (1994). Nitric oxide and cGMP cause vasorelaxation by activation of a charybdotoxin-sensitive K channel by cGMP-dependent protein kinase. *Proc Natl Acad Sci U S A*, **91**, 7583-7.

Armoni, M., Harel, C., Bar-Yoseph, F., Milo, S. and Karnieli, E. (2005). Free fatty acids repress the GLUT4 gene expression in cardiac muscle via novel response elements. *J Biol Chem*, **280**, 34786-95.

Armstrong, L., Lako, M., Lincoln, J., Cairns, P.M. and Hole, N. (2000). mTert expression correlates with telomerase activity during the differentiation of murine embryonic stem cells. *Mech Dev*, **97**, 109-16.

Asia\_Pacific\_Cohort\_Studies\_Collaboration (2006). The impact of cardiovascular risk factors on the age-related excess risk of coronary heart disease. pp. 1025-1033.

Assady, S., Maor, G., Amit, M., Itskovitz-Eldor, J., Skorecki, K.L. and Tzukerman, M. (2001). Insulin Production by Human Embryonic Stem Cells. *Diabetes*, **50**, 1691-1697.

Avilion, A.A., Nicolis, S.K., Pevny, L.H., Perez, L., Vivian, N. and Lovell-Badge, R. (2003). Multipotent cell lineages in early mouse development depend on SOX2 function. *Genes Dev*, **17**, 126-40.

Azpiazu, N. and Frasch, M. (1993). Tinman and bagpipe: Two homeo box genes that determine cell fates in the dorsal mesoderm of Drosophila. *Genes and Development*, **7**, 1325-1340.

Bafico, A., Gazit, A., Pramila, T., Finch, P.W., Yaniv, A. and Aaronson, S.A. (1999). Interaction of frizzled related protein (FRP) with Wnt ligands and the frizzled receptor suggests alternative mechanisms for FRP inhibition of Wnt signaling. *Journal of Biological Chemistry*, **274**, 16180-16187.

Bafico, A., Liu, G., Yaniv, A., Gazit, A. and Aaronson, S.A. (2001). Novel mechanism of Wnt signalling inhibition mediated by Dickkopf-1 interaction with LRP6/Arrow. *Nature Cell Biology*, **3**, 683-686.

Bai, X., Xiao, Z., Pan, Y., Hu, J., Pohl, J., Wen, J. and Li, L. (2004). Cartilagederived morphogenetic protein-1 promotes the differentiation of mesenchymal stem cells into chondrocytes. *Biochem Biophys Res Commun*, **325**, 453-60. Bailey, J. and Europe-Finner, G.N. (2005). Identification of human myometrial target genes of the c-Jun NH2-terminal kinase (JNK) pathway: the role of activating transcription factor 2 (ATF2) and a novel spliced isoform ATF2-small. *J Mol Endocrinol*, **34**, 19-35.

Bain, G., Kitchens, D., Yao, M., Huettner, J.E. and Gottlieb, D.I. (1995). Embryonic stem cells express neuronal properties in vitro. *Dev Biol*, **168**, 342-57.

Balazovich, K.J., Portnow, D., Boxer, L.A. and Prochownik, E.V. (1987). Changes in protein kinase C activity are associated with the differentiation of Friend erythroleukemia cells. *Biochimica et Biophysica Acta - Molecular Cell Research*, **927**, 247-255.

Baldi, A., Abbate, A., Bussani, R., Patti, G., Melfi, R., Angelini, A., Dobrina, A., Rossiello, R., Silvestri, F., Baldi, F. and Di Sciascio, G. (2002). Apoptosis and Post-infarction Left Ventricular Remodeling. *Journal of Molecular and Cellular Cardiology*, **34**, 165-174.

Balendran, A., Hare, G.R., Kieloch, A., Williams, M.R. and Alessi, D.R. (2000). Further evidence that 3-phosphoinositide-dependent protein kinase-1 (PDK1) is required for the stability and phosphorylation of protein kinase C (PKC) isoforms. *FEBS Letters*, **484**, 217-223.

Salh, B., Marotta, A., Wagey, R., Sayed, M., Pelech, S. (2002) Dysregulation of phosphatidylinositol 3-kinase and downstream effectors in human breast cancer. *International Journal of Cancer*, **98**, 148-154.

Bal-Price, A. and Brown, G.C. (2001). Inflammatory Neurodegeneration Mediated by Nitric Oxide from Activated Glia-Inhibiting Neuronal Respiration, Causing Glutamate Release and Excitotoxicity. *J. Neurosci.*, **21**, 6480-6491.

Barancik, M., Bohacova, V., Kvackajova, J., Hudecova, S., Krizanova, O. and Breier, A. (2001). SB203580, a specific inhibitor of p38-MAPK pathway, is a new reversal agent of P-glycoprotein-mediated multidrug resistance. *Eur J Pharm Sci*, **14**, 29-36.

Bardwell, A.J., Abdollahi, M. and Bardwell, L. (2003). Docking sites on mitogen-activated protein kinase (MAPK) kinases, MAPK phosphatases and the Elk-1 transcription factor compete for MAPK binding and are crucial for enzymic activity. *Biochem. J.*, **370**, 1077-1085.

Barry, F., Boynton, R., Murphy, M., Haynesworth, S. and Zaia, J. (2001). The SH-3 and SH-4 antibodies recognize distinct epitopes on CD73 from human mesenchymal stem cells. *Biochem Biophys Res Commun*, **289**, 519-24.

Barry, F.P., Boynton, R.E., Haynesworth, S., Murphy, J.M. and Zaia, J. (1999). The monoclonal antibody SH-2, raised against human mesenchymal stem cells, recognizes an epitope on endoglin (CD105). *Biochem Biophys Res Commun*, **265**, 134-9.

Bartunek, J., Croissant, J.D., Wijns, W., Gofflot, S., de Lavareille, A., Vanderheyden, M., Kaluzhny, Y., Mazouz, N., Willemsen, P., Penicka, M., Mathieu, M., Homsy, C., De Bruyne, B., McEntee, K., Lee, I.W. and Heyndrickx, G.R. (2007). Pretreatment of adult bone marrow mesenchymal stem cells with cardiomyogenic growth factors and repair of the chronically infarcted myocardium. *Am J Physiol Heart Circ Physiol*, **292**, H1095-1104.

Behfar, A., Zingman, L.V., Hodgson, D.M., Rauzier, J.-M., Kane, G.C., Terzic, A. and Puceat, M. (2002). Stem cell differentiation requires a paracrine pathway in the heart. *FASEB J.*, **16**, 1558-1566.

Beier, F. and LuValle, P. (1999). Serum induction of the collagen X promoter requires the Raf/MEK/ERK and p38 pathways. *Biochem Biophys Res Commun*, **262**, 50-4.

Belaguli, N.S., Sepulveda, J.L., Nigam, V., Charron, F., Nemer, M. and Schwartz, R.J. (2000). Cardiac tissue enriched factors serum response factor and GATA-4 are mutual coregulators. *Mol Cell Biol*, **20**, 7550-8.

Bell, L.M., Leong, M.L.L., Kim, B., Wang, E., Park, J., Hemmings, B.A. and Firestone, G.L. (2000). Hyperosmotic Stress Stimulates Promoter Activity and Regulates Cellular Utilization of the Serum- and Glucocorticoid-inducible Protein Kinase (Sgk) by a p38 MAPK-dependent Pathway. *J. Biol. Chem.*, **275**, 25262-25272.

Berrill, A., Tan, H.L., Wuang, S.C., Fong, W.J., Choo, A.B.H. and Oh, S.K.W. (2004). Assessment of Stem Cell Markers During Long-Term Culture of Mouse Embryonic Stem Cells. *Cytotechnology*, **44**, 77-91.

Bertolaso, L., Gibellini, D., Secchiero, P., Previati, M., Falgione, D., Visani, G., Rizzoli, R., Capitani, S. and Zauli, G. (1998). Accumulation of catalytically active PKC-zeta into the nucleus of HL-60 cell line plays a key role in the induction of granulocytic differentiation mediated by all-trans retinoic acid. *Br J Haematol*, **100**, 541-9.

Bhanot, P., Brink, M., Samos, C.H., Hsieh, J.-C., Wang, Y., Macke, J.P., Andrew, D., Nathans, J. and Nusse, R. (1996). A new member of the frizzled family from Drosophila functions as a Wingless receptor. *Nature*, **382**, 225-230.

Bhatia, M., Wang, J.C.Y., Kapp, U., Bonnet, D. and Dick, J.E. (1997). Purification of primitive human hematopoietic cells capable of repopulating immune-deficient mice. *Proc Natl Acad Sci U S A*, **94**, 5320-5325.

Bialik, S., Geenen, D.L., Sasson, I.E., Cheng, R., Horner, J.W., Evans, S.M., Lord, E.M., Koch, C.J. and Kitsis, R.N. (1997). Myocyte apoptosis during acute myocardial infarction in the mouse localizes to hypoxic regions but occurs independently of p53. *J Clin Invest*, **100**, 1363-72.

Biben, C. and Harvey, R.P. (1997). Homeodomain factor Nkx2-5 controls left/right asymmetric expression of bHLH gene eHAND during murine heart development. *Genes and Development*, **11**, 1357-1369.

Biben, C., Palmer, S., Elliott, D.A. and Harvey, R.P. (1997). Homebox genes and heart development. *Cold Spring Harbor Symposia on Quantitative Biology*, **62**, 395-403.

Bishay, K., Ory, K., Lebeau, J., Levalois, C., Olivier, M.F. and Chevillard, S. (2000). DNA damage-related gene expression as biomarkers to assess cellular response after gamma irradiation of a human lymphoblastoid cell line. *Oncogene*, **19**, 916-23.

Bloch, W., Andressen, C., Addicks, K., Fleischmann, B.K., Hops, B., Hescheler, J. and Lorke, D.E. (1999). Nitric oxide synthase expression and role during cardiomyogenesis. *Cardiovascular Research*, **43**, 675-684.

Bobick, B.E., Thornhill, T.M. and Kulyk, W.M. (2007). Fibroblast growth factors 2, 4, and 8 exert both negative and positive effects on limb, frontonasal, and mandibular chondrogenesis via MEK-ERK activation. *J Cell Physiol*, **211**, 233-43.

Boczan, J., Boros, S., Mechler, F., Kovacs, L. and Biro, T. (2000). Differential expressions of protein kinase C isozymes during proliferation and differentiation of human skeletal muscle cells in vitro. *Acta Neuropathol*, **99**, 96-104.

Bodero, A.J., Ye, R. and Lees-Miller, S.P. (2003). UV-light induces p38 MAPK-dependent phosphorylation of Bcl10. *Biochem Biophys Res Commun*, **301**, 923-6.

Bodmer, R. (1993). The gene tinman is required for specification of the heart and visceral muscles in Drosophila. *Development*, **118**, 719-729.

Boettger, T., Stein, S. and Kessel, M. (1997). The chicken NKX2.8 homeobox gene: A novel member of the NK-2 gene family. *Development Genes and Evolution*, **207**, 65-70.

Boeuf, H., Hauss, C., Graeve, F.D., Baran, N. and Kedinger, C. (1997). Leukemia Inhibitory Factor-dependent Transcriptional Activation in Embryonic Stem Cells. *J. Cell Biol.*, **138**, 1207-1217.

Bokui, N., Otani, T., Igarashi, K., Kaku, J., Oda, M., Nagaoka, T., Seno, M., Tatematsu, K., Okajima, T., Matsuzaki, T., Ting, K., Tanizawa, K. and Kuroda, S. (2008). Involvement of MAPK signaling molecules and Runx2 in the NELL1-induced osteoblastic differentiation. *FEBS Lett*, **582**, 365-71.

Bonaros, N., Rauf, R., Wolf, D., Margreiter, E., Tzankov, A., Schlechta, B., Kocher, A., Ott, H., Schachner, T., Hering, S., Bonatti, J. and Laufer, G.

(2006). Combined transplantation of skeletal myoblasts and angiopoietic progenitor cells reduces infarct size and apoptosis and improves cardiac function in chronic ischemic heart failure. *The Journal of Thoracic and Cardiovascular Surgery*, **132**, 1321-1328.e2.

Bonavita, F., Stefanelli, C., Giordano, E., Columbaro, M., Facchini, A., Bonafe, F., Caldarera, C.M. and Guarnieri, C. (2003). H9c2 cardiac myoblasts undergo apoptosis in a model of ischemia consisting of serum deprivation and hypoxia: inhibition by PMA. *FEBS Lett*, **536**, 85-91.

Bottazzi, M.E., Zhu, X., Bohmer, R.M. and Assoian, R.K. (1999). Regulation of p21(cip1) expression by growth factors and the extracellular matrix reveals a role for transient ERK activity in G1 phase. *J Cell Biol*, **146**, 1255-64.

Boyer, L.A., Lee, T.I., Cole, M.F., Johnstone, S.E., Levine, S.S., Zucker, J.P., Guenther, M.G., Kumar, R.M., Murray, H.L., Jenner, R.G., Gifford, D.K., Melton, D.A., Jaenisch, R. and Young, R.A. (2005). Core transcriptional regulatory circuitry in human embryonic stem cells. *Cell*, **122**, 947-56.

Boyle, M., Wong, C., Rocha, M. and Jones, D.L. (2007). Decline in Self-Renewal Factors Contributes to Aging of the Stem Cell Niche in the Drosophila Testis. *Cell Stem Cell*, **1**, 470-478.

Bradley, A., Evans, M., Kaufman, M.H. and Robertson, E. (1984). Formation of germ-line chimaeras from embryo-derived teratocarcinoma cell lines. *Nature*, **309**, 255-256.

Brand, T., Andrée, B., Schneider, A., Buchberger, A. and Arnold, H.H. (1997). Chicken NKx2-8, a novel homeobox gene expressed during early heart and foregut development. *Mechanisms of Development*, **64**, 53-59.

Brealey, D., Brand, M., Hargreaves, I., Heales, S., Land, J., Smolenski, R., Davies, N.A., Cooper, C.E. and Singer, M. (2002). Association between mitochondrial dysfunction and severity and outcome of septic shock. *Lancet*, **360**, 219-23.

Brewer, A.C., Alexandrovich, A., Shah, A.M., Mjaatvedt, C.H., Patient, R.K. and Pizzey, J.A. (2005). GATA factors lie upstream of Nkx 2.5 in the transcriptional regulatory cascade that effects cardiogenesis. *Stem Cells and Development*, **14**, 425-439.

Brockman, J.A., Scherer, D.C., McKinsey, T.A., Hall, S.M., Qi, X., Lee, W.Y. and Ballard, D.W. (1995). Coupling of a signal response domain in I kappa B alpha to multiple pathways for NF-kappa B activation. *Mol Cell Biol*, **15**, 2809-18.

Brook, F.A. and Gardner, R.L. (1997). The origin and efficient derivation of embryonic stem cells in the mouse. *Proc Natl Acad Sci U S A*, **94**, 5709-12.

Brown, C.O., III, Chi, X., Garcia-Gras, E., Shirai, M., Feng, X.-H. and

Schwartz, R.J. (2004). The Cardiac Determination Factor, Nkx2-5, Is Activated by Mutual Cofactors GATA-4 and Smad1/4 via a Novel Upstream Enhancer. *J. Biol. Chem.*, **279**, 10659-10669.

Bruneau, B.G., Bao, Z.Z., Tanaka, M., Schott, J.J., Izumo, S., Cepko, C.L., Seidman, J.G. and Seidman, C.E. (2000). Cardiac expression of the ventricle-specific homeobox gene Irx4 is modulated by Nkx2-5 and dHand. *Developmental Biology*, **217**, 266-277.

Bruneau, B.G., Logan, M., Davis, N., Levi, T., Tabin, C.J., Seidman, J.G. and Seidman, C.E. (1999). Chamber-specific cardiac expression of Tbx5 and heart defects in Holt- Oram syndrome. *Developmental Biology*, **211**, 100-108.

Bruneau, B.G., Nemer, G., Schmitt, J.P., Charron, F., Robitaille, L., Caron, S., Conner, D.A., Gessler, M., Nemer, M., Seidman, C.E. and Seidman, J.G. (2001). A murine model of Holt-Oram syndrome defines roles of the T-Box transcription factor Tbx5 in cardiogenesis and disease. *Cell*, **106**, 709-721.

Buchberger, A., Pabst, O., Brand, T., Seidl, K. and Arnold, H.H. (1996). Chick NKx-2.3 represents a novel family member of vertebrate homologues to the Drosophila homeobox gene tinman: Differential expression of cNKx-2.3 and cNKx-2.5 during heart and gut development. *Mechanisms of Development*, **56**, 151-163.

Buckingham, M., Meilhac, S. and Zaffran, S. (2005). Building the mammalian heart from two sources of myocardial cells. *Nature Reviews Genetics*, **6**, 826-835.

Bulavin, D.V., Saito, S., Hollander, M.C., Sakaguchi, K., Anderson, C.W., Appella, E. and Fornace, A.J., Jr. (1999). Phosphorylation of human p53 by p38 kinase coordinates N-terminal phosphorylation and apoptosis in response to UV radiation. *Embo J*, **18**, 6845-54.

Burch, P.M., Yuan, Z., Loonen, A. and Heintz, N.H. (2004). An extracellular signal-regulated kinase 1- and 2-dependent program of chromatin trafficking of c-Fos and Fra-1 is required for cyclin D1 expression during cell cycle reentry. *Mol Cell Biol*, **24**, 4696-709.

Burdon, T., Stracey, C., Chambers, I., Nichols, J. and Smith, A. (1999). Suppression of SHP-2 and ERK signalling promotes self-renewal of mouse embryonic stem cells. *Dev Biol*, **210**, 30-43.

Cai, C.L., Liang, X., Shi, Y., Chu, P.H., Pfaff, S.L., Chen, J. and Evans, S. (2003). Isl1 identifies a cardiac progenitor population that proliferates prior to differentiation and contributes a majority of cells to the heart. *Dev Cell*, **5**, 877-89.

Cai, C.-L., Zhou, W., Yang, L., Bu, L., Qyang, Y., Zhang, X., Li, X., Rosenfeld, M.G., Chen, J. and Evans, S. (2005). T-box genes coordinate regional rates of proliferation and regional specification during cardiogenesis. *Development*,

**132,** 2475-2487.

Calvi, L.M., Adams, G.B., Weibrecht, K.W., Weber, J.M., Olson, D.P., Knight, M.C., Martin, R.P., Schipani, E., Divieti, P., Bringhurst, F.R., Milner, L.A., Kronenberg, H.M. and Scadden, D.T. (2003). Osteoblastic cells regulate the haematopoietic stem cell niche. **425**, 841-846.

Cao, F., Wagner, R.A., Wilson, K.D., Xie, X., Fu, J.D., Drukker, M., Lee, A., Li, R.A., Gambhir, S.S., Weissman, I.L., Robbins, R.C. and Wu, J.C. (2008). Transcriptional and functional profiling of human embryonic stem cell-derived cardiomyocytes. *PLoS ONE*, **3**, e3474.

Caraglia, M., Marra, M., Leonetti, C., Meo, G., D'Alessandro, A.M., Baldi, A., Santini, D., Tonini, G., Bertieri, R., Zupi, G., Budillon, A. and Abbruzzese, A. (2007). R115777 (Zarnestra)/Zoledronic acid (Zometa) cooperation on inhibition of prostate cancer proliferation is paralleled by Erk/Akt inactivation and reduced Bcl-2 and bad phosphorylation. *J Cell Physiol*, **211**, 533-43.

Caspi, O., Huber, I., Kehat, I., Habib, M., Arbel, G., Gepstein, A., Yankelson, L., Aronson, D., Beyar, R. and Gepstein, L. (2007). Transplantation of human embryonic stem cell-derived cardiomyocytes improves myocardial performance in infarcted rat hearts. *J Am Coll Cardiol*, **50**, 1884-93.

Cavallo, R.A., Cox, R.T., Moline, M.M., Roose, J., Polevoy, G.A., Clevers, H., Peifer, M. and Bejsovec, A. (1998). Drosophila Tcf and Groucho interact to repress Wingless signalling activity. *Nature*, **395**, 604-608.

Cazaubon, S., Bornancin, F. and Parker, P.J. (1994). Threonine-497 is a critical site for permissive activation of protein kinase C alpha. *Biochem J*, **301** (Pt 2), 443-8.

Cenni, V., Doppler, H., Sonnenburg, E.D., Maraldi, N., Newton, A.C. and Toker, A. (2002). Regulation of novel protein kinase C epsilon by phosphorylation. *Biochem J*, **363**, 537-45.

Chae, H.J., Kim, H.R., Kwak, Y.G., Ko, J.K., Joo, C.U. and Chae, S.W. (2001). Signal transduction of nitric oxide donor-induced protection in hydrogen peroxide-mediated apoptosis in H9C2 cardiomyoblasts. *Immunopharmacol Immunotoxicol*, **23**, 187-204.

Chambers, I., Colby, D., Robertson, M., Nichols, J., Lee, S., Tweedie, S. and Smith, A. (2003). Functional expression cloning of Nanog, a pluripotency sustaining factor in embryonic stem cells. *Cell*, **113**, 643-55.

Chandrashekhar, Y. (2005). Role of apoptosis in ventricular remodeling. *Curr Heart Fail Rep*, **2**, 18-22.

Chang, J., Sonoyama, W., Wang, Z., Jin, Q., Zhang, C., Krebsbach, P.H., Giannobile, W., Shi, S. and Wang, C.-Y. (2007). Noncanonical Wnt-4 Signaling Enhances Bone Regeneration of Mesenchymal Stem Cells in
Craniofacial Defects through Activation of p38 MAPK. J. Biol. Chem., 282, 30938-30948.

Chapman, D.L., Garvey, N., Hancock, S., Alexiou, M., Agulnik, S.I., Gibson-Brown, J.J., Cebra-Thomas, J., Bollag, R.J., Silver, L.M. and Papaioannou, V.E. (1996). Expression of the T-box family genes, Tbx1-Tbx5, during early mouse development. *Developmental Dynamics*, **206**, 379-390.

Charron, F., Paradis, P., Bronchain, O., Nemer, G. and Nemer, M. (1999). Cooperative interaction between GATA-4 and GATA-6 regulates myocardial gene expression. *Molecular and Cellular Biology*, **19**, 4355-4365.

Charron, F., Tsimiklis, G., Arcand, M., Robitaille, L., Liang, Q., Molkentin, J.D., Meloche, S. and Nemer, M. (2001). Tissue-specific GATA factors are transcriptional effectors of the small GTPase RhoA. *Genes Dev.*, **15**, 2702-2719.

Charron, J., Malynn, B.A., Fisher, P., Stewart, V., Jeannotte, L., Goff, S.P., Robertson, E.J. and Alt, F.W. (1992). Embryonic lethality in mice homozygous for a targeted disruption of the N-myc gene. *Genes Dev.*, **6**, 2248-2257.

Chen, C.Y. and Schwartz, R.J. (1996). Recruitment of the tinman homolog Nkx-2.5 by serum response factor activates cardiac alpha-actin gene transcription. *Molecular and Cellular Biology*, **16**, 6372-6384.

Chen, F. and Weinberg, R.A. (1995). Biochemical evidence for the autophosphorylation and transphosphorylation of transforming growth factor beta receptor kinases. *Proc Natl Acad Sci U S A*, **92**, 1565-9.

Chen, J.N. and Fishman, M.C. (1996). Zebrafish tinman homolog demarcates the heart field and initiates myocardial differentiation. *Development*, **122**, 3809-3816.

Chen, L., Liu, L., Luo, Y. and Huang, S. (2008). MAPK and mTOR pathways are involved in cadmium-induced neuronal apoptosis. *Journal of Neurochemistry*, **105**, 251-261.

Chen, L.-L., Yin, H. and Huang, J. (2007). Inhibition of TGF-[beta]1 signaling by eNOS gene transfer improves ventricular remodeling after myocardial infarction through angiogenesis and reduction of apoptosis. *Cardiovascular Pathology*, **16**, 221-230.

Chiri, S., Bogliolo, S., Ehrenfeld, J. and Ciapa, B. (2004). Activation of extracellular signal-regulated kinase ERK after hypo-osmotic stress in renal epithelial A6 cells. *Biochim Biophys Acta*, **1664**, 224-9.

Cho, E.S., Lee, K.W. and Lee, H.J. (2008). Cocoa procyanidins protect PC12 cells from hydrogen-peroxide-induced apoptosis by inhibiting activation of p38 MAPK and JNK. *Mutation Research/Fundamental and Molecular Mechanisms of Mutagenesis*, **640**, 123-130.

Choi, S.C., Yoon, J., Shim, W.J., Ro, Y.M. and Lim, D.S. (2004). 5-azacytidine induces cardiac differentiation of P19 embryonic stem cells. *Exp Mol Med*, **36**, 515-23.

Chou, M.M., Hou, W., Johnson, J., Graham, L.K., Lee, M.H., Chen, C.S., Newton, A.C., Schaffhausen, B.S. and Toker, A. (1998). Regulation of protein kinase C zeta by PI 3-kinase and PDK-1. *Curr Biol*, **8**, 1069-77.

Chou, M.M., Hou, W., Johnson, J., Graham, L.K., Lee, M.H., Chen, C.-S., Newton, A.C., Schaffhausen, B.S. and Toker, A. (1998). Regulation of protein kinase C [zeta] by PI 3-kinase and PDK-1. *Current Biology*, **8**, 1069-1078.

Christoffels, V.M., Hoogaars, W.M., Tessari, A., Clout, D.E., Moorman, A.F.M. and Campione, M. (2004). T-box transcription factor Tbx2 represses differentiation and formation of the cardiac chambers. *Developmental Dynamics*, **229**, 763-770.

Chuang, S.-M., Wang, I.C. and Yang, J.-L. (2000). Roles of JNK, p38 and ERK mitogen-activated protein kinases in the growth inhibition and apoptosis induced by cadmium. *Carcinogenesis*, **21**, 1423-1432.

Chuva de Sousa Lopes, S.M., Hassink, R.J., Feijen, A., van Rooijen, M.A., Doevendans, P.A., Tertoolen, L., Brutel de la Riviere, A. and Mummery, C.L. (2006). Patterning the heart, a template for human cardiomyocyte development. *Dev Dyn*, **235**, 1994-2002.

Ciani, E., Guidi, S., Bartesaghi, R. and Contestabile, A. (2002). Nitric oxide regulates cGMP-dependent cAMP-responsive element binding protein phosphorylation and Bcl-2 expression in cerebellar neurons: implication for a survival role of nitric oxide. *J Neurochem*, **82**, 1282-9.

Ciani, E., Guidi, S., Della Valle, G., Perini, G., Bartesaghi, R. and Contestabile, A. (2002). Nitric Oxide Protects Neuroblastoma Cells from Apoptosis Induced by Serum Deprivation through cAMP-response Elementbinding Protein (CREB) Activation. *J. Biol. Chem.*, **277**, 49896-49902.

Ciani, E., Severi, S., Contestabile, A., Bartesaghi, R. and Contestabile, A. (2004). Nitric oxide negatively regulates proliferation and promotes neuronal differentiation through N-Myc downregulation. *J Cell Sci*, **117**, 4727-4737.

Clarke, D.L., Johansson, C.B., Wilbertz, J., Veress, B., Nilsson, E., Karlstr, ouml, m, H., Lendahl, U., Fris, eacute and n, J. (2000). Generalized Potential of Adult Neural Stem Cells. *Science*, **288**, 1660-1663.

Muscari, C., Bonafe, F., Carboni, M., Govoni, M., Stanic, I., Gamberini, C., Ricci, F., Tazzari, P. L., Caldarera, C. M., Guarnieri, C. (2008). Difluoromethylornithine stimulates early cardiac commitment of mesenchymal stem cells in a model of mixed culture with cardiomyocytes. *Journal of Cellular Biochemistry*, **103**, 1046-1052.

Cleaver, O.B., Patterson, K.D. and Krieg, P.A. (1996). Overexpression of the tinman-related genes XNkx-2.5 and XNkx-2.3 in Xenopus embryos results in myocardial hyperplasia. *Development*, **122**, 3549-3556.

Coleman, T.A., Kunsch, C., Maher, M., Ruben, S.M. and Rosen, C.A. (1993). Acquisition of NFKB1-selective DNA binding by substitution of four amino acid residues from NFKB1 into RelA. *Mol Cell Biol*, **13**, 3850-9.

Conrad, C., Gottgens, B., Kinston, S., Ellwart, J. and Huss, R. (2002). GATA transcription in a small rhodamine 123(low)CD34(+) subpopulation of a peripheral blood-derived CD34(-)CD105(+) mesenchymal cell line. *Exp Hematol*, **30**, 887-95.

Cote, F., Laflamme, L., Payet, M.D. and Gallo-Payet, N. (1998). Nitric oxide, a new second messenger involved in the action of angiotensin II on neuronal differentiation of NG108-15 cells. *Endocr Res*, **24**, 403-7.

Coussens, L., Parker, P.J., Rhee, L., Yang-Feng, T.L., Chen, E., Waterfield, M.D., Francke, U. and Ullrich, A. (1986). Multiple, distinct forms of bovine and human protein kinase C suggest diversity in cellular signaling pathways. *Science*, **233**, 859-66.

Cowley, S., Paterson, H., Kemp, P. and Marshall, C.J. (1994). Activation of MAP kinase kinase is necessary and sufficient for PC12 differentiation and for transformation of NIH 3T3 cells. *Cell*, **77**, 841-852.

Cuenda, A. and Cohen, P. (1999). Stress-activated Protein Kinase-2/p38 and a Rapamycin-sensitive Pathway Are Required for C2C12 Myogenesis. *J. Biol. Chem.*, **274**, 4341-4346.

Cuenda, A., Rouse, J., Doza, Y.N., Meier, R., Cohen, P., Gallagher, T.F., Young, P.R. and Lee, J.C. (1995). SB 203580 is a specific inhibitor of a MAP kinase homologue which is stimulated by cellular stresses and interleukin-1. *FEBS Lett*, **364**, 229-33.

Cui, L., Johkura, K., Yue, F., Ogiwara, N., Okouchi, Y., Asanuma, K. and Sasaki, K. (2004). Spatial distribution and initial changes of SSEA-1 and other cell adhesion-related molecules on mouse embryonic stem cells before and during differentiation. *J Histochem Cytochem*, **52**, 1447-57.

Danalache, B.A., Paquin, J., Donghao, W., Grygorczyk, R., Moore, J.C., Mummery, C.L., Gutkowska, J. and Jankowski, M. (2007). Nitric oxide signaling in oxytocin-mediated cardiomyogenesis. *Stem Cells*, **25**, 679-88.

Dani, C., Smith, A.G., Dessolin, S., Leroy, P., Staccini, L., Villageois, P., Darimont, C. and Ailhaud, G. (1997). Differentiation of embryonic stem cells into adipocytes in vitro. *J Cell Sci*, **110**, 1279-1285.

Dann, C.E., Hsieh, J.-C., Rattner, A., Sharma, D., Nathans, J. and Leahy, D.J.

(2001). Insights into Wnt binding and signalling from the structures of two Frizzled cysteine-rich domains. *Nature*, **412**, 86-90.

D'Ascenzo, M., Martinotti, G., Azzena, G.B. and Grassi, C. (2002). cGMP/protein kinase G-dependent inhibition of N-type Ca2+ channels induced by nitric oxide in human neuroblastoma IMR32 cells. *J Neurosci*, **22**, 7485-92.

Davidson, G., Mao, B., del Barco Barrantes, I. and Niehrs, C. (2002). Kremen proteins interact with Dickkopf1 to regulate anteroposterior CNS patterning. *Development*, **129**, 5587-5596.

Davidson, G., Wu, W., Shen, J., Bilic, J., Fenger, U., Stannek, P., Glinka, A. and Niehrs, C. (2005). Casein kinase 1 [gamma] couples Wnt receptor activation to cytoplasmic signal transduction. *Nature*, **438**, 867-872.

Davidson, S.M. and Morange, M. (2000). Hsp25 and the p38 MAPK pathway are involved in differentiation of cardiomyocytes. *Developmental Biology*, **218**, 146-160.

Davis, S., Aldrich, T.H., Stahl, N., Pan, L., Taga, T., Kishimoto, T., Ip, N.Y. and Yancopoulos, G.D. (1993). LIFR beta and gp130 as heterodimerizing signal transducers of the tripartite CNTF receptor. *Science*, **260**, 1805-1808.

de Angelis, L., Zhao, J., Andreucci, J.J., Olson, E.N., Cossu, G. and McDermott, J.C. (2005). Regulation of vertebrate myotome development by the p38 MAP kinase-MEF2 signaling pathway. *Dev Biol*, **283**, 171-9.

De Coppi, P., Bartsch, G., Siddiqui, M.M., Xu, T., Santos, C.C., Perin, L., Mostoslavsky, G., Serre, A.C., Snyder, E.Y., Yoo, J.J., Furth, M.E., Soker, S. and Atala, A. (2007). Isolation of amniotic stem cell lines with potential for therapy. *Nat Biotech*, **25**, 100-106.

De La Cruz, M.V., Sanchez Gomez, C., Arteaga, M.M. and Arguello, C. (1977). Experimental study of the development of the truncus and the conus in the chick embryo. *Journal of Anatomy*, **123**, 661-686.

Deak, M., Clifton, A.D., Lucocq, L.M. and Alessi, D.R. (1998). Mitogen- and stress-activated protein kinase-1 (MSK1) is directly activated by MAPK and SAPK2/p38, and may mediate activation of CREB. *Embo J*, **17**, 4426-41.

Degousee, N., Stefanski, E., Lindsay, T.F., Ford, D.A., Shahani, R., Andrews, C.A., Thuerauf, D.J., Glembotski, C.C., Nevalainen, T.J., Tischfield, J. and Rubin, B.B. (2001). p38 MAPK regulates group IIa phospholipase A2 expression in interleukin-1beta -stimulated rat neonatal cardiomyocytes. *J Biol Chem*, **276**, 43842-9.

Delcommenne, M., Tan, C., Gray, V., Rue, L., Woodgett, J. and Dedhar, S. (1998). Phosphoinositide-3-OH kinase-dependent regulation of glycogen synthase kinase 3 and protein kinase B/AKT by the integrin-linked kinase.

*Proc Natl Acad Sci U S A*, **95**, 11211-6.

Denning, C., Allegrucci, C., Priddle, H., Barbadillo-Munoz, M.D., Anderson, D., Self, T., Smith, N.M., Parkin, C.T. and Young, L.E. (2006). Common culture conditions for maintenance and cardiomyocyte differentiation of the human embryonic stem cell lines, BG01 and HUES-7. *Int J Dev Biol*, **50**, 27-37.

DeRossi, C., Laiosa, M.D., Silverstone, A.E. and Holdener, B.C. (2000). Mouse fzd4 maps within a region of chromosome 7 important for thymus and cardiac development. *Genesis*, **27**, 64-75.

Desbois-Mouthon, C., Blivet-Van Eggelpoel, M.J., Beurel, E., Boissan, M., Delelo, R., Cadoret, A. and Capeau, J. (2002). Dysregulation of glycogen synthase kinase-3beta signaling in hepatocellular carcinoma cells. *Hepatology*, **36**, 1528-36.

Di Lisi, R., Millino, C., Calabria, E., Altruda, F., Schiaffino, S. and Ausoni, S. (1998). Combinatorial cis-Acting Elements Control Tissue-specific Activation of the Cardiac Troponin I Gene in Vitro and in Vivo. *J. Biol. Chem.*, **273**, 25371-25380.

Diedrichs, H., Chi, M., Boelck, B., Mehlhorn, U. and Schwinger, R.H. (2004). Increased regulatory activity of the calcineurin/NFAT pathway in human heart failure. *Eur J Heart Fail*, **6**, 3-9.

Ding, G., Zhang, A., Huang, S., Pan, X., Zhen, G., Chen, R. and Yang, T. (2007). ANG II induces c-Jun NH2-terminal kinase activation and proliferation of human mesangial cells via redox-sensitive transactivation of the EGFR. *Am J Physiol Renal Physiol*, **293**, F1889-97.

Ding, V.W., Chen, R.H. and McCormick, F. (2000). Differential regulation of glycogen synthase kinase 3beta by insulin and Wnt signaling. *J Biol Chem*, **275**, 32475-81.

D'Ippolito, G., Diabira, S., Howard, G.A., Menei, P., Roos, B.A. and Schiller, P.C. (2004). Marrow-isolated adult multilineage inducible (MIAMI) cells, a unique population of postnatal young and old human cells with extensive expansion and differentiation potential. *J Cell Sci*, **117**, 2971-2981.

Diwan, A. (2007). Inhibition of ischemic cardiomyocyte apoptosis through targeted ablation of Bnip3 restrains postinfarction remodeling in mice. *J. Clin. Invest.*, **117**, 2825-2833.

Doetschman, T.C., Eistetter, H., Katz, M., Schmidt, W. and Kemler, R. (1985). The in vitro development of blastocyst-derived embryonic stem cell lines: formation of visceral yolk sac, blood islands and myocardium. *J Embryol Exp Morphol*, **87**, 27-45.

Dong, L.Q., Zhang, R.B., Langlais, P., He, H., Clark, M., Zhu, L. and Liu, F.

(1999). Primary Structure, Tissue Distribution, and Expression of Mouse Phosphoinositide-dependent Protein Kinase-1, a Protein Kinase That Phosphorylates and Activates Protein Kinase Czeta. *J. Biol. Chem.*, **274**, 8117-8122.

Dorion, S., Lambert, H. and Landry, J. (2002). Activation of the p38 signaling pathway by heat shock involves the dissociation of glutathione S-transferase Mu from Ask1. *J Biol Chem*, **277**, 30792-7.

Drab, M., Haller, H., Bychkov, R., Erdmann, B., Lindschau, C., Haase, H., Morano, I., Luft, F.C. and Wobus, A.M. (1997). From totipotent embryonic stem cells to spontaneously contracting smooth muscle cells: a retinoic acid and db-cAMP in vitro differentiation model. *FASEB J.*, **11**, 905-915.

Draper, J.S., Pigott, C., Thomson, J.A. and Andrews, P.W. (2002). Surface antigens of human embryonic stem cells: Changes upon differentiation in culture. *Journal of Anatomy*, **200**, 249-258.

Du, L., Smolewski, P., Bedner, E., Traganos, F. and Darzynkiewicz, Z. (2001). Selective protection of mitogenically stimulated human lymphocytes but not leukemic cells from cytosine arabinoside-induced apoptosis by LY294002, a phosphoinositol-3 kinase inhibitor. *Int J Oncol*, **19**, 811-9.

Dudek, H., Datta, S.R., Franke, T.F., Birnbaum, M.J., Yao, R., Cooper, G.M., Segal, R.A., Kaplan, D.R. and Greenberg, M.E. (1997). Regulation of neuronal survival by the serine-threonine protein kinase Akt. *Science*, **275**, 661-5.

Durocher, D., Charron, F., Warren, R., Nemer, M. and Schwartz, R.J. (1997). The cardiac transcription factors nkx2-5 and GATA-4 are mutual cofactors. *EMBO Journal*, **16**, 5687-5696.

Durocher, D., Chen, C.Y., Ardati, A., Schwartz, R.J. and Nemer, M. (1996). The atrial natriuretic factor promoter is a downstream target for Nkx- 2.5 in the myocardium. *Molecular and Cellular Biology*, **16**, 4648-4655.

Durocher, D. and Nemer, M. (1998). Combinatorial interactions regulating cardiac transcription. *Developmental Genetics*, **22**, 250-262.

Dutil, E.M., Toker, A. and Newton, A.C. (1998). Regulation of conventional protein kinase C isozymes by phosphoinositide-dependent kinase 1 (PDK-1). *Current Biology*, **8**, 1366-1375.

Eastman, Q. and Grosschedl, R. (1999). Regulation of LEF-1/TCF transcription factors by Wnt and other signals. *Current Opinion in Cell Biology*, **11**, 233-240.

Edling, C.E., Pedersen, M., Carlsson, L., Ronnstrand, L., Palmer, R.H. and Hallberg, B. (2007). Haematopoietic progenitor cells utilise conventional PKC to suppress PKB/Akt activity in response to c-Kit stimulation. *Br J Haematol*,

**136,** 260-8.

Eisenberg, C.A., Gourdie, R.G. and Eisenberg, L.M. (1997). Wnt-11 is expressed in early avian mesoderm and required for the differentiation of the quail mesoderm cell line QCE-6. *Development*, **124**, 525-536.

Ekhterae, D., Lin, Z., Lundberg, M.S., Crow, M.T., Brosius, F.C., 3rd and Nunez, G. (1999). ARC inhibits cytochrome c release from mitochondria and protects against hypoxia-induced apoptosis in heart-derived H9c2 cells. *Circ Res*, **85**, e70-7.

Eligini, S., Habib, A., Lebret, M., Creminon, C., Levy-Toledano, S. and Maclouf, J. (2001). Induction of cyclo-oxygenase-2 in human endothelial cells by SIN-1 in the absence of prostaglandin production. *Br J Pharmacol*, **133**, 1163-1171.

Ellert-Miklaszewska, A., Kaminska, B. and Konarska, L. (2005). Cannabinoids down-regulate PI3K/Akt and Erk signalling pathways and activate proapoptotic function of Bad protein. *Cell Signal*, **17**, 25-37.

Endo, T.A., Masuhara, M., Yokouchi, M., Suzuki, R., Sakamoto, H., Mitsui, K., Matsumoto, A., Tanimura, S., Ohtsubo, M., Misawa, H., Miyazaki, T., Leonor, N., Taniguchi, T., Fujita, T., Kanakura, Y., Komiya, S. and Yoshimura, A. (1997). A new protein containing an SH2 domain that inhibits JAK kinases. *Nature*, **387**, 921-4.

Endo, Y., Beauchamp, E., Woods, D., Taylor, W.G., Toretsky, J.A., Uren, A. and Rubin, J.S. (2008). Wnt-3a and Dickkopf-1 stimulate neurite outgrowth in Ewing tumor cells via a Frizzled3- and c-Jun N-terminal kinase-dependent mechanism. *Mol Cell Biol*, **28**, 2368-79.

Endres, M., Scott, G., Namura, S., Salzman, A.L., Huang, P.L., Moskowitz, M.A. and Szabó, C. (1998). Role of peroxynitrite and neuronal nitric oxide synthase in the activation of poly(ADP-ribose) synthetase in a murine model of cerebral ischemia-reperfusion. *Neuroscience Letters*, **248**, 41-44.

Engel, D., Peshock, R., Armstong, R.C., Sivasubramanian, N. and Mann, D.L. (2004). Cardiac myocyte apoptosis provokes adverse cardiac remodeling in transgenic mice with targeted TNF overexpression. *Am J Physiol Heart Circ Physiol*, **287**, H1303-1311.

Erlich, R.B., Kahn, S.A., Lima, F.R., Muras, A.G., Martins, R.A., Linden, R., Chiarini, L.B., Martins, V.R. and Moura Neto, V. (2007). STI1 promotes glioma proliferation through MAPK and PI3K pathways. *Glia*, **55**, 1690-8.

Etzion, S., Battler, A., Barbash, I.M., Cagnano, E., Zarin, P., Granot, Y., Kedes, L.H., Kloner, R.A. and Leor, J. (2001). Influence of embryonic cardiomyocyte transplantation on the progression of heart failure in a rat model of extensive myocardial infarction. *J Mol Cell Cardiol*, **33**, 1321-30.

Evans, M.J. and Kaufman, M.H. (1981). Establishment in culture of pluripotential cells from mouse embryos. *Nature*, **292**, 154-156.

Evans, S.M., Yan, W., Murillo, M.P., Ponce, J. and Papalopulu, N. (1995). tinman, a Drosophila homeobox gene required for heart and visceral mesoderm specification, may be represented by a family of genes in vertebrates: XNkx-2.3, a second vertebrate homologue of tinman. *Development*, **121**, 3889-3899.

Eves, E.M., Xiong, W., Bellacosa, A., Kennedy, S.G., Tsichlis, P.N., Rosner, M.R. and Hay, N. (1998). Akt, a target of phosphatidylinositol 3-kinase, inhibits apoptosis in a differentiating neuronal cell line. *Mol Cell Biol*, **18**, 2143-52.

Fathi, F., Murasawa, S., Hasegawa, S., Asahara, T., Kermani, A.J. and Mowla, S.J. (2008). Cardiac differentiation of P19CL6 cells by oxytocin. *Int J Cardiol.* 

Feng, Z. and Porter, A.G. (1999). NF-kappa B/Rel Proteins Are Required for Neuronal Differentiation of SH-SY5Y Neuroblastoma Cells. *J. Biol. Chem.*, **274**, 30341-30344.

Ferrero, R., Rodriguez-Pascual, F., Miras-Portugal, M.T. and Torres, M. (1999). Comparative effects of several nitric oxide donors on intracellular cyclic GMP levels in bovine chromaffin cells: correlation with nitric oxide production. *Br J Pharmacol*, **127**, 779-87.

Fleming, H.E., Janzen, V., Lo Celso, C., Guo, J., Leahy, K.M., Kronenberg, H.M. and Scadden, D.T. (2008). Wnt Signaling in the Niche Enforces Hematopoietic Stem Cell Quiescence and Is Necessary to Preserve Self-Renewal In Vivo. *Cell Stem Cell*, **2**, 274-283.

Fliss, H. and Gattinger, D. (1996). Apoptosis in Ischemic and Reperfused Rat Myocardium. *Circ Res*, **79**, 949-956.

Flynn, P., Wong, M., Zavar, M., Dean, N.M. and Stokoe, D. (2000). Inhibition of PDK-1 activity causes a reduction in cell proliferation and survival. *Current Biology*, **10**, 1439-1442.

Foehr, E.D., Bohuslav, J., Chen, L.-F., DeNoronha, C., Geleziunas, R., Lin, X., O'Mahony, A. and Greene, W.C. (2000). The NF-kappa B-inducing Kinase Induces PC12 Cell Differentiation and Prevents Apoptosis. *J. Biol. Chem.*, **275**, 34021-34024.

Fok, E.Y.L. and Zandstra, P.W. (2005). Shear-Controlled Single-Step Mouse Embryonic Stem Cell Expansion and Embryoid Body-Based Differentiation. *Stem Cells*, **23**, 1333-1342.

Franco, D., Lamers, W.H. and Moorman, A.F.M. (1998). Patterns of expression in the developing myocardium: Towards a morphologically

integrated transcriptional model. Cardiovascular Research, 38, 25-53.

Freedman, J.E., Sauter, R., Battinelli, E.M., Ault, K., Knowles, C., Huang, P.L. and Loscalzo, J. (1999). Deficient Platelet-Derived Nitric Oxide and Enhanced Hemostasis in Mice Lacking the NOSIII Gene. *Circ Res*, **84**, 1416-1421.

Frossi, B., Rivera, J., Hirsch, E. and Pucillo, C. (2007). Selective Activation of Fyn/PI3K and p38 MAPK Regulates IL-4 Production in BMMC under Nontoxic Stress Condition. *J Immunol*, **178**, 2549-2555.

Frost, J.A., Geppert, T.D., Cobb, M.H. and Feramisco, J.R. (1994). A requirement for extracellular signal-regulated kinase (ERK) function in the activation of AP-1 by Ha-Ras, phorbol 12-myristate 13-acetate, and serum. *Proc Natl Acad Sci U S A*, **91**, 3844-8.

Fu, Y., Yan, W., Mohun, T.J. and Evans, S.M. (1998). Vertebrate tinman homologues XNkx2-3 and XNkx2-5 are required for heart formation in a functionally redundant manner. *Development*, **125**, 4439-4449.

Fukuda, A., Hikita, A., Wakeyama, H., Akiyama, T., Oda, H., Nakamura, K. and Tanaka, S. (2005). Regulation of osteoclast apoptosis and motility by small GTPase binding protein Rac1. *J Bone Miner Res*, **20**, 2245-53.

Fukuda, H., Takahashi, J., Watanabe, K., Hayashi, H., Morizane, A., Koyanagi, M., Sasai, Y. and Hashimoto, N. (2006). Fluorescence-Activated Cell Sorting-Based Purification of Embryonic Stem Cell-Derived Neural Precursors Averts Tumor Formation after Transplantation. pp. 763-771.

Fukuda, K. (2001). Development of Regenerative Cardiomyocytes from Mesenchymal Stem Cells for Cardiovascular Tissue Engineering. *Artificial Organs*, **25**, 187-193.

Fukunaga, R. and Hunter, T. (1997). MNK1, a new MAP kinase-activated protein kinase, isolated by a novel expression screening method for identifying protein kinase substrates. *Embo J*, **16**, 1921-33.

Gaitanaki, C., Konstantina, S., Chrysa, S. and Beis, I. (2003). Oxidative stress stimulates multiple MAPK signalling pathways and phosphorylation of the small HSP27 in the perfused amphibian heart. *J Exp Biol*, **206**, 2759-2769.

Gaitanaki, C., Papatriantafyllou, M., Stathopoulou, K. and Beis, I. (2006). Effects of various oxidants and antioxidants on the p38-MAPK signalling pathway in the perfused amphibian heart. *Mol Cell Biochem*, **291**, 107-17.

Gallea, S., Lallemand, F., Atfi, A., Rawadi, G., Ramez, V., Spinella-Jaegle, S., Kawai, S., Faucheu, C., Huet, L., Baron, R. and Roman-Roman, S. (2001). Activation of mitogen-activated protein kinase cascades is involved in regulation of bone morphogenetic protein-2-induced osteoblast differentiation in pluripotent C2C12 cells. *Bone*, **28**, 491-8.

Gao, T., Furnari, F. and Newton, A.C. (2005). PHLPP: A Phosphatase that Directly Dephosphorylates Akt, Promotes Apoptosis, and Suppresses Tumor Growth. *Molecular Cell*, **18**, 13-24.

Garg, V. (2006). Insights into the genetic basis of congenital heart disease. *Cell Mol Life Sci*, **63**, 1141-8.

Garg, V., Kathiriya, I.S., Barnes, R., Schluterman, M.K., King, I.N., Butler, C.A., Rothrock, C.R., Eapen, R.S., Hirayama-Yamada, K., Joo, K., Matsuoka, R., Cohen, J.C. and Srivastava, D. (2003). GATA4 mutations cause human congenital heart defects and reveal an interaction with TBX5. *Nature*, **424**, 443-447.

Garrido, J.L., Godoy, J.A., Alvarez, A., Bronfman, M. and Inestrosa, N.C. (2002). Protein kinase C inhibits amyloid beta peptide neurotoxicity by acting on members of the Wnt pathway. *Faseb J*, **16**, 1982-4.

Garriock, R.J., D'Agostino, S.L., Pilcher, K.C. and Krieg, P.A. (2005). Wnt11-R, a protein closely related to mammalian Wnt11, is required for heart morphogenesis in Xenopus. *Developmental Biology*, **279**, 179-192.

Gassanov, N., Jankowski, M., Danalache, B., Wang, D., Grygorczyk, R., Hoppe, U.C. and Gutkowska, J. (2007). Arginine vasopressin-mediated cardiac differentiation: Insights into the role of its receptors and nitric oxide signaling. *J. Biol. Chem.*, **282**, 11255-11265.

Gearing, D., Thut, C., VandeBos, T., Gimpel, S., Delaney, P., King, J., Price, V., Cosman, D. and Beckmann, M. (1991). Leukemia inhibitory factor receptor is structurally related to the IL-6 signal transducer, gp130. *EMBO J.*, **10**, 2839-2848.

Gearing, D.P. and Bruce, A.G. (1992). Oncostatin M binds the high-affinity leukemia inhibitory factor receptor. *New Biologist*, **4**, 61-65.

Ghigo, D., Priotto, C., Migliorino, D., Geromin, D., Franchino, C., Todde, R., Costamagna, C., Pescarmona, G. and Bosia, A. (1998). Retinoic acid-induced differentiation in a human neuroblastoma cell line is associated with an increase in nitric oxide synthesis. *J Cell Physiol*, **174**, 99-106.

Giles, R.H., Van Es, J.H. and Clevers, H. (2003). Caught up in a Wnt storm: Wnt signaling in cancer. *Biochimica et Biophysica Acta - Reviews on Cancer*, **1653**, 1-24.

Gingras, A.C., Kennedy, S.G., O'Leary, M.A., Sonenberg, N. and Hay, N. (1998). 4E-BP1, a repressor of mRNA translation, is phosphorylated and inactivated by the Akt(PKB) signaling pathway. *Genes Dev*, **12**, 502-13.

Giorgione, J.R., Lin, J.-H., McCammon, J.A. and Newton, A.C. (2006). Increased Membrane Affinity of the C1 Domain of Protein Kinase C{delta} Compensates for the Lack of Involvement of Its C2 Domain in Membrane Recruitment. J. Biol. Chem., 281, 1660-1669.

Giusti, L., Gargini, C., Ceccarelli, F., Bacci, M., Italiani, P. and Mazzoni, M.R. (2004). Modulation of endothelin-A receptor, Galpha subunit, and RGS2 expression during H9c2 cardiomyoblast differentiation. *J Recept Signal Transduct Res*, **24**, 297-317.

Glinka, A., Wu, W., Delius, H., Monaghan, A.P., Blumenstock, C. and Niehrs, C. (1998). Dickkopf-1 is a member of a new family of secreted proteins and functions in head induction. *Nature*, **391**, 357-362.

Goetze, S., Xi, X.-P., Kawano, Y., Kawano, H., Fleck, E., Hsueh, W.A. and Law, R.E. (1999). TNF-{alpha}–Induced Migration of Vascular Smooth Muscle Cells Is MAPK Dependent. *Hypertension*, **33**, 183-189.

Golan, T., Yaniv, A., Bafico, A., Liu, G. and Gazit, A. (2004). The human Frizzled 6 (HFz6) acts as a negative regulator of the canonical Wnt. beta-catenin signaling cascade. *J Biol Chem*, **279**, 14879-88.

Gold, M.E., Wood, K.S., Byrns, R.E., Buga, G.M. and Ignarro, L.J. (1990). Larginine-dependent vascular smooth muscle relaxation and cGMP formation. *Am J Physiol*, **259**, H1813-21.

Gonzalez, A., Rota, M., Nurzynska, D., Misao, Y., Tillmanns, J., Ojaimi, C., Padin-Iruegas, M.E., Muller, P., Esposito, G., Bearzi, C., Vitale, S., Dawn, B., Sanganalmath, S.K., Baker, M., Hintze, T.H., Bolli, R., Urbanek, K., Hosoda, T., Anversa, P., Kajstura, J. and Leri, A. (2008). Activation of cardiac progenitor cells reverses the failing heart senescent phenotype and prolongs lifespan. *Circ Res*, **102**, 597-606.

Gooi, H.C., Feizi, T., Kapadia, A., Knowles, B.B., Solter, D. and Evans, M.J. (1981). Stage-specific embryonic antigen involves alpha 1 goes to 3 fucosylated type 2 blood group chains. *Nature*, **292**, 156-8.

Grauss, R.W., Winter, E.M., van Tuyn, J., Pijnappels, D.A., Steijn, R.V., Hogers, B., van der Geest, R.J., de Vries, A.A.F., Steendijk, P., van der Laarse, A., Gittenberger-de Groot, A.C., Schalij, M.J. and Atsma, D.E. (2007). Mesenchymal stem cells from ischemic heart disease patients improve left ventricular function after acute myocardial infarction. *Am J Physiol Heart Circ Physiol*, **293**, H2438-2447.

Greenlee, A.R., Kronenwetter-Koepel, T.A., Kaiser, S.J. and Liu, K. (2005). Comparison of Matrigel(TM) and gelatin substrata for feeder-free culture of undifferentiated mouse embryonic stem cells for toxicity testing. *Toxicology in Vitro*, **19**, 389-397.

Grepin, C., Dagnino, L., Robitaille, L., Haberstroh, L., Antakly, T. and Nemer, M. (1994). A hormone-encoding gene identifies a pathway for cardiac but not skeletal muscle gene transcription. *Molecular and Cellular Biology*, **14**, 3115-3129.

Grepin, C., Nemer, G. and Nemer, M. (1997). Enhanced cardiogenesis in embryonic stem cells overexpressing the GATA-4 transcription factor. *Development*, **124**, 2387-2395.

Gross, E.R., Hsu, A.K. and Gross, G.J. (2006). The JAK/STAT Pathway is Essential For Opioid-Induced Cardioprotection: JAK2 as a Mediator of STAT3, Akt and GSK3{beta}. *Am J Physiol Heart Circ Physiol*.

Gross, E.R., Hsu, A.K. and Gross, G.J. (2006). The JAK/STAT pathway is essential for opioid-induced cardioprotection: JAK2 as a mediator of STAT3, Akt, and GSK-3 beta. *Am J Physiol Heart Circ Physiol*, **291**, H827-34.

Grow, M.W. and Krieg, P.A. (1998). Tinman Function Is Essential for Vertebrate Heart Development: Elimination of Cardiac Differentiation by Dominant Inhibitory Mutants of thetinman-Related Genes,XNkx2-3andXNkx2-5. *Developmental Biology*, **204**, 187-196.

Gum, R.J., McLaughlin, M.M., Kumar, S., Wang, Z., Bower, M.J., Lee, J.C., Adams, J.L., Livi, G.P., Goldsmith, E.J. and Young, P.R. (1998). Acquisition of sensitivity of stress-activated protein kinases to the p38 inhibitor, SB 203580, by alteration of one or more amino acids within the ATP binding pocket. *J Biol Chem*, **273**, 15605-10.

Guttridge, D.C., Albanese, C., Reuther, J.Y., Pestell, R.G. and Baldwin, A.S., Jr. (1999). NF-kappa B Controls Cell Growth and Differentiation through Transcriptional Regulation of Cyclin D1. *Mol. Cell. Biol.*, **19**, 5785-5799.

Guttridge, D.C., Mayo, M.W., Madrid, L.V., Wang, C.Y. and Baldwin, A.S., Jr. (2000). NF-kappaB-induced loss of MyoD messenger RNA: possible role in muscle decay and cachexia. *Science*, **289**, 2363-6.

Habara-Ohkubo, A. (1996). Differentiation of beating cardiac muscle cells from a derivative of P19 embryonal carcinoma cells. *Cell Struct Funct*, **21**, 101-10.

Habets, P.E.M.H., Moorman, A.F.M., Clout, D.E.W., Van Roon, M.A., Lingbeek, M., Van Lohuizen, M., Campione, M. and Christoffels, V.M. (2002). Cooperative action of Tbx2 and Nkx2.5 inhibits ANF expression in the atrioventricular canal: Implications for cardiac chamber formation. *Genes and Development*, **16**, 1234-1246.

Hahn, J.-Y., Cho, H.-J., Kang, H.-J., Kim, T.-S., Kim, M.-H., Chung, J.-H., Bae, J.-W., Oh, B.-H., Park, Y.-B. and Kim, H.-S. (2008). Pre-Treatment of Mesenchymal Stem Cells With a Combination of Growth Factors Enhances Gap Junction Formation, Cytoprotective Effect on Cardiomyocytes, and Therapeutic Efficacy for Myocardial Infarction. *Journal of the American College of Cardiology*, **51**, 933-943.

Hakuno, D., Fukuda, K., Makino, S., Konishi, F., Tomita, Y., Manabe, T.,

Suzuki, Y., Umezawa, A. and Ogawa, S. (2002). Bone Marrow-Derived Regenerated Cardiomyocytes (CMG Cells) Express Functional Adrenergic and Muscarinic Receptors. *Circulation*, **105**, 380-386.

Hamazaki, T., Kehoe, S.M., Nakano, T. and Terada, N. (2006). The Grb2/Mek pathway represses Nanog in murine embryonic stem cells. *Mol Cell Biol*, **26**, 7539-49.

Hambardzumyan, D., Becher, O.J., Rosenblum, M.K., Pandolfi, P.P., Manova-Todorova, K. and Holland, E.C. (2008). PI3K pathway regulates survival of cancer stem cells residing in the perivascular niche following radiation in medulloblastoma in vivo. *Genes Dev*, **22**, 436-48.

Hansis, C., Grifo, J.A. and Krey, L.C. (2000). Oct-4 expression in inner cell mass and trophectoderm of human blastocysts. *Mol Hum Reprod*, **6**, 999-1004.

Harrelson, Z., Kelly, R.G., Goldin, S.N., Gibson-Brown, J.J., Bollag, R.J., Silver, L.M. and Papaioannou, V.E. (2004). Tbx2 is essential for patterning the atrioventricular canal and for morphogenesis of the outflow tract during heart development. *Development*, **131**, 5041-5052.

Hart, A.H., Hartley, L., Ibrahim, M. and Robb, L. (2004). Identification, cloning and expression analysis of the pluripotency promoting Nanog genes in mouse and human. *Dev Dyn*, **230**, 187-98.

Hatzoglou, A., Roussel, J., Bourgeade, M.F., Rogier, E., Madry, C., Inoue, J., Devergne, O. and Tsapis, A. (2000). TNF receptor family member BCMA (B cell maturation) associates with TNF receptor-associated factor (TRAF) 1, TRAF2, and TRAF3 and activates NF-kappa B, elk-1, c-Jun N-terminal kinase, and p38 mitogen-activated protein kinase. *J Immunol*, **165**, 1322-30.

Hayakawa, J., Depatie, C., Ohmichi, M. and Mercola, D. (2003). The activation of c-Jun NH2-terminal kinase (JNK) by DNA-damaging agents serves to promote drug resistance via activating transcription factor 2 (ATF2)-dependent enhanced DNA repair. *J Biol Chem*, **278**, 20582-92.

Hazan-Halevy, I. and Levy, R. (2000). Activation of cytosolic phospholipase A2 by opsonized zymosan in human neutrophils requires both ERK and p38 MAP-kinase. *Adv Exp Med Biol*, **479**, 115-23.

He, J.-Q., Ma, Y., Lee, Y., Thomson, J.A. and Kamp, T.J. (2003). Human Embryonic Stem Cells Develop Into Multiple Types of Cardiac Myocytes: Action Potential Characterization. *Circ Res*, **93**, 32-39.

He, X.C., Zhang, J., Tong, W.G., Tawfik, O., Ross, J., Scoville, D.H., Tian, Q., Zeng, X., He, X., Wiedemann, L.M., Mishina, Y. and Li, L. (2004). BMP signaling inhibits intestinal stem cell self-renewal through suppression of Wnt-beta-catenin signaling. *Nat Genet*, **36**, 1117-21.

Hebestreit, H., Dibbert, B., Balatti, I., Braun, D., Schapowal, A., Blaser, K. and Simon, H.U. (1998). Disruption of fas receptor signaling by nitric oxide in eosinophils. *J Exp Med*, **187**, 415-25.

Hedgepeth, C.M., Conrad, L.J., Zhang, J., Huang, H.C., Lee, V.M. and Klein, P.S. (1997). Activation of the Wnt signaling pathway: a molecular mechanism for lithium action. *Dev Biol*, **185**, 82-91.

Henderson, J.K., Draper, J.S., Baillie, H.S., Fishel, S., Thomson, J.A., Moore, H. and Andrews, P.W. (2002). Preimplantation human embryos and embryonic stem cells show comparable expression of stage-specific embryonic antigens. *Stem Cells*, **20**, 329-337.

Hertz, M.I., Taylor, D.O., Trulock, E.P., Boucek, M.M., Mohacsi, P.J., Edwards, L.B. and Keck, B.M. (2002). The Registry Of The International Society For Heart And Lung Transplantation: Nineteenth Official Report--2002. *The Journal of Heart and Lung Transplantation*, **21**, 950-970.

Hescheler, J., Meyer, R., Plant, S., Krautwurst, D., Rosenthal, W. and Schultz, G. (1991). Morphological, biochemical, and electrophysiological characterization of a clonal cell (H9c2) line from rat heart. *Circ Res*, **69**, 1476-1486.

Hibi, M., Murakami, M., Saito, M., Hirano, T., Taga, T. and Kishimoto, T. (1990). Molecular cloning and expression of an IL-6 signal transducer, gp130. *Cell*, **63**, 1149-57.

Hikiji, H., Shin, W.S., Oida, S., Takato, T., Koizumi, T. and Toyo-oka, T. (1997). Direct action of nitric oxide on osteoblastic differentiation. *FEBS Letters*, **410**, 238-242.

Hilton, D.J., Richardson, R.T., Alexander, W.S., Viney, E.M., Willson, T.A., Sprigg, N.S., Starr, R., Nicholson, S.E., Metcalf, D. and Nicola, N.A. (1998). Twenty proteins containing a C-terminal SOCS box form five structural classes. *Proc Natl Acad Sci U S A*, **95**, 114-9.

Hinz, M., Krappmann, D., Eichten, A., Heder, A., Scheidereit, C. and Strauss, M. (1999). NF-kappa B Function in Growth Control: Regulation of Cyclin D1 Expression and G0/G1-to-S-Phase Transition. *Mol. Cell. Biol.*, **19**, 2690-2698.

Hiroi, Y., Kudoh, S., Monzen, K., Ikeda, Y., Yazaki, Y., Nagai, R. and Komuro, I. (2001). Tbx5 associates with Nkx2-5 and synergistically promotes cardiomyocyte differentiation. *Nature Genetics*, **28**, 276-280.

Hodgson, D.M., Behfar, A., Zingman, L.V., Kane, G.C., Perez-Terzic, C., Alekseev, A.E., Puceat, M. and Terzic, A. (2004). Stable benefit of embryonic stem cell therapy in myocardial infarction. *Am J Physiol Heart Circ Physiol*, **287**, H471-479.

Hoffman, J.I.E. and Kaplan, S. (2002). The incidence of congenital heart

disease. Journal of the American College of Cardiology, **39**, 1890-1900.

Holtzinger, A. and Evans, T. (2007). Gata5 and Gata6 are functionally redundant in zebrafish for specification of cardiomyocytes. *Dev Biol*, **312**, 613-22.

Hong, F., Kwon, S.J., Jhun, B.S., Kim, S.S., Ha, J., Kim, S.-J., Sohn, N.W., Kang, C. and Kang, I. (2001). Insulin-like growth factor-1 protects H9c2 cardiac myoblasts from oxidative stress-induced apoptosis via phosphatidylinositol 3-kinase and extracellular signal-regulated kinase pathways. *Life Sciences*, **68**, 1095-1105.

Hotta, Y., Otsuka-Murakami, H., Fujita, M., Nakagawa, J., Yajima, M., Liu, W., Ishikawa, N., Kawai, N., Masumizu, T. and Kohno, M. (1999). Protective role of nitric oxide synthase against ischemia-reperfusion injury in guinea pig myocardial mitochondria. *Eur J Pharmacol*, **380**, 37-48.

House, C. and Kemp, B.E. (1987). Protein kinase C contains a pseudosubstrate prototope in its regulatory domain. *Science*, **238**, 1726-8.

House, C. and Kemp, B.E. (1990). Protein kinase C pseudosubstrate prototope: structure-function relationships. *Cell Signal*, **2**, 187-90.

Hovatta, O., Mikkola, M., Gertow, K., Stromberg, A.-M., Inzunza, J., Hreinsson, J., Rozell, B., Blennow, E., Andang, M. and Ahrlund-Richter, L. (2003). A culture system using human foreskin fibroblasts as feeder cells allows production of human embryonic stem cells. *Hum. Reprod.*, **18**, 1404-1409.

Hsieh, J.-C., Rattner, A., Smallwood, P.M. and Nathans, J. (1999). Biochemical characterization of Wnt-Frizzled interactions using a soluble, biologically active vertebrate Wnt protein. *PNAS*, **96**, 3546-3551.

Hu, T., Yamagishi, H., Maeda, J., McAnally, J., Yamagishi, C. and Srivastava, D. (2004). Tbx1 regulates fibroblast growth factors in the anterior heart field through a reinforcing autoregulatory loop involving forkhead transcription factors. *Development*, **131**, 5491-5502.

Hubert, H.B., Feinleib, M., McNamara, P.M. and Castelli, W.P. (1983). Obesity as an independent risk factor for cardiovascular disease: a 26-year follow-up of participants in the Framingham Heart Study. *Circulation*, **67**, 968-77.

Hunter, A.L., Zhang, J., Chen, S.C., Si, X., Wong, B., Ekhterae, D., Luo, H. and Granville, D.J. (2007). Apoptosis repressor with caspase recruitment domain (ARC) inhibits myogenic differentiation. *FEBS Letters*, **581**, 879-884.

Hurley, J.H., Newton, A.C., Parker, P.J., Blumberg, P.M. and Nishizuka, Y. (1997). Taxonomy and function of C1 protein kinase C homology domains. *Protein Sci*, **6**, 477-480.

ladecola, C., Pelligrino, D.A., Moskowitz, M.A. and Lassen, N.A. (1994). Nitric oxide synthase inhibition and cerebrovascular regulation. *J Cereb Blood Flow Metab*, **14**, 175-92.

Ibuki, Y., Mizuno, S. and Goto, R. (2003). [gamma]-Irradiation-induced DNA damage enhances NO production via NF-[kappa]B activation in RAW264.7 cells. *Biochimica et Biophysica Acta (BBA) - Molecular Cell Research*, **1593**, 159-167.

Inga, A., Reamon-Buettner, S.M., Borlak, J. and Resnick, M.A. (2005). Functional dissection of sequence-specific NKX2-5 DNA binding domain mutations associated with human heart septation defects using a yeast-based system. *Hum Mol Genet*, **14**, 1965-75.

Ip, H.S., Wilson, D.B., Heikinheimo, M., Tang, Z., Ting, C.N., Simon, M.C., Leiden, J.M. and Parmacek, M.S. (1994). The GATA-4 transcription factor transactivates the cardiac muscle-specific troponin C promoter-enhancer in nonmuscle cells. *Molecular and Cellular Biology*, **14**, 7517-7526.

Ip, N.Y., Nye, S.H., Boulton, T.G., Davis, S., Taga, T., Li, Y., Birren, S.J., Yasukawa, K., Kishimoto, T., Anderson, D.J., Stahl, N. and Yancopoulos, G.D. (1992). CNTF and LIF act on neuronal cells via shared signaling pathways that involve the IL-6 signal transducing receptor component gp130. *Cell*, **69**, 1121-1132.

Itoh, G., Tamura, J. and Suzuki, M. (1995). DNA fragmentation of human infarcted myocardial cells demonstrated by the nick end labeling method and DNA agarose gel electrophoresis. *Am J Pathol*, **146**, 1325-1331.

Itskovitz-Eldor, J., Schuldiner, M., Karsenti, D., Eden, A., Yanuka, O., Amit, M., Soreq, H. and Benvenisty, N. (2000). Differentiation of human embryonic stem cells into embryoid bodies compromising the three embryonic germ layers. *Molecular medicine (Cambridge, Mass.)*, **6**, 88-95.

Izadpanah, R., Trygg, C., Patel, B., Kriedt, C., Dufour, J., Gimble, J.M. and Bunnell, B.A. (2006). Biologic properties of mesenchymal stem cells derived from bone marrow and adipose tissue. *J Cell Biochem*, **99**, 1285-97.

Jackson, K.A., Mi, T. and Goodell, M.A. (1999). From the Cover: Hematopoietic potential of stem cells isolated from murine skeletal muscle. *PNAS*, **96**, 14482-14486.

Jacob, S.W. and Herschler, R. (1986). Pharmacology of DMSO. *Cryobiology*, **23**, 14-27.

Jadlowiec, J., Koch, H., Zhang, X., Campbell, P.G., Seyedain, M. and Sfeir, C. (2004). Phosphophoryn Regulates the Gene Expression and Differentiation of NIH3T3, MC3T3-E1, and Human Mesenchymal Stem Cells via the Integrin/MAPK Signaling Pathway. *J. Biol. Chem.*, **279**, 53323-53330.

Jamali, M., Rogerson, P.J., Wilton, S. and Skerjanc, I.S. (2001). Nkx2-5 Activity Is Essential for Cardiomyogenesis. *J. Biol. Chem.*, **276**, 42252-42258.

Jan, M.S., Liu, H.S. and Lin, Y.S. (1999). Bad overexpression sensitizes NIH/3T3 cells to undergo apoptosis which involves caspase activation and ERK inactivation. *Biochem Biophys Res Commun*, **264**, 724-9.

Jaspard, B., Couffinhal, T., Dufourcq, P., Moreau, C. and Duplaa, C. (2000). Expression pattern of mouse sFRP-1 and mWnt-8 gene during heart morphogenesis. *Mech Dev*, **90**, 263-7.

Jenum, A.K., Anderssen, S.A., Birkeland, K.I., Holme, I., Graff-Iversen, S., Lorentzen, C., Ommundsen, Y., Raastad, T., Odegaard, A.K. and Bahr, R. (2006). Promoting Physical Activity in a Low-Income Multiethnic District: Effects of a Community Intervention Study to Reduce Risk Factors for Type 2 Diabetes and Cardiovascular Disease: A community intervention reducing inactivity. pp. 1605-1612.

Jeon, E.S., Moon, H.J., Lee, M.J., Song, H.Y., Kim, Y.M., Bae, Y.C., Jung, J.S. and Kim, J.H. (2006). Sphingosylphosphorylcholine induces differentiation of human mesenchymal stem cells into smooth-muscle-like cells through a TGF-{beta}-dependent mechanism. *J Cell Sci*, **119**, 4994-5005.

Jerome, L.A. and Papaioannou, V.E. (2001). DiGeorge syndrome phenotype in mice mutant for the T-box gene, Tbx1. *Nat Genet*, **27**, 286-291.

Jia, G., Cheng, G., Gangahar, D.M. and Agrawal, D.K. (2008). Involvement of connexin 43 in angiotensin II-induced migration and proliferation of saphenous vein smooth muscle cells via the MAPK-AP-1 signaling pathway. *J Mol Cell Cardiol* **44**: 882-90.

Jiang, Q.S., Huang, X.N., Dai, Z.K., Yang, G.Z., Zhou, Q.X., Shi, J.S. and Wu, Q. (2007). Inhibitory effect of ginsenoside Rb1 on cardiac hypertrophy induced by monocrotaline in rat. *J Ethnopharmacol*, **111**, 567-72.

Jiang, Y. and Evans, T. (1996). The Xenopus GATA-4/5/6 genes are associated with cardiac specification and can regulate cardiac-specific transcription during embryogenesis. *Dev Biol*, **174**, 258-70.

Johnson, J.E., Giorgione, J. and Newton, A.C. (2000). The C1 and C2 Domains of Protein Kinase C Are Independent Membrane Targeting Modules, with Specificity for Phosphatidylserine Conferred by the C1 Domain. *Biochemistry*, **39**, 11360-11369.

Jones, P.A. and Taylor, S.M. (1980). Cellular differentiation, cytidine analogs and DNA methylation. *Cell*, **20**, 85-93.

Jones, P.A., Taylor, S.M. and Wilson, V. (1983). DNA modification,

differentiation, and transformation. J Exp Zool, 228, 287-95.

Jones, P.F., Jakubowicz, T., Pitossi, F.J., Maurer, F. and Hemmings, B.A. (1991). Molecular Cloning and Identification of a Serine/Threonine Protein Kinase of the Second-Messenger Subfamily. *Proc Natl Acad Sci U S A*, **88**, 4171-4175.

Jousilahti, P., Vartiainen, E., Tuomilehto, J. and Puska, P. (1999). Sex, Age, Cardiovascular Risk Factors, and Coronary Heart Disease : A Prospective Follow-Up Study of 14 786 Middle-Aged Men and Women in Finland. pp. 1165-1172.

Kageyama, K., Ihara, Y., Goto, S., Urata, Y., Toda, G., Yano, K. and Kondo, T. (2002). Overexpression of calreticulin modulates protein kinase B/Akt signaling to promote apoptosis during cardiac differentiation of cardiomyoblast H9c2 cells. *J Biol Chem*, **277**, 19255-64.

Kaltschmidt, B., Uherek, M., Wellmann, H., Volk, B. and Kaltschmidt, C. (1999). Inhibition of NF-kappa B potentiates amyloid beta -mediated neuronal apoptosis. *Proc Natl Acad Sci U S A*, **96**, 9409-9414.

Kanai, M., Konda, Y., Nakajima, T., Izumi, Y., Kanda, N., Nanakin, A., Kubohara, Y. and Chiba, T. (2003). Differentiation-inducing factor-1 (DIF-1) inhibits STAT3 activity involved in gastric cancer cell proliferation via MEK-ERK-dependent pathway. *Oncogene*, **22**, 548-54.

Kanakaraj, P., Schafer, P.H., Cavender, D.E., Wu, Y., Ngo, K., Grealish, P.F., Wadsworth, S.A., Peterson, P.A., Siekierka, J.J., Harris, C.A. and Fung-Leung, W.P. (1998). Interleukin (IL)-1 receptor-associated kinase (IRAK) requirement for optimal induction of multiple IL-1 signaling pathways and IL-6 production. *Journal of Experimental Medicine*, **187**, 2073-2079.

Kanno, S., Kim, P.K.M., Sallam, K., Lei, J., Billiar, T.R. and Shears, L.L., II (2004). Nitric oxide facilitates cardiomyogenesis in mouse embryonic stem cells. *PNAS*, **101**, 12277-12281.

Kaprio, J., Kujala, U.M., Koskenvuo, M. and Sarna, S. (2000). Physical activity and other risk factors in male twin-pairs discordant for coronary heart disease. *Atherosclerosis*, **150**, 193-200.

Karasawa, T., Yokokura, H., Kitajewski, J. and Lombroso, P.J. (2002). Frizzled-9 Is Activated by Wnt-2 and Functions in Wnt/beta -Catenin Signaling. *J. Biol. Chem.*, **277**, 37479-37486.

Kasahara, H., Lee, B., Schott, J.J., Benson, D.W., Seidman, J.G., Seidman, C.E. and Izumo, S. (2000). Loss of function and inhibitory effects of human CSX/NKX2.5 homeoprotein mutations associated with congenital heart disease. *J Clin Invest*, **106**, 299-308.

Kashour, T., Burton, T., Dibrov, A. and Amara, F. (2003). Myogenic signaling

by lithium in cardiomyoblasts is Akt independent but requires activation of the [beta]-catenin-Tcf/Lef pathway. *Journal of Molecular and Cellular Cardiology*, **35**, 937-951.

Katayama, Y., Yano, T., Bessho, A., Deguchi, S., Sunami, K., Mahmut, N., Shinagawa, K., Omoto, E., Makino, S., Miyamoto, T., Mizuno, S., Fukuda, T., Eto, T., Fujisaki, T., Ohno, Y., Inaba, S., Niho, Y. and Harada, M. (1997). The effects of a simplified method for cryopreservation and thawing procedures on peripheral blood stem cells. *Bone Marrow Transplant*, **19**, 283-7.

Kauffmann-Zeh, A., Rodriguez-Viciana, P., Ulrich, E., Gilbert, C., Coffer, P., Downward, J. and Evan, G. (1997). Suppression of c-Myc-induced apoptosis by Ras signalling through PI(3)K and PKB. *Nature*, **385**, 544-8.

Kawachi, H., Moriya, N., Korai, T., Tanaka, S.-y., Watanabe, M., Matsui, T., Kawada, T. and Yano, H. (2007). Nitric oxide suppresses preadipocyte differentiation in 3T3-L1 culture. *Molecular and Cellular Biochemistry*, **300**, 61-67.

Kawachi, I., Colditz, G.A., Stampfer, M.J., Willett, W.C., Manson, J.E., Rosner, B., Hunter, D.J., Hennekens, C.H. and Speizer, F.E. (1993). Smoking Cessation in Relation to Total Mortality Rates in Women: A Prospective Cohort Study. pp. 992-1000.

Kawakami, Y., Nishimoto, H., Kitaura, J., Maeda-Yamamoto, M., Kato, R.M., Littman, D.R., Leitges, M., Rawlings, D.J. and Kawakami, T. (2004). Protein kinase C betall regulates Akt phosphorylation on Ser-473 in a cell type- and stimulus-specific fashion. *J Biol Chem*, **279**, 47720-5.

Kehat, I., Kenyagin-Karsenti, D., Snir, M., Segev, H., Amit, M., Gepstein, A., Livne, E., Binah, O., Itskovitz-Eldor, J. and Gepstein, L. (2001). Human embryonic stem cells can differentiate into myocytes with structural and functional properties of cardiomyocytes. *J. Clin. Invest.*, **108**, 407-414.

Kehat, I., Khimovich, L., Caspi, O., Gepstein, A., Shofti, R. and Arbel, G. (2004). Electromechanical integration of cardiomyocytes derived from human embryonic stem cells. *Nat Biotechnol*, **22**, 1282-1289.

Keller, G.M. (1995). In vitro differentiation of embryonic stem cells. *Current Opinion in Cell Biology*, **7**, 862-869.

Kelly, R.G., Brown, N.A. and Buckingham, M.E. (2001). The arterial pole of the mouse heart forms from Fgf10-expressing cells in pharyngeal mesoderm. *Dev Cell*, **1**, 435-40.

Kemp, C., Willems, E., Abdo, S., Lambiv, L. and Leyns, L. (2005). Expression of all Wnt genes and their secreted antagonists during mouse blastocyst and postimplantation development. *Dev Dyn*, **233**, 1064-75.

Kiger, A.A., Jones, D.L., Schulz, C., Rogers, M.B. and Fuller, M.T. (2001).

Stem Cell Self-Renewal Specified by JAK-STAT Activation in Response to a Support Cell Cue. pp. 2542-2545.

Kiley, S.C. and Parker, P.J. (1995). Differential localization of protein kinase C isozymes in U937 cells: evidence for distinct isozyme functions during monocyte differentiation. *J Cell Sci*, **108**, 1003-16.

Kim, H.A., Lee, K.B. and Bae, S.C. (2005). The mechanism of lowconcentration sodium nitroprusside-mediated protection of chondrocyte death. *Arthritis Res Ther*, **7**, R526-35.

Kim, J.M., Yoon, M.Y., Kim, J., Kim, S.S., Kang, I., Ha, J. and Kim, S.S. (1999). Phosphatidylinositol 3-kinase regulates differentiation of H9c2 cardiomyoblasts mainly through the protein kinase B/Akt-independent pathway. *Arch Biochem Biophys*, **367**, 67-73.

Kim, S.H., Johnson, V.J., Shin, T.-Y. and Sharma, R.P. (2004). Selenium Attenuates Lipopolysaccharide-Induced Oxidative Stress Responses Through Modulation of p38 MAPK and NF-{kappa}B Signaling Pathways. *Experimental Biology and Medicine*, **229**, 203-213.

Kim, S.H., Oh, S.M. and Kim, T.S. (2005). Induction of human leukemia HL-60 cell differentiation via a PKC/ERK pathway by helenalin, a pseudoguainolide sesquiterpene lactone. *European Journal of Pharmacology*, **511**, 89-97.

Kim, Y. and Nirenberg, M. (1989). Drosophila NK-homeobox genes. *Proc Natl Acad Sci U S A*, **86**, 7716-7720.

Kim, Y.M., Talanian, R.V. and Billiar, T.R. (1997). Nitric oxide inhibits apoptosis by preventing increases in caspase-3-like activity via two distinct mechanisms. *J Biol Chem*, **272**, 31138-48.

Kispert, A., Koschorz, B. and Herrmann, B.G. (1995). The T protein encoded by Brachyury is a tissue-specific transcription factor. *Embo J*, **14**, 4763-72.

Klein, P.S. and Melton, D.A. (1996). A molecular mechanism for the effect of lithium on development. *Proc Natl Acad Sci U S A*, **93**, 8455-8459.

Klinz, F.J., Bloch, W., Addicks, K. and Hescheler, J. (1999). Inhibition of Phosphatidylinositol-3-kinase Blocks Development of Functional Embryonic Cardiomyocytes. *Experimental Cell Research*, **247**, 79-83.

Klippel, A., Kavanaugh, W.M., Pot, D. and Williams, L.T. (1997). A specific product of phosphatidylinositol 3-kinase directly activates the protein kinase Akt through its pleckstrin homology domain. *Mol. Cell. Biol.*, **17**, 338-344.

Kobayashi, T. and Cohen, P. (1999). Activation of serum- and glucocorticoidregulated protein kinase by agonists that activate phosphatidylinositide 3kinase is mediated by 3-phosphoinositide-dependent protein kinase-1 (PDK1) and PDK2. *Biochem. J.*, **339**, 319-328. Kobayashi, T., Deak, M., Morrice, N. and Cohen, P. (1999). Characterization of the structure and regulation of two novel isoforms of serum- and glucocorticoid-induced protein kinase. *Biochem. J.*, **344**, 189-197.

Komuro, I. and Izumo, S. (1993). Csx: A murine homeobox-containing gene specifically expressed in the developing heart. *Proc Natl Acad Sci U S A*, **90**, 8145-8149.

Kook, S.-H., Choi, K.-C., Son, Y.-O., Lee, K.-Y., Hwang, I.-H., Lee, H.-J., Chung, W.-T., Lee, C.-B., Park, J.-S. and Lee, J.-C. (2008). Involvement of p38 MAPK-mediated signaling in the calpeptin-mediated suppression of myogenic differentiation and fusion in C2C12 cells. *Molecular and Cellular Biochemistry*, **310**, 85-92.

Kosinski, C., Li, V.S., Chan, A.S., Zhang, J., Ho, C., Tsui, W.Y., Chan, T.L., Mifflin, R.C., Powell, D.W., Yuen, S.T., Leung, S.Y. and Chen, X. (2007). Gene expression patterns of human colon tops and basal crypts and BMP antagonists as intestinal stem cell niche factors. *Proc Natl Acad Sci USA*, **104**, 15418-23.

Koutsourakis, M., Langeveld, A., Patient, R., Beddington, R. and Grosveld, F. (1999). The transcription factor GATA6 is essential for early extraembryonic development. *Development*, **126**, 723-732.

Koyama, A., Otsuka, E., Inoue, A., Hirose, S. and Hagiwara, H. (2000). Nitric oxide accelerates the ascorbic acid-induced osteoblastic differentiation of mouse stromal ST2 cells by stimulating the production of prostaglandin E(2). *Eur J Pharmacol*, **391**, 225-31.

Kramer, R.M., Roberts, E.F., Um, S.L., Borsch-Haubold, A.G., Watson, S.P., Fisher, M.J. and Jakubowski, J.A. (1996). p38 mitogen-activated protein kinase phosphorylates cytosolic phospholipase A2 (cPLA2) in thrombin-stimulated platelets. Evidence that proline-directed phosphorylation is not required for mobilization of arachidonic acid by cPLA2. *J Biol Chem*, **271**, 27723-9.

Krebs, D.L. and Hilton, D.J. (2001). SOCS Proteins: Negative Regulators of Cytokine Signaling. *Stem Cells*, **19**, 378-387.

Kruger, H.S., Venter, C.S. and Vorster, H.H. (2003). Physical inactivity as a risk factor for cardiovascular disease in communities undergoing rural to urban transition: the THUSA study. *Cardiovasc J S Afr*, **14**, 16-23, quiz 23, 28.

Krumenacker, J.S., Katsuki, S., Kots, A. and Murad, F. (2006). Differential expression of genes involved in cGMP-dependent nitric oxide signaling in murine embryonic stem (ES) cells and ES cell-derived cardiomyocytes. *Nitric Oxide*, **14**, 1-11.

Krupnick, J.G., Damjanov, I., Danjanov, A., Zhu, Z.M. and Fenderson, B.A.

(1994). Globo-series carbohydrate antigens are expressed in different forms on human and murine teratocarcinoma-derived cells. *International Journal of Cancer*, **59**, 692-698.

Ku, M. and Melton, D.A. (1993). Xwnt-11: a maternally expressed Xenopus wnt gene. *Development*, **119**, 1161-73.

Kubes, P., Suzuki, M. and Granger, D.N. (1991). Nitric oxide: an endogenous modulator of leukocyte adhesion. *Proc Natl Acad Sci U S A*, **88**, 4651-5.

Kudo, M., Wang, Y., Wani, M.A., Xu, M., Ayub, A. and Ashraf, M. (2003). Implantation of bone marrow stem cells reduces the infarction and fibrosis in ischemic mouse heart. *Journal of Molecular and Cellular Cardiology*, **35**, 1113-1119.

Kuemmerle, J.F. (2003). IGF-I elicits growth of human intestinal smooth muscle cells by activation of PI3K, PDK-1, and p70S6 kinase. *Am J Physiol Gastrointest Liver Physiol*, **284**, G411-22.

Kuhl, M., Sheldahl, L.C., Malbon, C.C. and Moon, R.T. (2000). Ca2+/Calmodulin-dependent Protein Kinase II Is Stimulated by Wnt and Frizzled Homologs and Promotes Ventral Cell Fates in Xenopus. *J. Biol. Chem.*, **275**, 12701-12711.

Kuo, C.T., Morrisey, E.E., Anandappa, R., Sigrist, K., Lu, M.M., Parmacek, M.S., Soudais, C. and Leiden, J.M. (1997). GATA4 transcription factor is required for ventral morphogenesis and heart tube formation. *Genes and Development*, **11**, 1048-1060.

Kurata, S.-i. (2000). Selective Activation of p38 MAPK Cascade and Mitotic Arrest Caused by Low Level Oxidative Stress. *J. Biol. Chem.*, **275**, 23413-23416.

Kuroda, T., Tada, M., Kubota, H., Kimura, H., Hatano, S.-y., Suemori, H., Nakatsuji, N. and Tada, T. (2005). Octamer and Sox Elements Are Required for Transcriptional cis Regulation of Nanog Gene Expression. *Mol. Cell. Biol.*, **25**, 2475-2485.

Kwiecinska, P., Roszkiewicz, B., Lokociejewska, M. and Orzechowski, A. (2005). Elevated expression of NF-[kappa]B and Bcl-2 proteins in C2C12 myocytes during myogenesis is affected by PD98059, LY294002 and SB203580. *Cell Biology International*, **29**, 319-331.

Kwon, C., Arnold, J., Hsiao, E.C., Taketo, M.M., Conklin, B.R. and Srivastava, D. (2007). Canonical Wnt signaling is a positive regulator of mammalian cardiac progenitors. *Proc Natl Acad Sci U S A*, **104**, 10894-10899.

Kyaw, M., Yoshizumi, M., Tsuchiya, K., Kirima, K. and Tamaki, T. (2001). Antioxidants inhibit JNK and p38 MAPK activation but not ERK 1/2 activation by angiotensin II in rat aortic smooth muscle cells. *Hypertens Res*, **24**, 251-61. Labbaye, C., Valtieri, M., Barberi, T., Meccia, E., Masella, B., Pelosi, E., Condorelli, G.L., Testa, U. and Peschle, C. (1995). Differential expression and functional role of GATA-2, NF-E2, and GATA-1 in normal adult hematopoiesis. *Journal of Clinical Investigation*, **95**, 2346-2358.

Laflamme, M.A., Chen, K.Y., Naumova, A.V., Muskheli, V., Fugate, J.A., Dupras, S.K., Reinecke, H., Xu, C., Hassanipour, M., Police, S., O'Sullivan, C., Collins, L., Chen, Y., Minami, E., Gill, E.A., Ueno, S., Yuan, C., Gold, J. and Murry, C.E. (2007). Cardiomyocytes derived from human embryonic stem cells in pro-survival factors enhance function of infarcted rat hearts. *Nat Biotech*, **25**, 1015-1024.

Laflamme, M.A., Gold, J., Xu, C., Hassanipour, M., Rosler, E., Police, S., Muskheli, V. and Murry, C.E. (2005). Formation of Human Myocardium in the Rat Heart from Human Embryonic Stem Cells. *Am J Pathol*, **167**, 663-671.

Lam, N., Chesney, M.A. and Kimble, J. (2006). Wnt Signaling and CEH-22/tinman/Nkx2.5 Specify a Stem Cell Niche in C. elegans. *Current Biology*, **16**, 287-295.

Laugwitz, K.L., Moretti, A., Lam, J., Gruber, P., Chen, Y., Woodard, S., Lin, L.Z., Cai, C.L., Lu, M.M., Reth, M., Platoshyn, O., Yuan, J.X., Evans, S. and Chien, K.R. (2005). Postnatal isl1+ cardioblasts enter fully differentiated cardiomyocyte lineages. *Nature*, **433**, 647-53.

Laverriere, A.C., MacNeill, C., Mueller, C., Poelmann, R.E., Burch, J.B. and Evans, T. (1994). GATA-4/5/6, a subfamily of three transcription factors transcribed in developing heart and gut. *J Biol Chem*, **269**, 23177-84.

Lavon, N. and Benvenisty, N. (2003). Differentiation and Genetic Manipulation of Human Embryonic Stem Cells and the Analysis of the Cardiovascular System. *Trends in Cardiovascular Medicine*, **13**, 47-52.

Lawrence, M.C., McGlynn, K., Naziruddin, B., Levy, M.F. and Cobb, M.H. (2007). Inaugural Article: Differential regulation of CHOP-10/GADD153 gene expression by MAPK signaling in pancreatic beta-cells. *Proc Natl Acad Sci U S A*, **104**, 11518-11525.

Le Good, J.A., Ziegler, W.H., Parekh, D.B., Alessi, D.R., Cohen, P. and Parker, P.J. (1998). Protein kinase C isotypes controlled by phosphoinositide 3-kinase through the protein kinase PDK1. *Science*, **281**, 2042-2045.

Lee, K.H., Evans, S., Ruan, T.Y. and Lassar, A.B. (2004). SMAD-mediated modulation of YY1 activity regulates the BMP response and cardiac-specific expression of a GATA4/5/6-dependent chick Nkx2.5 enhancer. *Development*, **131**, 4709-4723.

Lee, K.H., Xu, Q. and Breitbart, R.E. (1996). A new tinman-related gene, nkx2.7, anticipates the expression of nkx2.5 and nkx2.3 in zebrafish heart and

pharyngeal endoderm. Developmental Biology, **180**, 722-731.

Lee, M.E., Temizer, D.H., Clifford, J.A. and Quertermous, T. (1991). Cloning of the GATA-binding protein that regulates endothelin-1 gene expression in endothelial cells. *Journal of Biological Chemistry*, **266**, 16188-16192.

Lee, Y., Shioi, T., Kasahara, H., Izumo, S., Jobe, S.M., Wiese, R.J. and Markham, B.E. (1998). The cardiac tissue-restricted homeobox protein Csx/Nkx2.5 Physically associates with the zinc finger protein GATA4 and cooperatively activates atrial natriuretic factor gene expression. *Molecular and Cellular Biology*, **18**, 3120-3129.

Lefer, D.J., Jones, S.P., Girod, W.G., Baines, A., Grisham, M.B., Cockrell, A.S., Huang, P.L. and Scalia, R. (1999). Leukocyte-endothelial cell interactions in nitric oxide synthase-deficient mice. *Am J Physiol*, **276**, H1943-50.

Lehtinen, S.K., Rahkila, P., Helenius, M., Korhonen, P. and Salminen, A. (1996). Down-Regulation of Transcription Factors AP-1, Sp-1, and NF-[kappa]B Precedes Myocyte Differentiation. *Biochemical and Biophysical Research Communications*, **229**, 36-43.

Lemonnier, J., Ghayor, C., Guicheux, J. and Caverzasio, J. (2004). Protein kinase C-independent activation of protein kinase D is involved in BMP-2-induced activation of stress mitogen-activated protein kinases JNK and p38 and osteoblastic cell differentiation. *J Biol Chem*, **279**, 259-64.

Leor, J., Gerecht, S., Cohen, S., Miller, L., Holbova, R., Ziskind, A., Shachar, M., Feinberg, M.S., Guetta, E. and Itskovitz-Eldor, J. (2007). Human embryonic stem cell transplantation to repair the infarcted myocardium. *Heart*, **93**, 1278-1284.

Levenberg, S., Golub, J.S., Amit, M., Itskovitz-Eldor, J. and Langer, R. (2002). Endothelial cells derived from human embryonic stem cells. *PNAS*, **99**, 4391-4396.

Li, C.C., Dai, R.M., Chen, E. and Longo, D.L. (1994). Phosphorylation of NF-KB1-p50 is involved in NF-kappa B activation and stable DNA binding. *J. Biol. Chem.*, **269**, 30089-30092.

Li, C.Y., Zhan, Y.Q., Xu, C.W., Xu, W.X., Wang, S.Y., Lv, J., Zhou, Y., Yue, P.B., Chen, B. and Yang, X.M. (2004). EDAG regulates the proliferation and differentiation of hematopoietic cells and resists cell apoptosis through the activation of nuclear factor-kappa B. *Cell Death Differ*, **11**, 1299-308.

Li, G., Chen, X., Jang, Y., Wang, J., Xing, X., Yang, W. and Hu, Y. (2002). Obesity, coronary heart disease risk factors and diabetes in Chinese: an approach to the criteria of obesity in the Chinese population. *Obes Rev*, **3**, 167-72.

Li, H. and Forstermann, U. (2000). Nitric oxide in the pathogenesis of vascular disease. *J Pathol*, **190**, 244-54.

Li, H., Yu, B., Zhang, Y., Pan, Z., Xu, W. and Li, H. (2006). Jagged1 protein enhances the differentiation of mesenchymal stem cells into cardiomyocytes. *Biochemical and Biophysical Research Communications*, **341**, 320-325.

Li, J., Bombeck, C.A., Yang, S., Kim, Y.M. and Billiar, T.R. (1999). Nitric oxide suppresses apoptosis via interrupting caspase activation and mitochondrial dysfunction in cultured hepatocytes. *J Biol Chem*, **274**, 17325-33.

Li, Q.Y., Newbury-Ecob, R.A., Terrett, J.A., Yi, C.H., Brook, J.D., Young, I.D., Raeburn, J.A., Wilson, D.I., Curtis, A.R.J., Bullen, P.J., Strachan, T., Gebuhr, T., Law, D.J., Robson, S.C., Bonnet, D., Lyonnet, S. and Buckler, A.J. (1997). Holt-Oram syndrome is caused by mutations in TBX5, a member of the Brachyury (T) gene family. *Nature Genetics*, **15**, 21-29.

Li, Y., Liu, L., Barger, S.W. and Griffin, W.S. (2003). Interleukin-1 mediates pathological effects of microglia on tau phosphorylation and on synaptophysin synthesis in cortical neurons through a p38-MAPK pathway. *J Neurosci*, **23**, 1605-11.

Li, Y., Tennekoon, G.I., Birnbaum, M., Marchionni, M.A. and Rutkowski, J.L. (2001). Neuregulin signaling through a PI3K/Akt/Bad pathway in Schwann cell survival. *Mol Cell Neurosci*, **17**, 761-7.

Liang, Q., Wiese, R.J., Bueno, O.F., Dai, Y.-S., Markham, B.E. and Molkentin, J.D. (2001). The Transcription Factor GATA4 Is Activated by Extracellular Signal-Regulated Kinase 1- and 2-Mediated Phosphorylation of Serine 105 in Cardiomyocytes. *Mol. Cell. Biol.*, **21**, 7460-7469.

Lim, J.Y., Park, S.I., Oh, J.H., Kim, S.M., Jeong, C.H., Jun, J.A., Lee, K.S., Oh, W., Lee, J.K. and Jeun, S.S. (2008). Brain-derived neurotrophic factor stimulates the neural differentiation of human umbilical cord blood-derived mesenchymal stem cells and survival of differentiated cells through MAPK/ERK and PI3K/Akt-dependent signaling pathways. *J Neurosci Res*, **86**, 2168-78.

Lin, H.Y., Wang, X.F., Ng-Eaton, E., Weinberg, R.A. and Lodish, H.F. (1992). Expression cloning of the TGF-beta type II receptor, a functional transmembrane serine/threonine kinase. *Cell*, **68**, 775-85.

Lindsley, R.C., Gill, J.G., Kyba, M., Murphy, T.L. and Murphy, K.M. (2006). Canonical Wnt signaling is required for development of embryonic stem cellderived mesoderm. *Development*, **133**, 3787-3796.

Lints, T.J., Parsons, L.M., Hartley, L., Lyons, I. and Harvey, R.P. (1993). Nkx-2.5: A novel murine homeobox gene expressed in early heart progenitor cells and their myogenic descendants. *Development*, **119**, 419-431.

Liu, A., Hoffman, P.W., Lu, W. and Bai, G. (2004). NF-{kappa}B Site Interacts with Sp Factors and Up-regulates the NR1 Promoter during Neuronal Differentiation. *J. Biol. Chem.*, **279**, 17449-17458.

Liu, C.T., Yang, Y.J., Yin, F., Wang, X., Yu, X.H., Wang, Q.H., Wang, X.L. and Xie, M. (2006). The immunobiological development of human bone marrow mesenchymal stem cells in the course of neuronal differentiation. *Cell Immunol*, **244**, 19-32.

Liu, F., Ventura, F., Doody, J. and Massague, J. (1995). Human type II receptor for bone morphogenic proteins (BMPs): extension of the two-kinase receptor model to the BMPs. *Mol Cell Biol*, **15**, 3479-86.

Liu, S., Liu, H., Pan, Y., Tang, S., Xiong, J., Hui, N., Wang, S., Qi, Z. and Li, L. (2004). Human embryonic germ cells isolation from early stages of post-implantation embryos. *Cell and Tissue Research*, **318**, 525-531.

Liu, Y., Asakura, M., Inoue, H., Nakamura, T., Sano, M., Niu, Z., Chen, M., Schwartz, R.J. and Schneider, M.D. (2007). Sox17 is essential for the specification of cardiac mesoderm in embryonic stem cells. *Proc Natl Acad Sci U S A*, **104**, 3859-64.

Lloyd-Jones, D.M., Larson, M.G., Beiser, A. and Levy, D. (1999). Lifetime risk of developing coronary heart disease. *The Lancet*, **353**, 89-92.

Logan, C.Y. and Nusse, R. (2004). The Wnt signaling pathway in development and disease. *Annual Review of Cell and Developmental Biology*, **20**, 781-810.

Lomonosova, E.E., Kirsch, M., Rauen, U. and de Groot, H. (1998). The critical role of Hepes in SIN-1 cytotoxicity, peroxynitrite versus hydrogen peroxide. *Free Radic Biol Med*, **24**, 522-8.

Loomes, K.M., Taichman, D.B., Glover, C.L., Williams, P.T., Markowitz, J.E., Piccoli, D.A., Baldwin, H.S. and Oakey, R.J. (2002). Characterization of Notch receptor expression in the developing mammalian heart and liver. *Am J Med Genet*, **112**, 181-9.

Louhelainen, M., Vahtola, E., Kaheinen, P., Leskinen, H., Merasto, S., Kyto, V., Finckenberg, P., Colucci, W.S., Levijoki, J., Pollesello, P., Haikala, H. and Mervaala, E.M.A. (2007). Effects of levosimendan on cardiac remodeling and cardiomyocyte apoptosis in hypertensive Dahl//Rapp rats. *Br J Pharmacol*, **150**, 851-861.

Lowry, J.A. and Atchley, W.R. (2000). Molecular evolution of the GATA family of transcription factors: Conservation within the DNA-binding domain. *Journal of Molecular Evolution*, **50**, 103-115.

Lu, F.Z., Fujino, M., Kitazawa, Y., Uyama, T., Hara, Y., Funeshima, N., Jiang, J.Y., Umezawa, A. and Li, X.K. (2005). Characterization and gene transfer in

mesenchymal stem cells derived from human umbilical-cord blood. *J Lab Clin Med*, **146**, 271-8.

Ludolph, D.C. and Konieczny, S.F. (1995). Transcription factor families: Muscling in on the myogenic program. *FASEB Journal*, **9**, 1595-1604.

Lyall, F., Jablonka-Shariff, A., Johnson, R.D., Olson, L.M. and Nelson, D.M. (1998). Gene expression of nitric oxide synthase in cultured human term placental trophoblast during in vitro differentiation. *Placenta*, **19**, 253-60.

Lyons, I., Parsons, L.M., Hartley, L., Li, R., Andrews, J.E., Robb, L. and Harvey, R.P. (1995). Myogenic and morphogenetic defects in the heart tubes of murine embryos lacking the homeo box gene Nkx2-5. *Genes and Development*, **9**, 1654-1666.

Maehama, T. and Dixon, J.E. (1998). The Tumor Suppressor, PTEN/MMAC1, Dephosphorylates the Lipid Second Messenger, Phosphatidylinositol 3,4,5-Trisphosphate. *J. Biol. Chem.*, **273**, 13375-13378.

Makino, S., Fukuda, K., Miyoshi, S., Konishi, F., Kodama, H., Pan, J., Sano, M., Takahashi, T., Hori, S., Abe, H., Hata, J.-i., Umezawa, A. and Ogawa, S. (1999). Cardiomyocytes can be generated from marrow stromal cells in vitro. *J. Clin. Invest.*, **103**, 697-705.

Maltsev, V.A., Rohwedel, J., Hescheler, J. and Wobus, A.M. (1993). Embryonic stem cells differentiate in vitro into cardiomyocytes representing sinusnodal, atrial and ventricular cell types. *Mech Dev*, **44**, 41-50.

Mangi, A.A., Noiseux, N., Kong, D., He, H., Rezvani, M., Ingwall, J.S. and Dzau, V.J. (2003). Mesenchymal stem cells modified with Akt prevent remodeling and restore performance of infarcted hearts. *Nat Med*, **9**, 1195-1201.

Manolio, T.A., Pearson, T.A., Wenger, N.K., Barrett-Connor, E., Payne, G.H. and Harlan, W.R. (1992). Cholesterol and heart disease in older persons and women. Review of an NHLBI workshop. *Ann Epidemiol*, **2**, 161-76.

Mao, B., Wu, W., Davidson, G., Marhold, J., Li, M., Mechler, B.M., Dellus, H., Hoppe, D., Stannek, P., Walter, C., Glinka, A. and Niehrs, C. (2002). Kremen proteins are Dickkopf receptors that regulate Wnt/beta-catenin signalling. *Nature*, **417**, 664-667.

Mao, B., Wu, W., Li, Y., Hoppe, D., Stannek, P., Glinka, A. and Niehrs, C. (2001). LDL-receptor-related protein 6 is a receptor for Dickkopf proteins. *Nature*, **411**, 321-325.

Marchisio, M., Santavenere, E., Paludi, M., Gaspari, A.R., Lanuti, P., Bascelli, A., Di Baldassarre, A., Miscia, S. and Ercolino, E. (2005). Erythroid cell differentiation is characterized by nuclear matrix localization and

phosphorylation of protein kinases C (PKC) alpha, delta, and zeta. *J Cell Physiol*, **205**, 32-36.

Marchisio, M., Santavenere, E., Paludi, M., Gaspari, A.R., Lanuti, P., Bascelli, A., Ercolino, E., Di Baldassarre, A. and Miscia, S. (2005). Erythroid cell differentiation is characterized by nuclear matrix localization and phosphorylation of protein kinases C (PKC) alpha, delta, and zeta. *J Cell Physiol*, **205**, 32-36.

Martin, G.R. (1981). Isolation of a pluripotent cell line from early mouse embryos cultured in medium conditioned by teratocarcinoma stem cells. *Proc Natl Acad Sci U S A*, **78**, 7634-8.

Martiny-Baron, G., Kazanietz, M.G., Mischak, H., Blumberg, P.M., Kochs, G., Hug, H., Marme, D. and Schachtele, C. (1993). Selective inhibition of protein kinase C isozymes by the indolocarbazole Go 6976. *J Biol Chem*, **268**, 9194-7.

Marvin, M.J., Di Rocco, G., Gardiner, A., Bush, S.M. and Lassar, A.B. (2001). Inhibition of Wnt activity induces heart formation from posterior mesoderm. *Genes Dev*, **15**, 316-27.

Masui, S., Nakatake, Y., Toyooka, Y., Shimosato, D., Yagi, R., Takahashi, K., Okochi, H., Okuda, A., Matoba, R., Sharov, A.A., Ko, M.S.H. and Niwa, H. (2007). Pluripotency governed by Sox2 via regulation of Oct3/4 expression in mouse embryonic stem cells. *Nat Cell Biol*, **9**, 625-635.

Mathy-Hartert, M., Mouithys-Mickalad, A., Kohnen, S., Deby-Dupont, G., Lamy, M. and Hans, P. (2000). Effects of propofol on endothelial cells subjected to a peroxynitrite donor (SIN-1). *Anaesthesia*, **55**, 1066-1071.

Matsuda, T., Nakamura, T., Nakao, K., Arai, T., Katsuki, M., Heike, T. and Yokota, T. (1999). STAT3 activation is sufficient to maintain an undifferentiated state of mouse embryonic stem cells. *Embo J*, **18**, 4261-9.

Maurus, D., Heligon, C., Burger-Schwarzler, A., Brandli, A.W. and Kuhl, M. (2005). Noncanonical Wnt-4 signaling and EAF2 are required for eye development in Xenopus laevis. *Embo J*, **24**, 1181-91.

Maxon, M.E. and Herskowitz, I. (2001). Ash1p is a site-specific DNA-binding protein that actively represses transcription. *Proc Natl Acad Sci U S A*, **98**, 1495-500.

Maye, P., Zheng, J., Li, L. and Wu, D. (2004). Multiple mechanisms for Wnt11-mediated repression of the canonical Wnt signaling pathway. *Journal of Biological Chemistry*, **279**, 24659-24665.

McBride, K. and Nemer, M. (2001). Regulation of the ANF and BNP promoters by GATA factors: Lessons learned for cardiac transcription. *Canadian Journal of Physiology and Pharmacology*, **79**, 673-681.

McBurney, M.W., Jones-Villeneuve, E.M., Edwards, M.K. and Anderson, P.J. (1982). Control of muscle and neuronal differentiation in a cultured embryonal carcinoma cell line. *Nature*, **299**, 165-7.

McBurney, M.W. and Rogers, B.J. (1982). Isolation of male embryonal carcinoma cells and their chromosome replication patterns. *Dev Biol*, **89**, 503-8.

McCarthy, S.A., Samuels, M.L., Pritchard, C.A., Abraham, J.A. and McMahon, M. (1995). Rapid induction of heparin-binding epidermal growth factor/diphtheria toxin receptor expression by Raf and Ras oncogenes. *Genes and Development*, **9**, 1953-1964.

McFarland, D.C. and Pesall, J.E. (2008). Phospho-MAPK as a marker of myogenic satellite cell responsiveness to growth factors. *Comparative Biochemistry and Physiology Part B: Biochemistry and Molecular Biology*, **149**, 463-467.

McGrew, M.J., Bogdanova, N., Hasegawa, K., Hughes, S.H., Kitsis, R.N. and Rosenthal, N. (1996). Distinct gene expression patterns in skeletal and cardiac muscle are dependent on common regulatory sequences in the MLC1/3 locus. *Mol. Cell. Biol.*, **16**, 4524-4534.

McManus, E.J., Sakamoto, K., Armit, L.J., Ronaldson, L., Shpiro, N., Marquez, R. and Alessi, D.R. (2005). Role that phosphorylation of GSK3 plays in insulin and Wnt signalling defined by knockin analysis. *Embo J*, **24**, 1571-83.

Meilhac, S.M., Esner, M., Kelly, R.G., Nicolas, J.F. and Buckingham, M.E. (2004). The clonal origin of myocardial cells in different regions of the embryonic mouse heart. *Developmental Cell*, **6**, 685-698.

Mello, C.C., Schubert, C., Draper, B., Zhang, W., Lobel, R. and Priess, J.R. (1996). The PIE-1 protein and germline specification in C. elegans embryos. *Nature*, **382**, 710-712.

Melloni, E., Pontremoli, S., Michetti, M., Sacco, O., Cakiroglu, A.G., Jackson, J.F., Rifkind, R.A. and Marks, P.A. (1987). Protein kinase C activity and hexamethylenebisacetamide-induced erythroleukemia cell differentiation. *Proc Natl Acad Sci U S A*, **84**, 5282-5286.

Menard, C., Pupier, S., Mornet, D., Kitzmann, M., Nargeot, J. and Lory, P. (1999). Modulation of L-type calcium channel expression during retinoic acidinduced differentiation of H9C2 cardiac cells. *J Biol Chem*, **274**, 29063-70.

Merika, M. and Orkin, S.H. (1993). DNA-binding specificity of GATA family transcription factors. *Molecular and Cellular Biology*, **13**, 3999-4010.

Miagkov, A.V., Kovalenko, D.V., Brown, C.E., Didsbury, J.R., Cogswell, J.P.,

Stimpson, S.A., Baldwin, A.S. and Makarov, S.S. (1998). NF-kappaB activation provides the potential link between inflammation and hyperplasia in the arthritic joint. *Proc Natl Acad Sci U S A*, **95**, 13859-64.

Michurina, T., Krasnov, P., Balazs, A., Nakaya, N., Vasilieva, T., Kuzin, B., Khrushchov, N., Mulligan, R.C. and Enikolopov, G. (2004). Nitric oxide is a regulator of hematopoietic stem cell activity. *Mol Ther*, **10**, 241-8.

Min, I.M., Pietramaggiori, G., Kim, F.S., Passegué, E., Stevenson, K.E. and Wagers, A.J. (2008). The Transcription Factor EGR1 Controls Both the Proliferation and Localization of Hematopoietic Stem Cells. *Cell Stem Cell*, **2**, 380-391.

Minami, M., Inoue, M., Wei, S., Takeda, K., Matsumoto, M., Kishimoto, T. and Akira, S. (1996). STAT3 activation is a critical step in gp130-mediated terminal differentiation and growth arrest of a myeloid cell line. *Proc Natl Acad Sci U S A*, **93**, 3963-6.

Miskolczi-McCallum, C.M., Scavetta, R.J., Svendsen, P.C., Soanes, K.H. and Brook, W.J. (2005). The Drosophila melanogaster T-box genes midline and H15 are conserved regulators of heart development. *Developmental Biology*, **278**, 459-472.

Mistry, D.K. and Garland, C.J. (1998). Nitric oxide (NO)-induced activation of large conductance Ca2+-dependent K+ channels (BKCa) in smooth muscle cells isolated from the rat mesenteric artery. *Br J Pharmacol*, **124**, 1131-1140.

Mitsui, K., Tokuzawa, Y., Itoh, H., Segawa, K., Murakami, M., Takahashi, K., Maruyama, M., Maeda, M. and Yamanaka, S. (2003). The homeoprotein Nanog is required for maintenance of pluripotency in mouse epiblast and ES cells. *Cell*, **113**, 631-42.

Mizukami, Y., Hirata, T. and Yoshida, K. (1997). Nuclear translocation of PKC zeta during ischemia and its inhibition by wortmannin, an inhibitor of phosphatidylinositol 3-kinase. *FEBS Lett*, **401**, 247-51.

Mjaatvedt, C.H., Nakaoka, T., Moreno-Rodriguez, R., Norris, R.A., Kern, M.J., Eisenberg, C.A., Turner, D. and Markwald, R.R. (2001). The outflow tract of the heart is recruited from a novel heart-forming field. *Dev Biol*, **238**, 97-109.

Mochizuki, K., Sakaguchi, N., Takabe, S. and Goda, T. (2007). Dephosphorylation of TR[alpha]-1 by p44/42 MAPK inhibition enhances T3mediated GLUT5 gene expression in the intestinal cell line Caco-2 cells. *Biochemical and Biophysical Research Communications*, **359**, 979-984.

Molenaar, M., van de Wetering, M., Oosterwegel, M., Peterson-Maduro, J., Godsave, S., Korinek, V., Roose, J., Destree, O. and Clevers, H. (1996). XTcf-3 Transcription Factor Mediates [beta]-Catenin-Induced Axis Formation in Xenopus Embryos. *Cell*, **86**, 391-399.

Molkentin, J.D., Lin, Q., Duncan, S.A. and Olson, E.N. (1997). Requirement of the transcription factor GATA4 for heart tube formation and ventral morphogenesis. *Genes Dev*, **11**, 1061-72.

Monje, P., Marinissen, M.J. and Gutkind, J.S. (2003). Phosphorylation of the carboxyl-terminal transactivation domain of c-Fos by extracellular signal-regulated kinase mediates the transcriptional activation of AP-1 and cellular transformation induced by platelet-derived growth factor. *Mol Cell Biol*, **23**, 7030-43.

Monkley, S.J., Delaney, S.J., Pennisi, D.J., Christiansen, J.H. and Wainwright, B.J. (1996). Targeted disruption of the Wnt2 gene results in placentation defects. *Development*, **122**, 3343-53.

Monzen, K., Hiroi, Y., Kudoh, S., Akazawa, H., Oka, T., Takimoto, E., Hayashi, D., Hosoda, T., Kawabata, M., Miyazono, K., Ishii, S., Yazaki, Y., Nagai, R. and Komuro, I. (2001). Smads, TAK1, and their common target ATF-2 play a critical role in cardiomyocyte differentiation. *Journal of Cell Biology*, **153**, 687-698.

Monzen, K., Shiojima, I., Hiroi, Y., Kudoh, S., Oka, T., Takimoto, E., Hayashi, D., Hosoda, T., Habara-Ohkubo, A., Nakaoka, T., Fujita, T., Yazaki, Y. and Komuro, I. (1999). Bone Morphogenetic Proteins Induce Cardiomyocyte Differentiation through the Mitogen-Activated Protein Kinase Kinase Kinase TAK1 and Cardiac Transcription Factors Csx/Nkx-2.5 and GATA-4. *Mol. Cell. Biol.*, **19**, 7096-7105.

Moore, J.C., Spijker, R., Martens, A.C., de Boer, T., Rook, M.B., van der Heyden, M.A., Tertoolen, L.G. and Mummery, C.L. (2004). A P19Cl6 GFP reporter line to quantify cardiomyocyte differentiation of stem cells. *Int J Dev Biol*, **48**, 47-55.

Moore, M.L., Wang, G.L., Belaguli, N.S., Schwartz, R.J. and McMillin, J.B. (2001). GATA-4 and serum response factor regulate transcription of the muscle-specific carnitine palmitoyltransferase I beta in rat heart. *J Biol Chem*, **276**, 1026-33.

Moretti, A., Lam, J., Evans, S.M. and Laugwitz, K.L. (2007). Cardiovascular development: towards biomedical applicability : Biology of IsI1 (+) cardiac progenitor cells in development and disease. *Cell Mol Life Sci*, **64**, 674-82.

Morin, S., Charron, F., Robitaille, L. and Nemer, M. (2000). GATA-dependent recruitment of MEF2 proteins to target promoters. *Embo J*, **19**, 2046-55.

Moriuchi, A., Hirono, S., Ido, A., Ochiai, T., Nakama, T., Uto, H., Hori, T., Hayashi, K. and Tsubouchi, H. (2001). Additive and Inhibitory Effects of Simultaneous Treatment with Growth Factors on DNA Synthesis through MAPK Pathway and G1 Cyclins in Rat Hepatocytes. *Biochemical and Biophysical Research Communications*, **280**, 368-373.

Moriwaki, J., Kajita, E., Kirikoshi, H., Koike, J., Sagara, N., Yasuhiko, Y., Saitoh, T., Hirai, M., Katoh, M. and Shiokawa, K. (2000). Isolation of Xenopus frizzled-10A and frizzled-10B genomic clones and their expression in adult tissues and embryos. *Biochem Biophys Res Commun*, **278**, 377-84.

Morooka, T. and Nishida, E. (1998). Requirement of p38 Mitogen-activated Protein Kinase for Neuronal Differentiation in PC12 Cells. *J. Biol. Chem.*, **273**, 24285-24288.

Morrisey, E.E., Ip, H.S., Lu, M.M. and Parmacek, M.S. (1996). GATA-6: A zinc finger transcription factor that is expressed in multiple cell lineages derived from lateral mesoderm. *Developmental Biology*, **177**, 309-322.

Mosior, M. and Epand, R.M. (1993). Mechanism of activation of protein kinase C: roles of diolein and phosphatidylserine. *Biochemistry*, **32**, 66-75.

Mosmann, T. (1983). Rapid colorimetric assay for cellular growth and survival: Application to proliferation and cytotoxicity assays. *Journal of Immunological Methods*, **65**, 55-63.

Mouthon, M.A., Bernard, O., Mitjavila, M.T., Romeo, P.H., Vainchenker, W. and Mathieu-Mahul, D. (1993). Expression of tal-1 and GATA-binding proteins during human hematopoiesis. *Blood*, **81**, 647-655.

Muller, J.M., Cahill, M.A., Rupec, R.A., Baeuerle, P.A. and Nordheim, A. (1997). Antioxidants as well as oxidants activate c-fos via Ras-dependent activation of extracellular-signal-regulated kinase 2 and Elk-1. *Eur J Biochem*, **244**, 45-52.

Muller-Ehmsen, J., Peterson, K.L., Kedes, L., Whittaker, P., Dow, J.S., Long, T.I., Laird, P.W. and Kloner, R.A. (2002). Rebuilding a damaged heart: long-term survival of transplanted neonatal rat cardiomyocytes after myocardial infarction and effect on cardiac function. *Circulation*, **105**, 1720-6.

Müller-Ehmsen, J., Peterson, K.L., Kedes, L., Whittaker, P., Dow, J.S., Long, T.I., Laird, P.W. and Kloner, R.A. (2002). Rebuilding a damaged heart: Long-term survival of transplanted neonatal rat cardiomyocytes after myocardial infarction and effect on cardiac function. *Circulation*, **105**, 1720-1726.

Mummery, C., Ward-van Oostwaard, D., Doevendans, P., Spijker, R., van den, B.S. and Hassink, R. (2003). Differentiation of human embryonic stem cells to cardiomyocytes: role of coculture with visceral endoderm-like cells. *Circulation*, **107**, 2733-2740.

Murakami, M., Hibi, M., Nakagawa, N., Nakagawa, T., Yasukawa, K., Yamanishi, K., Taga, T. and Kishimoto, T. (1993). IL-6-induced homodimerization of gp130 and associated activation of a tyrosine kinase. *Science*, **260**, 1808-1810.

Murphy, A.M., Thompson, W.R., Peng, L.F. and Jones, L. (1997). Regulation

of the rat cardiac troponin I gene by the transcription factor GATA-4. *Biochem. J.*, **322**, 393-384.

Murray, C.J. and Lopez, A.D. (1997). Mortality by cause for eight regions of the world: Global Burden of Disease Study. *Lancet*, **349**, 1269-76.

Musashi, M. (2000). The role of protein kinase c isoforms in cell proliferation and apoptosis. *International Journal of Hematology*, **72**, 12-19.

Nagy, A., Gocza, E., Diaz, E.M., Prideaux, V.R., Ivanyi, E., Markkula, M. and Rossant, J. (1990). Embryonic stem cells alone are able to support fetal development in the mouse. *Development*, **110**, 815-21.

Naiche, L.A., Harrelson, Z., Kelly, R.G. and Papaioannou, V.E. (2005). T-box genes in vertebrate development. *Annual Review of Genetics*, **39**, 219-239.

Naito, A.T., Shiojima, I., Akazawa, H., Hidaka, K., Morisaki, T., Kikuchi, A. and Komuro, I. (2006). Developmental stage-specific biphasic roles of Wnt/betacatenin signaling in cardiomyogenesis and hematopoiesis. *Proc Natl Acad Sci U S A*, **103**, 19812-7.

Naito, A.T., Tominaga, A., Oyamada, M., Oyamada, Y., Shiraishi, I., Monzen, K., Komuro, I. and Takamatsu, T. (2003). Early stage-specific inhibitions of cardiomyocyte differentiation and expression of Csx/Nkx-2.5 and GATA-4 by phosphatidylinositol 3-kinase inhibitor LY294002. *Experimental Cell Research*, **291**, 56-69.

Nakagawa, H., Yoshida, M. and Miyamoto, S. (2000). Nitric oxide underlies the differentiation of PC12 cells induced by depolarization with high KCI. *J Biochem (Tokyo)*, **127**, 113-9.

Nakajima, K., Yamanaka, Y., Nakae, K., Kojima, H., Ichiba, M., Kiuchi, N., Kitaoka, T., Fukada, T., Hibi, M. and Hirano, T. (1996). A central role for Stat3 in IL-6-induced regulation of growth and differentiation in M1 leukemia cells. *Embo J*, **15**, 3651-8.

Nakamura, T., Arai, T., Takagi, M., Sawada, T., Matsuda, T., Yokota, T. and Heike, T. (1998). A selective switch-on system for self-renewal of embryonic stem cells using chimeric cytokine receptors. *Biochem Biophys Res Commun*, **248**, 22-7.

Nakamura, T., Sano, M., Songyang, Z. and Schneider, M.D. (2003). A Wntand beta -catenin-dependent pathway for mammalian cardiac myogenesis. *Proc Natl Acad Sci U S A*, **100**, 5834-9.

Nakao, T., Shiota, M., Tatemoto, Y., Izumi, Y. and Iwao, H. (2007). Pravastatin induces rat aortic endothelial cell proliferation and migration via activation of PI3K/Akt/mTOR/p70 S6 kinase signaling. *J Pharmacol Sci*, **105**, 334-41.

Nalefski, E.A. and Newton, A.C. (2001). Membrane Binding Kinetics of Protein Kinase C βII Mediated by the C2 Domain. *Biochemistry*, **40**, 13216-13229.

Narazaki, M., Witthuhn, B.A., Yoshida, K., Silvennoinen, O., Yasukawa, K., Ihle, J.N., Kishimoto, T. and Taga, T. (1994). Activation of JAK2 Kinase Mediated by the Interleukin 6 Signal Transducer gp130. *Proc Natl Acad Sci U S A*, **91**, 2285-2289.

Natarajan, K., Singh, S., Burke, T.R., Jr., Grunberger, D. and Aggarwal, B.B. (1996). Caffeic acid phenethyl ester is a potent and specific inhibitor of activation of nuclear transcription factor NF-kappa B. *Proc Natl Acad Sci U S A*, **93**, 9090-5.

Naumann, M. and Scheidereit, C. (1994). Activation of NF-kappa B in vivo is regulated by multiple phosphorylations. *EMBO Journal*, **13**, 4597-4607.

Navas-Nacher, E.L., Colangelo, L., Beam, C. and Greenland, P. (2001). Risk Factors for Coronary Heart Disease in Men 18 to 39 Years of Age. pp. 433-439.

Nelson, T.J., Ge, Z.D., Van Orman, J., Barron, M., Rudy-Reil, D., Hacker, T.A., Misra, R., Duncan, S.A., Auchampach, J.A. and Lough, J.W. (2006). Improved cardiac function in infarcted mice after treatment with pluripotent embryonic stem cells. *Anat Rec A Discov Mol Cell Evol Biol*, **288**, 1216-24.

Neri, L.M., Marchisio, M., Colamussi, M.L. and Bertagnolo, V. (1999). Monocytic differentiation of HL-60 cells is characterized by the nuclear translocation of phosphatidylinositol 3-kinase and of definite phosphatidylinositol-specific phospholipase C isoforms. *Biochem Biophys Res Commun*, **259**, 314-20.

Newman, C.S. and Krieg, P.A. (1998). tinman-related genes expressed during heart development in Xenopus. *Developmental Genetics*, **22**, 230-238.

Newton, A.C. (2003). Regulation of the ABC kinases by phosphorylation: protein kinase C as a paradigm. *Biochem. J.*, **370**, 361-371.

Nichols, J. (1998). Formation of pluripotent stem cells in the mammalian embryo depends on the POU transcription factor Oct-4. *Cell*, **95**, 379-391.

Nichols, J., Chambers, I. and Smith, A. (1994). Derivation of germline competent embryonic stem cells with a combination of interleukin-6 and soluble interleukin-6 receptor. *Experimental Cell Research*, **215**, 237-239.

Nicholson, S.E., De Souza, D., Fabri, L.J., Corbin, J., Willson, T.A., Zhang, J.-G., Silva, A., Asimakis, M., Farley, A., Nash, A.D., Metcalf, D., Hilton, D.J., Nicola, N.A. and Baca, M. (2000). Suppressor of cytokine signaling-3 preferentially binds to the SHP-2-binding site on the shared cytokine receptor subunit gp130. *Proc Natl Acad Sci U S A*, **97**, 6493-6498.

Nicholson, S.E., Willson, T.A., Farley, A., Starr, R., Zhang, J.G., Baca, M., Alexander, W.S., Metcalf, D., Hilton, D.J. and Nicola, N.A. (1999). Mutational analyses of the SOCS proteins suggest a dual domain requirement but distinct mechanisms for inhibition of LIF and IL-6 signal transduction. *Embo J*, **18**, 375-85.

Nick, J.A., Avdi, N.J., Young, S.K., Lehman, L.A., McDonald, P.P., Frasch, S.C., Billstrom, M.A., Henson, P.M., Johnson, G.L. and Worthen, G.S. (1999). Selective activation and functional significance of p38alpha mitogen-activated protein kinase in lipopolysaccharide-stimulated neutrophils. *J Clin Invest*, **103**, 851-8.

Niedbala, W., Wei, X.-q., Campbell, C., Thomson, D., Komai-Koma, M. and Liew, F.Y. (2002). Nitric oxide preferentially induces type 1 T cell differentiation by selectively up-regulating IL-12 receptor beta 2 expression via cGMP. *Proc Natl Acad Sci U S A*, **99**, 16186-16191.

Nieto, A., Cabrera, C.M., Catalina, P., Cobo, F., Barnie, A., Cortés, J.L., Barroso del Jesus, A., Montes, R. and Concha, A. (2007). Effect of mitomycin-C on human foreskin fibroblasts used as feeders in human embryonic stem cells: Immunocytochemistry MIB1 score and DNA ploidy and apoptosis evaluated by flow cytometry. *Cell Biology International*, **31**, 269-278.

Niisato, N., Taruno, A. and Marunaka, Y. (2007). Involvement of p38 MAPK in hypotonic stress-induced stimulation of  $\beta$ - and  $\gamma$ -ENaC expression in renal epithelium *Biochemical and Biophysical Research Communications*, **358**, 819-824.

Nikolova, M., Chen, X. and Lufkin, T. (1997). Nkx2.6 expression is transiently and specifically restricted to the branchial region of pharyngeal-stage mouse embryos. *Mechanisms of Development*, **69**, 215-218.

Nishimoto, M., Fukushima, A., Okuda, A. and Muramatsu, M. (1999). The gene for the embryonic stem cell coactivator UTF1 carries a regulatory element which selectively interacts with a complex composed of Oct-3/4 and Sox-2. *Mol Cell Biol*, **19**, 5453-65.

Nishizuka, Y. (1995). Protein kinase C and lipid signaling for sustained cellular responses. *FASEB Journal*, **9**, 484-496.

Niwa, H., Burdon, T., Chambers, I. and Smith, A. (1998). Self-renewal of pluripotent embryonic stem cells is mediated via activation of STAT3. *Genes Dev*, **12**, 2048-60.

Niwa, H., Miyazaki, J.-i. and Smith, A.G. (2000). Quantitative expression of Oct-3/4 defines differentiation, dedifferentiation or self-renewal of ES cells. *Nat Genet*, **24**, 372-376.

Noden, D.M. (1991). Origins and patterning of avian outflow tract

endocardium. Development, **111**, 867-876.

Noordman, Y.E., Jansen, P.A. and Hendriks, W.J. (2006). Tyrosine-specific MAPK phosphatases and the control of ERK signaling in PC12 cells. *J Mol Signal*, **1**, 4.

Nussbaum, J., Minami, E., Laflamme, M.A., Virag, J.A.I., Ware, C.B., Masino, A., Muskheli, V., Pabon, L., Reinecke, H. and Murry, C.E. (2007). Transplantation of undifferentiated murine embryonic stem cells in the heart: teratoma formation and immune response. *FASEB J.*, **21**, 1345-1357.

Oancea, E. and Meyer, T. (1998). Protein kinase C as a molecular machine for decoding calcium and diacylglycerol signals. *Cell*, **95**, 307-318.

Obregon, E., Punzon, M.C., Gonzalez-Nicolas, J., Fernandez-Cruz, E., Fresno, M. and Munoz-Fernandez, M.A. (1997). Induction of adhesion/differentiation of human neuroblastoma cells by tumour necrosis factor-alpha requires the expression of an inducible nitric oxide synthase. *Eur J Neurosci*, **9**, 1184-93.

Odabas, S., Sayar, F., Guven, G., Yanikkaya-Demirel, G. and Piskin, E. (2008). Separation of mesenchymal stem cells with magnetic nanosorbents carrying CD105 and CD73 antibodies in flow-through and batch systems. *J Chromatogr B Analyt Technol Biomed Life Sci*, **861**, 74-80.

Ohtani, T., Ishihara, K., Atsumi, T., Nishida, K., Kaneko, Y., Miyata, T., Itoh, S., Narimatsu, M., Maeda, H., Fukada, T., Itoh, M., Okano, H., Hibi, M. and Hirano, T. (2000). Dissection of signaling cascades through gp130 in vivo: reciprocal roles for STAT3- and SHP2-mediated signals in immune responses. *Immunity*, **12**, 95-105.

Okada, S., Nakauchi, H., Nagayoshi, K., Nishikawa, S., Nishikawa, S., Miura, Y. and Suda, T. (1991). Enrichment and characterization of murine hematopoietic stem cells that express c-kit molecule. *Blood*, **78**, 1706-1712.

Okada, S., Takehara, Y., Yabuki, M., Yoshioka, T., Yasuda, T., Inoue, M. and Utsumi, K. (1996). Nitric oxide, a physiological modulator of mitochondrial function. *Physiol Chem Phys Med NMR*, **28**, 69-82.

Okamoto, K., Okazawa, H., Okuda, A., Sakai, M., Muramatsu, M. and Hamada, H. (1990). A novel octamer binding transcription factor is differentially expressed in mouse embryonic cells. *Cell*, **60**, 461-72.

Okuda, A., Fukushima, A., Nishimoto, M., Orimo, A., Yamagishi, T., Nabeshima, Y., Kuro-o, M., Nabeshima, Y., Boon, K., Keaveney, M., Stunnenberg, H.G. and Muramatsu, M. (1998). UTF1, a novel transcriptional coactivator expressed in pluripotent embryonic stem cells and extraembryonic cells. *Embo J*, **17**, 2019-32.

Okuyama, M., Yamaguchi, S., Yamaoka, M., Nitobe, J., Fujii, S., Yoshimura,
T. and Tomoike, H. (2000). Nitric Oxide Enhances Expression and Shedding of Tumor Necrosis Factor Receptor I (p55) in Endothelial Cells. *Arterioscler Thromb Vasc Biol*, **20**, 1506-1511.

Orkin, S.H. (1992). GATA-binding transcription factors in hematopoietic cells. *Blood*, **80**, 575-581.

Orlic, D., Anderson, S., Biesecker, L.G., Sorrentino, B.P. and Bodine, D.M. (1995). Pluripotent Hematopoietic Stem Cells Contain High Levels of mRNA for c-kit, GATA-2, p45 NF-E2, and c-myb and Low Levels or no mRNA for c-fms and the Receptors for Granulocyte Colony-Stimulating Factor and Interleukins 5 and 7. *PNAS*, **92**, 4601-4605.

Orr, J.W., Keranen, L.M. and Newton, A.C. (1992). Reversible exposure of the pseudosubstrate domain of protein kinase C by phosphatidylserine and diacylglycerol. *J. Biol. Chem.*, **267**, 15263-15266.

Orr, J.W. and Newton, A.C. (1994). Intrapeptide regulation of protein kinase C. *J. Biol. Chem.*, **269**, 8383-8387.

Oswald, J., Boxberger, S., Jorgensen, B., Feldmann, S., Ehninger, G., Bornhauser, M. and Werner, C. (2004). Mesenchymal stem cells can be differentiated into endothelial cells in vitro. *Stem Cells*, **22**, 377-84.

Ouwens, D.M., de Ruiter, N.D., van der Zon, G.C., Carter, A.P., Schouten, J., van der Burgt, C., Kooistra, K., Bos, J.L., Maassen, J.A. and van Dam, H. (2002). Growth factors can activate ATF2 via a two-step mechanism: phosphorylation of Thr71 through the Ras-MEK-ERK pathway and of Thr69 through RalGDS-Src-p38. *Embo J*, **21**, 3782-93.

Ozawa, M., Muramatsu, T. and Solter, D. (1985). SSEA-1, a stage-specific embryonic antigen of the mouse, is carried by the glycoprotein-bound large carbohydrate in embryonal carcinoma cells. *Cell Differ*, **16**, 169-73.

Pagano, M., Naviglio, S., Spina, A., Chiosi, E., Castoria, G., Romano, M., Sorrentino, A., Illiano, F. and Illiano, G. (2004). Differentiation of H9c2 cardiomyoblasts: The role of adenylate cyclase system. *J Cell Physiol*, **198**, 408-16.

Pages, G., Lenormand, P., L'Allemain, G., Chambard, J.C., Meloche, S. and Pouyssegur, J. (1993). Mitogen-activated protein kinases p42(mapk) and p44(mapk) are required for fibroblast proliferation. *Proc Natl Acad Sci U S A*, **90**, 8319-8323.

Pallante, B.A., Duignan, I., Okin, D., Chin, A., Bressan, M.C., Mikawa, T. and Edelberg, J.M. (2007). Bone Marrow Oct3/4+ Cells Differentiate Into Cardiac Myocytes via Age-Dependent Paracrine Mechanisms. *Circ Res*, **100**, e1-11.

Palmer, L.A., Gaston, B. and Johns, R.A. (2000). Normoxic Stabilization of Hypoxia-Inducible Factor-1 Expression and Activity: Redox-Dependent Effect

of Nitrogen Oxides. Mol Pharmacol, 58, 1197-1203.

Palmieri, S.L., Peter, W., Hess, H. and Scholer, H.R. (1994). Oct-4 transcription factor is differentially expressed in the mouse embryo during establishment of the first two extraembryonic cell lineages involved in implantation. *Dev Biol*, **166**, 259-67.

Palojoki, E., Saraste, A., Eriksson, A., Pulkki, K., Kallajoki, M., Voipio-Pulkki, L.-M. and Tikkanen, I. (2001). Cardiomyocyte apoptosis and ventricular remodeling after myocardial infarction in rats. *Am J Physiol Heart Circ Physiol*, **280**, H2726-2731.

Pan, Z.-Z., Bruening, W., Giasson, B.I., Lee, V.M.Y. and Godwin, A.K. (2002). gamma -Synuclein Promotes Cancer Cell Survival and Inhibits Stress- and Chemotherapy Drug-induced Apoptosis by Modulating MAPK Pathways. *J. Biol. Chem.*, **277**, 35050-35060.

Pandur, P., Lasche, M., Eisenberg, L.M. and Kuhl, M. (2002). Wnt-11 activation of a non-canonical Wnt signalling pathway is required for cardiogenesis. *Nature*, **418**, 636-41.

Park, S.-P., Lee, Y.J., Lee, K.S., Shin, H.A., Cho, H.Y., Chung, K.S., Kim, E.Y. and Lim, J.H. (2004). Establishment of human embryonic stem cell lines from frozen-thawed blastocysts using STO cell feeder layers. *Hum. Reprod.*, **19**, 676-684.

Parker, P.J., Coussens, L., Totty, N., Rhee, L., Young, S., Chen, E., Stabel, S., Waterfield, M.D. and Ullrich, A. (1986). The complete primary structure of protein kinase C--the major phorbol ester receptor. *Science*, **233**, 853-9.

Partovian, C. and Simons, M. (2004). Regulation of protein kinase B/Akt activity and Ser473 phosphorylation by protein kinase Calpha in endothelial cells. *Cell Signal*, **16**, 951-7.

Pasha, Z., Wang, Y., Sheikh, R., Zhang, D., Zhao, T. and Ashraf, M. (2008). Preconditioning enhances cell survival and differentiation of stem cells during transplantation in infarcted myocardium. *Cardiovasc Res*, **77**, 134-142.

Patient, R.K. and McGhee, J.D. (2002). The GATA family (vertebrates and invertebrates). *Current Opinion in Genetics & Development*, **12**, 416-422.

Pellegrino, M.J. and Stork, P.J. (2006). Sustained activation of extracellular signal-regulated kinase by nerve growth factor regulates c-fos protein stabilization and transactivation in PC12 cells. *J Neurochem*, **99**, 1480-93.

Pera, E.M. and De Robertis, E.M. (2000). A direct screen for secreted proteins in Xenopus embryos identifies distinct activities for the Wnt antagonists Crescent and Frzb-1. *Mechanisms of Development*, **96**, 183-195.

Perin, L., Sedrakyan, S., Da Sacco, S., De Filippo, R. and Dr. Jennie, P.M. (2008). Characterization of human amniotic fluid stem cells and their pluripotential capability. *Methods Cell Biol* **86**: 85-99.

Persad, S., Attwell, S., Gray, V., Delcommenne, M., Troussard, A., Sanghera, J. and Dedhar, S. (2000). Inhibition of integrin-linked kinase (ILK) suppresses activation of protein kinase B/Akt and induces cell cycle arrest and apoptosis of PTEN-mutant prostate cancer cells. *Proc Natl Acad Sci U S A*, **97**, 3207-12.

Person, A.D., Garriock, R.J., Krieg, P.A., Runyan, R.B. and Klewer, S.E. (2005). Frzb modulates Wnt-9a-mediated [beta]-catenin signaling during avian atrioventricular cardiac cushion development. *Developmental Biology*, **278**, 35-48.

Pesant, M., Sueur, S., Dutartre, P., Tallandier, M., Grimaldi, P.A., Rochette, L. and Connat, J.-L. (2006). Peroxisome proliferator-activated receptor [delta] (PPAR[delta]) activation protects H9c2 cardiomyoblasts from oxidative stress-induced apoptosis. *Cardiovascular Research*, **69**, 440-449.

Peterkin, T., Gibson, A. and Patient, R. (2003). GATA-6 maintains BMP-4 and Nkx2 expression during cardiomyocyte precursor maturation. *EMBO Journal*, **22**, 4260-4273.

Peterkin, T., Gibson, A. and Patient, R. (2007). Redundancy and evolution of GATA factor requirements in development of the myocardium. *Dev Biol*, **311**, 623-35.

Phung, Y.T., Bekker, J.M., Hallmark, O.G. and Black, S.M. (1999). Both neuronal NO synthase and nitric oxide are required for PC12 cell differentiation: a cGMP independent pathway. *Molecular Brain Research*, **64**, 165-178.

Pineda, J., Kogan, J.H. and Aghajanian, G.K. (1996). Nitric oxide and carbon monoxide activate locus coeruleus neurons through a cGMP-dependent protein kinase: involvement of a nonselective cationic channel. *J Neurosci*, **16**, 1389-99.

Pinson, K.I., Brennan, J., Monkley, S., Avery, B.J. and Skarnes, W.C. (2000). An LDL-receptor-related protein mediates Wnt signalling in mice. *Nature*, **407**, 535-538.

Pintus, G., Tadolini, B., Posadino, A.M., Sanna, B., Debidda, M., Bennardini, F., Sava, G. and Ventura, C. (2002). Inhibition of the MEK/ERK signaling pathway by the novel antimetastatic agent NAMI-A down regulates c-myc gene expression and endothelial cell proliferation. *Eur J Biochem*, **269**, 5861-70.

Pittenger, M.F., Mackay, A.M., Beck, S.C., Jaiswal, R.K., Douglas, R., Mosca, J.D., Moorman, M.A., Simonetti, D.W., Craig, S. and Marshak, D.R. (1999).

Multilineage Potential of Adult Human Mesenchymal Stem Cells. *Science*, **284**, 143-147.

Polisetty, N., Fatima, A., Madhira, S.L., Sangwan, V.S. and Vemuganti, G.K. (2008). Mesenchymal cells from limbal stroma of human eye. *Mol Vis*, **14**, 431-42.

Preciado, D., Caicedo, E., Jhanjee, R., Silver, R., Harris, G., Juhn, S.K., Choo, D.I. and Ondrey, F. (2005). Pseudomonas aeruginosa Lipopolysaccharide Induction of Keratinocyte Proliferation, NF-{kappa}B, and Cyclin D1 Is Inhibited by Indomethacin. *J Immunol*, **174**, 2964-2973.

Prugger, C., Wellmann, J., Heidrich, J., Brand-Herrmann, S.M. and Keil, U. (2008). Cardiovascular risk factors and mortality in patients with coronary heart disease. *Eur J Epidemiol*, **23**, 731-7.

Prusa, A.-R., Marton, E., Rosner, M., Bernaschek, G. and Hengstschlager, M. (2003). "Oct-4-expressing cells in human amniotic fluid: a new source for stem cell research?" *Hum Reprod*, **18**: 1489-93.

Pullen, N., Dennis, P.B., Andjelkovic, M., Dufner, A., Kozma, S.C., Hemmings, B.A. and Thomas, G. (1998). Phosphorylation and Activation of p70s6k by PDK1. *Science*, **279**, 707-710.

Qi, J.-H., Matsumoto, T., Huang, K., Olausson, K., Christofferson, R. and Claesson-Welsh, L. (1999). Phosphoinositide 3 kinase is critical for survival, mitogenesis and migration but not for differentiation of endothelial cells. *Angiogenesis*, **3**, 371-380.

Qin, F., Patel, R., Yan, C. and Liu, W. (2006). NADPH oxidase is involved in angiotensin II-induced apoptosis in H9C2 cardiac muscle cells: Effects of apocynin. *Free Radical Biology and Medicine*, **40**, 236-246.

Qin, F., Simeone, M. and Patel, R. (2007). Inhibition of NADPH oxidase reduces myocardial oxidative stress and apoptosis and improves cardiac function in heart failure after myocardial infarction. *Free Radical Biology and Medicine*, **43**, 271-281.

Raftopoulos, C., Bermingham, M.A. and Steinbeck, K.S. (1999). Coronary heart disease risk factors in male adolescents, with particular reference to smoking and blood lipids. *Journal of Adolescent Health*, **25**, 68-74.

Ramljak, D., Coticchia, C.M., Nishanian, T.G., Saji, M., Ringel, M.D., Conzen, S.D. and Dickson, R.B. (2003). Epidermal growth factor inhibition of c-Mycmediated apoptosis through Akt and Erk involves Bcl-xL upregulation in mammary epithelial cells. *Exp Cell Res*, **287**, 397-410.

Ramos-Nino, M.E., Blumen, S.R., Sabo-Attwood, T., Pass, H., Carbone, M., Testa, J.R., Altomare, D.A. and Mossman, B.T. (2008). HGF mediates cell proliferation of human mesothelioma cells through a PI3K/MEK5/Fra-1

pathway. Am J Respir Cell Mol Biol, 38, 209-17.

Rao, A.S., Kremenevskaja, N., Resch, J. and Brabant, G. (2005). Lithium stimulates proliferation in cultured thyrocytes by activating Wnt/beta-catenin signalling. *Eur J Endocrinol*, **153**, 929-38.

Rattner, A., Hsieh, J.C., Smallwood, P.M., Gilbert, D.J., Copeland, N.G., Jenkins, N.A. and Nathans, J. (1997). A family of secreted proteins contains homology to the cysteine-rich ligand-binding domain of frizzled receptors. *Proc Natl Acad Sci U S A*, **94**, 2859-2863.

Raz, R., Lee, C.K., Cannizzaro, L.A., d'Eustachio, P. and Levy, D.E. (1999). Essential role of STAT3 for embryonic stem cell pluripotency. *Proc Natl Acad Sci U S A*, **96**, 2846-51.

Reddy, A.M., Seo, J.H., Ryu, S.Y., Kim, Y.S., Kim, Y.S., Min, K.R. and Kim, Y. (2004). Cinnamaldehyde and 2-methoxycinnamaldehyde as NF-kappaB inhibitors from Cinnamomum cassia. *Planta Med*, **70**, 823-7.

Redkar, A., Montgomery, M. and Litvin, J. (2001). Fate map of early avian cardiac progenitor cells. *Development*, **128**, 2269-2279.

Reecy, J.M., Yamada, M., Cummings, K., Chen, C.Y., Schwartz, R.J., Eichele, G., Sosic, D. and Olson, E.N. (1997). Chicken Nkx-2.8: A novel homeobox gene expressed in early heart progenitor cells and pharyngeal pouch-2 and -3 endoderm. *Developmental Biology*, **188**, 295-311.

Reim, I. and Frasch, M. (2005). The Dorsocross T-box genes are key components of the regulatory network controlling early cardiogenesis in Drosophila. *Development*, **132**, 4911-25.

Reim, I., Mohler, J.P. and Frasch, M. (2005). Tbx20-related genes, mid and H15, are required for tinman expression, proper patterning, and normal differentiation of cardioblasts in Drosophila. *Mech Dev*, **122**, 1056-69.

Reubinoff, B.E., Itsykson, P., Turetsky, T., Pera, M.F., Reinhartz, E., Itzik, A. and Ben-Hur, T. (2001). Neural progenitors from human embryonic stem cells. *Nat Biotech*, **19**, 1134-1140.

Reubinoff, B.E., Pera, M.F., Fong, C.Y., Trounson, A. and Bongso, A. (2000). Embryonic stem cell lines from human blastocysts: somatic differentiation in vitro. *Nat Biotechnol*, **18**, 399-404.

Reyes, J.C., Muro-Pastor, M.I. and Florencio, F.J. (2004). The GATA family of transcription factors in arabidopsis and rice. *Plant Physiology*, **134**, 1718-1732.

Reyes, M., Lund, T., Lenvik, T., Aguiar, D., Koodie, L. and Verfaillie, C.M. (2001). Purification and ex vivo expansion of postnatal human marrow mesodermal progenitor cells. *Blood*, **98**, 2615-2625.

Rink, S.M., Lipman, R., Alley, S.C., Hopkins, P.B. and Tomasz, M. (1996). Bending of DNA by the Mitomycin C-Induced, GpG Intrastrand Cross-Link. pp. 382-389.

Rodda, D.J., Chew, J.-L., Lim, L.-H., Loh, Y.-H., Wang, B., Ng, H.-H. and Robson, P. (2005). Transcriptional Regulation of Nanog by OCT4 and SOX2. *J. Biol. Chem.*, **280**, 24731-24737.

Rodriguez-Niedenfuhr, M., Dathe, V., Jacob, H.J., Prols, F. and Christ, B. (2003). Spatial and temporal pattern of Wnt-6 expression during chick development. *Anat Embryol (Berl)*, **206**, 447-51.

Rohwedel, J., Maltsev, V., Bober, E., Arnold, H.H., Hescheler, J. and Wobus, A.M. (1994). Muscle Cell Differentiation of Embryonic Stem Cells Reflects Myogenesis in Vivo: Developmentally Regulated Expression of Myogenic Determination Genes and Functional Expression of Ionic Currents. *Developmental Biology*, **164**, 87-101.

Roman-Roman, S., Shi, D.L., Stiot, V., Hay, E., Vayssiere, B., Garcia, T., Baron, R. and Rawadi, G. (2004). Murine Frizzled-1 behaves as an antagonist of the canonical Wnt/beta-catenin signaling. *J Biol Chem*, **279**, 5725-33.

Rosamond, W.D., Chambless, L.E., Folsom, A.R., Cooper, L.S., Conwill, D.E., Clegg, L., Wang, C.-H. and Heiss, G. (1998). Trends in the Incidence of Myocardial Infarction and in Mortality Due to Coronary Heart Disease, 1987 to 1994. pp. 861-867.

Rosenzweig, B.L., Imamura, T., Okadome, T., Cox, G.N., Yamashita, H., ten Dijke, P., Heldin, C.H. and Miyazono, K. (1995). Cloning and characterization of a human type II receptor for bone morphogenetic proteins. *Proc Natl Acad Sci U S A*, **92**, 7632-6.

Rosenzweig, K.E., Youmell, M.B., Palayoor, S.T. and Price, B.D. (1997). Radiosensitization of human tumor cells by the phosphatidylinositol3-kinase inhibitors wortmannin and LY294002 correlates with inhibition of DNAdependent protein kinase and prolonged G2-M delay. *Clin Cancer Res*, **3**, 1149-1156.

Rosner, M.H., Vigano, M.A., Ozato, K., Timmons, P.M., Poirier, F., Rigby, P.W. and Staudt, L.M. (1990). A POU-domain transcription factor in early stem cells and germ cells of the mammalian embryo. *Nature*, **345**, 686-92.

Rosseland, C.M., Wierod, L., Flinder, L.I., Oksvold, M.P., Skarpen, E. and Huitfeldt, H.S. (2008). Distinct functions of H-Ras and K-Ras in proliferation and survival of primary hepatocytes due to selective activation of ERK and PI3K. *J Cell Physiol*, **215**, 818-26.

Rota, M., Padin-Iruegas, M.E., Misao, Y., De Angelis, A., Maestroni, S., Ferreira-Martins, J., Fiumana, E., Rastaldo, R., Arcarese, M.L., Mitchell, T.S.,

Boni, A., Bolli, R., Urbanek, K., Hosoda, T., Anversa, P., Leri, A. and Kajstura, J. (2008). Local activation or implantation of cardiac progenitor cells rescues scarred infarcted myocardium improving cardiac function. *Circ Res*, **103**, 107-16.

Roura, S., Farre, J., Soler-Botija, C., Llach, A., Hove-Madsen, L., Cairo, J.J., Godia, F., Cinca, J. and Bayes-Genis, A. (2006). Effect of aging on the pluripotential capacity of human CD105+ mesenchymal stem cells. *Eur J Heart Fail*, **8**, 555-63.

Rousseau, S., Dolado, I., Beardmore, V., Shpiro, N., Marquez, R., Nebreda, A.R., Arthur, J.S., Case, L.M., Tessier-Lavigne, M., Gaestel, M., Cuenda, A. and Cohen, P. (2006). CXCL12 and C5a trigger cell migration via a PAK1/2-p38alpha MAPK-MAPKAP-K2-HSP27 pathway. *Cell Signal*, **18**, 1897-905.

Rutan, G.H., McDonald, R.H. and Kuller, L.H. (1989). A historical perspective of elevated systolic vs diastolic blood pressure from an epidemiological and clinical trial viewpoint. *Journal of Clinical Epidemiology*, **42**, 663-673.

Sabapathy, K., Kallunki, T., David, J.P., Graef, I., Karin, M. and Wagner, E.F. (2001). c-Jun NH2-terminal kinase (JNK)1 and JNK2 have similar and stagedependent roles in regulating T cell apoptosis and proliferation. *J Exp Med*, **193**, 317-28.

Sagara, N., Toda, G., Hirai, M., Terada, M. and Katoh, M. (1998). Molecular cloning, differential expression, and chromosomal localization of human frizzled-1, frizzled-2, and frizzled-7. *Biochem Biophys Res Commun*, **252**, 117-22.

Saitoh, T., Hirai, M. and Katoh, M. (2001). Molecular cloning and characterization of human Frizzled-8 gene on chromosome 10p11.2. *Int J Oncol*, **18**, 991-6.

Sakai, N., Sasaki, K., Ikegaki, N., Shirai, Y., Ono, Y. and Saito, N. (1997). Direct visualization of the translocation of the  $\gamma$ -subspecies of protein kinase C in living cells using fusion proteins with green fluorescent protein. *Journal of Cell Biology*, **139**, 1465-1476.

Salvatorelli, E., Guarnieri, S., Menna, P., Liberi, G., Calafiore, A.M., Mariggio, M.A., Mordente, A., Gianni, L. and Minotti, G. (2006). Defective one or two electron reduction of the anticancer anthracycline epirubicin in human heart: Relative importance of vesicular sequestration and impaired efficiency of electron addition. *J Biol Chem*.

Salvatorelli, E., Guarnieri, S., Menna, P., Liberi, G., Calafiore, A.M., Mariggio, M.A., Mordente, A., Gianni, L. and Minotti, G. (2006). Defective one- or twoelectron reduction of the anticancer anthracycline epirubicin in human heart. Relative importance of vesicular sequestration and impaired efficiency of electron addition. *J Biol Chem*, **281**, 10990-11001. Sanai, N., Tramontin, A.D., Quinones-Hinojosa, A., Barbaro, N.M., Gupta, N., Kunwar, S., Lawton, M.T., McDermott, M.W., Parsa, A.T., Manuel-Garcia Verdugo, J., Berger, M.S. and Alvarez-Buylla, A. (2004). Unique astrocyte ribbon in adult human brain contains neural stem cells but lacks chain migration. *Nature*, **427**, 740-744.

Sanna, B., Bueno, O.F., Dai, Y.S., Wilkins, B.J. and Molkentin, J.D. (2005). Direct and indirect interactions between calcineurin-NFAT and MEK1-extracellular signal-regulated kinase 1/2 signaling pathways regulate cardiac gene expression and cellular growth. *Mol Cell Biol*, **25**, 865-78.

Sanyal, M., Nag, T.C. and Das, C. (2000). Localization of nitric oxide synthase in human trophoblast cells: role of nitric oxide in trophoblast proliferation and differentiation. *Am J Reprod Immunol*, **43**, 70-7.

Saraste, A., Pulkki, K., Kallajoki, M., Henriksen, K., Parvinen, M. and Voipio-Pulkki, L.M. (1997). Apoptosis in human acute myocardial infarction. *Circulation*, **95**, 320-3.

Sasaki, N., Sato, T., Ohler, A., O'Rourke, B. and Marban, E. (2000). Activation of Mitochondrial ATP-Dependent Potassium Channels by Nitric Oxide. *Circulation*, **101**, 439-445.

Sato, N., Meijer, L., Skaltsounis, L., Greengard, P. and Brivanlou, A.H. (2004). Maintenance of pluripotency in human and mouse embryonic stem cells through activation of Wnt signaling by a pharmacological GSK-3-specific inhibitor. *Nat Med*, **10**, 55-63.

Sato, N., Sanjuan, I.M., Heke, M., Uchida, M., Naef, F. and Brivanlou, A.H. (2003). Molecular signature of human embryonic stem cells and its comparison with the mouse. *Dev Biol*, **260**, 404-13.

Sauer, H., Rahimi, G., Hescheler, J. and Wartenberg, M. (2000). Role of reactive oxygen species and phosphatidylinositol 3-kinase in cardiomyocyte differentiation of embryonic stem cells. *FEBS Letters*, **476**, 218-223.

Sawatzky, D.A., Willoughby, D.A., Colville-Nash, P.R. and Rossi, A.G. (2006). The Involvement of the Apoptosis-Modulating Proteins ERK 1/2, Bcl-xL and Bax in the Resolution of Acute Inflammation in Vivo. *Am J Pathol*, **168**, 33-41.

Scavo, L.M., Karas, M., Murray, M. and Leroith, D. (2004). Insulin-Like Growth Factor-I Stimulates Both Cell Growth and Lipogenesis during Differentiation of Human Mesenchymal Stem Cells into Adipocytes. *J Clin Endocrinol Metab*, **89**, 3543-3553.

Scheid, M.P., Schubert, K.M. and Duronio, V. (1999). Regulation of bad phosphorylation and association with Bcl-x(L) by the MAPK/Erk kinase. *J Biol Chem*, **274**, 31108-13.

Schlange, T., Andrée, B., Arnold, H.H. and Brand, T. (2000). BMP2 is

required for early heart development during a distinct time period. *Mech Dev*, **91**, 259-70.

Schmitt, R.M., Bruyns, E. and Snodgrass, H.R. (1991). Hematopoietic development of embryonic stem cells in vitro: cytokine and receptor gene expression. *Genes Dev*, **5**, 728-40.

Schneider, V.A. and Mercola, M. (2001). Wnt antagonism initiates cardiogenesis in Xenopus laevis. *Genes Dev*, **15**, 304-15.

Schuldiner, M., Yanuka, O., Itskovitz-Eldor, J., Melton, D.A. and Benvenisty, N. (2000). From the Cover: Effects of eight growth factors on the differentiation of cells derived from human embryonic stem cells. *Proc Natl Acad Sci U S A*, **97**, 11307-11312.

Schulte-Merker, S., Ho, R.K., Herrmann, B.G. and Nusslein-Volhard, C. (1992). The protein product of the zebrafish homologue of the mouse T gene is expressed in nuclei of the germ ring and the notochord of the early embryo. *Development*, **116**, 1021-32.

Schultheiss, T.M., Lassar, A.B. and Burch, J.B.E. (1997). A role for bone morphogenetic proteins in the induction of cardiac myogenesis. *Genes and Development*, **11**, 451-462.

Schuringa, J.J., van der Schaaf, S., Vellenga, E., Eggen, B.J. and Kruijer, W. (2002). LIF-induced STAT3 signaling in murine versus human embryonal carcinoma (EC) cells. *Exp Cell Res*, **274**, 119-29.

Schwanke, K., Wunderlich, S., Reppel, M., Winkler, M.E., Matzkies, M., Groos, S., Itskovitz-Eldor, J., Simon, A.R., Hescheler, J., Haverich, A. and Martin, U. (2006). Generation and Characterization of Functional Cardiomyocytes from Rhesus Monkey Embryonic Stem Cells. pp. 1423-1432.

Semenov, M.V., Tamai, K., Brott, B.K., Kuhl, M., Sokol, S. and He, X. (2001). Head inducer Dickkopf-1 is a ligand for Wnt coreceptor LRP6. *Curr Biol*, **11**, 951-61.

Sepulveda, J.L., Belaguli, N., Nigam, V., Chen, C.Y.I., Nemer, M. and Schwartz, R.J. (1998). GATA-4 and Nkx-2.5 Coactivate Nkx-2 DNA binding targets: Role for regulating early cardiac gene expression. *Molecular and Cellular Biology*, **18**, 3405-3415.

Serra, J.M., Gutierrez, A., Alemany, R., Navarro, M., Ros, T., Saus, C., Gines, J., Sampol, A., Amat, J.C., Serra-Moises, L., Martin, J., Galmes, A., Vogler, O. and Besalduch, J. (2008). Inhibition of c-Myc down-regulation by sustained ERK activation prevents the antimetabolite methotrexate- and gemcitabine-induced differentiation in non-small cell lung cancer cells. *Mol Pharmacol.* 

Sethi, G. and Sodhi, A. (2004). Role of p38 mitogen-activated protein kinase and caspases in UV-B-induced apoptosis of murine peritoneal macrophages.

Photochem Photobiol, **79**, 48-54.

Seydoux, G., Mello, C.C., Pettitt, J., Wood, W.B., Priess, J.R. and Fire, A. (1996). Repression of gene expression in the embryonic germ lineage of C. elegans. *Nature*, **382**, 713-716.

Shaik, Z.P., Fifer, E.K. and Nowak, G. (2008). Akt activation improves oxidative phosphorylation in renal proximal tubular cells following nephrotoxicant injury. *Am J Physiol Renal Physiol*, **294**, F423-32.

Shamblott, M.J., Axelman, J., Wang, S., Bugg, E.M., Littlefield, J.W., Donovan, P.J., Blumenthal, P.D., Huggins, G.R. and Gearhart, J.D. (1998). Derivation of pluripotent stem cells from cultured human primordial germ cells. *Proc Natl Acad Sci U S A*, **95**, 13726-13731.

Shami, P.J., Sauls, D.L. and Weinberg, J.B. (1998). Schedule and concentration-dependent induction of apoptosis in leukemia cells by nitric oxide. *Leukemia*, **12**, 1461-6.

Sharma, P. and Mondal, A.K. (2006). Evidence that the MAPK-docking site in MAPKK Dpbs2p is essential for its function. *Biochemical and Biophysical Research Communications*, **346**, 562-566.

Sheldahl, L.C., Park, M., Malbon, C.C. and Moon, R.T. (1999). Protein kinase C is differentially stimulated by Wnt and Frizzled homologs in a G-proteindependent manner. *Current Biology*, **9**, 695-698.

Sheldahl, L.C., Slusarski, D.C., Pandur, P., Miller, J.R., Kuhl, M. and Moon, R.T. (2003). Dishevelled activates Ca2+ flux, PKC, and CamKII in vertebrate embryos. *J Cell Biol*, **161**, 769-77.

Shen, Q., Goderie, S.K., Jin, L., Karanth, N., Sun, Y., Abramova, N., Vincent, P., Pumiglia, K. and Temple, S. (2004). Endothelial Cells Stimulate Self-Renewal and Expand Neurogenesis of Neural Stem Cells. *Science*, **304**, 1338-1340.

Shibata, H., Ageyama, N., Tanaka, Y., Kishi, Y., Sasaki, K., Nakamura, S., Muramatsu, S.-i., Hayashi, S., Kitano, Y., Terao, K. and Hanazono, Y. (2006). Improved Safety of Hematopoietic Transplantation with Monkey Embryonic Stem Cells in the Allogeneic Setting. *Stem Cells*, **24**, 1450-1457.

Shibata, M., Ono, H., Hikasa, H., Shinga, J. and Taira, M. (2000). Xenopus crescent encoding a Frizzled-like domain is expressed in the Spemann organizer and pronephros. *Mechanisms of Development*, **96**, 243-246.

Shimizu, R. and Yamamoto, M. (2005). Gene expression regulation and domain function of hematopoietic GATA factors. *Seminars in Cell and Developmental Biology*, **16**, 129-136.

Singh, M.K., Christoffels, V.M., Dias, J.M., Trowe, M.-O., Petry, M., Schuster-

Gossler, K., Burger, A., Ericson, J. and Kispert, A. (2005). Tbx20 is essential for cardiac chamber differentiation and repression of Tbx2. *Development*, **132**, 2697-2707.

Singla, D.K., Hacker, T.A., Ma, L., Douglas, P.S., Sullivan, R., Lyons, G.E. and Kamp, T.J. (2006). Transplantation of embryonic stem cells into the infarcted mouse heart: formation of multiple cell types. *Journal of Molecular and Cellular Cardiology*, **40**, 195-200.

Sipido, K.R. and Marban, E. (1991). L-type calcium channels, potassium channels, and novel nonspecific cation channels in a clonal muscle cell line derived from embryonic rat ventricle. *Circ Res*, **69**, 1487-99.

Skarsgard, P.L., Wang, X., McDonald, P., Lui, A.H., Lam, E.K., McManus, B.M., van Breemen, C. and Laher, I. (2000). Profound inhibition of myogenic tone in rat cardiac allografts is due to eNOS- and iNOS-based nitric oxide and an intrinsic defect in vascular smooth muscle contraction. *Circulation*, **101**, 1303-10.

Skerjanc, I.S., Petropoulos, H., Ridgeway, A.G. and Wilton, S. (1998). Myocyte enhancer factor 2C and Nkx2-5 up-regulate each other's expression and initiate cardiomyogenesis in P19 cells. *Journal of Biological Chemistry*, **273**, 34904-34910.

Slepak, T.I., Webster, K.A., Zang, J., Prentice, H., O'Dowd, A., Hicks, M.N. and Bishopric, N.H. (2001). Control of cardiac-specific transcription by p300 through myocyte enhancer factor-2D. *J Biol Chem*, **276**, 7575-85.

Slusarski, D.C., Yang-Snyder, J., Busa, W.B. and Moon, R.T. (1997). Modulation of embryonic intracellular Ca2+ signaling by Wnt-5A. *Developmental Biology*, **182**, 114-120.

Smith, A.G., Heath, J.K., Donaldson, D.D., Wong, G.G., Moreau, J., Stahl, M. and Rogers, D. (1988). Inhibition of pluripotential embryonic stem cell differentiation by purified polypeptides. *Nature*, **336**, 688-690.

Smith, A.G. and Hooper, M.L. (1987). Buffalo rat liver cells produce a diffusible activity which inhibits the differentiation of murine embryonal carcinoma and embryonic stem cells. *Dev Biol*, **121**, 1-9.

Smith, J.A., Poteet-Smith, C.E., Malarkey, K. and Sturgill, T.W. (1999). Identification of an extracellular signal-regulated kinase (ERK) docking site in ribosomal S6 kinase, a sequence critical for activation by ERK in vivo. *J Biol Chem*, **274**, 2893-8.

Smith, P.K., Krohn, R.I., Hermanson, G.T., Mallia, A.K., Gartner, F.H., Provenzano, M.D., Fujimoto, E.K., Goeke, N.M., Olson, B.J. and Klenk, D.C. (1985). Measurement of protein using bicinchoninic acid. *Anal Biochem*, **150**, 76-85.

Smyth, D.C., Kerr, C., Li, Y., Tang, D. and Richards, C.D. (2008). Oncostatin M induction of eotaxin-1 expression requires the convergence of PI3'K and ERK1/2 MAPK signal transduction pathways. *Cell Signal*, **20**, 1142-50.

Solter, D. and Knowles, B.B. (1978). Monoclonal antibody defining a stagespecific mouse embryonic antigen (SSEA-1). *Proc Natl Acad Sci U S A*, **75**, 5565-9.

Song, J.J., Cho, J.G., Hwang, S.J., Cho, C.G., Park, S.W. and Chae, S.W. (2008). Inhibitory effect of caffeic acid phenethyl ester (CAPE) on LPS-induced inflammation of human middle ear epithelial cells. *Acta Otolaryngol*, **128**, 1303-7.

Song, X., Call, G.B., Kirilly, D. and Xie, T. (2007). Notch signaling controls germline stem cell niche formation in the Drosophila ovary. *Development*, **134**, 1071-80.

Sonnenburg, E.D., Gao, T. and Newton, A.C. (2001). The Phosphoinositidedependent Kinase, PDK-1, Phosphorylates Conventional Protein Kinase C Isozymes by a Mechanism That Is Independent of Phosphoinositide 3-Kinase. *Journal of Biological Chemistry*, **276**, 45289-45297.

Souza, R.F., Shewmake, K., Pearson, S., Sarosi, G.A., Jr., Feagins, L.A., Ramirez, R.D., Terada, L.S. and Spechler, S.J. (2004). Acid increases proliferation via ERK and p38 MAPK-mediated increases in cyclooxygenase-2 in Barrett's adenocarcinoma cells. *Am J Physiol Gastrointest Liver Physiol*, **287**, G743-748.

Spangrude, G.J., Heimfeld, S. and Weissman, I.L. (1988). Purification and characterization of mouse hematopoietic stem cells. *Science*, **241**, 58-62.

Srivastava, D. (2006). Making or Breaking the Heart: From Lineage Determination to Morphogenesis. *Cell*, **126**, 1037-1048.

Stahl, N., Boulton, T.G., Farruggella, T., Ip, N.Y., Davis, S., Witthuhn, B.A., Quelle, F.W., Silvennoinen, O., Barbieri, G., Pellegrini, S. and et, a. (1994). Association and activation of Jak-Tyk kinases by CNTF-LIF-OSM-IL-6 beta receptor components. *Science*, **263**, 92-95.

Stamler, J., Dyer, A.R., Shekelle, R.B., Neaton, J. and Stamler, R. (1993). Relationship of baseline major risk factors to coronary and all-cause mortality, and to longevity: findings from long-term follow-up of Chicago cohorts. *Cardiology*, **82**, 191-222.

Standaert, M.L., Bandyopadhyay, G., Kanoh, Y., Sajan, M.P. and Farese, R.V. (2001). Insulin and PIP3 activate PKC-zeta by mechanisms that are both dependent and independent of phosphorylation of activation loop (T410) and autophosphorylation (T560) sites. *Biochemistry*, **40**, 249-55.

Stennard, F.A., Costa, M.W., Elliott, D.A., Rankin, S., Haast, S.J.P., Lai, D.,

McDonald, L.P.A., Niederreither, K., Dolle, P., Bruneau, B.G., Zorn, A.M. and Harvey, R.P. (2003). Cardiac T-box factor Tbx20 directly interacts with Nkx2-5, GATA4, and GATA5 in regulation of gene expression in the developing heart. *Developmental Biology*, **262**, 206-224.

Sun, S., Guo, Z., Xiao, X., Liu, B., Liu, X., Tang, P.H. and Mao, N. (2003). Isolation of mouse marrow mesenchymal progenitors by a novel and reliable method. *Stem Cells*, **21**, 527-35.

Sundaram, A.A., Ayala, C., Greenlund, K.J. and Keenan, N.L. (2005). Differences in the Prevalence of Self-Reported Risk Factors for Coronary Heart Disease Among American Women by Race/Ethnicity and Age: Behavioral Risk Factor Surveillance System, 2001. *American Journal of Preventive Medicine*, **29**, 25-30.

Suzuki, K., Hino, M., Kutsuna, H., Hato, F., Sakamoto, C., Takahashi, T., Tatsumi, N. and Kitagawa, S. (2001). Selective activation of p38 mitogenactivated protein kinase cascade in human neutrophils stimulated by IL-1beta. *J Immunol*, **167**, 5940-7.

Suzuki, Y.J., Ikeda, T., Shi, S.S., Kitta, K., Kobayashi, Y.M., Morad, M., Jones, L.R. and Blumberg, J.B. (1999). Regulation of GATA-4 and AP-1 in transgenic mice overexpressing cardiac calsequestrin. *Cell Calcium*, **25**, 401-7.

Svensson, E.C., Tufts, R.L., Polk, C.E. and Leiden, J.M. (1999). Molecular cloning of FOG-2: A modulator of transcription factor GATA-4 in cardiomyocytes. *Proc Natl Acad Sci U S A*, **96**, 956-961.

Swijnenburg, R.-J., Tanaka, M., Vogel, H., Baker, J., Kofidis, T., Gunawan, F., Lebl, D.R., Caffarelli, A.D., de Bruin, J.L., Fedoseyeva, E.V. and Robbins, R.C. (2005). Embryonic Stem Cell Immunogenicity Increases Upon Differentiation After Transplantation Into Ischemic Myocardium. *Circulation*, **112**, I-166-172.

Tada, M. and Smith, J.C. (2000). Xwnt11 is a target of Xenopus Brachyury: regulation of gastrulation movements via Dishevelled, but not through the canonical Wnt pathway. *Development*, **127**, 2227-38.

Takahashi, K., Tanabe, K., Ohnuki, M., Narita, M., Ichisaka, T., Tomoda, K. and Yamanaka, S. (2007). Induction of pluripotent stem cells from adult human fibroblasts by defined factors. *Cell*, **131**, 861-72.

Takahashi, Y., Carpino, N., Cross, J.C., Torres, M., Parganas, E. and Ihle, J.N. (2003). SOCS3: an essential regulator of LIF receptor signaling in trophoblast giant cell differentiation. *Embo J*, **22**, 372-84.

Takehara, Y., Kanno, T., Yoshioka, T., Inoue, M. and Utsumi, K. (1995). Oxygen-dependent regulation of mitochondrial energy metabolism by nitric oxide. *Arch Biochem Biophys*, **323**, 27-32.

Tamai, K., Semenov, M., Kato, Y., Spokony, R., Liu, C., Katsuyama, Y., Hess, F., Saint-Jeannet, J.-P. and He, X. (2000). LDL-receptor-related proteins in Wnt signal transduction. *Nature*, **407**, 530-535.

Tamai, K., Zeng, X., Liu, C., Zhang, X., Harada, Y., Chang, Z. and He, X. (2004). A Mechanism for Wnt Coreceptor Activation. *Molecular Cell*, **13**, 149-156.

Tamir, Y. and Bengal, E. (2000). Phosphoinositide 3-Kinase Induces the Transcriptional Activity of MEF2 Proteins during Muscle Differentiation. *J. Biol. Chem.*, **275**, 34424-34432.

Tan, S.M., Wang, S.T., Hentze, H. and Droge, P. (2007). A UTF1-based selection system for stable homogeneously pluripotent human embryonic stem cell cultures. *Nucl. Acids Res.*, **35**, e118-.

Tanaka, M., Chen, Z., Bartunkova, S., Yamasaki, N. and Izumo, S. (1999). The cardiac homeobox gene Csx/Nkx2.5 lies genetically upstream of multiple genes essential for heart development. *Development*, **126**, 1269-1280.

Tatebayashi, K., Takekawa, M. and Saito, H. (2003). A docking site determining specificity of Pbs2 MAPKK for Ssk2/Ssk22 MAPKKKs in the yeast HOG pathway. *Embo J*, **22**, 3624-34.

Taylor, D.O., Edwards, L.B., Boucek, M.M., Trulock, E.P., Aurora, P., Christie, J., Dobbels, F., Rahmel, A.O., Keck, B.M. and Hertz, M.I. (2007). Registry of the International Society for Heart and Lung Transplantation: twenty-fourth official adult heart transplant report--2007. *J Heart Lung Transplant*, **26**, 769-81.

Taylor, D.O., Edwards, L.B., Boucek, M.M., Trulock, E.P., Waltz, D.A., Keck, B.M. and Hertz, M.I. (2006). Registry of the International Society for Heart and Lung Transplantation: twenty-third official adult heart transplantation report-2006. *J Heart Lung Transplant*, **25**, 869-79.

Taylor, S.M. and Jones, P.A. (1982). Changes in phenotypic expression in embryonic and adult cells treated with 5-azacytidine. *J Cell Physiol*, **111**, 187-94.

ten Dijke, P., Yamashita, H., Ichijo, H., Franzen, P., Laiho, M., Miyazono, K. and Heldin, C.H. (1994). Characterization of type I receptors for transforming growth factor-beta and activin. *Science*, **264**, 101-4.

Terada, N., Hamazaki, T., Oka, M., Hoki, M., Mastalerz, D.M., Nakano, Y., Meyer, E.M., Morel, L., Petersen, B.E. and Scott, E.W. (2002). Bone marrow cells adopt the phenotype of other cells by spontaneous cell fusion. *Nature*, **416**, 542-545.

Thiels, E., Kanterewicz, B.I., Norman, E.D., Trzaskos, J.M. and Klann, E.

(2002). Long-term depression in the adult hippocampus in vivo involves activation of extracellular signal-regulated kinase and phosphorylation of Elk-1. *J Neurosci*, **22**, 2054-62.

Thippeswamy, T., McKay, J.S. and Morris, R. (2001). Bax and caspases are inhibited by endogenous nitric oxide in dorsal root ganglion neurons in vitro. *Eur J Neurosci*, **14**, 1229-36.

Thom, T., Haase, N., Rosamond, W., Howard, V.J., Rumsfeld, J., Manolio *et al.* (2006).

Heart disease and stroke statistics--2006 update: a report from the American Heart Association Statistics Committee and Stroke Statistics Subcommittee. *Circulation* **113**: e85-151.

Thomson, J.A., Itskovitz-Eldor, J., Shapiro, S.S., Waknitz, M.A., Swiergiel, J.J., Marshall, V.S. and Jones, J.M. (1998). Embryonic Stem Cell Lines Derived from Human Blastocysts. *Science*, **282**, 1145-1147.

Thomson, J.A., Kalishman, J., Golos, T.G., Durning, M., Harris, C.P., Becker, R.A. and Hearn, J.P. (1995). Isolation of a Primate Embryonic Stem Cell Line. pp. 7844-7848.

Timmerman, L.A., Grego-Bessa, J., Raya, A., Bertran, E., Perez-Pomares, J.M., Diez, J., Aranda, S., Palomo, S., McCormick, F., Izpisua-Belmonte, J.C. and de la Pompa, J.L. (2004). Notch promotes epithelial-mesenchymal transition during cardiac development and oncogenic transformation. *Genes Dev.*, **18**, 99-115.

Tokuzawa, Y., Kaiho, E., Maruyama, M., Takahashi, K., Mitsui, K., Maeda, M., Niwa, H. and Yamanaka, S. (2003). Fbx15 Is a Novel Target of Oct3/4 but Is Dispensable for Embryonic Stem Cell Self-Renewal and Mouse Development. *Mol. Cell. Biol.*, **23**, 2699-2708.

Toledano, M.B., Ghosh, D., Trinh, F. and Leonard, W.J. (1993). N-terminal DNA-binding domains contribute to differential DNA-binding specificities of NF-kappa B p50 and p65. *Mol. Cell. Biol.*, **13**, 852-860.

Toma, C., Pittenger, M.F., Cahill, K.S., Byrne, B.J. and Kessler, P.D. (2002). Human Mesenchymal Stem Cells Differentiate to a Cardiomyocyte Phenotype in the Adult Murine Heart. *Circulation*, **105**, 93-98.

Toma, J.G., Akhavan, M., Fernandes, K.J., Barnabe-Heider, F., Sadikot, A., Kaplan, D.R. and Miller, F.D. (2001). Isolation of multipotent adult stem cells from the dermis of mammalian skin. *Nat Cell Biol*, **3**, 778-84.

Tomita, Y., Makino, S., Hakuno, D., Hattan, N., Kimura, K., Miyoshi, S., Murata, M., Ieda, M. and Fukuda, K. (2007). Application of mesenchymal stem cell-derived cardiomyocytes as bio-pacemakers: current status and problems to be solved. *Med Biol Eng Comput*.

Tonissen, K.F., Drysdale, T.A., Krieg, P.A., Lints, T.J. and Harvey, R.P. (1994). XNkx-2.5, a Xenopus gene related to Nkx-2.5 and tinman: Evidence for a conserved role in cardiac development. *Developmental Biology*, **162**, 325-328.

Topol, L., Jiang, X., Choi, H., Garrett-Beal, L., Carolan, P.J. and Yang, Y. (2003). Wnt-5a inhibits the canonical Wnt pathway by promoting GSK-3-independent b-catenin degradation. *Journal of Cell Biology*, **162**, 899-908.

Torres, M.A., Yang-Snyder, J.A., Purcell, S.M., DeMarais, A.A., McGrew, L.L. and Moon, R.T. (1996). Activities of the Wnt-1 class of secreted signaling factors are antagonized by the Wnt-5A class and by a dominant negative cadherin in early Xenopus development. *Journal of Cell Biology*, **133**, 1123-1137.

Toullec, D., Pianetti, P., Coste, H., Bellevergue, P., Grand-Perret, T., Ajakane, M., Baudet, V., Boissin, P., Boursier, E. and Loriolle, F. (1991). The bisindolylmaleimide GF 109203X is a potent and selective inhibitor of protein kinase C. *J. Biol. Chem.*, **266**, 15771-15781.

Traenckner, E.B., Pahl, H.L., Henkel, T., Schmidt, K.N., Wilk, S. and Baeuerle, P.A. (1995). Phosphorylation of human I kappa B-alpha on serines 32 and 36 controls I kappa B-alpha proteolysis and NF-kappa B activation in response to diverse stimuli. *Embo J*, **14**, 2876-83.

Trigo-Rocha, F., Aronson, W.J., Hohenfellner, M., Ignarro, L.J., Rajfer, J. and Lue, T.F. (1993). Nitric oxide and cGMP: mediators of pelvic nerve-stimulated erection in dogs. *Am J Physiol*, **264**, H419-22.

Trivedi, P. and Hematti, P. (2007). Simultaneous generation of CD34+ primitive hematopoietic cells and CD73+ mesenchymal stem cells from human embryonic stem cells cocultured with murine OP9 stromal cells. *Exp Hematol*, **35**, 146-54.

Tropel, P., Platet, N., Platel, J.C., Noel, D., Albrieux, M., Benabid, A.L. and Berger, F. (2006). Functional neuronal differentiation of bone marrow-derived mesenchymal stem cells. *Stem Cells*, **24**, 2868-76.

Trulock, E.P., Edwards, L.B., Taylor, D.O., Boucek, M.M., Mohacsi, P.J., Keck, B.M. and Hertz, M.I. (2003). The registry of the international society for heart and lung transplantation: twentieth official adult lung and heart-lung transplant report--2003. *The Journal of Heart and Lung Transplantation*, **22**, 625-635.

Tsai, F.-Y. and Orkin, S.H. (1997). Transcription Factor GATA-2 Is Required for Proliferation/Survival of Early Hematopoietic Cells and Mast Cell Formation, But Not for Erythroid and Myeloid Terminal Differentiation. *Blood*, **89**, 3636-3643.

Tsai, S.F., Martin, D.I.K., Zon, L.I., D'Andrea, A.D., Wong, G.G. and Orkin, S.H. (1989). Cloning of cDNA for the major DNA-binding protein of the erythroid lineage through expression in mammalian cells. *Nature*, **339**, 446-451.

Tsao, P.W., Suzuki, T., Totsuka, R., Murata, T., Takagi, T., Ohmachi, Y., Fujimura, H. and Takata, I. (1997). The effect of dexamethasone on the expression of activated NF-kappa B in adjuvant arthritis. *Clin Immunol Immunopathol*, **83**, 173-8.

Tse, A.K.-W., Wan, C.-K., Shen, X.-L., Zhu, G.-Y., Cheung, H.-Y., Yang, M. and Fong, W.-F. (2007). 1,25-Dihydroxyvitamin D3 induces biphasic NF-[kappa]B responses during HL-60 leukemia cells differentiation through protein induction and PI3K/Akt-dependent phosphorylation/degradation of I[kappa]B. *Experimental Cell Research*, **313**, 1722-1734.

Tuli, R., Tuli, S., Nandi, S., Huang, X., Manner, P.A., Hozack, W.J., Danielson, K.G., Hall, D.J. and Tuan, R.S. (2003). Transforming Growth Factor-{beta}-mediated Chondrogenesis of Human Mesenchymal Progenitor Cells Involves N-cadherin and Mitogen-activated Protein Kinase and Wnt Signaling Cross-talk. *J. Biol. Chem.*, **278**, 41227-41236.

Tulina, N. and Matunis, E. (2001). Control of Stem Cell Self-Renewal in Drosophila Spermatogenesis by JAK-STAT Signaling. pp. 2546-2549.

Tumbar, T., Guasch, G., Greco, V., Blanpain, C., Lowry, W.E., Rendl, M. and Fuchs, E. (2004). Defining the Epithelial Stem Cell Niche in Skin 10.1126/science.1092436. *Science*, **303**, 359-363.

Turkson, J., Bowman, T., Adnane, J., Zhang, Y., Djeu, J.Y., Sekharam, M., Frank, D.A., Holzman, L.B., Wu, J., Sebti, S. and Jove, R. (1999). Requirement for Ras/Rac1-mediated p38 and c-Jun N-terminal kinase signaling in Stat3 transcriptional activity induced by the Src oncoprotein. *Mol Cell Biol*, **19**, 7519-28.

Turnpenny, L., Brickwood, S., Spalluto, C.M., Piper, K., Cameron, I.T., Wilson, D.I. and Hanley, N.A. (2003). Derivation of Human Embryonic Germ Cells: An Alternative Source of Pluripotent Stem Cells. *Stem Cells*, **21**, 598-609.

Tzahor, E. and Lassar, A.B. (2001). Wnt signals from the neural tube block ectopic cardiogenesis. *Genes Dev*, **15**, 255-60.

Uchida, S., Fuke, S. and Tsukahara, T. (2007). Upregulations of Gata4 and oxytocin receptor are important in cardiomyocyte differentiation processes of P19CL6 cells. *Journal of Cellular Biochemistry*, **100**, 629-641.

Vaira, S., Johnson, T., Hirbe, A.C., Alhawagri, M., Anwisye, I., Sammut, B., O'Neal, J., Zou, W., Weilbaecher, K.N., Faccio, R. and Novack, D.V. (2008). RelB is the NF-kappaB subunit downstream of NIK responsible for osteoclast differentiation. *Proc Natl Acad Sci U S A*, **105**, 3897-3902.

Van Berkel, T.F.M., Boersma, H., Roos-Hesselink, J.W., Erdman, R.A.M. and Simoons, M.L. (1999). Impact of smoking cessation and smoking interventions in patients with coronary heart disease. pp. 1773-1782.

van de Wetering, M., Cavallo, R., Dooijes, D., van Beest, M., van Es, J., Loureiro, J., Ypma, A., Hursh, D., Jones, T., Bejsovec, A., Peifer, M., Mortin, M. and Clevers, H. (1997). Armadillo Coactivates Transcription Driven by the Product of the Drosophila Segment Polarity Gene dTCF. *Cell*, **88**, 789-799.

van de Wetering, M., Sancho, E., Verweij, C., de Lau, W., Oving, I., Hurlstone, A., van der Horn, K., Batlle, E., Coudreuse, D., Haramis, A.P., Tjon-Pon-Fong, M., Moerer, P., van den Born, M., Soete, G., Pals, S., Eilers, M., Medema, R. and Clevers, H. (2002). The beta-catenin/TCF-4 complex imposes a crypt progenitor phenotype on colorectal cancer cells. *Cell*, **111**, 241-50.

van Gijn, M.E., Blankesteijn, W.M., Smits, J.F., Hierck, B. and Gittenberger-de Groot, A.C. (2001). Frizzled 2 is transiently expressed in neural crest-containing areas during development of the heart and great arteries in the mouse. *Anat Embryol (Berl)*, **203**, 185-92.

Vanhoutte, P., Barnier, J.V., Guibert, B., Pages, C., Besson, M.J., Hipskind, R.A. and Caboche, J. (1999). Glutamate induces phosphorylation of Elk-1 and CREB, along with c-fos activation, via an extracellular signal-regulated kinase-dependent pathway in brain slices. *Mol Cell Biol*, **19**, 136-46.

Venkatakrishnan, C.D., Tewari, A.K., Moldovan, L., Cardounel, A.J., Zweier, J.L., Kuppusamy, P. and Ilangovan, G. (2006). Heat shock protects cardiac cells from doxorubicin-induced toxicity by activating p38 MAPK and phosphorylation of small heat shock protein 27. *Am J Physiol Heart Circ Physiol*, **291**, H2680-91.

Viswanathan, S., Benatar, T., Rose-John, S., Lauffenburger, D.A. and Zandstra, P.W. (2002). Ligand/Receptor Signaling Threshold (LIST) Model Accounts for gp130-Mediated Embryonic Stem Cell Self-Renewal Responses to LIF and HIL-6. *Stem Cells*, **20**, 119-138.

Vitelli, F., Taddei, I., Morishima, M., Meyers, E.N., Lindsay, E.A. and Baldini, A. (2002). A genetic link between Tbx1 and fibroblast growth factor signaling. *Development*, **129**, 4605-4611.

Vlahos, C.J., Matter, W.F., Hui, K.Y. and Brown, R.F. (1994). A specific inhibitor of phosphatidylinositol 3-kinase, 2-(4- morpholinyl)-8-phenyl-4H-1-benzopyran-4-one (LY294002). *J. Biol. Chem.*, **269**, 5241-5248.

Volonte, D., Galbiati, F., Pestell, R.G. and Lisanti, M.P. (2001). Cellular stress induces the tyrosine phosphorylation of caveolin-1 (Tyr(14)) via activation of p38 mitogen-activated protein kinase and c-Src kinase. Evidence for caveolae, the actin cytoskeleton, and focal adhesions as mechanical sensors of osmotic stress. *J. Biol. Chem.*, **276**, 8094-8103.

Volpato, M., Seargent, J., Loadman, P.M. and Phillips, R.M. (2005). Formation of DNA interstrand cross-links as a marker of Mitomycin C bioreductive activation and chemosensitivity. *European Journal of Cancer*, **41**, 1331-1338.

Wang, C.-Y., Mayo, M.W., Korneluk, R.G., Goeddel, D.V. and Baldwin, A.S., Jr. (1998). NF-B Antiapoptosis: Induction of TRAF1 and TRAF2 and c-IAP1 and c-IAP2 to Suppress Caspase-8 Activation. *Science*, **281**, 1680-1683.

Wang, T., Xu, Z., Jiang, W. and Ma, A. (2006). Cell-to-cell contact induces mesenchymal stem cell to differentiate into cardiomyocyte and smooth muscle cell. *Int J Cardiol*, **109**, 74-81.

Wang, Y.K., Sporle, R., Paperna, T., Schughart, K. and Francke, U. (1999). Characterization and expression pattern of the frizzled gene Fzd9, the mouse homolog of FZD9 which is deleted in Williams-Beuren syndrome. *Genomics*, **57**, 235-48.

Waskiewicz, A.J., Flynn, A., Proud, C.G. and Cooper, J.A. (1997). Mitogenactivated protein kinases activate the serine/threonine kinases Mnk1 and Mnk2. *Embo J*, **16**, 1909-20.

Weber, J.D., Phillips, P.J., Baldassare, J.J. and Raben, D.M. (1997). Sustained activation of extracellular-signal-regulated kinase 1 (ERK1) is required for the continued expression of cyclin D1 in G1 phase. *Biochemical Journal*, **326**, 61-68.

Weeraratna, A.T., Jiang, Y., Hostetter, G., Rosenblatt, K., Duray, P., Bittner, M. and Trent, J.M. (2002). Wnt5a signaling directly affects cell motility and invasion of metastatic melanoma. *Cancer Cell*, **1**, 279-288.

Wehrli, M., Dougan, S.T., Caldwell, K., O'Keefe, L., Schwartz, S., Vaizel-Ohayon, D., Schejter, E., Tomlinson, A. and DiNardo, S. (2000). arrow encodes an LDL-receptor-related protein essential for Wingless signalling. *Nature*, **407**, 527-530.

Wen, H.C., Huang, W.C., Ali, A., Woodgett, J.R. and Lin, W.W. (2003). Negative regulation of phosphatidylinositol 3-kinase and Akt signalling pathway by PKC. *Cell Signal*, **15**, 37-45.

Wen, J., Xia, Q., Lu, C., Yin, L., Hu, J., Gong, Y., Yin, B., Monzen, K., Yuan, J., Qiang, B., Zhang, X. and Peng, X. (2007). Proteomic analysis of cardiomyocytes differentiation in mouse embryonic carcinoma P19CL6 cells. *J Cell Biochem*, **102**, 149-60.

Westfall, T.A., Brimeyer, R., Twedt, J., Gladon, J., Olberding, A., Furutani-Seiki, M. and Slusarski, D.C. (2003). Wnt-5/pipetail functions in vertebrate axis formation as a negative regulator of Wnt/beta-catenin activity. *Journal of Cell Biology*, **162**, 889-898.

Wexler, E.M., Stanton, P.K. and Nawy, S. (1998). Nitric oxide depresses GABAA receptor function via coactivation of cGMP-dependent kinase and phosphodiesterase. *J Neurosci*, **18**, 2342-9.

Wheeler, G.N. and Hoppler, S. (1999). Two novel Xenopus frizzled genes expressed in developing heart and brain. *Mechanisms of Development*, **86**, 203-207.

Wiechelman, K.J., Braun, R.D. and Fitzpatrick, J.D. (1988). Investigation of the bicinchoninic acid protein assay: identification of the groups responsible for color formation. *Anal Biochem*, **175**, 231-7.

Wiesmann, A., Buhring, H.J., Mentrup, C. and Wiesmann, H.P. (2006). Decreased CD90 expression in human mesenchymal stem cells by applying mechanical stimulation. *Head Face Med*, **2**, 8.

Willert, K., Brown, J.D., Danenberg, E., Duncan, A.W., Weissman, I.L., Reya, T., Yates, J.R. and Nusse, R. (2003). Wnt proteins are lipid-modified and can act as stem cell growth factors. *Nature*, **423**, 448-452.

Williams, R.L., Hilton, D.J., Pease, S., Willson, T.A., Stewart, C.L., Gearing, D.P., Wagner, E.F., Metcalf, D., Nicola, N.A. and Gough, N.M. (1988). Myeloid leukaemia inhibitory factor maintains the developmental potential of embryonic stem cells. *Nature*, **336**, 684-7.

Wobus, A.M., Wallukat, G. and Hescheler, J. (1991). Pluripotent mouse embryonic stem cells are able to differentiate into cardiomyocytes expressing chronotropic responses to adrenergic and cholinergic agents and Ca2+ channel blockers. *Differentiation*, **48**, 173-182.

Woodward, M., Martiniuk, A., Ying Lee, C.M., Lam, T.H., Vanderhoorn, S., Ueshima, H., Fang, X., Kim, H.C., Rodgers, A., Patel, A., Jamrozik, K. and Huxley, R. (2008). Elevated total cholesterol: its prevalence and population attributable fraction for mortality from coronary heart disease and ischaemic stroke in the Asia-Pacific region. *Eur J Cardiovasc Prev Rehabil*, **15**, 397-401.

Worster, A.A., Brower-Toland, B.D., Fortier, L.A., Bent, S.J., Williams, J. and Nixon, A.J. (2001). Chondrocytic differentiation of mesenchymal stem cells sequentially exposed to transforming growth factor-beta1 in monolayer and insulin-like growth factor-I in a three-dimensional matrix. *J Orthop Res*, **19**, 738-49.

Wozney, J.M., Rosen, V., Celeste, A.J., Mitsock, L.M., Whitters, M.J., Kriz, R.W., Hewick, R.M. and Wang, E.A. (1988). Novel regulators of bone formation: molecular clones and activities. *Science*, **242**, 1528-34.

Wrana, J.L., Attisano, L., Wieser, R., Ventura, F. and Massague, J. (1994). Mechanism of activation of the TGF-beta receptor. *Nature*, **370**, 341-7.

Wu, W.-T., Chi, K.-H., Ho, F.-M., Tsao, W.-C. and Lin, W.-W. (2004). Proteasome inhibitors up-regulate haem oxygenase-1 gene expression: requirement of p38 MAPK (mitogen-activated protein kinase) activation but not of NF-kappaB (nuclear factor kappaB) inhibition. *Biochem. J.*, **379**, 587-593.

Xin, M., Davis, C.A., Molkentin, J.D., Lien, C.-L., Duncan, S.A., Richardson, J.A. and Olson, E.N. (2006). A threshold of GATA4 and GATA6 expression is required for cardiovascular development. *Proc Natl Acad Sci U S A*, **103**, 11189-11194.

Xu, C., Adamson, E.D., Liguori, G. and Persico, M.G. (1998). Specific arrest of cardiogenesis in cultured embryonic stem cells lacking Cripto-1. *Developmental Biology*, **196**, 237-247.

Xu, H., Morishima, M., Wylie, J.N., Schwartz, R.J., Bruneau, B.G., Lindsay, E.A. and Baldini, A. (2004). Tbx1 has a dual role in the morphogenesis of the cardiac outflow tract. *Development*, **131**, 3217-3227.

Xu, W., Zhang, X., Qian, H., Zhu, W., Sun, X., Hu, J., Zhou, H. and Chen, Y. (2004). Mesenchymal Stem Cells from Adult Human Bone Marrow Differentiate into a Cardiomyocyte Phenotype In Vitro. *Experimental Biology and Medicine*, **229**, 623-631.

Xu, X.Q., Zweigerdt, R., Soo, S.Y., Ngoh, Z.X., Tham, S.C., Wang, S.T., Graichen, R., Davidson, B., Colman, A. and Sun, W. (2008). Highly enriched cardiomyocytes from human embryonic stem cells. *Cytotherapy*, **10**, 376-89.

Yamagishi, H., Maeda, J., Hu, T., McAnally, J., Conway, S.J., Kume, T., Meyers, E.N., Yamagishi, C. and Srivastava, D. (2003). Tbx1 is regulated by tissue-specific forkhead proteins through a common Sonic hedgehog-responsive enhancer. *Genes Dev.*, **17**, 269-281.

Yamaguchi, T.P., Takada, S., Yoshikawa, Y., Wu, N. and McMahon, A.P. (1999). T (Brachyury) is a direct target of Wnt3a during paraxial mesoderm specification. *Genes Dev.*, **13**, 3185-3190.

Yan, H., Aziz, E., Shillabeer, G., Wong, A., Shanghavi, D., Kermouni, A., Abdel-Hafez, M. and Lau, D.C.W. (2002). Nitric oxide promotes differentiation of rat white preadipocytes in culture. *J. Lipid Res.*, **43**, 2123-2129.

Yanada, S., Ochi, M., Adachi, N., Nobuto, H., Agung, M. and Kawamata, S. (2006). Effects of CD44 antibody-- or RGDS peptide--immobilized magnetic beads on cell proliferation and chondrogenesis of mesenchymal stem cells. *J Biomed Mater Res A*, **77**, 773-84.

Yanazume, T., Morimoto, T., Wada, H., Kawamura, T. and Hasegawa, K. (2003). Biological role of p300 in cardiac myocytes. *Mol Cell Biochem*, **248**, 115-9.

Yang, C., Przyborski, S., Cooke, M.J., Zhang, X., Stewart, R., Anyfantis, G., Atkinson, S.P., Saretzki, G., Armstrong, L. and Lako, M. (2008). A key role for telomerase reverse transcriptase unit in modulating human embryonic stem cell proliferation, cell cycle dynamics, and in vitro differentiation. *Stem Cells*, **26**, 850-63.

Yang, L., Bryder, D., Adolfsson, J., Nygren, J., Mansson, R., Sigvardsson, M. and Jacobsen, S.E.W. (2005). Identification of Lin-Sca1+kit+CD34+Flt3-short-term hematopoietic stem cells capable of rapidly reconstituting and rescuing myeloablated transplant recipients. *Blood*, **105**, 2717-2723.

Yang, L., Li, S., Hatch, H., Ahrens, K., Cornelius, J.G., Petersen, B.E. and Peck, A.B. (2002). In vitro trans-differentiation of adult hepatic stem cells into pancreatic endocrine hormone-producing cells. *PNAS*, **99**, 8078-8083.

Yang, S.H., Galanis, A. and Sharrocks, A.D. (1999). Targeting of p38 mitogen-activated protein kinases to MEF2 transcription factors. *Mol Cell Biol*, **19**, 4028-38.

Yasukawa, H., Hoshijima, M., Gu, Y., Nakamura, T., Pradervand, S., Hanada, T., Hanakawa, Y., Yoshimura, A., Ross, J., Jr. and Chien, K.R. (2001). Suppressor of cytokine signaling-3 is a biomechanical stress-inducible gene that suppresses gp130-mediated cardiac myocyte hypertrophy and survival pathways. *J Clin Invest*, **108**, 1459-67.

Yeom, Y.I., Fuhrmann, G., Ovitt, C.E., Brehm, A., Ohbo, K., Gross, M., Hubner, K. and Scholer, H.R. (1996). Germline regulatory element of Oct-4 specific for the totipotent cycle of embryonal cells. *Development*, **122**, 881-94.

Yoshida, K., Chambers, J., Nichols, J., Smith, A., Saito, M., Yasukawa, K., Shoyab, M., Taga, T. and Kishimoto, T. (1994). Maintenance of the pluripotential phenotype of embryonic stem cells through direct activation of gp130 signalling pathways. *Mechanisms of Development*, **45**, 163-171.

Yoshida, K., Taga, T., Saito, M., Suematsu, S., Kumanogoh, A., Tanaka, T., Fujiwara, H., Hirata, M., Yamagami, T., Nakahata, T., Hirabayashi, T., Yoneda, Y., Tanaka, K., Wang, W.-Z., Mori, C., Shiota, K., Yoshida, N. and Kishimoto, T. (1996). Targeted disruption of gp130, a common signal transducer for the interleukin 6 family of cytokines, leads to myocardial and hematological disorders. *PNAS*, **93**, 407-411.

Yoshida-Koide, U., Matsuda, T., Saikawa, K., Nakanuma, Y., Yokota, T., Asashima, M. and Koide, H. (2004). Involvement of Ras in extraembryonic endoderm differentiation of embryonic stem cells. *Biochemical and Biophysical Research Communications*, **313**, 475-481.

You, H.J., Woo, C.H., Choi, E.Y., Cho, S.H., Yoo, Y.J. and Kim, J.H. (2005). Roles of Rac and p38 kinase in the activation of cytosolic phospholipase A2 in response to PMA. *Biochem J*, **388**, 527-35.

Young, P.R., McLaughlin, M.M., Kumar, S., Kassis, S., Doyle, M.L., McNulty, D., Gallagher, T.F., Fisher, S., McDonnell, P.C., Carr, S.A., Huddleston, M.J., Seibel, G., Porter, T.G., Livi, G.P., Adams, J.L. and Lee, J.C. (1997). Pyridinyl imidazole inhibitors of p38 mitogen-activated protein kinase bind in the ATP site. *J Biol Chem*, **272**, 12116-21.

Yu, J., Vodyanik, M.A., Smuga-Otto, K., Antosiewicz-Bourget, J., Frane, J.L., Tian, S., Nie, J., Jonsdottir, G.A., Ruotti, V., Stewart, R., Slukvin, II and Thomson, J.A. (2007). Induced pluripotent stem cell lines derived from human somatic cells. *Science*, **318**, 1917-20.

Yuan, H., Corbi, N., Basilico, C. and Dailey, L. (1995). Developmental-specific activity of the FGF-4 enhancer requires the synergistic action of Sox2 and Oct-3. *Genes Dev.*, **9**, 2635-2645.

Zaffran, S. and Frasch, M. (2005). The homeodomain of Tinman mediates homo- and heterodimerization of NK proteins. *Biochemical and Biophysical Research Communications*, **334**, 361-369.

Zandi, E., Chen, Y. and Karin, M. (1998). Direct phosphorylation of IkappaB by IKKalpha and IKKbeta: discrimination between free and NF-kappaB-bound substrate. *Science*, **281**, 1360-3.

Zandstra, P.W., Bauwens, C., Yin, T., Liu, Q., Schiller, H., Zweigerdt, R., Pasumarthi, K.B. and Field, L.J. (2003). Scalable production of embryonic stem cell-derived cardiomyocytes. *Tissue Eng*, **9**, 767-78.

Zauli, G., Visani, G., Bassini, A., Caramelli, E., Ottaviani, E., Bertolaso, L., Bertagnolo, V., Borgatti, P. and Capitani, S. (1996). Nuclear translocation of protein kinase C-alpha and -zeta isoforms in HL-60 cells induced to differentiate along the granulocytic lineage by all-trans retinoic acid. *Br J Haematol*, **93**, 542-50.

Zetser, A., Gredinger, E. and Bengal, E. (1999). p38 Mitogen-activated Protein Kinase Pathway Promotes Skeletal Muscle Differentiation. PARTICIPATION OF THE MEF2C TRANSCRIPTION FACTOR. *J. Biol. Chem.*, **274**, 5193-5200.

Zhang, A., Ding, G., Huang, S., Wu, Y., Pan, X., Guan, X., Chen, R. and Yang, T. (2005). c-Jun NH2-terminal kinase mediation of angiotensin IIinduced proliferation of human mesangial cells. *Am J Physiol Renal Physiol*, **288**, F1118-24.

Zhang, H. and Bradley, A. (1996). Mice deficient for BMP2 are nonviable and have defects in amnion/chorion and cardiac development. *Development*, **122**, 2977-2986.

Zhang, H., Toyofuku, T., Kamei, J. and Hori, M. (2003). GATA-4 regulates cardiac morphogenesis through transactivation of the N-cadherin gene. *Biochemical and Biophysical Research Communications*, **312**, 1033-1038.

Zhang, J., Niu, C., Ye, L., Huang, H., He, X., Tong, W.-G., Ross, J., Haug, J., Johnson, T., Feng, J.Q., Harris, S., Wiedemann, L.M., Mishina, Y. and Li, L. (2003). Identification of the haematopoietic stem cell niche and control of the niche size. **425**, 836-841.

Zhang, L., Yang, S.H. and Sharrocks, A.D. (2007). Rev7/MAD2B links c-Jun N-terminal protein kinase pathway signaling to activation of the transcription factor Elk-1. *Mol Cell Biol*, **27**, 2861-9.

Zhang, S.-C., Wernig, M., Duncan, I.D., Brustle, O. and Thomson, J.A. (2001). In vitro differentiation of transplantable neural precursors from human embryonic stem cells. *Nat Biotech*, **19**, 1129-1133.

Zhang, W.J., Park, C., Arentson, E. and Choi, K. (2005). Modulation of hematopoietic and endothelial cell differentiation from mouse embryonic stem cells by different culture conditions. *Blood*, **105**, 111-4.

Zhao, M., New, L., Kravchenko, V.V., Kato, Y., Gram, H., di Padova, F., Olson, E.N., Ulevitch, R.J. and Han, J. (1999). Regulation of the MEF2 family of transcription factors by p38. *Mol Cell Biol*, **19**, 21-30.

Zhao, R., Watt, A.J., Battle, M.A., Li, J., Bondow, B.J. and Duncan, S.A. (2008). Loss of both GATA4 and GATA6 blocks cardiac myocyte differentiation and results in acardia in mice. *Developmental Biology*, **317**, 614-619.

Zhao, Y., Glesne, D. and Huberman, E. (2003). A human peripheral blood monocyte-derived subset acts as pluripotent stem cells. *PNAS*, **100**, 2426-2431.

Zheng, B., Han, M., Wen, J.K. and Zhang, R. (2008). Human heart LIM protein activates atrial-natriuretic-factor gene expression by interacting with the cardiac-restricted transcription factor Nkx2.5. *Biochem J*, **409**, 683-90.

Zheng, Y.-B., Gao, Z.-L., Xie, C., Zhu, H.-P., Peng, L., Chen, J.-H. and Chong, Y.T. (2008). Characterization and hepatogenic differentiation of mesenchymal stem cells from human amniotic fluid and human bone marrow: A comparative study. *Cell Biology International*, **32**, 1439-1448.

Zhong, S., Fromm, J. and Johnson, D.L. (2007). TBP is differentially regulated by c-Jun N-terminal kinase 1 (JNK1) and JNK2 through Elk-1, controlling c-Jun expression and cell proliferation. *Mol Cell Biol*, **27**, 54-64.

Zhou, J., An, H., Xu, H., Liu, S. and Cao, X. (2005). Heat shock up-regulates expression of Toll-like receptor-2 and Toll-like receptor-4 in human monocytes via p38 kinase signal pathway. *Immunology*, **114**, 522-30.

Zhou, P., Qian, L. and Iadecola, C. (2005). Nitric oxide inhibits caspase activation and apoptotic morphology but does not rescue neuronal death. *J* 

Cereb Blood Flow Metab, 25, 348-57.

Zhou, X., Quann, E. and Gallicano, G.I. (2003). Differentiation of nonbeating embryonic stem cells into beating cardiomyocytes is dependent on downregulation of PKC[beta] and [zeta] in concert with upregulation of PKC[var epsilon]. *Developmental Biology*, **255**, 407-422.

Zhu, H., Mitsuhashi, N., Klein, A., Barsky, L.W., Weinberg, K., Barr, M.L., Demetriou, A. and Wu, G.D. (2006). The role of the hyaluronan receptor CD44 in mesenchymal stem cell migration in the extracellular matrix. *Stem Cells*, **24**, 928-35.

Zhu, J., Heid, P.J., Fukuyama, M., Sugimoto, A., Rothman, J.H., Hill, R.J. and Priess, J.R. (1997). end-1 Encodes an apparent GATA factor that specifies the endoderm precursor in Caenorhabditis elegans embryos. *Genes and Development*, **11**, 2883-2896.

Zhu, X., Rottkamp, C.A., Boux, H., Takeda, A., Perry, G. and Smith, M.A. (2000). Activation of p38 kinase links tau phosphorylation, oxidative stress, and cell cycle-related events in Alzheimer disease. *J Neuropathol Exp Neurol*, **59**, 880-8.

Zima, A., Martynyuk, A.E., Seubert, C.N., Morey, T.E., Sumners, C., Cucchiara, R.F. and Dennis, D.M. (2000). Antagonism of the positive dromotropic effect of isoproterenol by adenosine: role of nitric oxide, cGMP-dependent cAMP-phosphodiesterase and protein kinase G. *J Mol Cell Cardiol*, **32**, 1609-19.

Zimmerman, L.B., De Jesús-Escobar, J.M. and Harland, R.M. (1996). The Spemann Organizer Signal noggin Binds and Inactivates Bone Morphogenetic Protein 4. *Cell*, **86**, 599-606.

Zou, Y., Evans, S., Chen, J., Kuo, H.C., Harvey, R.P. and Chien, K.R. (1997). CARP, a cardiac ankyrin repeat protein, is downstream in the Nkx2-5 homeobox gene pathway. *Development*, **124**, 793-804.

Zuk, P.A., Zhu, M., Ashjian, P., De Ugarte, D.A., Huang, J.I., Mizuno, H., Alfonso, Z.C., Fraser, J.K., Benhaim, P. and Hedrick, M.H. (2002). Human Adipose Tissue Is a Source of Multipotent Stem Cells. *Mol. Biol. Cell*, **13**, 4279-4295.

#### **APPENDIX**

#### **APPENDIX I: CELL LINES**

## Appendix I (a):H9c2 (2-1) Cell Line

H9c2 (2-1)
Rat BDIX heart myoblast . Cell form myotubes in 1%
serum (FBS)
Rat
heart
Myoblast
Adherent
DMEM + 2mM Glutamine + 10 % FBS
Split cells at 70-80% confluency using trypsin/EDTA.
Important to passage at sub-confluence to avoid
spontaneous cell fusion to form myotubes
European Collection of Cell Cultures (ECACC)
http://www.hpacultures.org.uk/collections/ecacc.jsp
88092904

#### Appendix I (b):P19 Embryonal Carcinoma Stem Cell Line

Cell Line Name	P19 embryonal carcinoma stem cells
Cell Line Description	The mouse teratocarcinoma pluripotent cell line was derived from an embryonal carcinoma induced in a C3H/He strain mouse. The P19 cells c differentiate into neuronal and glial cells in the presence of retinoic acid. The P19 cell Aggregates differentiate into cardiac and skeletal muscle in the presence of DMSO
Species	Mouse
Tissue	embryo
Growth Mode	Adherent
Culture Medium	Alpha MEM + 2mM Glutamine + 10 % FBS
Subculture Routine	Split cells at 70-80% confluency using trypsin/EDTA.
Karyotype:	Euploid
Supplier	European Collection of Cell Cultures (ECACC)
	http://www.hpacultures.org.uk/collections/ecacc.jsp
Catalogue No	95102107

#### APPENDIX II: CELL CULTURE REAGENTS

Reagents	Description	Storage	Supplier
Minimal Essential Medium (MEM) Alpha Medium (α- MEM) (Product code:22561021)	Without but Ribonucleosides and Deoxyribonucleo sides but Contains Earle's Salts and L- glutamine	Fridge 4 ℃	Invitrogen, UK http://www.invitrogen.com
Dulbecco's Modified Eagle Medium (D-MEM) (1X) (Low Glucose) (Product code: 22320022)	Contains L- Glutamine, 1000 mg/L D-Glucose, Sodium Pyruvate, 25 mM HEPES.	Fridge 4℃,	Invitrogen, UK http://www.invitrogen.com
Foetal Bovine Serum (Product code: 10500064)	Heat inactivated Foetal Bovine Serum	Stock: -20℃,	Invitrogen, UK http://www.invitrogen.com
Penicillin- Streptomycin, liquid (10,000 units penicillin;10,000 µg streptomycin) (Product code: 15140122)	Contains 10,000 units of penicillin and 10,000 µg of streptomycin kills Gram-ve and +ve bacteria	Stock: -20℃,	Invitrogen, UK http://www.invitrogen.com
Trypsin, 0.5% (10x) with EDTA 4Na, (Product code: 15400054)	Contains 5.0 g/L of trypsin, 2.0 g/L of EDTA.4Na, 8.5 g/L of NaCl	Stock : -20 ℃, dilutedstock solution Aliquots: Fridge 4 ℃,	Invitrogen, UK http://www.invitrogen.com

#### APPENDIX III: NITRIC OXIDE DONORS AND 8-BROMO-CGMP

Appendix III (a): NOC-5	
Alternate Names	3-[2-Hydroxy-1-(1-methylethyl)-2-nitrosohydrazino]-1-
	propanamine
Size:	10 mg
Description	A nitric oxide (NO) donor. Half-life $(t_{1/2})$ of NO release is 93
	min (PBS, pH 7.4, 22℃).
CAS Number	146724-82-5
Molecular Weight	176.2
Molecular Formula	$C_6H_{16}N_4O_2$
Structure:	$H_{3}C$ $H_{3}C$ $H_{3}C$ $H_{3}C$ $H_{3}C$ $H_{3}C$ $H_{3}C$
Solubility	H <sub>2</sub> O or NaOH (0.1 N, ≥ pH 10)
Stock Concentration Prepared	100mM (5mg/284µl of NaOH)
Storage	Protect from light
	Solid: -20℃
	Alkaline stock solution Aliquots: -20°C
Supplier	Calbiochem
	http://www.merckbiosciences.co.uk
Catalogue Number	487950

Appendix III (b): NOC-18	
Alternate Names:	DETA NONOate; DETA/NO; 2,2'-
	(Hydroxynitrosohydrazino) <i>bis</i> -ethanamine
Description	A nitric oxide (NO) donor. Half-life for NO release ( $t_{1/2}$ =
	3400 min; PBS, pH 7.4, 22 ℃)
CAS Number	146724-94-9
Molecular Weight	163.2
Molecular Formula	$C_4H_{13}N_5O_2$
Structure:	0
Solubility	H <sub>2</sub> O or NaOH
Stock Concentration	100mM (5mg/306µl of NaOH)
Prepared	
Storage	Protect from light
	Solid: -20 ℃
	Stock solution Aliquots: -20 ℃
Supplier	Calbiochem, http://www.merckbiosciences.co.uk
Catalogue Number	487957

Appendix III (c): SNAP	
Alternate Names	(±)-S-Nitroso-N-acetylpenicillamine
Size:	20 mg
Description	A Nitric oxide donor with a half-life ( $t_{1/2} = 10$ h). SNAP begins to evolve nitric oxide immediately in aqueous
	buffers
CAS Number	79032-48-7
Molecular Weight	220.2
Molecular Formula	$C_7H_{12}N_2O_4S$
Structure	O=N-SH NHCOCH3
Solubility	DMSO, Ethanol or Methanol
Stock Concentration Prepared	100mM (5mg/227µl of DMSO)
Storage	Protect from light store Solid: -20 ℃
	Stock solution Aliquots: -20 ℃
Supplier	Calbiochem
	http://www.merckbiosciences.co.uk
Catalogue Number	487910

Appendix III (d) : SIN-1, Hydrochloride		
Alternate Names	3-Morpholinosydnonimine, HCI	
Size:	20 mg	
Description	Nitric oxide (NO) donor. Spontaneously decomposes to yield NO and superoxide anion radicals.	
CAS Number	16142-27-1	
Molecular Weight	206.7	
Molecular Formula	$C_6H_{10}N_4O_2 \cdot HCI$	
Structure		
Solubility	DMSO or Ethanol	
Stock Concentration Prepared	100mM (5mg/242µl of DMSO)	
Storage	Protect from light Solid: -20℃ Stock solution Aliquots: -20℃	
Supplier	Calbiochem http://www.merckbiosciences.co.uk	
Catalogue Number	567028	

Appendix III (e): Guanos	ine 3',5'-cyclic Monophosphate	
Alternate Names	8-Bromo-cGMP; Br-cGMP, Na	
Description	Cell-permeable cGMP analog	
CAS Number	51116-01-9	
Molecular Weight	446.1	
Molecular Formula	$C_{10}H_{10}BrN_5O_7P \cdot Na$	
Structure	HN HN HN Br H2N N H2N O H2N O H	
Solubility	H <sub>2</sub> O	
Stock Concentration	100mM (5mg/112µl of NaOH)	
Prepared		
Storage	Solid: -20℃,stock solution Aliquots: -20℃	
Supplier	Calbiochem	
	http://www.merckbiosciences.co.uk	
Catalogue Number	203820	

Г

#### **APPENDIX VI: INHIBITORS**

Appendix VI (a): Bisindolylmaleimide I		
Alternate Names	Gö 6850; GF109203X, 2-[1-(3-Dimethylaminopropyl)-1H- indol-3-yl]-3-(1H-indol-3-yl)-maleimide	
Description	A reversible protein kinase C (PKC) inhibitor that acts as a competitive inhibitor for the ATP-binding site	
CAS Number	133052-90-1	
Molecular Weight	412.5	
Molecular Formula	$C_{25}H_{24}N_4O_2$	
Structure:		
Solubility	Anhydrous DMSO	
Stock Concentration Prepared	10mM (1mg/242µl of DMSO)	
Storage	Solid: -20℃,stock solution Aliquots: -20℃	
Supplier	Calbiochem	
	http://www.merckbiosciences.co.uk	
Catalogue Number	203290	

Appendix VI (b): LY294002		
Alternate Names	2-(4-Morpholinyl)-8-phenyl-4H-1-benzopyran-4-one	
Description	A phosphatidylinositol 3-kinase (PI 3-kinase) inhibitor that	
	acts on the ATP-binding site of the enzyme	
CAS Number	154447-36-6	
Molecular Weight	307.4	
Molecular Formula	C <sub>19</sub> H <sub>17</sub> NO <sub>3</sub>	
Structure:		
Solubility	DMSO	
Stock Concentration Prepared	10mM (5mg/1627µl) of DMSO	
Storage	Solid: -20℃,stock solution Aliquots: -20℃, Protect from light	
Supplier	Calbiochem	
	http://www.merckbiosciences.co.uk	
Catalogue Number	440202	

Appendix VI (c): SB2035	80
Alternate Names	4-(4-Fluorophenyl)-2-(4-methylsulfinylphenyl)-5-(4- pyridyl)1H-imidazole
Description	A highly specific, cell-permeable inhibitor of p38 kinase $(IC_{50} = 34 \text{ nM} \text{ in vitro}, 600 \text{ nM} \text{ in cells}).$
CAS Number	152121-47-6
Molecular Weight	377.4
Molecular Formula	C <sub>21</sub> H <sub>16</sub> N <sub>3</sub> OSF
Structure:	F NH CH <sub>3</sub>
Solubility	DMSO
Stock Concentration Prepared	10mM (1mg/265μl of DMSO)
Storage	Protect from light Solid: -20 ℃ Stock solution Aliquots: -20 ℃
Supplier	Calbiochem http://www.merckbiosciences.co.uk
Catalogue Number	559389

Appendix VI (d): CAPE		
Alternate Names	Caffeic Acid Phenethyl Ester	
Description	An active component of propolis from honeybee hive and	
	a potent and specific inhibitor of NF-κB activation	
CAS Number	104594-70-9	
Molecular Weight	284.3	
Molecular Formula	C <sub>17</sub> H <sub>16</sub> O <sub>4</sub>	
Structure		
Solubility	DMSO	
Stock Concentration Prepared	100mM (5mg/176μl of DMSO)	
Storage	Solid: -20 ℃,stock solution Aliquots: -20 ℃, Protect from light	
Supplier	Calbiochem http://www.merckbiosciences.co.uk	
Catalogue Number	211200	

Г

#### **APPENDIX VII: ANTIBODIES**

#### Appendix VII (a): Monoclonal Anti-Cardiac Myosin Light Chain I Antibody (MLC-1v) Antibody

Antibody Name	Cardiac Myosin light chain I antibody [MLM527]
Antibody Type	Primary antibodies
Description	Mouse monoclonal [MLM527] to Cardiac Myosin light chain
Immunogen	Human ventricle myosin light chain I
Reacts with	Human, Mouse, Rat, Cow, Pig
Specificity	Antibody reacts with ventricle myosin light chain I (MLC-1v)
Application used	Western blotting (can also be used for ELISA, ICC/IF)
Band size	23 to 27kDa
Raised in	Mouse
Clonality	Monoclonal
Clone number	MLM527
Isotype	lgG2b
Purity	Protein A purified
Storage	Aliquot and store at -20 °C or -80 °
Concentration	1.0 mg/ml
Dilution used	1:2500 in blocking buffer
Secondary antibody	Anti-mouse antibody conjugated
required	to horseradish peroxidase (HRP)
Supplier	Abcam, UK
	http://www.abcam.com
Catalogue Number	ab680

# Appendix VII (b): Monoclonal Anti-Cardiac Myosin Heavy Chain (α& β isoforms) Antibody

Antibody Name	Heavy Chain Cardiac Myosin antibody [3-48]
Antibody Type	Primary antibodies
Description	Mouse monoclonal [3-48] to heavy chain cardiac Myosin
Immunogen	Full length native protein (purified) (Human).
Reacts with	Human, Mouse, Rat, Cow, Dog, Pig, Rabbit
Specificity	Antibody reacts with alpha and beta isoform of myosin heavy
	chain
Application used	Western blotting (can also be used for ELISA, Flow Cyt,
	ICC/IF, IHC)
Band size	223 kDa
Raised in	Mouse
Clonality	Monoclonal
Clone number	3-48
Isotype	lgG1
Purity	IgG fraction
Storage	Aliquot and store at -20℃ or -80℃
Concentration	1.0mg/ml
Dilution used	1:2500 in blocking buffer
Secondary antibody	Anti-mouse antibody conjugated
required	to horseradish peroxidase (HRP)
Supplier	Abcam, UK
	http://www.abcam.com
Catalogue Number	ab15

## Appendix VII (c): Monoclonal Anti-cardiac Troponin I Antibody

Antibody Name	Cardiac Troponin I antibody [284 (19C7)]
Antibody Type	Primary antibodies
Description	Mouse monoclonal [284 (19C7)] to cardiac Troponin
Immunogen	Native troponin complex
Reacts with	Human, Mouse, Rat, Cat, Cow, Dog, Fish, Goat, Monkey, Pig,
	Rabbit
Specificity	Reacts equally with free cardiac troponin I and cardiac troponin
	I forming complexes with other troponin components. No cross-
	reactivity with skeletal muscle troponin I.
Application used	Western blotting (can also be used for AP, ICC/IF, IHC-P,
	ELISA)
Band size	24 kDa
Raised in	Mouse
Clonality	Monoclonal
Clone number	284 (19C7)
Isotype	IgG2b
Purity	Protein A purified
Storage	Aliquot and store at -20℃ or -80℃
Concentration	3.2 mg/ml
Dilution used	1:2500 in blocking buffer
Secondary antibody	Anti-mouse antibody conjugated
required	to horseradish peroxidase (HRP)
Supplier	Abcam, UK
	http://www.abcam.com
Catalogue Number	ab19615

## Appendix VII (d): Monoclonal Anti- phospho-Akt (Ser473) Antibody

Antibody Name	Mouse Phospho-Akt (Ser473) (587F11) monoclonal antibody
Antibody Type	Primary antibody
Description	A mouse monoclonal antibody to Akt phosphorylated at serine 473 (Ser473)
Immunogen	A synthetic phospho-peptide (KLH-coupled) corresponding to residues around Ser47 of mouse Akt.
Reacts with	Human, mouse Rat Hamster
Specificity	Antibody reacts with Akt only when phosphorylated at serine 473 and does not react with related kinases such as PKC and p70 S6 kinase.
Application used	Western blotting (can also be used IP)
Band size	60 kDa
Raised in	mouse
Clonality	monoclonal
Clone number	587F11
Isotype	lgG2b
Purity	Protein A purified
Storage	Whole tube (i.e. no <i>aliquot) o</i> f the antibody was stored at - 20 °C. Required portions of the antibody were taken from -20 °C.
Concentration	1.0 mg/ml
Dilution used	1:1000
Secondary antibody required	Anti-mouse antibody conjugated to horseradish peroxidase (HRP)
Supplier	Cell Signaling Technology
	http://www.cellsignal.com
Catalogue Number	4051
Appendix VII (e): Monoclonal Anti-β-Actin antibody conjugated to horseradish peroxidise (HRP)	
--	---
Antibody Name	Monoclonal Anti-β-Actin-Peroxidase, Clone AC-15
Antibody Type	Primary antibody
Description	A Protein A purified fraction of mouse monoclonal anti-b-actin isolated from ascites fluid of the AC-15 and conjugated horseradish peroxidise (HRP)
Immunogen	A synthetic peptide corresponding to a slightly modified b- cytoplasmic actin N-terminal sequence: Ac-Asp-Asp-Asp-Ile-Ala-Ala-Leu-Val-Ile- Asp-Asn-Gly-Ser-Gly-Lys conjugated to KLH
Reacts with	Human, mouse, rat, bovine, sheep, pig, rabbit, cat, dog, guinea pig, chicken, carp
Specificity	Antibody reacts specifically with $\beta$ -actin found many tissues and species
Application used	Loading control for western blotting
Band size	42 kDa
Raised in	mouse
Clonality	Monoclonal
Clone number	AC-15
Isotype	lgG1
Purity	Protein A purified
Storage	Aliquot and store at -20℃ or -80℃
Concentration	Conjugate concentration: 2.5–6 mg/mL Antibody Concentration: 2–4 mg/mL
Dilution used	1:25000
Secondary antibody required	None required
Supplier	Sigma (Sigma-Aldrich), UK http://www.sigmaaldrich.com
Catalogue Number	A3854

## **APPENDIX VIII: PREPARATION OF STOCK SOLUTIONS**

Calculation of mole of a compound

Mole of compound

$$Mole = \frac{mass(g)}{mwt(molecular weight)g/mole}$$

Relationship between the mole, concentration and volume of a compound in solution

Concentration (mole/L(M)) = 
$$\frac{mole}{volume(L)} = \frac{mass/_{mwt}}{Volume(L)}$$
  
 $\therefore$  Concentration (M) =  $\frac{mass}{mwt \ x \ volume(L)}$ 

Calculation of required volume for dissolving a compound

The volume parameter from the above equation was made the subject and volume ( $\mu$ I) required to achieve a selected stock concentration of each compound was calculated.

$$Volume (L) = \frac{mass}{mwt \ x \ concentration (M)}$$

$$Volume (\mu l) = \frac{mass (g)}{mwt x mole/L} = \frac{mass (g)}{mwt x mole/10^{-6} \mu l}$$
$$\therefore Volume (\mu l) = \frac{mass (g)}{mwt x mole/L} = \frac{mass (g)}{mwt x mole/\mu l} \times 10^{6}$$

An Example Calculation

## Example: Calculation of the volume of DMSO required for preparing the 10mM stock concentration of LY294002.

Mass of LY294002 used =  $1 \text{ mg} = 1 \times 10^{-3} \text{ g}$ Required Stock concentration =  $10\text{ mM} = 10 \times 10^{-3} \text{ M} = 10^{-2} \text{ M}$ Molecular weight of LY294002 = 307.4 g/mole

$$\therefore Volume \ (\mu l) = \frac{mass \ (g)}{mwt \ x \ mole/L} = \frac{mass \ (g)}{mwt \ x \ mole/\mu l} \ x \ 10^6$$

Volume (
$$\mu L$$
) =  $\frac{1 \times 10^{-3} g}{307.4 \ g/_{mole} \times 10^{-2} \ mole/_{10^{-6} \mu L}}$ 

$$Volume (\mu L) = \frac{1 \times 10^{-3} g}{307.4 \ g/_{mole} \times 10^{-2} \ mole/_{\mu L}} \times 10^{6}$$
$$Volume (\mu L) of DMSO = 325.3 \mu L$$

In a tissue culture hood (i.e. laminar flow cabinet),  $325.3\mu$ L of DMSO was added to 1mg of LY294002. The LY294002 solid was through dissolved, aliquots of 10 $\mu$ L and 20  $\mu$ L portions were made and stored at -20°C.